

Minimax and adaptive tests for detecting abrupt and possibly transitory changes in a Poisson process

Magalie Fromont, Fabrice Grela, Ronan Le Guével

▶ To cite this version:

Magalie Fromont, Fabrice Grela, Ronan Le Guével. Minimax and adaptive tests for detecting abrupt and possibly transitory changes in a Poisson process. 2021. hal-03249762v2

HAL Id: hal-03249762 https://hal.science/hal-03249762v2

Preprint submitted on 1 Sep 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Minimax and adaptive tests for detecting abrupt and possibly transitory changes in a Poisson process

Magalie Fromont

magalie.fromont@univ-rennes2.fr Fabrice Grela

fabrice.grela @univ-rennes 2.fr

and

Ronan Le Guével

ronan.leguevel@univ-rennes2.fr Univ. Rennes, Equipe de Statistique Irmar, UMR CNRS 6625 Place du Recteur Henri Le Moal, CS 24307, 35043 RENNES Cedex, France

Abstract

Motivated by applications in cybersecurity and epidemiology, we consider the problem of detecting an abrupt change in the intensity of a Poisson process, characterised by a jump (non transitory change) or a bump (transitory change) from constant. We propose a complete study from the nonasymptotic minimax testing point of view, when the constant baseline intensity is known or unknown. The question of minimax adaptation with respect to each parameter (height, location, length) of the change is tackled, leading to a comprehensive overview of the various minimax separation rate regimes. We exhibit three such regimes and identify the factors of the two phase transitions by giving the cost of adaptation to each parameter. For any alternative hypothesis, depending on the knowledge or not of the change parameters, we propose minimax or minimax adaptive tests based on simple linear counting statistics derived from Neyman-Pearson tests, or quadratic statistics. When the change location or length is unknown, minimax adaptation is obtained from a scan aggregation principle combined with a Bonferroni or min-p level correction, and a conditioning trick when the baseline intensity is unknown.

1 Introduction

As explained in the introduction of the book of Daley and Vere-Jones [32], historically the theory of point processes seems to emerge with the study of the first life tables and renewal processes, and of counting problems in the research of Poisson [87]. Since recently, point processes are largely deployed in the epidemiology, genetics, neuroscience and communications engineering literature. At the origin of this work, we were actually interested in some applications in public health and healthcare surveillance, where a point process on a bounded interval may represent occurrences of a medical event in a particular context, and in cybersecurity, where it may represent a packet or session arrival process in internet traffic or occurrences of certain cyber attacks or intrusions. In these contexts, being able to define conditions for abnormal behaviours to be detectable and to detect such anomalies as efficiently as possible is of particular importance.

Change detection in a Poisson process model Despite rather widespread debates regarding the real nature of the point process that can model observations in the above applications, the Poisson process model is the most frequently encountered in the dedicated articles, probably due to its convenient theoretical properties as well as its ability to fit the data. An abrupt change in the intensity of the Poisson process may reveal a significant health phenomenon when the process models epidemiological data (see [100] for a review), malicious activity or intrusion attempt when it models packet or session arrival processes in internet traffic (see [88], [20], [67], [106] or [99]), or a change of attack pattern when it models the occurrences of cyber attacks (see [33] and [61]). Another cybersecurity problem, considered in [97], [98] and [108], concerns communication over Poisson packet channels. In such a channel, an authorised transmitter sends packets to an authorised receiver according to a Poisson process, and a covert transmitter wishes to communicate some informations to a covert receiver on the same channel without being detected by a watchful adversary. Different models of covert transmissions have been studied by authors, treating the cases where the covert transmitter is restricted to packet insertion or where he or she can only alter the packet timing by slowing down the incoming process to a lower rate to convey the information. The question of detectability of such covert transmissions, translated as a question of detectability of a bump in the Poisson process intensity, is clearly related to the minimax testing point of view adopted here and described below.

Considering a Poisson process observed on a bounded fixed interval, we are thus interested in the problem of detecting an abrupt change in its distribution, characterised by a jump or a bump in its intensity. This problem comes within the more general framework of statistical change-point analysis. In view of the long history, going back to the 1940-1950's with the seminal works of Wald [107], Girshick and Rubin [55], Page [84], Fisher [49], and the extensive literature on change-point analysis, we can not pretend to present a comprehensive state of the art. Detailed overviews will be found in the monographs of Basseville and Nikirov [10], Carlstein et al. [21], Csörgö and Horváth [26], Brodsky and Darkhovsky [16, 15], Tartakovsky et al. [103], and a structured and annotated bibliography in the paper by Lee [72].

Statistical change-point problems can essentially be classified into two main classes, depending on whether they are formulated as on-line or off-line change-point problems.

On-line change-point analysis, also referred to as sequential analysis or disorder problems, generally deals with time sequences of random variables or stochastic processes, and aims at constructing a stopping time as close as possible to an unknown time of disorder or change in the distribution. For presentations of the most common performance measures and optimisation criteria used to this end, see for instance [69], [81] or [88], and references therein.

Off-line change-point analysis, also referred to as a posteriori change-point analysis, in fact raises two distinct questions: the one of detecting a given number of change-points or estimating the change-points number, and the one of estimating some or all the parameters of such change-points (jump locations and/or heights), once detected.

Though most of these questions can be, as explained in [83], formulated or interpreted as

single or multiple hypotheses testing problems, since they are usually all treated together, rather few attention seems to be paid to the testing performances themselves: detection rates results are not always explicitly stated and well formalised in the literature.

A nonasymptotic minimax testing point of view Our work, which focuses on the question of detecting a jump or a bump in the intensity of a Poisson process, precisely aims at proposing a nonasymptotic minimax testing set-up and a guided progressive approach to construct minimax and minimax adaptive detection procedures. It can thus also be viewed as a necessary preliminary step towards a further rigorous minimax study of multiple testing procedures designed for change-point localisation as in [83].

Let us consider a (possibly inhomogeneous) Poisson process $N = (N_t)_{t \in [0,1]}$ observed on the interval [0, 1], with intensity λ with respect to some measure Λ on [0, 1], and whose distribution is denoted by P_{λ} . As in [50] and [51], we assume that the measure Λ satisfies $d\Lambda(t) = Ldt$, where L is a positive number. Note that when L is an integer, this assumption amounts to considering the Poisson process N as L pooled i.i.d. Poisson processes with the same intensity λ , with respect to dt: L can therefore be seen as a growing number when comparisons with asymptotic existing results in other frequentist models are needed.

Depending on the intended application, and the level of knowledge on the baseline intensity of the process N it induces, the questions of detecting a jump or a bump in λ are here formulated as problems of testing the null hypothesis (H_0) " $\lambda \in S_0$ " versus the alternative (H_1) " $\lambda \in S_1$ ", where S_0 is either the set of a single known constant intensity, or the set of all constant intensities on [0, 1], and S_1 is a set of alternative intensities defined as positive piecewise constant functions, with one jump or one bump. As mentioned above, the point of view that we adopt here for our theoretical study is nonasymptotic, based on minimax criteria in accordance with the Neyman-Pearson principle. So, given a first kind error level α in (0, 1), any of our (nonrandomised) tests ϕ , with values in $\{0, 1\}$ and rejecting (H_0) when $\phi(N) = 1$, is primarily required to be of level α , that is to satisfy

$$\sup_{\lambda \in \mathcal{S}_0} P_\lambda\left(\phi(N) = 1\right) \le \alpha$$

Then, given a second kind error level β in (0,1), any of our level α tests ϕ_{α} is secondarily required to achieve, over the considered set of alternatives S_1 , the (α, β) -minimax separation rate defined as follows.

Considering the usual metric d_2 of $\mathbb{L}_2([0,1])$, and a level α test ϕ_{α} of (H_0) versus (H_1) , the β -uniform separation rate of ϕ_{α} over S_1 is defined by

$$\operatorname{SR}_{\beta}(\phi_{\alpha}, \mathcal{S}_{1}) = \inf \left\{ r > 0, \sup_{\lambda \in \mathcal{S}_{1}, \ d_{2}(\lambda, \mathcal{S}_{0}) \ge r} P_{\lambda}(\phi_{\alpha}(N) = 0) \le \beta \right\} .$$
(1)

The corresponding (α, β) -minimax separation rate over S_1 is defined by

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{1}\right) = \inf_{\phi_{\alpha}, \ \sup_{\lambda \in \mathcal{S}_{0}} P_{\lambda}\left(\phi_{\alpha}(N)=1\right) \leq \alpha} \mathrm{SR}_{\beta}\left(\phi_{\alpha}, \mathcal{S}_{1}\right) , \qquad (2)$$

where the infimum is taken over all possible nonrandomised level α tests.

A level α test ϕ_{α} is said to be β -minimax over S_1 if $SR_{\beta}(\phi_{\alpha}, S_1)$ achieves $mSR_{\alpha,\beta}(S_1)$, possibly up to a multiplicative constant depending on α and β .

These definitions due to Baraud [8] translate, in a nonasymptotic framework, the (asymptotic) minimax testing criteria that originate in Ingster's work [63, 64, 65], and that have now several variants in the literature, among them the asymptotic minimax testing with exact separation constants criteria introduced in [74].

For each choice of S_0 , several sets of alternatives S_1 are investigated, according to whether the jump or bump parameters are known or not. Following the terminology adopted since Spokoiny's paper [101], a complete minimax adaptivity study of the problem is therefore conducted: when one of the alternative parameters is unknown at least, the corresponding minimax tests are said to be minimax adaptive with respect to this unknown parameter. After determining lower bounds for the (α, β) -minimax separation rates over all these alternative sets, we construct nonasymptotic minimax and minimax adaptive detection tests. To the best of our knowledge, no such minimax results in the present Poisson process model have already been established.

Change-point detection procedures in Poisson processes models References dealing with change-point detection in a Poisson process are actually mainly dedicated to the construction of optimal on-line detection rules (see e.g. [86], [57], [18], and [11] for Bayesian approaches; [38], [39], [80] or [40, 41] and references therein for non-Bayesian approaches), or asymptotic off-line detection tests. On the one hand, a few off-line procedures are derived from the Bayesian perspective, such as the ones in [3], [91] and [90] dealing with the single change-point case, [56], [112] or [94] dealing with the multiple change-points case. On the other hand, non-Bayesian off-line procedures are numerous, due to the variety of Poisson processes convenient properties. Since the earliest procedures of Neyman and Pearson [82], Sukhatme [102], Maguire, Pearson and Wynn [78], many contributions have been made considering the exponential distribution of the homogeneous Poisson process inter-arrivals, like in [79], [109], [96], or more recently [4]. Recalling that the Poisson process N is homogeneous if and only if for every positive integer n, given $N_1 = n$, the points of the process are independent and uniformly distributed on [0, 1], any test of uniformity on [0, 1] in a density model can be directly applied conditionally to N_1 , or used as a source of inspiration to obtain a Poisson process adapted test of homogeneity. Closer to the tests we propose in the present work, many other existing tests for the single change-point problem are thus based on or inspired from the historical likelihood ratio, Cramér von-Mises or Kolmogorov-Smirnov statistics, with various weighting or other transforming strategies, as the ones of Rubin [93], Lewis [75], or Kendall and Kendall [68]. Deshaves and Picard [35, 36] study the optimality of weighted Kolmogorov-Smirnov and likelihood ratio tests in the non local asymptotic sense of Bahadur [6] and Brown [17], and their equivalence in the local asymptotic sense of Le Cam [70]. Asymptotic properties of point and interval change-point estimators deduced from these tests can be found in [2], [77], and [53] where Galeano also integrates these tests in a binary segmentation algorithm to further address multiple change-points detection. More recently, Dachian et al. [30] (see also Yang's [110] PhD thesis, and [31] and [111] where tests derived from the Bayesian perspective are proposed) and Farinetto [46] consider a single change-point detection problem in the general framework of inhomogeneous Poisson processes.

Testing procedures for close purposes in Poisson processes models On related topics, it may be worth first mentioning the foundational paper by Davies [34], whose

goodness-of-fit test is also discussed in [32]. Several procedures for testing goodnessof-fit or homogeneity of a Poisson process versus the alternative hypothesis that it has an increasing intensity have then been introduced and explored through experimental comparative studies in a series of papers by Bain et al. [7], Engelhardt et al. [42], Cohen and Sackrowitz [23], [59], [60]. Though these procedures are not initially designed to handle the change-point detection problem, they can nevertheless be applied to this end. Among them, the so-called Laplace and Z tests introduced by Cox [24] and Crown [25], whose extensions are proposed in [85], [1], and [13], stand out when they are used to detect a positive jump. Fazli and Kutoyants [48], Fazli [47], and more recently Dachian et al. [29] consider the goodness-of-fit testing problem where the null hypothesis corresponds to a given inhomogeneous Poisson process, and the alternatives correspond to single or onesided parametric Poisson processes families. The problem of testing that a point process is a given homogeneous Poisson process versus it belongs to a stationary self-exciting or stress-release point processes family is treated in [27] and [28].

Related minimax studies Focusing now on the minimax point of view, one can cite Ingster and Kutoyants [66] and Fromont et al.'s studies of goodness-of-fit or homogeneity tests, where the alternative hypotheses [50], corresponding to Poisson processes with non-parametric intensities in Sobolev and Besov spaces with known and unknown smoothness parameters respectively, are however not suited for change-points detection problems.

To find minimax tests devoted to change-points detection problems in the existing literature, it is actually needed to switch to other statistical models.

Using the conditioning trick explained above, which enables to treat change-points detection problems in the Poisson model as particular change-points detection problems in the classical density model, Rivera and Walther [92] propose two positive bump detection tests based on scan or average aggregation of likelihood ratio statistics. Though their optimality results are not directly transposable to the minimax set-up that we consider here due to deconditioning difficulties, they nevertheless give preview of possible approaches towards more Poisson processes-specific minimax tests. In the classical density model, Dümbgen and Walther [37] had already tackled the problem of detecting local increases and decreases of the density or the failure rate. The introduced procedures, based on aggregation of local order statistics and spacings, were proved to satisfy asymptotic minimax adaptation properties.

Of course, the most complete bibliography on jump or bump detection from the minimax testing point of view lies in the basic Gaussian framework, where the observation is modelled by a Gaussian vector $Y = (Y_1, \ldots, Y_n)$ with variance $\sigma^2 I_n$. Arias-Castro et al. [5] first studied the minimax separation rate for the problem of detecting a bump, that is a change in mean from zero over an interval, when considering the ℓ_2 -metric on the mean vectors, related to the \mathbb{L}_2 -distance between the corresponding Gaussian distributions and also to the signal to noise ratio or signal energy often mentioned in regression models analysis. When the height and the length of the change are unknown, they exhibited a minimax separation rate of order $\sqrt{\log n}$ with an exact constant equal to $\sqrt{2}$. In other words, they proved that no test can reliably detect Y such that $\mathbb{E}[Y_i] = \delta \mathbb{1}_{\{i \in [\tau, \tau+\ell]\}}$ (with $\tau \in \{1, \ldots, n\}, \ell \in \{1, \ldots, n+1-\tau\}$) unless the condition $|\delta|\sqrt{\ell} \ge \sqrt{2(1+\eta)\log n}$ with $\eta > 0$ is satisfied, and they introduced minimax adaptive tests based on a scan aggregation of the Neyman-Pearson test statistics $\ell^{-1/2} \sum_{i=\tau}^{\tau+\ell-1} Y_i$ designed to detect non-zero mean, either over all the possible intervals $[\tau, \tau + \ell]$ or over intervals of dyadic type

 $[k2^{j}, (k+1)2^{j}]$. Chan and Walther [22] constructed three other tests, based on the same Neyman-Pearson test statistics, but combined according to different aggregation schemes. All these tests were proved to be consistent as soon as the refined condition $|\delta|\sqrt{\ell} \geq \sqrt{2\log(n/\ell)} + b_n$ with $b_n \to +\infty$ holds, which slightly improves Arias-Castro et al.'s lower bound at least when $\ell/n := \ell_n/n$ is allowed to tend to 0 with n tending to $+\infty$. A nonasymptotic counterpart of this improved lower bound has been very recently provided by Verzelen et al. [105]. But the procedures introduced in this work go beyond the scope of the present minimax testing study as they further address the twin problems of detecting and localising multiple change-points. In the case where the change height is known, equal to 1, Brunel [19] constructed a test based on a scanning of the shifted test statistic $\sum_{i=\tau}^{\tau+\ell-1} Y_i - \ell/2$, which is consistent as soon as $\ell/\log n \to +\infty$. Still considering the ℓ_2 -metric on the mean vectors, but considering, among piecewise monotone signals estimation problems, the special problem of detecting a jump from an unknown constant mean, Gao et al. [54] obtained a lower bound of order $\sqrt{\log \log n}$. More precisely, they proved that no test can reliably detect Y such that $\mathbb{E}[Y_i] = \delta \mathbb{1}_{\{i \in [\tau,n]\}}$ (with $\tau \in \{2, \ldots, n\}$) unless $|\delta| \sqrt{(\tau - 1)(n + 1 - \tau)/n} \ge c \sqrt{\log \log(16n)}$. Verzelen et al. [105] provide a nonasymptotic lower bound equal to $\sqrt{2(1-c)(1-n^{-1/2})\log\log n}$ for c in (0, 2/3) and n large enough. As for a corresponding upper bound, Gao et al. [54] refer to the asymptotic test of Csörgö and Horváth [26], based on the scan statistic $\max_{1 \le \tau \le n} \sqrt{n/(\tau(n-\tau))} \sum_{i=1}^{\tau} Y_i - (k/n) \sum_{i=1}^{n} Y_i$, which is proved to be optimal from asymptotic inequalities in the spirit of the Iterated Logarithm Law. Notice that this scan statistic is closely related to the well-known CUSUM statistics, which have a long history in the single change-point analysis literature from Hinkley's [58] work, as well as in multiple change-points analysis references, where they are at the core of binary segmentation approaches. Verzelen et al. [105] introduce a test based on a max penalized CUSUM statistic, with location-dependent penalties, whose separation rate is of the optimal order $\sqrt{2 \log \log n}$ (thus proving, combined with their lower bound, that the exact constant is $\sqrt{2}$ as in the bump detection case), with possible refinement when restricting to particular change locations.

In more complex Gaussian models, with sparse high dimensional, heterogeneous or dependence properties, it is worth mentioning at least the work of Enikeeva and Harchaoui [43], Enikeeva et al. [45], Liu et al. [76] and Enikeeva et al. [44], addressing bump detection problems from asymptotic minimax points of view, that are quite close to the one we adopt here.

Notice that we do not tackle the problem of detecting multiple change-points with more than two change-points, nor the problem of localising change-points, that we consider as out of the scope of the present paper and a basis for future work.

Our contribution The present work addresses the question of detecting a jump or a bump in the intensity of a Poisson process from the nonasymptotic minimax point of view described above. At this end, we will determine the minimax separation rates that correspond to :

- the detection of a change from a known or an unknown constant baseline intensity,
- the detection of a non transitory change formalised as a jump in the intensity, or a transitory change formalised as a bump in the intensity,

- the detection of a change with known or unknown height, length and location.

We will thus provide a comprehensive overview of the various minimax separation rate regimes, with a special focus on the phase transitions, and their determining factors. Considering each parameter as known or unknown, one by one, indeed enables us to precisely identify what causes such phase transitions, and the precise cost of minimax adaptation to each of these parameters. Among the main results of this study, we find a phase transition from a $\sqrt{\log \log L/L}$ order minimax separation rate for jump detection to a $\sqrt{\log L/L}$ order minimax separation rate for bump detection, when the jump or bump height, the jump or bump location and the bump length are together unknown: this phase transition is similar to the Gaussian one. But we also exhibit minimax separation rates that are not even known in the basic Gaussian model, up to our knowledge.

For the bump detection problem, we indeed prove that the minimax separation rate is of order $\sqrt{\log L/L}$ when both location and length of the bump are unknown, whether the height is known or not, of order $\sqrt{\log \log L/L}$ as in the jump detection problem when the only location of the bump is known (with height and length unknown), of order $\sqrt{1/L}$ in the other cases.

For the jump detection problem, the results could be more easily anticipated: we prove that the minimax separation rate is of order $\sqrt{\log \log L/L}$ when both height and location of the jump are unknown, as in the Gaussian model, and of order $\sqrt{1/L}$ in the other cases.

Such minimax separation rates are as usual obtained in two steps. Lower bounds are first deduced from classical Bayesian arguments, originating from Le Cam's theory, and clearly outlined by Ingster [63, 64, 65] and Baraud [8] in an asymptotic and a nonasymptotic framework respectively. Combined with these Bayesian arguments, the Poisson processes properties, and mainly Girsanov's Lemma, are key points of the proofs. Then, matching upper bounds are derived from the construction of minimax or minimax adaptive tests. The tests that we propose are based on either linear statistics adapted from the Neyman-Pearson test in the case where all the bump or jump parameters are known, and close to the CUSUM like statistics used in the Gaussian framework, or quadratic statistics that we felt better suitable for the estimation of the distance d_2 , considered here, between λ and \mathcal{S}_0 . The links between these two types of statistics are discussed. This discussion raises new interesting questions about general aggregation of bilateral tests, that are furthermore supported by our short simulation study. Minimax adaptation when some change parameters are unknown is obtained from scan aggregation approaches, that all differ depending on which parameters are unknown. The critical values involved in the scan aggregation approaches are also differently adjusted, with an additional crucial conditioning trick already used in [50] when the baseline intensity is unknown, to lead to a nonasymptotic level α and nonasymptotic minimax optimality. Upper bounding these critical values often was the main and most difficult point of the proofs. We had to use a wide variety of exponential and concentration inequalities, from historical ones due to [89] to very recent ones due to Le Guével [71] which are specific to suprema of counting processes and their related square martingales when dealing with detection of a change from a known baseline intensity, plus exponential inequalities for suprema or oscillations of empirical processes and U-statistics due to Mason, Shorack and Wellner (see [95]) and Houdré and Reynaud-Bouret [62], or obtained from Bernstein and Bennett's inequalities as stated in [12], refined through combination with chaining techniques.

Organisation of the paper Section 2 of the paper is devoted to the problem of detecting a change from a known baseline intensity, while Section 3 deals with the problem of detecting a change from an unknown intensity. For each problem, all the possible sets of alternatives according to whether each parameter of the change (height, location and length) is known or not, including the special case where the change is non transitory (jump detection), are handled. And for each of the resulting ten sets of alternatives, lower bounds for minimax separation rates are provided, as a preliminary basis for corresponding upper bounds (when appropriate, that is when at least the height or the length is unknown). As explained above, these upper bounds are obtained by constructing minimax or minimax adaptive tests, which are mainly based on aggregation of either linear or quadratic statistics, coupled with adjusted critical values. A short simulation study is presented in Section 4, whose aim is to compare linear and quadratic type tests, and also to compare them with standard tests used to detect inhomogeneity of Poisson processes in practice. Proofs of the core results are postponed to Section 5, and proofs of technical results mostly based on exponential inequalities and devoted to quantiles and critical values upper bounds are postponed to Section 6, which also contains fundamental and general results for lower bounds.

Notation Concerning the Poisson Process $N = (N_t)_{t \in [0,1]}$, we use the notation $N(\tau_1, \tau_2]$ for $N_{\tau_2} - N_{\tau_1}$ for every τ_1, τ_2 in [0, 1]. As usual, dN stands for the point measure associated with N, and E_{λ} and $\operatorname{Var}_{\lambda}$ respectively stand for the expectation and the variance under P_{λ} , that is when N has λ as intensity with respect to $d\Lambda(t) = Ldt$. The distance d_2 has been introduced above, and associated with this distance, we consider the usual scalar product and norm of $\mathbb{L}_2([0,1])$ respectively denoted by $\langle ., . \rangle_2$ and $\|.\|_2$. For all x and y in $\mathbb{R}, x \vee y$ (resp. $x \wedge y$) denotes the maximum (resp. minimum) between x and y, and the sign function sgn is defined by $\operatorname{sgn}(x) = \mathbbm{1}_{x>0} - \mathbbm{1}_{x<0}$.

All along the article, we will introduce some positive constants denoted by $C(\alpha, \beta, ...)$ and $L_0(\alpha, \beta, ...)$, meaning that they may depend on $(\alpha, \beta, ...)$. Though they are denoted in the same way, they may vary from one line to another. When they appear in the main results about lower and upper bounds, we do not intend to precisely evaluate them. However, some possible, probably pessimistic, explicit expressions for them are proposed in the proofs.

2 Detecting an abrupt, possibly transitory, change in a known baseline intensity

As a first step of work, and because this also addresses particular applications, we are here interested in the problem of detecting an abrupt change in the intensity of the Poisson process N, when its baseline is assumed to be known, equal to a positive constant function λ_0 on [0, 1]. For the sake of simplicity, the constant function λ_0 and its value on [0, 1] are often confused in the following. The null hypothesis of the present section can therefore be expressed as (H_0) " $\lambda \in S_0[\lambda_0] = {\lambda_0}$ ", while the alternative hypothesis varies according to the height, length and location of the intensity jump or bump knowledge.

In order to further cover the full range of alternatives in a unified notation, we introduce for δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$, τ^* in (0, 1), ℓ^* in $(0, 1 - \tau^*]$ the set $S_{\delta^*, \tau^*, \ell^*}[\lambda_0]$ of intensities with a change of height δ^* , location τ^* and length ℓ^* from λ_0 :

$$[Alt.1] \ \mathcal{S}_{\delta^*,\tau^*,\ell^*}[\lambda_0] = \{\lambda : [0,1] \to (0,+\infty), \ \forall t \in [0,1] \ \lambda(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau^*,\tau^*+\ell^*]}(t)\} \ . (3)$$

Testing (H_0) versus (H_1) " $\lambda \in S_{\delta^*, \tau^*, \ell^*}[\lambda_0]$ " falls within the scope of the Neyman-Pearson fundamental lemma and an Uniformly Most Powerful (UMP) test exists, thus achieving the minimax separation rate over $S_{\delta^*, \tau^*, \ell^*}[\lambda_0]$. Details are provided below.

Then, when the question of adaptivity with respect to unknown parameters is tackled, the unknown parameters are replaced by single, double or triple dots in the notation $S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$.

Notice that for any alternative intensity $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell]}$ with δ in $(-\lambda_0, +\infty) \setminus \{0\}$, τ in (0, 1), and ℓ in $(0, 1 - \tau]$, $d_2(\lambda, S_0[\lambda_0]) = |\delta|\sqrt{\ell}$. Hence, as soon as λ has a known change height $\delta = \delta^*$ and a known change length $\ell = \ell^*$, the distance $d_2(\lambda, S_0[\lambda_0])$ is fixed, equal to $|\delta^*|\sqrt{\ell^*}$. The β -uniform separation rate of any level α test over $S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ or $S_{\delta^*,\dots,\ell^*}[\lambda_0]$ as defined by (1) is therefore either 0 or $+\infty$ (with the usual convention inf $\emptyset = +\infty$), as well as the minimax separation rate. In these only two cases, studying our tests from the minimax point of view would have no sense. Nevertheless, once having ensured that their first kind error rate is at most α , in order to follow the same line as the minimax results obtained in the other cases, we establish conditions expressed as a sufficient minimal distance $d_2(\lambda, S_0[\lambda_0])$, guaranteeing that their second kind error rate is at most equal to some prescribed level β .

2.1 Uniformly most powerful detection of a possibly transitory change with known location and length

Let us now give more details about the above problem of testing the simple null hypothesis (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus the simple alternative hypothesis (H_1) " $\lambda \in S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ " with $S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ defined by (3). Notice that for any λ in $S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$, then

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) = |\delta^*| \sqrt{\ell^*}$$

Given α in (0, 1), Neyman-Pearson tests of (H_0) versus (H_1) " $\lambda \in \mathcal{S}_{\delta^*, \tau^*, \ell^*}[\lambda_0]$ " of size α can be constructed. To this end, we recall Girsanov's lemma (see [14] for a proof).

Lemma 1 (Girsanov). Let $N = (N_t)_{t \in [0,1]}$ be an inhomogeneous Poisson process with jump locations $(X_j)_{j\geq 1}$, with bounded intensity λ with respect to some measure Λ on [0,1], and with distribution denoted by P_{λ} under the probability \mathbb{P} . Assume that λ_0 is a bounded nonnegative function such that for every $j \geq 1$, $\lambda_0(X_j) > 0$ \mathbb{P} -almost surely. Then

$$\frac{dP_{\lambda}}{dP_{\lambda_0}}(N) = \exp\left[\int_0^1 \ln\left(\frac{\lambda(t)}{\lambda_0(t)}\right) dN_t - \int_0^1 (\lambda(t) - \lambda_0(t)) d\Lambda_t\right] .$$

From this fundamental lemma, we deduce the likelihood ratio, for λ in $\mathcal{S}_{\delta^*,\tau^*,\ell^*}[\lambda_0]$,

$$\frac{dP_{\lambda}}{dP_{\lambda_0}}(N) = \exp\left[\ln\left(1 + \frac{\delta^*}{\lambda_0}\right)N(\tau^*, \tau^* + \ell^*] - \delta^*\ell^*L\right] \quad , \tag{4}$$

which leads to the following size α Neyman-Pearson tests:

$$\begin{cases} \phi_{1,\alpha}^{-}(N) = \mathbb{1}_{N(\tau^{*},\tau^{*}+\ell^{*}] < p_{\lambda_{0}\ell^{*}L}(\alpha)} + \gamma^{-}(\alpha) \mathbb{1}_{N(\tau^{*},\tau^{*}+\ell^{*}] = p_{\lambda_{0}\ell^{*}L}(\alpha)} & \text{if } \delta^{*} < 0 , \\ \phi_{1,\alpha}^{+}(N) = \mathbb{1}_{N(\tau^{*},\tau^{*}+\ell^{*}] > p_{\lambda_{0}\ell^{*}L}(1-\alpha)} + \gamma^{+}(1-\alpha) \mathbb{1}_{N(\tau^{*},\tau^{*}+\ell^{*}] = p_{\lambda_{0}\ell^{*}L}(1-\alpha)} & \text{if } \delta^{*} > 0 , \end{cases}$$
(5)

where $p_{\xi}(u)$ denotes the *u*-quantile of the Poisson distribution with parameter ξ , and

$$\gamma^{-}(u) = \frac{u - P_{\lambda_0}(N(\tau^*, \tau^* + \ell^*] < p_{\lambda_0}\ell^*L(u))}{P_{\lambda_0}(N(\tau^*, \tau^* + \ell^*] = p_{\lambda_0\ell^*L}(u))}, \quad \gamma^{+}(u) = 1 - \gamma^{-}(u) \quad .$$
(6)

Proposition 2 (Second kind error rates control for [Alt.1]). Let $L \ge 1$, α and β in $(0,1), \lambda_0 > 0, \delta^*$ in $(-\lambda_0, +\infty) \setminus \{0\}, \tau^*$ in (0,1) and ℓ^* in $(0,1-\tau^*]$. Considering the problem of testing (H_0) versus (H_1) " $\lambda \in S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ ", let $\phi_{1,\alpha}$ be the test $\phi_{1,\alpha}^-$ if $\delta^* < 0$, $\phi_{1,\alpha}^+$ if $\delta^* > 0$ (see (5) for definitions of the tests). The test $\phi_{1,\alpha}$ is a UMP test of size α . Moreover, $P_{\lambda}(\phi_{1,\alpha}(N) = 0) \le \beta$ as soon as λ belongs to $S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ with

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \left(\sqrt{(\lambda_0 + \delta^*)/\beta} + \sqrt{\lambda_0/\alpha}\right)/\sqrt{L} \quad .$$
(7)

Comments. Notice first that the same result holds with $P_{\lambda}(\phi_{1,\alpha}(N) = 0)$ replaced by the second kind error rate $E_{\lambda}[1 - \phi_{1,\alpha}(N)]$. Then, as explained above, studying the present test from the minimax point is not really relevant. One can however notice that the uniform separation rate of a UMP test necessarily provides the minimax separation rate over any set of alternatives. Since for λ in $S_{\delta^*,\tau^*,\ell^*}[\lambda_0], d_2(\lambda, S_0[\lambda_0]) = |\delta^*|\sqrt{\ell^*}$, the above proposition implies that if $L \geq (\sqrt{(\lambda_0 + \delta^*)/\beta} + \sqrt{\lambda_0/\alpha})^2/(\delta^{*2}\ell^*)$, then $P_{\lambda}(\phi_{1,\alpha}(N) = 0) \leq \beta$. Therefore, in this case, the β -uniform separation rate of $\phi_{1,\alpha}$ over $S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ is equal to 0, and consequently, the (α, β) -minimax separation rate mSR_{α,β} $(S_{\delta^*,\tau^*,\ell^*}[\lambda_0])$.

To this end, we introduce, for τ^* in (0,1) and ℓ^* in $(0,1-\tau^*]$ the set

$$\begin{bmatrix} \mathbf{Alt.2} \end{bmatrix} \quad \mathcal{S}_{\cdot,\tau^*,\ell^*}[\lambda_0] = \{\lambda : [0,1] \to (0,+\infty), \ \exists \delta \in (-\lambda_0,+\infty) \setminus \{0\}, \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^*+\ell^*]}(t) \} \ , \quad (8)$$

and we consider the problem of testing (H_0) versus (H_1) " $\lambda \in S_{,\tau^*,\ell^*}[\lambda_0]$ ".

The following result gives a nonasymptotic lower bound for the (α, β) -minimax separation rate over the set of alternatives $S_{,\tau^*,\ell^*}[\lambda_0]$ of the parametric order $L^{1/2}$, which is obtained from a now classical Bayesian approach that originates in Le Cam's theory and Ingster's work [65] in an asymptotic perspective, and that has been next adapted to the nonasymptotic perspective by Baraud [8]. For the sake of clarity and completeness, the main points of this approach are recalled in Section 6.1, and the complete proof can be found in Section 5.

Proposition 3 (Minimax lower bound for [Alt.2]). Let α, β in (0, 1) such that $\alpha + \beta < 1$, $\lambda_0 > 0, \tau^*$ in (0, 1) and ℓ^* in $(0, 1 - \tau^*]$. For all $L \ge 1$, the following lower bound holds:

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0]\right) \ge \sqrt{\lambda_0 \log C_{\alpha,\beta}/L}, \text{ with } C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2.$$

In order to prove that this lower bound is sharp with respect to L, we introduce a first bilateral test based on the counting statistic $N(\tau^*, \tau^* + \ell^*]$ of the above Neyman-Pearson tests. So let

$$\phi_{2,\alpha}^{(1)}(N) = \mathbb{1}_{\left\{N(\tau^*,\tau^*+\ell^*] > p_{\lambda_0\ell^*L}(1-\alpha_1)\right\}} + \gamma^+(1-\alpha_1)\mathbb{1}_{\left\{N(\tau^*,\tau^*+\ell^*] = p_{\lambda_0\ell^*L}(1-\alpha_1)\right\}} \\ + \mathbb{1}_{\left\{N(\tau^*,\tau^*+\ell^*] < p_{\lambda_0\ell^*L}(\alpha_2)\right\}} + \gamma^-(\alpha_2)\mathbb{1}_{\left\{N(\tau^*,\tau^*+\ell^*] = p_{\lambda_0\ell^*L}(\alpha_2)\right\}} , \quad (9)$$

where α_1 and α_2 in (0, 1) are determined by

$$\alpha_1 + \alpha_2 = \alpha$$
 and $E_{\lambda_0}[N(\tau^*, \tau^* + \ell^*]\phi_{2,\alpha}^{(1)}(N)] = \alpha E_{\lambda_0}[N(\tau^*, \tau^* + \ell^*)]$. (10)

Since our testing problem amounts to a problem of testing " $\delta = 0$ " versus " $\delta \neq 0$ " in the exponential model $dP_{\lambda}/dP_{\lambda_0}(N) = \exp\left[\ln\left(1 + \delta/\lambda_0\right)N(\tau^*, \tau^* + \ell^*] - \delta\ell^*L\right]$, applying the result of Chapter 4.2 in [73] allows to see that $\phi_{2,\alpha}^{(1)}$ is an Uniformly Most Powerful Unbiased (UMPU) test of size α . We also prove that it is β -minimax (up to a possible multiplicative constant) over the set of alternatives $S_{,\tau^*,\ell^*}[\lambda_0]$ with the following result.

Proposition 4 (Minimax upper bound for [Alt.2]). Let $L \ge 1$, α, β in (0, 1), $\lambda_0 > 0$, τ^* in (0, 1) and ℓ^* in $(0, 1 - \tau^*]$. The test $\phi_{2,\alpha}^{(1)}$ of (H_0) versus (H_1) " $\lambda \in S_{,\tau^*,\ell^*}[\lambda_0]$ " defined by (9)-(10) is of level α , that is $P_{\lambda_0}(\phi_{2,\alpha}^{(1)}(N) = 1) \le \alpha$ (it is even of size α , that is $E_{\lambda_0}[\phi_{2,\alpha}^{(1)}(N)] = \alpha$). Moreover, there exists $C(\alpha, \beta, \lambda_0, \tau^*, \ell^*) > 0$ such that

$$\operatorname{SR}_{\beta}(\phi_{2,\alpha}^{(1)}, \mathcal{S}_{\tau^{*},\ell^{*}}[\lambda_{0}]) \leq C(\alpha, \beta, \lambda_{0}, \ell^{*})/\sqrt{L}$$
,

which entails in particular $\operatorname{mSR}_{\alpha,\beta}(\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0]) \leq C(\alpha,\beta,\lambda_0,\ell^*)/\sqrt{L}.$

It is interesting to notice that the test $\phi_{2,\alpha}^{(1)}$ can also be written as

$$\phi_{2,\alpha}^{(1)}(N) = \mathbb{1}_{\left\{\bar{N}(\tau^*,\tau^*+\ell^*] > \bar{p}_{\lambda_0\ell^*L}(1-\alpha_1)\right\}} + \gamma^+(1-\alpha_1)\mathbb{1}_{\left\{\bar{N}(\tau^*,\tau^*+\ell^*] = \bar{p}_{\lambda_0\ell^*L}(1-\alpha_1)\right\}} \\ + \mathbb{1}_{\left\{\bar{N}(\tau^*,\tau^*+\ell^*] < \bar{p}_{\lambda_0\ell^*L}(\alpha_2)\right\}} + \gamma^-(\alpha_2)\mathbb{1}_{\left\{\bar{N}(\tau^*,\tau^*+\ell^*] = \bar{p}_{\lambda_0\ell^*L}(\alpha_2)\right\}} , \quad (11)$$

where $\bar{N}(\tau^*, \tau^* + \ell^*] = N(\tau^*, \tau^* + \ell^*] - \lambda_0 \ell^* L$ and $\bar{p}_{\lambda_0 \ell^* L}(u)$ is the *u*-quantile of this recentered Poisson variable under (H_0) . The statistic $\bar{N}(\tau^*, \tau^* + \ell^*]$ being an unbiased estimator of $L\langle\lambda - \lambda_0, \mathbb{1}_{(\tau^*, \tau^* + \ell^*]}\rangle_2$ when λ belongs to $\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0]$, it is intuitively natural to reject (H_0) " $\lambda = \lambda_0$ " when it is too small or too large, with critical values determining what "too small" or "too large" means based on the quantiles of $\bar{N}(\tau^*, \tau^* + \ell^*]$ under (H_0) .

Since the distance considered in our minimax criteria is the L₂-distance d_2 , a more natural idea here is to construct a test based on a statistic estimating $d_2(\lambda, \lambda_0)$ or $d_2^2(\lambda, \lambda_0)$ when $\lambda \in S_{,\tau^*,\ell^*}[\lambda_0]$. Such an unbiased estimator of $d_2^2(\lambda, \lambda_0)$ is in fact given by the quadratic statistic $T_{\tau^*,\tau^*+\ell^*}(N)$, where for τ_1, τ_2 such that $0 \leq \tau_1 < \tau_2 \leq 1$,

$$T_{\tau_1,\tau_2}(N) = \frac{1}{L^2(\tau_2 - \tau_1)} \left(N\left(\tau_1, \tau_2\right)^2 - (1 + 2\lambda_0 L(\tau_2 - \tau_1)) N\left(\tau_1, \tau_2\right] + \lambda_0^2 L^2(\tau_2 - \tau_1)^2 \right)$$
(12)

This leads us to consider a second minimax test, defined by

$$\phi_{2,\alpha}^{(2)}(N) = \mathbb{1}_{T_{\tau^*,\tau^*+\ell^*}(N) > t_{\lambda_0,\tau^*,\tau^*+\ell^*}(1-\alpha)} , \qquad (13)$$

where $t_{\lambda_0,\tau_1,\tau_2}(u)$ denotes the *u*-quantile of the distribution of $T_{\tau_1,\tau_2}(N)$ under (H_0) .

Proposition 5 (Alternate minimax upper bound for [Alt.2]). Let $L \ge 1$, α, β in (0, 1), $\lambda_0 > 0$, τ^* in (0, 1) and ℓ^* in $(0, 1 - \tau^*]$. The test $\phi_{2,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{,\tau^*,\ell^*}[\lambda_0]$ " defined by (13) is of level α , that is $P_{\lambda_0}(\phi_{2,\alpha}^{(2)}(N) = 1) \le \alpha$, and there exists $C(\alpha, \beta, \lambda_0, \tau^*, \ell^*) > 0$ such that $\mathrm{SR}_{\beta}(\phi_{2,\alpha}^{(2)}, S_{,\tau^*,\ell^*}[\lambda_0]) \le C(\alpha, \beta, \lambda_0, \ell^*)/\sqrt{L}$.

Comments. Both tests $\phi_{2,\alpha}^{(1)}$ and $\phi_{2,\alpha}^{(2)}$ are therefore β -minimax over the set of alternatives $S_{\cdot,\tau^*,\ell^*}[\lambda_0]$, where the change height of the change is unknown, with an optimal uniform separation rate of the parametric order $1/\sqrt{L}$. Despite the bilateral form of the test $\phi_{2,\alpha}^{(1)}$ and the apparent unilateral form of $\phi_{2,\alpha}^{(2)}$, these tests have very close links that we detail below.

Notice that the present study involves the particular non transitory change or jump detection problem, with a known change location, taking $\ell^* = 1 - \tau^*$. The study of the general non transitory change or jump detection problem (of unknown location) is conducted in Section 2.5.1 as a particular case of change detection problem with unknown location and length.

2.2 Choice of the test statistics and links between the corresponding procedures

Let us focus on the test $\phi_{2,\alpha}^{(2)}$ and contrast it to the UMPU test $\phi_{2,\alpha}^{(1)}$. To this end, we notice that

$$\phi_{2,\alpha}^{(2)}(N) = \mathbb{1}_{N(\tau^*,\tau^*+\ell^*]-\lambda_0\ell^*L > \frac{1}{2}\left(1+\sqrt{1+4\lambda_0\ell^*L+4\ell^*L^2t_{\lambda_0,\tau^*,\tau^*+\ell^*}(1-\alpha)}\right)} \\ + \mathbb{1}_{N(\tau^*,\tau^*+\ell^*]-\lambda_0L\ell^* < \frac{1}{2}\left(1-\sqrt{1+4\lambda_0\ell^*L+4\ell^*L^2t_{\lambda_0,\tau^*,\tau^*+\ell^*}(1-\alpha)}\right)}$$

The test $\phi_{2,\alpha}^{(2)}$ can therefore be related to the test $\phi_{2,\alpha}^{(1)}$ expressed with the recentered statistic $\overline{N}(\tau^*, \tau^* + \ell^*]$ as in (11), with the only difference that its critical values are differently chosen. The choice made for $\phi_{2,\alpha}^{(2)}$ has two main consequences: a negative one, which is that the test is not UMPU contrary to $\phi_{2,\alpha}^{(1)}$, and a positive one which is that it does not involve calibration of two levels α_1 and α_2 as in $\phi_{2,\alpha}^{(1)}$ where they have to satisfy (10), equivalent to

$$E_{\lambda_{0}}\left[\left(\bar{N}(\tau^{*},\tau^{*}+\ell^{*}]-\bar{p}_{\lambda_{0}\ell^{*}L}(1-\alpha_{1})\right)\mathbb{1}_{\left\{\bar{N}(\tau^{*},\tau^{*}+\ell^{*}]-\bar{p}_{\lambda_{0}\ell^{*}L}(1-\alpha_{1})>0\right\}}\right] + E_{\lambda_{0}}\left[\left(\bar{N}(\tau^{*},\tau^{*}+\ell^{*}]-\bar{p}_{\lambda_{0}\ell^{*}L}(\alpha_{2})\right)\mathbb{1}_{\left\{\bar{N}(\tau^{*},\tau^{*}+\ell^{*}]-\bar{p}_{\lambda_{0}\ell^{*}L}(\alpha_{2})<0\right\}}\right] + \alpha_{1}\bar{p}_{\lambda_{0}\ell^{*}L}(1-\alpha_{1}) + \alpha_{2}\bar{p}_{\lambda_{0}\ell^{*}L}(\alpha_{2}) = 0$$

This calibration, which is already not so easy to execute when using such single bilateral tests, leads to an additional difficulty when considering their aggregation. Correcting the individual levels of some tests of the form of $\phi_{2,\alpha}^{(2)}$ when they are aggregated is quite a classical question, that can be solved by using principles of Bonferroni or min-*p* multiple tests. This question becomes particularly tricky for tests of the form of $\phi_{2,\alpha}^{(1)}$: the arbitrary choice of $\alpha_1 = \alpha_2$ makes the aggregation more convenient in the following, at the price to slightly degrade the final power of the aggregated test, especially when the distribution of the involved recentered statistics $\bar{N}(\tau_1, \tau_2]$ is not symmetric with respect to 0. More details are given in Section 2.6.

This last consequence, together with the fact that the test statistic $T_{\tau_1,\tau_2}(N)$ is an unbiased estimator of $\|\Pi_{V_{\tau_1,\tau_2}}(\lambda-\lambda_0)\|_2^2$ where $V_{\tau_1,\tau_2} = \operatorname{Vect}\left(\varphi_{(\tau_1,\tau_2]} = \mathbb{1}_{(\tau_1,\tau_2]}/\sqrt{\tau_2-\tau_1}\right)$ and $\Pi_{V_{\tau_1,\tau_2}}$ denotes the orthogonal projection onto V_{τ_1,τ_2} in $\mathbb{L}_2([0,1])$, which can be a starting point to generalise the present detection tests to more complex change-point detection problems (for instance in nonconstant baseline intensities), has led us to keep both types of tests all along our study whenever possible. Moreover, the tools used to obtain the minimax results for both tests are very different, hence the proofs for the two tests can be viewed as real alternate proofs of minimax separation rates upper bounds.

2.3 Minimax detection of a transitory change with known length

The present subsection is dedicated to the problem of testing (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus alternatives where the length of the change from the baseline intensity is known, with adaptation with respect to the change location, and with or without adaptation with respect to the height of the change. We therefore introduce, for ℓ^* in (0, 1) and δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$, the two following sets:

$$\begin{bmatrix} \mathbf{Alt.3} \end{bmatrix} \mathcal{S}_{\delta^*, \dots, \ell^*}[\lambda_0] = \left\{ \lambda : [0, 1] \to (0, +\infty), \ \exists \tau \in (0, 1 - \ell^*), \\ \forall t \in [0, 1] \quad \lambda(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau, \tau + \ell^*]}(t) \right\} , \quad (14)$$

$$[\mathbf{Alt.4}] \ \mathcal{S}_{,\cdot,\ell^*}[\lambda_0] = \left\{ \lambda \ [0,1] \to (0,+\infty), \ \exists \delta \in (-\lambda_0,+\infty) \setminus \{0\}, \ \exists \tau \in (0,1-\ell^*), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell^*]}(t) \right\} .$$
(15)

As seen in the above subsection, the knowledge of the change height δ^* is not necessary to construct an UMP test of (H_0) versus (H_1) " $\lambda \in S_{\delta^*, \tau^*, \ell^*}[\lambda_0]$ " as the test statistic, which is the exhaustive statistic in the considered exponential model, does not depend on the value of δ^* . This enables to directly extend it to an UMPU test of (H_0) versus (H_1) " $\lambda \in S_{\tau^*, \ell^*}[\lambda_0]$ " based on the same exhaustive statistic $N(\tau^*, \tau^* + \ell^*]$.

The only significant question is hence the one of adaptation to the change location τ^* . A natural approach to handle this question is to take the same linear and quadratic statistics as the ones used for testing (H_0) versus (H_1) " $\lambda \in S_{\delta^*,\tau^*,\ell^*}[\lambda_0]$ " or (H_1) " $\lambda \in S_{\cdot,\tau^*,\ell^*}[\lambda_0]$ ", but making τ^* varying in the whole set of possible change locations, or an appropriate restricted set of possible change locations. This approach, known as statistics scanning in the signal and image processing literature or statistics aggregation in the minimax testing literature, has close connections with multiple tests that were investigated in [52] (see Section 2.6), and that will be exploited in a further work dedicated to the change localisation problem.

We therefore first introduce the following linear statistic based aggregated tests:

$$\begin{cases} \phi_{3,\alpha}^{(1)-}(N) = \mathbb{1}_{\left\{\min_{\tau \in [0,1-\ell^*]} N(\tau,\tau+\ell^*] < p_{\lambda_0,\ell^*}^-(\alpha)\right\}}, \\ \phi_{3,\alpha}^{(1)+}(N) = \mathbb{1}_{\left\{\max_{\tau \in [0,1-\ell^*]} N(\tau,\tau+\ell^*] > p_{\lambda_0,\ell^*}^+(1-\alpha)\right\}}, \end{cases}$$
(16)

where $p_{\lambda_0,\ell^*}^-(u)$ and $p_{\lambda_0,\ell^*}^+(u)$ respectively denote the *u*-quantiles of the distributions of $\min_{\tau \in [0,1-\ell^*]} N(\tau,\tau+\ell^*]$ and $\max_{\tau \in [0,1-\ell^*]} N(\tau,\tau+\ell^*]$ under (H_0) . From these unilateral tests, we construct the bilateral test

$$\phi_{4,\alpha}^{(1)}(N) = \phi_{3,\alpha/2}^{(1)-}(N) \lor \phi_{3,\alpha/2}^{(1)+}(N) \quad , \tag{17}$$

intended to address the change height adaptation issue.

Finally, considering $M = \lceil 2/\ell^* \rceil$ and $u_{\alpha} = \alpha/\lceil (1-\ell^*)M \rceil$, the test statistic $T_{k/M,k/M+\ell^*}(N)$ defined by (12) and its *u*-quantile $t_{\lambda_0,k/M,k/M+\ell^*}(u)$ under (H_0) , we introduce the quadratic statistic based aggregated test

$$\phi_{3/4,\alpha}^{(2)}(N) = \mathbb{1}_{\left\{\max_{k \in \{0,\dots,\lceil (1-\ell^*)M\rceil - 1\}} \left(T_{\frac{k}{M},\frac{k}{M}+\ell^*}(N) - t_{\lambda_0,\frac{k}{M},\frac{k}{M}+\ell^*}(1-u_\alpha)\right) > 0\right\}}$$
(18)

As the set $S_{\delta^*,..,\ell^*}[\lambda_0]$ defined in (14) is composed of alternatives with known change height δ^* and length ℓ^* , the distance between any of its elements and $S_0[\lambda_0] = \{\lambda_0\}$ is fixed, equal to $|\delta^*|\sqrt{\ell^*}$. Therefore, it is not discussed from the minimax point of view. We only provide sufficient conditions for the tests $\phi_{3,\alpha}^{(1)+}$, $\phi_{3,\alpha}^{(1)-}$ and $\phi_{3/4,\alpha}^{(2)}$ to have a second kind error rate controlled by a prescribed level β under P_{λ} when $\lambda \in S_{\delta^*,..,\ell^*}[\lambda_0]$. The proofs of these results, postponed to Section 5, mainly rely on sharp bounds for the quantiles $p_{\overline{\lambda_0},\ell^*}(\alpha)$, $p_{\lambda_0,\ell^*}^+(1-\alpha)$ and $t_{\lambda_0,k/M,k/M+\ell^*}(1-u_{\alpha})$, that are deduced from two very recent exponential inequalities for the supremum of a counting process and the oscillation modulus of the square martingale associated with a counting process due to Le Guével [71]. Recall that the technical proofs of such quantiles bounds are detailed in Section 6.

Proposition 6 (Second kind error rate control for [Alt.3]). Let $L \ge 1$, α and β in $(0,1), \lambda_0 > 0, \delta^*$ in $(-\lambda_0, +\infty) \setminus \{0\}$ and ℓ^* in (0,1). Considering the problem of testing (H_0) v.s. (H_1) " $\lambda \in S_{\delta^*, \dots, \ell^*}[\lambda_0]$ ", let $\phi_{3,\alpha}^{(1/2)}$ be one of the tests $\phi_{3,\alpha}^{(1)+}$ or $\phi_{3/4,\alpha}^{(2)}$ if $\delta^* > 0$, and one of the tests $\phi_{3,\alpha}^{(1)-}$ or $\phi_{3/4,\alpha}^{(2)}$ if $\delta^* < 0$ (see (16) and (18) for definitions of the tests). The test $\phi_{3,\alpha}^{(1/2)}$ is of level α , that is $P_{\lambda_0}(\phi_{3,\alpha}^{(1/2)}(N) = 1) \le \alpha$. Moreover, there exists $C(\alpha, \beta, \lambda_0, \delta^*, \ell^*) > 0$ such that $P_{\lambda}\left(\phi_{3,\alpha}^{(1/2)}(N) = 0\right) \le \beta$ as soon as λ belongs to $S_{\delta^*, \dots, \ell^*}[\lambda_0]$ with

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge C(\alpha, \beta, \lambda_0, \delta^*, \ell^*) / \sqrt{L}$$

Comments. Remarking that for λ in $S_{\delta^*,..,\ell^*}[\lambda_0]$, $d_2(\lambda, S_0[\lambda_0]) = |\delta^*|\sqrt{\ell^*}$, Proposition 6 leads to exhibit a sufficient minimal value $L_0(\alpha, \beta, \lambda_0, \delta^*, \ell^*)$ for L so that the second kind error rates of the above tests are controlled by β . Anecdotally, it furthermore shows that if $L \geq L_0(\alpha, \beta, \lambda_0, \delta^*, \ell^*)$, the β -uniform separation rate of the above tests over $S_{\delta^*,..,\ell^*}[\lambda_0]$ is equal to 0, as well as the (α, β) -minimax separation rate mSR_{α,β} $(S_{\delta^*,..,\ell^*}[\lambda_0])$.

Turning now to the change height adaptation issue, the lower bound for $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0]\right)$ given in Proposition 3 directly leads, using the monotonicity property of the minimum separation rate recalled in Lemma 42, to the following lower bound for $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{,..,\ell^*}[\lambda_0]\right)$.

Corollary 7 (Minimax lower bound for [Alt.4]). Let α, β in (0, 1) such that $\alpha + \beta < 1$, $\lambda_0 > 0$ and ℓ^* in (0, 1). For $L \ge 1$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\cdot,\cdot,\ell^*}[\lambda_0]\right) \ge \sqrt{\lambda_0 \log C_{\alpha,\beta}/L}, \text{ with } C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2.$$

Proposition 8 (Minimax upper bounds for [Alt.4]). Let $L \geq 1$, α and β in (0,1), $\lambda_0 > 0$, and ℓ^* in (0,1). Let $\phi_{4,\alpha}^{(1/2)}$ be one of the tests $\phi_{4,\alpha}^{(1)}$ and $\phi_{3/4,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\dots,\ell^*}[\lambda_0]$ " defined respectively by (17) and (18). Then $\phi_{4,\alpha}^{(1/2)}$ is of level α , that is $P_{\lambda_0}(\phi_{4,\alpha}^{(1/2)}(N) = 1) \leq \alpha$. Moreover, there exists $C(\alpha, \beta, \lambda_0, \ell^*) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{4,\alpha}^{(1/2)}, \mathcal{S}_{\cdot,\cdot,\ell^{*}}[\lambda_{0}]\right) \leq C(\alpha, \beta, \lambda_{0}, \ell^{*})/\sqrt{L}$$

which entails in particular $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{\cdot,\cdot,\ell^*}[\lambda_0]) \leq C(\alpha,\beta,\lambda_0,\ell^*)/\sqrt{L}$.

Comments. The proof of Proposition 8 mainly relies as the proof of Proposition 6 on the quantile controls of lemmas 49 and 48, respectively deduced from theorems 4 and 2 in [71]. This result with Corollary 7 means that the tests $\phi_{4,\alpha}^{(1)}$ and $\phi_{3/4,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\dots,\ell^*}[\lambda_0]$ " are β -minimax. Moreover and importantly, regarding the results obtained for [Alt.2], Proposition 8 with Corollary 7 also means that minimax adaptation with respect to the change location can be achieved with a minimax separation rate of the parametric order, hence without any additional price to pay (possibly except multiplicative constants), as soon as the only change length is known. This may contrast with the common idea (maybe spread by results in the jump detection problem where adaptation to the change location is equivalent to adaptation to the change length) that adaptation to the change location is the main cause of an unavoidable logarithmic cost. Here, by considering all the cases separately and step by step, we aim at precisely exhibiting the various regimes of minimax separation rates: this allows us in particular to specify - where relevant - the price to pay for adaptation to the different alternative parameters.

2.4 Minimax detection of a transitory change with known location

In this subsection, we consider the problem of testing the null hypothesis (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus alternative hypotheses where the location of the change from the baseline intensity is known, with adaptation with respect to the change length, and with or without adaptation with respect to the height of the change. Contrary to the study of Section 2.3, while adaptation to the change length only can be done without any incidence on the minimax separation rate order, adaptation to the change height in addition to the change length has a non-negligible impact. We therefore examine these two questions in two separate subsections.

2.4.1 Known change height

Let us first investigate the problem of testing (H_0) versus (H_1) " $\lambda \in S_{\delta^*, \tau^*, \dots}[\lambda_0]$ ", where the set $S_{\delta^*, \tau^*, \dots}[\lambda_0]$ is defined for δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$ and τ^* in (0, 1) by

$$\begin{bmatrix} \mathbf{Alt.5} \end{bmatrix} \mathcal{S}_{\delta^*,\tau^*,\cdots}[\lambda_0] = \left\{ \lambda : [0,1] \to (0,+\infty), \ \exists \ell \in (0,1-\tau^*), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau^*,\tau^*+\ell]}(t) \right\} .$$
(19)

As explained above, we will see that the minimax separation rate over this alternative set remains unchanged, of the parametric order $L^{-1/2}$. A lower bound is easily obtained from the key arguments given in Section 6.1. Therefore the major point here is the construction of a minimax adaptive test, which has to take the knowledge of the change height δ^* into account. In order to determine the most relevant way to integrate this knowledge, we have used an exact expression for the probability distribution function as well as an exponential inequality for the supremum of Poisson processes with shift, both due to Pyke [89, Equation (6) and Theorem 3]. This has led to a new procedure which is rather atypical regarding the other tests of this paper, and which can be related to Brunel's [19] scan test in the Gaussian set-up. **Proposition 9** (Minimax lower bound for [Alt.5]). Let α, β in (0,1) such that $\alpha + \beta < 1$, $\lambda_0 > 0, \ \delta^* \ in \ (-\lambda_0, +\infty) \setminus \{0\} \ and \ \tau^* \ in \ (0, 1). For \ all \ L \ge \lambda_0 \log C_{\alpha, \beta} / (\delta^{*2}(1 - \tau^*)),$

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^*,\tau^*,\dots}[\lambda_0]\right) \ge \sqrt{\lambda_0 \log C_{\alpha,\beta}/L}, \text{ with } C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2$$

Let us now introduce the aggregated test

$$\phi_{5,\alpha}(N) = \mathbb{1}_{\left\{\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N) > s^+_{\lambda_0,\delta^*,\tau^*,L}(1-\alpha)\right\}},$$
(20)

where $S_{\delta^*,\tau_1,\tau_2}(N)$ is the statistic defined for $0 \leq \tau_1 < \tau_2 \leq 1$ by

$$S_{\delta^*,\tau_1,\tau_2}(N) = \operatorname{sgn}(\delta^*) \Big(N(\tau_1,\tau_2] - \lambda_0 L(\tau_2 - \tau_1) \Big) - |\delta^*| L(\tau_2 - \tau_1)/2 , \qquad (21)$$

and $s^+_{\lambda_0,\delta^*,\tau^*,L}(u)$ is the *u*-quantile of $\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N)$ under (H_0) . Lemma 50 provides a control of the quantile $s^+_{\lambda_0,\delta^*,\tau^*,L}(1-\alpha)$, which is deduced from Pyke's results [89], and which is the main argument to prove that $\phi_{5,\alpha}$ has an uniform separation rate of parametric order $L^{-1/2}$ and thus show that the lower bound of Proposition 9 is sharp.

Proposition 10 (Minimax upper bound for [Alt.5]). Let $L \ge 1$, α and β in (0, 1), $\lambda_0 > 0$, δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$ and τ^* in (0, 1). Let $\phi_{5,\alpha}$ be the test of (H_0) versus (H_1) " $\lambda \in$ $\mathcal{S}_{\delta^*,\tau^*,\dots}[\lambda_0]$ " defined by (20). Then $\phi_{5,\alpha}$ is of level α , that is $P_{\lambda_0}(\phi_{5,\alpha}(N)=1) \leq \alpha$. Moreover, there exists a constant $C(\alpha, \beta, \lambda_0, \delta^*) > 0$ such that

$$\operatorname{SR}_{\beta}(\phi_{5,\alpha}, \mathcal{S}_{\delta^*,\tau^*,\dots}[\lambda_0]) \leq C(\alpha, \beta, \lambda_0, \delta^*)/\sqrt{L}$$
,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^*,\tau^*,\dots}[\lambda_0]\right) \leq C(\alpha,\beta,\lambda_0,\delta^*)/\sqrt{L}$.

Unknown change height 2.4.2

Now addressing the question of adaptation to the change height together with the change length, we consider for τ^* in (0,1) the alternative set

$$\mathcal{S}_{,\tau^*,\dots}[\lambda_0] = \left\{ \lambda : [0,1] \to (0,+\infty), \ \exists \delta \in (-\lambda_0,+\infty) \setminus \{0\}, \ \exists \ell \in (0,1-\tau^*), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^*+\ell]}(t) \right\} .$$
(22)

A first preliminary result in fact shows that this set of alternatives is too large to be relevantly studied from the minimax point of view: the minimax separation rate is infinite over it.

Lemma 11. Let α and β in (0,1) such that $\alpha + \beta < 1$, $\lambda_0 > 0$, and τ^* in (0,1). For the problem of testing (H_0) " $\lambda \in \mathcal{S}_0[\lambda_0] = \{\lambda_0\}$ " versus (H_1) " $\lambda \in \mathcal{S}_{\cdot,\tau^*,\cdots}[\lambda_0]$ ", with $\mathcal{S}_{\tau^*,\dots}[\lambda_0]$ defined by (22), one has $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\tau^*,\dots}[\lambda_0]\right) = +\infty$.

This preliminary result leads us to consider, for $R > \lambda_0$, the restricted set of alternatives

$$\begin{bmatrix} \mathbf{Alt.6} \end{bmatrix} \mathcal{S}_{,\tau^*,\cdots}[\lambda_0, R] = \left\{ \lambda : [0, 1] \to (0, R], \ \exists \delta \in (-\lambda_0, R - \lambda_0] \setminus \{0\}, \ \exists \ell \in (0, 1 - \tau^*), \\ \forall t \in [0, 1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau^*, \tau^* + \ell]}(t) \right\} .$$
(23)

For the problem of testing (H_0) " $\lambda \in \mathcal{S}_0[\lambda_0] = \{\lambda_0\}$ " versus (H_1) " $\lambda \in \mathcal{S}_{\tau^*,\dots}[\lambda_0, R]$ ", we then obtain the following lower bound.

Proposition 12 (Minimax lower bound for [Alt.6]). Let α, β in (0, 1) with $\alpha + \beta < 1/2$, $\lambda_0 > 0$, $R > \lambda_0$, τ^* in (0, 1). There exists $L_0(\alpha, \beta, \lambda_0, R) > 0$ such that for $L \ge L_0(\alpha, \beta, \lambda_0, R)$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\cdot,\tau^*,\cdots}[\lambda_0,R]\right) \geq \sqrt{\lambda_0 \log \log L/L}$$
.

Let us now assume that $L \ge 3$. In order to prove that the above lower bound is of sharp order (with respect to L), we construct two aggregated tests: a first one based on a linear statistic and a second one based on quadratic statistic as in Section 2.3. We thus consider the discrete subset of $(0, 1 - \tau^*)$ of the dyadic form

$$\{\ell_{\tau^*,k} = (1 - \tau^*) 2^{-k}; k \in \{1, \dots, \lfloor \log_2 L \rfloor\}\}$$
,

and the corrected level $u_{\alpha} = \alpha / \lfloor \log_2(L) \rfloor$, which allow to define the two following tests:

where $p_{\xi}(u)$ stands for the *u*-quantile of the Poisson distribution of parameter ξ as in (5), and

$$\phi_{6,\alpha}^{(2)}(N) = \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(T_{\tau^*,\tau^*+\ell_{\tau^*,k}}(N) - t_{\lambda_0,\tau^*,\tau^*+\ell_{\tau^*,k}}(1-u_\alpha)\right) > 0\right\}},$$
(25)

where $T_{\tau_1,\tau_2}(N)$ is the quadratic statistic (12), and $t_{\lambda_0,\tau_1,\tau_2}(u)$ its *u*-quantile under (H_0) .

Proposition 13 (Minimax upper bound for [Alt.6]). Let α and β in (0, 1), $\lambda_0 > 0$, $R > \lambda_0$ and τ^* in (0, 1). Let $\phi_{6,\alpha}^{(1/2)}$ be one of the tests $\phi_{6,\alpha}^{(1)}$ and $\phi_{6,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{,\tau^*},...[\lambda_0, R]$ " respectively defined by (24) and (25). Then $\phi_{6,\alpha}^{(1/2)}$ is of level α , that is $P_{\lambda_0}(\phi_{6,\alpha}^{(1/2)}(N) = 1) \leq \alpha$. Moreover, there exists $C(\alpha, \beta, \lambda_0, R) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{6,\alpha}^{(1/2)}, \mathcal{S}_{\cdot,\tau^{*},\dots}[\lambda_{0},R]\right) \leq C(\alpha,\beta,\lambda_{0},R)\sqrt{\log\log L/L}$$

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\cdot,\tau^*,\cdots}[\lambda_0,R]\right) \leq C(\alpha,\beta,\lambda_0,R)\sqrt{\log\log L/L}$.

Comments. The proofs of these upper bounds are mainly based on a sharp control of the quantiles $p_{\lambda_0 \ell_{\tau^*,k}L}(u)$ derived from the Cramér-Chernov inequality (see Lemma 46), and of the quantile $t_{\lambda_0,\tau_1,\tau_2}(u)$ already used in the proof of Proposition 6. This result, combined with its corresponding lower bound, brings out a first phase transition in the minimax separation rates orders, from the parametric rate order $1/\sqrt{L}$ to $\sqrt{\log \log L/L}$. This means that adaptation with respect to both height and length of the bump has a $\sqrt{\log \log L}$ cost, while adaptation to only one of these parameters does not cause any additional price, nor adaptation to both height and location as noticed above. Though a comparable phase transition has already been observed in Gaussian models when dealing with the jump detection problem (where adaptation with respect to the location is equivalent to adaptation with respect to the length), up to our knowledge, such results did not appear yet in the bump detection literature.

2.5 Minimax detection of a possibly transitory change with unknown location and length

In this subsection, we address the final problem of testing the null hypothesis (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus alternative hypotheses where both location and length of the change from the baseline intensity are unknown, distinguishing the case where the change is transitory from the particular case where it is not transitory.

Still adopting the minimax point of view, we will see that when considering the transitory change detection problem, adaptation to both change location and length has a minimax separation rate cost of order $\sqrt{\log L}$, and this whether the change height is known or not. This highly contrasts with the study of the particular non transitory change or jump detection problem, which makes two different regimes of minimax separation rates appear, with a maximal cost of order $\sqrt{\log \log L}$ for change height adaptation.

Let us underline that the non transitory change or jump detection problem can be viewed as perfectly symmetrical to the above transitory change with known location detection problem (see Section 2.4). In the first problem, one can consider that the length of the change is unknown but the endpoint of the change is known, while in the second problem the length of the change is unknown but the starting point of the change is known. The study of the first non transitory change detection problem will therefore use very similar arguments as the study of the second transitory change with known location detection problem, finally leading to the same minimax separation rates. This is why we conduct it first here.

2.5.1 Non transitory change

In order to investigate the problem of detecting a non-transitory change or jump with unknown location, but known height, we introduce for δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$ the alternative set

$$[Alt.7] \mathcal{S}_{\delta^*,\dots,1-\dots}[\lambda_0] = \{\lambda : [0,1] \to (0,+\infty), \ \exists \tau \in (0,1), \ \lambda(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau,1]}(t) \} \ . \ (26)$$

This allows us to formalise the considered detection problem as a problem of testing the null hypothesis (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus the alternative (H_1) " $\lambda \in S_{\delta^*,..,1-..}[\lambda_0]$ ", with the corresponding minimax lower bound stated below.

Proposition 14 (Minimax lower bound for [Alt.7]). Let α and β in (0,1) such that $\alpha + \beta < 1$, $\lambda_0 > 0$ and δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$. For all $L \ge \lambda_0 \log C_{\alpha,\beta} / \delta^{*2}$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^*,\dots,1-\dots}[\lambda_0]\right) \ge \sqrt{\lambda_0 \log C_{\alpha,\beta}/L}, \text{ with } C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2.$$

Following the study and the notation of Section 2.4, we define the test

$$\phi_{7,\alpha}(N) = \mathbb{1}_{\left\{\sup_{\tau \in (0,1)} S_{\delta^*,\tau,1}(N) > s^+_{\lambda_0,\delta^*,L}(1-\alpha)\right\}} , \qquad (27)$$

where $S_{\delta^*,\tau_1,\tau_2}(N)$ is the statistic defined by (21) and $s^+_{\lambda_0,\delta^*,L}(u)$ stands for the *u*-quantile of $\sup_{\tau \in (0,1)} S_{\delta^*,\tau,1}(N)$ under (H_0) .

Proposition 15 (Minimax upper bound for [Alt.7]). Let $L \ge 1$, α and β in (0, 1), $\lambda_0 > 0$ and δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$. Let $\phi_{7,\alpha}$ be the test of (H_0) versus (H_1) " $\lambda \in S_{\delta^*,..,1-..}[\lambda_0]$ " defined by (27). Then $\phi_{7,\alpha}$ is of level α , that is $P_{\lambda_0}(\phi_{7,\alpha}(N) = 1) \le \alpha$. Moreover, there exists a constant $C(\alpha, \beta, \lambda_0, \delta^*) > 0$ such that

$$\operatorname{SR}_{\beta}(\phi_{7,\alpha}, \mathcal{S}_{\delta^*, \dots, 1-\dots}[\lambda_0]) \leq C(\alpha, \beta, \lambda_0, \delta^*) / \sqrt{L} ,$$

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^*,\dots,1-\dots}[\lambda_0]\right) \leq C(\alpha,\beta,\lambda_0,\delta^*)/\sqrt{L}.$

Let us now tackle the question of adaptation with respect to the change height and therefore introduce to this end a preliminary alternative set

$$\mathcal{S}_{,\dots,1-\dots}[\lambda_0] = \{\lambda : \exists \delta \in (-\lambda_0, +\infty) \setminus \{0\}, \exists \tau \in (0,1), \ \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau,1]}(t)\} \ . \tag{28}$$

As in Section 2.4, we underline that the minimax separation rate over this set is infinite.

Lemma 16. Let α, β in (0, 1) such that $\alpha + \beta < 1$. For the problem of testing (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus (H_1) " $\lambda \in S_{.,.,1-..}[\lambda_0]$ ", with $S_{.,.,1-..}[\lambda_0]$ defined by (28), one has $\mathrm{mSR}_{\alpha,\beta}(S_{.,.,1-..}[\lambda_0]) = +\infty$.

We thus consider for $R > \lambda_0$ the more suitable set of alternatives bounded by R, defined by

$$[\mathbf{Alt.8}] \ \mathcal{S}_{\cdot,\cdot,1-\cdot\cdot}[\lambda_0, R] = \left\{ \lambda : [0, 1] \to (0, R], \ \exists \delta \in (-\lambda_0, R - \lambda_0] \setminus \{0\}, \ \exists \tau \in (0, 1), \\ \forall t \in [0, 1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau, 1]}(t) \right\} .$$
(29)

Considering the problem of testing (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus (H_1) " $\lambda \in S_{\cdot,\dots,1-\dots}[\lambda_0, R]$ ", we then obtain the following lower bound.

Proposition 17 (Minimax lower bound for [Alt.8]). Let α, β in (0, 1) with $\alpha + \beta < 1/2$, $\lambda_0 > 0$ and $R > \lambda_0$. There exists $L_0(\alpha, \beta, \lambda_0, R) > 0$ such that for $L \ge L_0(\alpha, \beta, \lambda_0, R)$,

$$\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\cdot,\cdot,1-\cdot}[\lambda_{0},R]\right) \geq \sqrt{\lambda_{0}\log\log L/L}$$

Again, following the study and the notation of Section 2.4, we assume now that $L \ge 3$, we consider the discrete subset of (0, 1) of the dyadic form

$$\left\{\tau_k = 1 - 2^{-k}; \ k \in \{1, \dots, \lfloor \log_2 L \rfloor\}\right\} ,$$

and we set $u_{\alpha} = \alpha / \lfloor \log_2(L) \rfloor$, which allows to define the two following tests:

$$\phi_{8,\alpha}^{(1)}(N) = \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(N(\tau_k, 1] - p_{\lambda_0(1-\tau_k)L}\left(1 - \frac{u_\alpha}{2}\right)\right) > 0\right\}} \\ \vee \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(p_{\lambda_0(1-\tau_k)L}\left(\frac{u_\alpha}{2}\right) - N(\tau_k, 1]\right) > 0\right\}}, \quad (30)$$

where $p_{\xi}(u)$ stands for the *u*-quantile of the Poisson distribution of parameter ξ , and

$$\phi_{8,\alpha}^{(2)}(N) = \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(T_{\tau_k,1}(N) - t_{\lambda_0,\tau_k,1}(1-u_\alpha) \right) > 0 \right\}},$$
(31)

where $T_{\tau_1,\tau_2}(N)$ is the quadratic statistic (12) and $t_{\lambda_0,\tau_1,\tau_2}(u)$ its *u*-quantile under (H_0) .

Proposition 18 (Minimax upper bound for [Alt.8]). Let α, β in (0, 1) with $\alpha + \beta < 1$, $\lambda_0 > 0$ and $R > \lambda_0$. Let $\phi_{8,\alpha}^{(1/2)}$ be one of the tests $\phi_{8,\alpha}^{(1)}$ and $\phi_{8,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\dots,1-\dots}[\lambda_0, R]$ " respectively defined by (30) and (31). Then $\phi_{8,\alpha}^{(1/2)}$ is of level α , that is $P_{\lambda_0}(\phi_{8,\alpha}^{(1/2)}(N) = 1) \leq \alpha$. Moreover, there exists a constant $C(\alpha, \beta, \lambda_0, R) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{8,\alpha}^{(1/2)}, \mathcal{S}_{\cdot, \cdot, 1- \cdot, \cdot}[\lambda_{0}, R]\right) \leq C(\alpha, \beta, \lambda_{0}, R)\sqrt{\log \log L/L}$$

which entails in particular $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{\cdot,\cdot,1-\cdot,\cdot}[\lambda_0,R]) \leq C(\alpha,\beta,\lambda_0,R)\sqrt{\log\log L/L}.$

2.5.2 Transitory change

In this section, we address the transitory change detection problem, focusing here on the question of adaptation to unknown location and length.

As explained above, we will see that minimax adaptation to these both parameters has the most important cost in the present study, whose order is as large as $\sqrt{\log L}$, so that adaptation to the height will have no additional cost.

Let us first give lower bounds for the minimax separation rates, focusing on the case where the change height is known since the general case where all three parameters, location, length and height of the change are unknown then follows easily.

Hence, we introduce for δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$ the alternative set

$$\begin{bmatrix} \mathbf{Alt.9} \end{bmatrix} \mathcal{S}_{\delta^*,\dots,\dots}[\lambda_0] = \left\{ \lambda : [0,1] \to (0,+\infty), \ \exists \tau \in (0,1), \ \exists \ell \in (0,1-\tau), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau,\tau+\ell]}(t) \right\} .$$
(32)

Proposition 19 (Minimax lower bound for [Alt.9]). Let α, β in $(0, 1), \lambda_0 > 0$ and δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$. There exists $L_0(\alpha, \beta, \lambda_0, \delta^*) > 0$ such that for all $L \ge L_0(\alpha, \beta, \lambda_0, \delta^*)$,

$$\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^*,\dots,\dots}[\lambda_0]\right) \ge \sqrt{\lambda_0 \log L/(2L)}$$

Now considering the very general alternative set

$$\mathcal{S}_{,..,.}[\lambda_0] = \left\{ \lambda : [0,1] \to (0,+\infty), \ \exists \delta \in (-\lambda_0,+\infty) \setminus \{0\}, \ \exists \tau \in (0,1), \ \exists \ell \in (0,1-\tau), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell]}(t) \right\},$$
(33)

since it contains $\mathcal{S}_{\tau^*,\dots}[\lambda_0]$ defined by (22) for any τ^* in (0, 1), Lemma 11 straightforwardly leads to an infinite minimax separation rate lower bound.

Corollary 20. Let α, β in (0, 1) such that $\alpha + \beta < 1$. For the problem of testing (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus (H_1) " $\lambda \in S_{\cdot, \dots, \dots}[\lambda_0]$ ", with $S_{\cdot, \dots, \dots}[\lambda_0]$ defined by (33), one has $\mathrm{mSR}_{\alpha, \beta}(S_{\cdot, \dots, \dots}[\lambda_0]) = +\infty$.

We therefore restrict the alternative set to the one defined for $R \ge \lambda_0$ by

$$\begin{bmatrix} \mathbf{Alt.10} \end{bmatrix} \mathcal{S}_{\cdot,\cdot\cdot,\cdot\cdot} \begin{bmatrix} \lambda_0, R \end{bmatrix} = \left\{ \lambda : [0, 1] \to (0, R], \ \exists \delta \in (-\lambda_0, R - \lambda_0] \setminus \{0\}, \ \exists \tau \in (0, 1), \\ \exists \ell \in (0, 1 - \tau), \ \forall t \in [0, 1] \quad \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau, \tau + \ell]}(t) \right\},$$
(34)

and deal with the problem of testing (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus (H_1) " $\lambda \in S_{\dots,\dots}[\lambda_0, R]$ ". This alternative set $S_{\dots,\dots}[\lambda_0, R]$ includes $S_{R-\lambda_0,\dots,\dots}[\lambda_0]$ defined by (32) when $R > \lambda_0$, and $S_{-\lambda_0/2,\dots,\dots}[\lambda_0]$ when $\lambda_0 = R$. Therefore, Proposition 19 has the following direct corollary, whose proof as well as the proof of Corollary 20 is omitted for simplicity.

Corollary 21 (Minimax lower bound for [Alt10]). Let α, β in (0, 1) with $\alpha + \beta < 1$, $\lambda_0 > 0$ and $R \ge \lambda_0$. There exists $L_0(\alpha, \beta, \lambda_0, R) > 0$ such that for all $L \ge L_0(\alpha, \beta, \lambda_0, R)$,

$$\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\dots,\dots}[\lambda_0,R]\right) \geq \sqrt{\lambda_0 \log L/(2L)}$$
.

In order to prove that the above lower bounds are sharp, we secondly construct two novel minimax adaptive tests according to a scanning aggregation principle again. Since the lower bound shows an additional cost for adaptation to change location and length of order $\sqrt{\log L}$ instead of at most $\sqrt{\log \log L}$ when dealing with adaptation to only one of these parameters, we do not necessarily need to consider a dyadic set of aggregated tests. More precisely, setting $u_{\alpha} = 2\alpha/(\lceil L \rceil (\lceil L \rceil + 1))$, we define

$$\phi_{9/10,\alpha}^{(1)}(N) = \mathbb{1}_{\left\{\max_{k \in \{0,\dots,\lceil L \rceil - 1\}, k' \in \{1,\dots,\lceil L \rceil - k\}} \left(N\left(\frac{k}{\lceil L \rceil}, \frac{k+k'}{\lceil L \rceil}\right) - p_{\lambda_0 k' L/\lceil L \rceil}\left(1 - \frac{u_{\alpha}}{2}\right)\right) > 0\right\}} \\ \vee \mathbb{1}_{\left\{\max_{k \in \{0,\dots,\lceil L \rceil - 1\}, k' \in \{1,\dots,\lceil L \rceil - k\}} \left(p_{\lambda_0 k' L/\lceil L \rceil}\left(\frac{u_{\alpha}}{2}\right) - N\left(\frac{k}{\lceil L \rceil}, \frac{k+k'}{\lceil L \rceil}\right)\right) > 0\right\}}, \quad (35)$$

and

$$\phi_{9/10,\alpha}^{(2)}(N) = \mathbb{1}_{\left\{\max_{k \in \{0,\dots,\lceil L \rceil - 1\}, k' \in \{1,\dots,\lceil L \rceil - k\}} \left(T_{\frac{k}{\lceil L \rceil}, \frac{k+k'}{\lceil L \rceil}(N) - t_{\lambda_0, \frac{k}{\lceil L \rceil}, \frac{k+k'}{\lceil L \rceil}(1-u_{\alpha})}\right) > 0}\right\}}$$
(36)

Proposition 22 (Minimax upper bound for [Alt.10]). Let α, β in $(0, 1), \lambda_0 > 0$ and $R \geq \lambda_0$. Let $\phi_{9/10,\alpha}^{(1/2)}$ be one of the tests $\phi_{9/10,\alpha}^{(1)}$ and $\phi_{9/10,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\dots,\dots}[\lambda_0, R]$ " respectively defined by (35) and (36). Then $\phi_{9/10,\alpha}^{(1/2)}$ is of level α , that is $P_{\lambda_0}\left(\phi_{9/10,\alpha}^{(1/2)}(N) = 1\right) \leq \alpha$. Moreover, there exists a constant $C(\alpha, \beta, \lambda_0, R) > 0$ such that $SR_{\beta}\left(\phi_{9/10,\alpha}^{(1/2)}, S_{\dots,\dots}[\lambda_0, R]\right) \leq C(\alpha, \beta, \lambda_0, R)\sqrt{\log L/L}$,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{\cdot,\ldots,\ldots}[\lambda_0,R]) \leq C(\alpha,\beta,\lambda_0,R)\sqrt{\log L/L}$.

As in particular $S_{\delta^*,..,..}[\lambda_0]$ is included in $S_{.,.,..}[\lambda_0, \lambda_0 + \delta^*]$ for any $\delta^* > 0$ and $S_{.,.,..}[\lambda_0, \lambda_0]$ for any δ^* in $(-\lambda_0, 0)$, Proposition 22 has the following immediate corollary, which closes the study of possibly transitory change in a known baseline intensity detection.

Corollary 23 (Minimax upper bound for [Alt.9]). Let α, β in $(0,1), \lambda_0 > 0$ and δ^* in $(-\lambda_0, +\infty) \setminus \{0\}$. Let $\phi_{9/10,\alpha}^{(1/2)}$ be one of the tests $\phi_{9/10,\alpha}^{(1)}$ and $\phi_{9/10,\alpha}^{(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\delta^*,\dots,\dots}[\lambda_0]$ " respectively defined by (35) and (36). Then $\phi_{9/10,\alpha}^{(1/2)}$ is of level α , that is $P_{\lambda_0}\left(\phi_{9/10,\alpha}^{(1/2)}(N)=1\right) \leq \alpha$. Moreover, there exists $C(\alpha, \beta, \lambda_0, \delta^*) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{9/10,\alpha}^{(1/2)}, \mathcal{S}_{\delta^{*}, \dots, \dots}[\lambda_{0}]\right) \leq C(\alpha, \beta, \lambda_{0}, \delta^{*})\sqrt{\log L/L} ,$$

which entails in particular $\operatorname{mSR}_{\alpha,\beta}(\mathcal{S}_{\delta^*,\dots,\dots}[\lambda_0]) \leq C(\alpha,\beta,\lambda_0,\delta^*)\sqrt{\log L/L}.$

Comment. The upper bounds in Proposition 22 and Corollary 23, combined with their corresponding lower bounds, bring out a second phase transition in the minimax separation rate orders, from the rate order $\sqrt{\log \log L/L}$ when considering adaptation with respect to both bump height and length to $\sqrt{\log L/L}$, obtained when dealing with adaptation to at least bump location and length (with no additional cost when adapting to the bump height). As comparable minimax separation rates were already known in Gaussian models with [5] and [19], these results were more expected that some of the above ones.

2.6 Choice of individual levels for aggregated tests and links with multiple tests

Except the tests $\phi_{1,\alpha}^-$, $\phi_{1,\alpha}^+$, $\phi_{2,\alpha}^{(1)}$ and $\phi_{2,\alpha}^{(2)}$ that are classical single tests, all the tests introduced in the above study are based on aggregation principles. Among them, we can essentially distinguish two kinds of such aggregated tests.

The first aggregated test type is of the form $\phi_{agg1,\alpha}(N) = \mathbb{1}_{\{\sup_{\theta \in \Theta} S_{\theta}(N) > s^{+}_{\lambda_{0}}(1-\alpha)\}}$ or $\mathbb{1}_{\{\sup_{\theta \in \Theta} S_{\theta}(N) > s^{+}_{\lambda_{0}}(1-\alpha/2)\}} \vee \mathbb{1}_{\{\inf_{\theta \in \Theta} S_{\theta}(N) < s^{-}_{\lambda_{0}}(\alpha/2)\}}$, where:

- θ is one possible parameter or couple of parameters among the location τ or length ℓ of the bump/jump in the alternative intensity, and Θ is a subset of possible values for θ ,
- $S_{\theta}(N)$ is a statistic designed to test (H_0) " $\lambda = \lambda_0$ " versus (H_1) " λ has a jump or a bump with parameter or parameters θ ", such that $\sup_{\theta \in \Theta} S_{\theta}(N)$ and $\inf_{\theta \in \Theta} S_{\theta}(N)$ have computable, exactly or by a Monte Carlo method, *u*-quantiles $s_{\lambda_0}^+(u)$ and $s_{\lambda_0}^-(u)$ under (H_0) .

The tests $\phi_{3,\alpha}^{(1)-}$, $\phi_{3,\alpha}^{(1)+}$, $\phi_{4,\alpha}^{(1)}$, $\phi_{5,\alpha}$ and $\phi_{7,\alpha}$ can all be written in this way. The second aggregated test type is of the form $\phi_{agg2,\alpha}(N) = \mathbb{1}_{\left\{\sup_{\theta \in \Theta} \left(S_{\theta}(N) - s_{\lambda_{0},\theta}(1-u_{\alpha})\right) > 0\right\}}$ or $\mathbb{1}_{\left\{\sup_{\theta \in \Theta} \left(S_{\theta}(N) - s_{\lambda_{0},\theta}(1-u_{\alpha}/2)\right) > 0\right\}} \vee \mathbb{1}_{\left\{\sup_{\theta \in \Theta} \left(s_{\lambda_{0},\theta}(u_{\alpha}/2) - S_{\theta}(N)\right) > 0\right\}}$, where:

- θ is one possible parameter or couple of parameters among the location τ or length ℓ of the bump/jump in the alternative intensity, and Θ is a finite subset of possible values for θ ,
- $S_{\theta}(N)$ is a statistic designed to test (H_0) " $\lambda = \lambda_0$ " versus (H_1) " λ has a jump/bump with parameter or parameters θ ", with computable (1 u)-quantiles $s_{\lambda_0,\theta}(1 u)$ under (H_0) ,
- u_{α} is an individual, adjusted and smaller than α , level of test.

The tests $\phi_{3/4,\alpha}^{(2)}$, $\phi_{6,\alpha}^{(1)}$, $\phi_{8,\alpha}^{(2)}$, $\phi_{8,\alpha}^{(1)}$, $\phi_{9/10,\alpha}^{(2)}$, $\phi_{9/10,\alpha}^{(2)}$ can all be written in this way. Both $\phi_{agg1,\alpha}$ and $\phi_{agg2,\alpha}$ aggregated test types can thus be expressed as $\mathbb{1}_{\left\{\exists \theta \in \Theta, \ S_{\theta}(N) > c_{\lambda_0,\theta,\alpha}\right\}}$

Both $\psi_{agg1,\alpha}$ and $\psi_{agg2,\alpha}$ aggregated test types can thus be expressed as $\mathbb{I}_{\{\exists \theta \in \Theta, S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}^{1} \text{ or } S_{\theta}(N) < c_{\lambda_{0},\theta,\alpha}^{2}\}}$, where $c_{\lambda_{0},\theta,\alpha} = s_{\lambda_{0}}^{+}(1-\alpha)$, $c_{\lambda_{0},\theta,\alpha}^{1} = s_{\lambda_{0}}^{+}(1-\alpha/2)$ and $c_{\lambda_{0},\theta,\alpha}^{2} = s_{\lambda_{0}}^{-}(\alpha/2)$ do not vary with θ in $\phi_{agg1,\alpha}$ and $c_{\lambda_{0},\theta,\alpha} = s_{\lambda_{0},\theta}(1-u_{\alpha})$, $c_{\lambda_{0},\theta,\alpha}^{1} = s_{\lambda_{0},\theta}(1-u_{\alpha}/2)$ and $c_{\lambda_{0},\theta,\alpha}^{2} = s_{\lambda_{0},\theta}(u_{\alpha}/2)$ vary with θ in $\phi_{agg2,\alpha}$. This therefore means that these tests reject (H_{0}) when, scanning all the parameters or couples of parameters θ in Θ , at least one single test in the collection $\{\mathbbm{1}_{S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}}, \theta \in \Theta\}$ or $\{\mathbbm{1}_{S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}^{1} \lor \mathbbm{1}_{S_{\theta}(N) < c_{\lambda_{0},\theta,\alpha}^{2}}, \theta \in \Theta\}$ rejects (H_{0}) , which explains the name of scan aggregation principle. All the single tests $\mathbbm{1}_{S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}}$ or $\mathbbm{1}_{S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}^{1}} \lor \mathbbm{1}_{S_{\theta}(N) < c_{\lambda_{0},\theta,\alpha}^{2}}$ in the considered collections are of level α , but they can be in fact, individually, very conservative, otherwise their aggregation would not preserve the level α property *in fine*. In the particular case of $\phi_{agg2,\alpha}$, the single tests are of individual level u_{α} , taken here equal to $u_{\alpha} = \alpha/|\Theta|$.

A better choice for u_{α} , leading to a less conservative aggregated test, was first proposed in another context by Baraud et al. [9]. In our context, this choice corresponds to u'_{α} equal to

$$u_{\alpha}' = \begin{cases} \sup \left\{ u \in (0,1), \ P_{\lambda_{0}}\left(\sup_{\theta \in \Theta}\left(S_{\theta}(N) - s_{\lambda_{0},\theta}(1-u)\right) > 0\right) \leq \alpha \right\} \text{ or} \\ \sup \left\{ u \in (0,1), \\ P_{\lambda_{0}}\left(\sup_{\theta \in \Theta}\left(S_{\theta}(N) - s_{\lambda_{0},\theta}(1-u/2)\right) \lor \left(s_{\lambda_{0},\theta}(u/2) - S_{\theta}(N)\right) > 0\right) \leq \alpha \right\}. \end{cases}$$

$$(37)$$

Since $u_{\alpha} \leq u'_{\alpha}$, by definition, $s_{\lambda_{0},\theta}(1-u'_{\alpha}) \leq s_{\lambda_{0},\theta}(1-u_{\alpha})$, $s_{\lambda_{0},\theta}(1-u'_{\alpha}/2) \leq s_{\lambda_{0},\theta}(1-u_{\alpha}/2)$ and $s_{\lambda_{0},\theta}(u_{\alpha}/2) \leq s_{\lambda_{0},\theta}(u'_{\alpha}/2)$. Any upper bound for $s_{\lambda_{0},\theta}(1-u_{\alpha})$ or $s_{\lambda_{0},\theta}(1-u_{\alpha}/2)$, or lower bound for $s_{\lambda_{0},\theta}(u_{\alpha}/2)$, such as those used in the proofs of the minimax separation rates upper bounds and deduced from the quantiles bounds of Section 6.2, therefore remain valid for $s_{\lambda_{0},\theta}(1-u'_{\alpha})$, $s_{\lambda_{0},\theta}(1-u'_{\alpha}/2)$ or $s_{\lambda_{0},\theta}(u'_{\alpha}/2)$ respectively. As a consequence, all the above tests of type $\phi_{agg2,\alpha}$ but with u'_{α} instead of u_{α} , that we can denote by $\phi'_{agg2,\alpha}$, satisfy the same minimax properties as $\phi_{agg2,\alpha}$.

The fact that such adjusted aggregated tests $\phi'_{agg2,\alpha}$ are more powerful than $\phi_{agg2,\alpha}$ is not discernable in minimax results whereas it is clearly noticeable in practice, is a known shortcoming of the present nonasymptotic minimax point of view, where exact constants (making lower and upper bound match, up to a possible negligible term) are not expected, which is not solved yet up to our knowledge in any testing framework. Our simulation study presented in Section 4 focuses on the performances of adjusted aggregated tests of the form $\phi'_{agg2,\alpha}$.

Let us now turn to the links that can be highlighted between such minimax adaptive, aggregated tests $\phi_{agg2,\alpha}$ or adjusted aggregated tests $\phi'_{agg2,\alpha}$ and multiple tests. The parallel between such aggregated tests and multiple tests has been established in [52], as the foundation of a minimax theory for multiple tests. Notice that when each single test $\mathbb{1}_{S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}} \text{ in the collection } \left\{ \mathbb{1}_{S_{\theta}(N) > c_{\lambda_{0},\theta,\alpha}}, \ \theta \in \Theta \right\} \text{ can be interpreted as a test of a null }$ hypothesis $(H_{0,\theta})$ $\lambda \in S_{0,\theta}$ versus $(H_{1,\theta})$ $\lambda \notin S_{0,\theta}$, with $S_0 \subset \cap_{\theta \in \Theta} S_{0,\theta}$, it appears that our first choice of individual level $u_{\alpha} = \alpha/|\Theta|$ can be related to a Bonferroni type multiple test of the set of hypotheses $\{(H_{0,\theta}), \theta \in \Theta\}$, while our second choice u'_{α} defined by (37) can be related to a min-p type multiple test of the same set of hypotheses. As an example, recall that $T_{\tau_1,\tau_2}(N)$ defined by (12) is an unbiased estimator of $\|\Pi_{V_{\tau_1,\tau_2}}(\lambda-\lambda_0)\|_2^2$ (see Section 2.2). Therefore, the single tests $\mathbb{1}_{T_{k/[L],(k+k')/[L]}(N) > t_{\lambda_0,k/[L],(k+k')/[L]}(1-u)}$ involved in $\phi_{9/10,\alpha}^{(2)}(N)$ can be viewed as single tests of $(H_{0,(k/\lceil L\rceil,k'/\lceil L\rceil)}) \prod_{V_{k/\lceil L\rceil,(k+k')/\lceil L\rceil}} (\lambda - \lambda_0) = 0$ versus $(H_{1,(k/\lceil L\rceil,k'/\lceil L\rceil)}) \prod_{V_{k/\lceil L\rceil,(k+k')/\lceil L\rceil}} (\lambda - \lambda_0) \neq 0$. Our aggregated tests of the form $\phi_{agg2,\alpha}$ are thus clearly aggregated tests constructed from Bonferroni multiple tests, while the corresponding adjusted aggregated tests of the form $\phi_{agg2,\alpha}'$ are constructed from min-pmultiple tests of such collections of hypotheses (see [52] for some detailed study).

2.7 Summary and discussion

We present below a summary of the results stated above in a tabular form. Recall (c.f. (12) and (21)) that for $0 \le \tau_1 < \tau_2 \le 1$,

$$T_{\tau_1,\tau_2}(N) = \frac{1}{L^2(\tau_2 - \tau_1)} \left(N\left(\tau_1, \tau_2\right)^2 - \left(1 + 2\lambda_0 L(\tau_2 - \tau_1)\right) N\left(\tau_1, \tau_2\right] + \lambda_0^2 L^2(\tau_2 - \tau_1)^2 \right) ,$$

 $S_{\delta^*,\tau_1,\tau_2}(N) = \operatorname{sgn}(\delta^*)(N(\tau_1,\tau_2] - \lambda_0 L(\tau_2 - \tau_1)) - |\delta^*|L(\tau_2 - \tau_1)/2$, and that $p_{\xi}(u)$ and $t_{\lambda_0,\tau_1,\tau_2}(u)$ stand for the *u*-quantiles of the Poisson distribution of parameter ξ and the distribution of $T_{\tau_1,\tau_2}(N)$ under (H_0) respectively.

Non transitory change or jump detection		
Alternative set	$\mathrm{mSR}_{\alpha,\beta}$	Test statistics
$\frac{\mathcal{S}_{\delta^*,\tau^*,1-\tau^*}[\lambda_0]}{\mathcal{S}_{\cdot,\tau^*,1-\tau^*}[\lambda_0]}$	-	$N(\tau^*, 1]$
$\mathcal{S}_{\cdot, au^*,1- au^*}[\lambda_0]$	$L^{-1/2}$	$\frac{N(\tau^*, 1]}{N(\tau^*, 1]}$
		$T_{\tau^*,1}(N)$
$\mathcal{S}_{\delta^*,,1}[\lambda_0]$	$L^{-1/2}$	$\sup_{\tau \in (0,1)} S_{\delta^*,\tau,1}(N)$
$\mathcal{S}_{\cdot,\cdot,1-\cdot\cdot}[\lambda_0,R]$	$\sqrt{\frac{\log \log L}{L}}$	$\max_{k \in \{1, \dots, \lfloor \log_2 L \rfloor\}} \left(N(1 - 2^{-k}, 1] - p_{\lambda_0 2^{-k} L} \left(1 - u_\alpha / 2 \right) \right)$
		$\vee \left(p_{\lambda_0 2^{-k}L} \left(u_{\alpha}/2 \right) - N(1 - 2^{-k}, 1] \right)$
		$\max_{k \in \{1, \dots, \lfloor \log_2 L \rfloor\}} \left(T_{1-2^{-k}, 1}(N) - t_{\lambda_0, 1-2^{-k}, 1}(1-u_\alpha) \right)$
Transitory change or bump detection		
Alternative set	$\mathrm{mSR}_{\alpha,\beta}$	Test statistics
$\frac{\mathcal{S}_{\delta^*,\tau^*,\ell^*}[\lambda_0]}{\mathcal{S}_{\cdot,\tau^*,\ell^*}[\lambda_0]}$	-	$\frac{N(\tau^*, \tau^* + \ell^*]}{N(\tau^*, \tau^* + \ell^*]}$
$\mathcal{S}_{\boldsymbol{\cdot}, au^*,\ell^*}[\lambda_0]$	$L^{-1/2}$	
		$T_{\tau^*,\tau^*+\ell^*}(N)$
$\mathcal{S}_{\delta^*, \cdots, \ell^*}[\lambda_0]$	-	$\max_{\tau \in [0, 1-\ell^*]} N(\tau, \tau + \ell^*], \min_{\tau \in [0, 1-\ell^*]} N(\tau, \tau + \ell^*]$
		$\max_{k \in \{0,\dots,\lceil (1-\ell^*)\lceil 2/\ell^*\rceil\rceil-1\}} \left(T_{\frac{k}{\lceil 2/\ell^*\rceil},\frac{k}{\lceil 2/\ell^*\rceil}+\ell^*}(N) - t_{\lambda_0,\frac{k}{\lceil 2/\ell^*\rceil},\frac{k}{\lceil 2/\ell^*\rceil}+\ell^*}(1-u_\alpha) \right)$
$\mathcal{S}_{\cdot,\cdot\cdot,\ell^*}[\lambda_0]$	$L^{-1/2}$	$\max_{\tau \in [0, 1-\ell^*]} N(\tau, \tau + \ell^*], \min_{\tau \in [0, 1-\ell^*]} N(\tau, \tau + \ell^*]$
		$\max_{k \in \{0,\dots,\lceil (1-\ell^*)\lceil 2/\ell^*\rceil\rceil-1\}} \left(T_{\frac{k}{\lceil 2/\ell^*\rceil},\frac{k}{\lceil 2/\ell^*\rceil}+\ell^*}(N) - t_{\lambda_0,\frac{k}{\lceil 2/\ell^*\rceil},\frac{k}{\lceil 2/\ell^*\rceil}+\ell^*}(1-u_\alpha) \right) $
$\mathcal{S}_{\delta^*, au^*,}[\lambda_0]$	$L^{-1/2}$	$\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N)$
$\mathcal{S}_{\cdot,\tau^*,\dots}[\lambda_0,R]$	$\sqrt{\frac{\log \log L}{L}}$	$\max_{k \in \{1, \dots, \lfloor \log_2 L \rfloor\}} \left(N(\tau^*, \tau^* + (1 - \tau^*) 2^{-k}] - p_{\lambda_0(1 - \tau^*) 2^{-k}L} (1 - u_\alpha/2) \right)$
		$\vee \left(p_{\lambda_0(1-\tau^*)2^{-k}L} \left(u_{\alpha}/2 \right) - N(\tau^*, \tau^* + (1-\tau^*)2^{-k}] \right)$
		$\max_{k \in \{1, \dots, \lfloor \log_2 L \rfloor\}} \left(T_{\tau^*, \tau^* + (1 - \tau^*) 2^{-k}}(N) - t_{\lambda_0, \tau^*, \tau^* + (1 - \tau^*) 2^{-k}}(1 - u_\alpha) \right)$
$\mathcal{S}_{\delta^*,,}[\lambda_0]$	$\sqrt{\frac{\log L}{L}}$	$\max_{k \in \{0,\dots,\lceil L\rceil-1\}, k' \in \{1,\dots,\lceil L\rceil-k\}} \left(N\left(\frac{k}{\lceil L\rceil}, \frac{k+k'}{\lceil L\rceil}\right) - p_{\lambda_0 k' L/\lceil L\rceil} \left(1 - \frac{u_\alpha}{2}\right) \right)$
$\mathcal{S}_{\cdot,\cdot\cdot,\dots}[\lambda_0,R]$		$\vee \left(p_{\lambda_0 k'L/\lceil L \rceil} \left(\frac{u_{\alpha}}{2} \right) - N\left(\frac{k}{\lceil L \rceil}, \frac{k+k'}{\lceil L \rceil} \right) \right)$
		$\max_{k \in \{0,\dots,\lceil L\rceil-1\}, k' \in \{1,\dots,\lceil L\rceil-k\}} \left(T_{\frac{k}{\lceil L\rceil},\frac{k+k'}{\lceil L\rceil}}(N) - t_{\lambda_0,\frac{k}{\lceil L\rceil},\frac{k+k'}{\lceil L\rceil}}(1-u_\alpha) \right)$

The present overview notably enables to highlight two main phase transitions in minimax separation rates. A phase transition from the smallest parametric rate order $1/\sqrt{L}$ to the intermediate rate order $\sqrt{\log \log L/L}$, due to adaptation to both height and length of the bump when dealing with the bump detection problem (BDP), or both height and location of the jump (which is in fact equivalent to adaptation to the bump length here) when dealing with the jump detection problem (JDP). A similar phase transition was already known in the independent Gaussian model when dealing with the JDP as explained in the introduction (see [54] and [105]). But the tools used in this Gaussian model, mainly based on Law of Iterated Logarithm exponential inequalities could not be used here, which led us to circumvent the difficulty via new exponential inequalities of Le Guével [71] combined with a dyadic type scan aggregation approach. Two points which seem important to us here are: first, in the BDP, adaptation to both bump height and location can be conducted without any additional cost as soon as the bump length is known; second, when adaptation to the length in the BDP or the location in the JDP is considered, the knowledge of the bump or jump height suffices to cancel any price to pay for adaptation. Up to our knowledge, such results were not known, even in classical Gaussian models. Constructing minimax adaptive tests actually required a careful analysis of the shifted

Poisson process. Then, a phase transition from the intermediate rate order $\sqrt{\log \log L/L}$ to the largest rate order $\sqrt{\log L/L}$, due to adaptation to both position and length of the bump when dealing with the BDP. Notice that this rate is so large that additional adaptation to the height has no supplementary cost. Notice also that similar minimax separation rates were already known in the independent Gaussian model: Arias-Castro et al. [5] handled the case where the height is unknown, while Brunel [19] handled the case where the height is unknown, while Brunel [19] handled the case where the height is unknown CUSUM statistics. From this angle, our study provides nonasymptotic and Poisson processes counterparts for the Gaussian tools used in [5] and [19]. But we furthermore introduce, in the unknown height case, a novel scan aggregated quadratic statistic, whose interest is discussed in Section 2.2.

3 Detecting an abrupt, possibly transitory, change in an unknown intensity

We now turn to the problem of detecting an abrupt change in the intensity of the Poisson process N when its constant baseline is not assumed to be known anymore. This detection problem probably more largely fits applications, especially with epidemiological data for which it is often more realistic not to assume that the baseline intensity of the underlying Poisson process is known. In the present section, we therefore consider the null hypothesis expressed as (H_0) " $\lambda \in S_0^u[R]$ ", where $S_0^u[R]$ is the set of all possible constant intensities upper bounded by a given R > 0. As in the above section, we consider various alternative hypotheses, that are defined according to the persistent or transitory nature of the change, and its height, location and length knowledge. In order to further cover the full range of alternatives in a unified notation, we introduce for δ^* in $(-R, R) \setminus \{0\}, \tau^*$ in (0, 1) and ℓ^* in $(0, 1 - \tau^*]$ the set $S_{\delta^*, \tau^*, \ell^*}^u[R]$ of intensities with a change of height δ^* , location τ^* and length ℓ^* from an unknown λ_0 in $S_0^u[R]$, and still upper bounded by R,

$$\begin{bmatrix} \mathbf{Alt^{u}.1} \end{bmatrix} \quad \mathcal{S}^{u}_{\delta^{*},\tau^{*},\ell^{*}}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (-\delta^{*} \lor 0, (R-\delta^{*}) \land R], \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta^{*} \mathbb{1}_{(\tau^{*},\tau^{*}+\ell^{*}]}(t) \right\} .$$
(38)

Though it is not as immediate as in Section 2, testing (H_0) versus (H_1) " $\lambda \in \mathcal{S}^u_{\delta^*,\tau^*,\ell^*}[R]$ " also falls within the scope of Neyman-Pearson tests, and an Uniformly Most Powerful Unbiased (UMPU) test can be constructed by using a conditioning trick (see details below). As above, when the question of adaptivity w.r.t. some unknown parameters is tackled, the unknown parameters are replaced by single, double or triple dots in the notation $\mathcal{S}^u_{\delta^*,\tau^*,\ell^*}[R]$.

Notice that for any intensity λ such that $\lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell]}(t)$ for δ in $(-R, R) \setminus \{0\}, \tau$ in $(0, 1), \ell$ in $(0, 1 - \tau]$ and λ_0 in $(-\delta \lor 0, (R - \delta) \land R],$

$$d_2(\lambda, \mathcal{S}_0^u[R]) = |\delta| \sqrt{\ell(1-\ell)}$$
.

Hence, as soon as an alternative intensity has known change height $\delta = \delta^*$ and length $\ell = \ell^*$, the distance $d_2(\lambda, \mathcal{S}_0^u[R])$ is fixed, equal to $|\delta^*| \sqrt{\ell^*(1-\ell^*)}$. Hence, the β -uniform

separation rate of any level α test over $S^{u}_{\delta^*,\tau^*,\ell^*}[R]$ or $S^{u}_{\delta^*,\cdot,\ell^*}[R]$ is either 0 or $+\infty$, and so is the (α,β) -minimax separation rate. In these cases, our tests are not studied from the minimax point of view. As in Section 2, we nevertheless establish conditions, expressed as a sufficient minimal distance $d_2(\lambda, S^u_0[R])$, guaranteeing that their second kind error rate is controlled by β .

3.1 Uniformly most powerful detection of a possibly transitory change with known location and length

Let us first focus on the problem of testing (H_0) " $\lambda \in S_0^u[R]$ " versus (H_1) " $\lambda \in S_{\delta^*,\tau^*,\ell^*}^u[R]$ " with $S_{\delta^*,\tau^*,\ell^*}^u[R]$ defined by (38) for δ^* in $(-R,R) \setminus \{0\}, \tau^*$ in (0,1) and ℓ^* in $(0,1-\tau^*]$. Assume here that λ belongs to $\{\lambda : [0,1] \to (0,R], \exists \delta \in (-R,R), \exists \lambda_0 \in (-\delta \vee 0, (R-\delta) \wedge R], \lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^*+\ell^*]}\} \supset S_0^u[R] \cup S_{\delta^*,\tau^*,\ell^*}^u[R]$. In this model parametrised by (δ,λ_0) in $\{(\delta,\lambda_0), \delta \in (-R,R), \lambda_0 \in (-\delta \vee 0, (R-\delta) \wedge R]\}$, the distribution P_{λ} is dominated by P_1 (see Lemma 1), with a Likelihood Ratio given by

$$(dP_{\lambda}/dP_{1})(N) = e^{(\log(1+\delta/\lambda_{0})N(\tau^{*},\tau^{*}+\ell^{*}]+\log(\lambda_{0})N(0,1]-L(\lambda_{0}+\delta\ell^{*}-1))}$$

Reparametrising the model by $\theta_1 = \log(1 + \delta/\lambda_0)$ and $\theta_2 = \log(\lambda_0)$, this LR becomes

$$(dP_{\lambda}/dP_{1})(N) = e^{-L(e^{\theta_{2}}(1+(e^{\theta_{1}}-1)\ell^{*})-1)}e^{\theta_{1}N(\tau^{*},\tau^{*}+\ell^{*}]+\theta_{2}N(0,1]} .$$
(39)

Our testing problem can then be viewed as a problem of testing (H_0) " $\theta_1 = 0$ " versus (H_1) " $\theta_1 < 0$ " or (H_1) " $\theta_1 > 0$ " (depending on the sign of δ^*) in an exponential model with natural parameters $\theta = (\theta_1, \theta_2)$ and sufficient statistics $(N(\tau^*, \tau^* + \ell^*], N(0, 1])$, and where θ_2 can be interpreted as a nuisance parameter. From (39) and Lemma 2.7.2 of [73] we can deduce that given $N_1 = n$, the conditional distribution of $N(\tau^*, \tau^* + \ell^*]$ defines an exponential family with respect to some measure ν_n , with natural parameter θ_1 , and is in particular free of θ_2 . In this conditional framework, one knows that there exists an UMP test of (H_0) versus (H_1) of the Neyman-Pearson form. Recalling that given $N_1 = n$, $N(\tau^*, \tau^* + \ell^*]$ has the same distribution as a binomial random variable Y_{n,ℓ^*} with parameters (n, ℓ^*) , such conditional Neyman-Pearson tests lead us to consider the unilateral tests defined by

$$\begin{cases} \phi_{1,\alpha}^{u,-}(N) = \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] < b_{N_1,\ell^*}(\alpha)} + \gamma_{(N_1,\ell^*)}^{-}(\alpha) \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] = b_{N_1,\ell^*}(\alpha)} \text{ if } \delta^* < 0 , \\ \phi_{1,\alpha}^{u,+}(N) = \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] > b_{N_1,\ell^*}(1-\alpha)} + \gamma_{(N_1,\ell^*)}^{+}(1-\alpha) \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] = b_{N_1,\ell^*}(1-\alpha)} \text{ if } \delta^* > 0 , \end{cases}$$

$$(40)$$

where for all n in \mathbb{N} , $b_{n,\ell^*}(u)$ denotes the *u*-quantile of the distribution of Y_{n,ℓ^*} , and

$$\gamma_{(n,\ell^*)}^-(u) = (u - \mathbb{P}(Y_{n,\ell^*} < b_{n,\ell^*}(u))) / \mathbb{P}(Y_{n,\ell^*} = b_{n,\ell^*}(u)), \ \gamma_{(n,\ell^*)}^+(u) = 1 - \gamma_{(n,\ell^*)}^-(u) \ . \tag{41}$$

From Theorem 4.4.1 in [73] and the remark below its proof, we obtain the following result.

Proposition 24 (Uniformly Most Powerful Unbiased tests). Let $L \ge 1$, α in (0,1), δ^* in $(-R, R) \setminus \{0\}$, τ^* in (0,1) and ℓ^* in $(0,1-\tau^*]$. For the problem of testing (H_0) v.s. (H_1) " $\lambda \in S^u_{\delta^*,\tau^*,\ell^*}[R]$ ", let $\phi^u_{1,\alpha}$ be the test $\phi^{u,-}_{1,\alpha}$ if $\delta^* < 0$, $\phi^{u,+}_{1,\alpha}$ if $\delta^* > 0$ (see (40)). Then $E_{\lambda} \left[\phi^u_{1,\alpha}(N) \middle| N_1 = n \right] = \alpha$ for all λ in $S^u_0[R]$ and $\phi^u_{1,\alpha}$ is an UMPU test. In order to follow the same line as the minimax results obtained when regarding other alternative hypotheses with unknown height and/or length change, we further study which minimal distance $d_2(\lambda, \mathcal{S}_0^u[R])$ guarantees a second kind error rate control.

Proposition 25 (Second kind error rates control for [Alt^u.1]). Let $L \ge 1$, α in (0,1), δ^* in $(-R, R) \setminus \{0\}$, τ^* in (0, 1) and ℓ^* in $(0, 1 - \tau^*]$. For the problem of testing (H_0) v.s. (H_1) " $\lambda \in S^u_{\delta^*, \tau^*, \ell^*}[R]$ ", let $\phi^u_{1,\alpha}$ be the test $\phi^{u,-}_{1,\alpha}$ if $\delta^* < 0$, $\phi^{u,+}_{1,\alpha}$ if $\delta^* > 0$. Then there exists $C(\alpha, \beta, R, \ell^*) > 0$ such that $P_{\lambda}(\phi^u_{1,\alpha}(N) = 0) \le \beta$ if λ belongs to $S^u_{\delta^*, \tau^*, \ell^*}[R]$ with

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge C(\alpha, \beta, R, \ell^*) / \sqrt{L} \quad . \tag{42}$$

Comments. Noticing that for any λ in $S^{u}_{\delta^{*},\tau^{*},\ell^{*}}[R]$, $d_{2}(\lambda, S^{u}_{0}[R]) = |\delta^{*}|\sqrt{\ell^{*}(1-\ell^{*})}$, the above proposition implies that if $L \geq C^{2}(\alpha, \beta, R, \ell^{*})/(\delta^{*2}\ell^{*}(1-\ell^{*}))$, then $P_{\lambda}(\phi^{u}_{1,\alpha}(N) = 0) \leq \beta$. Therefore, in this case, the β -uniform separation rate of $\phi^{u}_{1,\alpha}$ over $S^{u}_{\delta^{*},\tau^{*},\ell^{*}}[R]$ is equal to 0, and so is the corresponding (α, β) -minimax separation rate $\mathrm{mSR}_{\alpha,\beta}(S^{u}_{\delta^{*},\tau^{*},\ell^{*}}[R])$. In order to address the question of adaptation to the change height, we consider the problem of testing (H_{0}) v.s. (H_{1}) " $\lambda \in S^{u}_{\cdot,\tau^{*},\ell^{*}}[R]$ ", where for $R > 0, \tau^{*}$ in (0, 1) and ℓ^{*} in $(0, 1 - \tau^{*}]$,

$$\begin{bmatrix} \mathbf{Alt}^{\mathbf{u}} \cdot \mathbf{2} \end{bmatrix} \quad \mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (0,R], \ \exists \delta \in (-\lambda_{0}, R - \lambda_{0}] \setminus \{0\}, \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta \mathbb{1}_{(\tau^{*},\tau^{*}+\ell^{*}]}(t) \right\} .$$
(43)

Unsurprisingly, with the Bayesian arguments already used to prove Proposition 3, one obtains a lower bound for the minimax separation rate over $S^{u}_{;\tau^{*},\ell^{*}}[R]$ of the parametric order $1/\sqrt{L}$.

Proposition 26 (Minimax lower bound for [Alt^u.2]). Let α and β in (0,1), R > 0, τ^* in (0,1) and ℓ^* in $(0, 1 - \tau^*]$. For all $L \ge (2 \log C_{\alpha,\beta}/(R\ell^*))$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R]\right) \geq \sqrt{R(1-\ell^{*})\log C_{\alpha,\beta}/(2L)}, \text{ with } C_{\alpha,\beta} = 1 + 4(1-\alpha-\beta)^{2}.$$

In order to prove that this lower bound is sharp, we construct two minimax adaptive tests. The first one is based on the linear statistic $N(\tau^*, \tau^* + \ell^*]$ and is very similar in spirit to the test $\phi_{2,\alpha}^{(1)}$ defined by (9), except that the associated critical values are based on the conditional distribution of $N(\tau^*, \tau^* + \ell^*]$ given $N_1 = n$ under (H_0) instead of the unconditional distribution (which is not free from the unknown constant baseline intensity under (H_0)). Let

$$\phi_{2,\alpha}^{u(1)}(N) = \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] > b_{N_1,\ell^*}(1-\alpha_1)} + \gamma_{(N_1,\ell^*)}^+(1-\alpha_1) \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] = b_{N_1,\ell^*}(1-\alpha_1)} \\ + \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] < b_{N_1,\ell^*}(\alpha_2)} + \gamma_{(N_1,\ell^*)}^-(\alpha_2) \mathbb{1}_{N(\tau^*,\tau^*+\ell^*] = b_{N_1,\ell^*}(\alpha_2)} , \quad (44)$$

where $\gamma_{(n,\ell^*)}^+$ and $\gamma_{(n,\ell^*)}^-$ are defined by (41), $b_{n,\ell^*}(u)$ denotes the *u*-quantile of the binomial distribution with parameters (n,ℓ^*) for all n in \mathbb{N} , and α_1 and α_2 in (0,1) are determined by

$$\begin{cases} \alpha_1 + \alpha_2 = \alpha , \\ E_{\lambda} \big[N(\tau^*, \tau^* + \ell^*] \phi_{2,\alpha}^{u(1)}(N) \big| N_1 = n \big] = \alpha E_{\lambda} \big[N(\tau^*, \tau^* + \ell^*] \big| N_1 = n \big] \forall \lambda \in \mathcal{S}_0^u[R] . \end{cases}$$
(45)

Theorem 4.4.1 of [73] again (but considering the bilateral test) shows that $\phi_{2,\alpha}^{u(1)}$ is UMPU. The second one is based on a quadratic statistic deduced from an estimation of the \mathbb{L}_2 distance between λ in $S^u_{,\tau^*,\ell^*}[R]$ and $S^u_0[R]$. For $0 < \tau_1 < \tau_2 \leq 1$, we define $\psi_0 = \mathbb{1}_{[0,1]}$ and $\psi_{\tau_1,\tau_2} = (\mathbb{1}_{(\tau_1,\tau_2]} - (\tau_2 - \tau_1)\psi_0)/\sqrt{(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)}$, so that $(\psi_0, \psi_{\tau_1,\tau_2})$ is an orthonormal family. Denoting by Π_{W_0} and $\Pi_{W_{\tau_1,\tau_2}}$ the orthogonal projections onto $W_0 =$ Vect (ψ_0) and $W_{\tau_1,\tau_2} = \text{Vect}(\psi_0, \psi_{\tau_1,\tau_2})$ in $\mathbb{L}_2([0,1])$ respectively, the quadratic statistic

$$T_{\tau_1,\tau_2}'(N) = \frac{1}{L^2(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)} \Big((N(\tau_1, \tau_2] - (\tau_2 - \tau_1)N(0, 1])^2 + (\tau_2 - \tau_1)(N(\tau_1, \tau_2] - (\tau_2 - \tau_1)N(0, 1]) - (1 - (\tau_2 - \tau_1))N(\tau_1, \tau_2] \Big) , \quad (46)$$

is an unbiased estimator of $\|\Pi_{W_{\tau_1,\tau_2}}(\lambda - \Pi_{W_0}(\lambda))\|_2^2$. We therefore consider the particular statistic $T'_{\tau^*,\tau^*+\ell^*}(N)$ which is an unbiased estimator of the squared \mathbb{L}_2 -distance between λ in $\mathcal{S}^u_{\cdot,\tau^*,\ell^*}[R]$ and the set of constant intensities, leading to the test defined by

$$\phi_{2,\alpha}^{u(2)}(N) = \mathbb{1}_{T'_{\tau^*,\tau^*+\ell^*}(N) > t'_{N_1,\tau^*,\tau^*+\ell^*}(1-\alpha)} , \qquad (47)$$

where $t'_{n,\tau_1,\tau_2}(u)$ is the *u*-quantile of the distribution of $T'_{\tau_1,\tau_2}(N)$ given $N_1 = n$ under (H_0) . Since this conditional distribution under (H_0) is the distribution of the renormalised U-statistic $L^{-2} \sum_{i\neq j=1}^{n} \psi_{\tau_1,\tau_2}(U_i) \psi_{\tau_1,\tau_2}(U_j)$ based on a *n*-sample (U_1,\ldots,U_n) of i.i.d. uniform random variables, we use an exponential inequality for *U*-statistics of order 2 due to Reynaud-Bouret and Houdré [62] to control the quantiles $t'_{n,\tau_1,\tau_2}(u)$ in theory (see Lemma 53), and Monte-Carlo methods to evaluate them in practice.

Proposition 27 (Minimax upper bound for [Alt^u.2]). Let $L \ge 1$, α , β in (0,1), R > 0, τ^* in (0,1) and ℓ^* in $(0,1-\tau^*]$. Let $\phi_{2,\alpha}^{u(1/2)}$ be one of the tests $\phi_{2,\alpha}^{u(1)}$ and $\phi_{2,\alpha}^{u(2)}$ of (H_0) versus (H_1) " $\lambda \in S^u_{:,\tau^*,\ell^*}[R]$ " respectively defined by (44)-(45) and (47). Then $\phi_{2,\alpha}^{u(1/2)}$ is of level α , that is $\sup_{\lambda \in S_0^u[R]} P_{\lambda}(\phi_{2,\alpha}^{u(1/2)}(N) = 1) \le \alpha$ ($\phi_{2,\alpha}^{u(1)}$ is even of size α). Moreover, there exists $C(\alpha, \beta, R, \ell^*) > 0$ such that

$$\operatorname{SR}_{\beta}(\phi_{2,\alpha}^{u(1/2)}, \mathcal{S}^{u}_{;\tau^{*},\ell^{*}}[R]) \leq C(\alpha, \beta, R, \ell^{*})/\sqrt{L}$$
,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R]) \leq C(\alpha,\beta,R,\ell^{*})/\sqrt{L}.$

Comments. This result proves that both tests $\phi_{2,\alpha}^{u(1)}$ and $\phi_{2,\alpha}^{u(2)}$ are β -minimax over the set of alternatives $S^{u}_{;\tau^{*},\ell^{*}}[R]$, where the height of the change is unknown, with an optimal uniform separation rate of order $1/\sqrt{L}$, as expected regarding results for $\phi_{2,\alpha}^{(1)}$ and $\phi_{2,\alpha}^{(2)}$ in Section 2.

Notice that this study involves the particular non transitory change or jump detection problem, with a known change location, taking $\ell^* = 1 - \tau^*$. Following the same layout as in Section 2, we investigate the jump detection problem with unknown location in Section 3.4.1.

3.2 Minimax detection of a transitory change with known length

In this subsection, we deal with the problem of testing the null hypothesis (H_0) " $\lambda \in S_0^u[R]$ " versus alternatives where the length of the change from the unknown baseline

intensity is known, with adaptation to the change location, and with or without adaptation to the change height. We therefore introduce for ℓ^* in (0, 1) and δ^* in $(-R, R) \setminus \{0\}$ the two following sets:

$$\begin{bmatrix} \mathbf{Alt}^{\mathbf{u}}.\mathbf{3} \end{bmatrix} \mathcal{S}^{u}_{\delta^{*},\cdots,\ell^{*}}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (-\delta^{*} \lor 0, (R-\delta^{*}) \land R], \\ \exists \tau \in (0,1-\ell^{*}), \ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta^{*} \mathbb{1}_{(\tau,\tau+\ell^{*}]}(t) \right\} , \quad (48)$$

$$\begin{bmatrix} \mathbf{Alt}^{\mathbf{u}}.4 \end{bmatrix} \mathcal{S}^{u}_{\cdot,\cdot\cdot,\ell^{*}}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (0,R], \ \exists \delta \in (-\lambda_{0}, R - \lambda_{0}] \setminus \{0\}, \\ \exists \tau \in (0,1-\ell^{*}), \ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta \mathbb{1}_{(\tau,\tau+\ell^{*}]}(t) \right\}.$$
(49)

Adapting the ideas of Section 2.3, we handle the question of adaptation to the change location τ^* by introducing aggregated tests based on the same linear and quadratic statistics as those used for testing (H_0) versus (H_1) " $\lambda \in \mathcal{S}^u_{\delta^*,\tau^*,\ell^*}[R]$ " above. We thus set on the one hand

$$\begin{cases} \phi_{3,\alpha}^{u(1)-}(N) = \mathbb{1}_{\min_{\tau \in [0,1-\ell^* \wedge (1/2)]} N(\tau,\tau+\ell^* \wedge (1/2)] < b^-_{N_1,\ell^* \wedge (1/2)}(\alpha) \\ \phi_{3,\alpha}^{u(1)+}(N) = \mathbb{1}_{\max_{\tau \in [0,1-\ell^* \wedge (1/2)]} N(\tau,\tau+\ell^* \wedge (1/2)] > b^+_{N_1,\ell^* \wedge (1/2)}(1-\alpha) \\ \end{cases}$$
(50)

$$\phi_{4,\alpha}^{u(1)}(N) = \phi_{3,\alpha/2}^{u(1)-}(N) \lor \phi_{3,\alpha/2}^{u(1)+}(N) \quad , \tag{51}$$

where $b_{n,\ell}^{-}(u)$ and $b_{n,\ell}^{+}(u)$ respectively denote the *u*-quantiles of the conditional distributions of $\min_{\tau \in [0,1-\ell]} N(\tau, \tau + \ell]$ and $\max_{\tau \in [0,1-\ell]} N(\tau, \tau + \ell]$ given $N_1 = n$ under (H_0) , for all n in \mathbb{N} and ℓ in (0, 1/2]. Then, we introduce on the other hand the aggregated test

$$\phi_{3/4,\alpha}^{u(2)}(N) = \mathbb{1}_{\left\{\max_{k \in \{0,\dots,\lceil (1-\ell^*)M\rceil - 1\}} \left(T'_{\frac{k}{M},\frac{k}{M}+\ell^*}(N) - t'_{N_1,\frac{k}{M},\frac{k}{M}+\ell^*}(1-u_{\alpha})\right) > 0\right\}},$$
(52)

where $M = \lceil 2/(\ell^*(1-\ell^*)) \rceil$, $u_{\alpha} = \alpha/\lceil (1-\ell^*)M \rceil$, $T'_{k/M,k/M+\ell^*}$ is defined by (46) and $t'_{n,k/M,k/M+\ell^*}(u)$ is the *u*-quantile of $T'_{k/M,k/M+\ell^*}(N)$ given $N_1 = n$ under (H_0) .

Since the set $S^u_{\delta^*,..,\ell^*}[R]$ of (48) is composed of alternatives with known change height δ^* and length ℓ^* , the distance between any of its elements and $S^u_0[R]$ is fixed, equal to $|\delta^*|\sqrt{\ell^*(1-\ell^*)}$. Hence, for this set, we only provide sufficient conditions for the tests $\phi^{u(1)+}_{3,\alpha}$, $\phi^{u(1)-}_{3,\alpha}$ and $\phi^{u(2)}_{3/4,\alpha}$ to have a second kind error rate controlled by a prescribed level β when $\lambda \in S^u_{\delta^*,..,\ell^*}[R]$. As in Proposition 6, the key points of the proofs of the following results for $\phi^{u(1)-}_{3,\alpha}$ and $\phi^{u(1)+}_{3,\alpha}$ are sharp lower or upper bounds for the involved quantiles $b^-_{n,\ell^*\wedge(1/2)}(\alpha)$ and $b^+_{n,\ell^*\wedge(1/2)}(\alpha)$, which are deduced from inequalities for oscillations of empirical processes found in [95] (see Lemma 54 for details). The result for $\phi^{u(2)}_{3/4,\alpha}$ relies on the control of the quantile $t'_{n,\tau_1,\tau_2}(u_{\alpha})$ obtained in Lemma 53 via the exponential inequality for U-statistics of order 2 due to Reynaud-Bouret and Houdré [62], as in Proposition 27.

Proposition 28 (Second kind error rate control for [Alt^u.3]). Let $L \geq 1$, α and β in (0,1), ℓ^* in (0,1) and δ^* in $(-R,R) \setminus \{0\}$, and consider the problem of testing (H_0) v.s. (H_1) " $\lambda \in S^u_{\delta^*, \dots, \ell^*}[R]$ ". Let $\phi^{u(1/2)}_{3,\alpha}$ be one of the tests $\phi^{u(1)+}_{3,\alpha}$ or $\phi^{u(2)}_{3/4,\alpha}$ if $\delta^* > 0$, and one of the tests $\phi^{u(1)-}_{3,\alpha}$ or $\phi^{u(2)}_{3/4,\alpha}$ if $\delta^* < 0$ (see (50) and (52)). The test $\phi^{u(1/2)}_{3,\alpha}$ is of level α , that is $\sup_{\lambda_0 \in S^u_0[R]} P_{\lambda_0}(\phi^{u(1/2)}_{3,\alpha}(N) = 1) \leq \alpha$. Moreover, there exists $C(\alpha, \beta, R, \delta^*, \ell^*) > 0$ such that $P_{\lambda}(\phi^{u(1/2)}_{3,\alpha}(N) = 0) \leq \beta$ as soon as λ belongs to $S^u_{\delta^*, \dots, \ell^*}[R]$ with

$$d_2(\lambda, \mathcal{S}^u_0[R]) \ge C(\alpha, \beta, R, \delta^*, \ell^*) / \sqrt{L} .$$

Comments. Remarking that for λ in $\mathcal{S}^{u}_{\delta^{*},\dots,\ell^{*}}[\lambda_{0}], d_{2}(\lambda, \mathcal{S}^{u}_{0}[R]) = |\delta^{*}|\sqrt{\ell^{*}(1-\ell^{*})}$, Proposition 28 provides a sufficient value $L_{0}(\alpha, \beta, R, \delta^{*}, \ell^{*})$ for L so that the second kind error rates of the three tests is controlled by β . If $L \geq L_{0}(\alpha, \beta, R, \delta^{*}, \ell^{*})$, their β -uniform separation rates over $\mathcal{S}^{u}_{\delta^{*},\dots,\ell^{*}}[R]$ is equal to 0, as well as the (α, β) -minimax separation rate.

Now considering the alternative set $S^{u}_{\cdot,\cdot,\ell^*}[R]$, that is the change height adaptation issue, the following lower bound is directly deduced from the lower bound for $\mathrm{mSR}_{\alpha,\beta}(S^{u}_{\cdot,\tau^*,\ell^*}[R])$ and the monotonicity property of the minimax separation rate recalled in Lemma 42.

Corollary 29 (Minimax lower bound for [Alt^u.4]). Let α and β in (0,1), R > 0 and ℓ^* in (0,1). For all $L \ge (2 \log C_{\alpha,\beta}/(R\ell^*))$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\cdot\cdot,\ell^{*}}[R]\right) \geq \sqrt{R(1-\ell^{*})\log C_{\alpha,\beta}/(2L)}, \text{ with } C_{\alpha,\beta} = 1 + 4(1-\alpha-\beta)^{2}.$$

Proposition 30 (Minimax upper bounds for [Alt^u.4]). Let $L \ge 1$, α, β in (0, 1), R > 0and ℓ^* in (0, 1). Let $\phi_{4,\alpha}^{u(1/2)}$ be one of the tests $\phi_{4,\alpha}^{u(1)}$ and $\phi_{3/4,\alpha}^{u(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\cdot,\cdot\cdot,\ell^*}^u[R]$ ", defined by (51) and (52). $\phi_{4,\alpha}^{u(1/2)}$ is of level α , that is $\sup_{\lambda_0 \in S_0^u[R]} P_{\lambda_0}(\phi_{4,\alpha}^{u(1/2)}(N) = 1) \le \alpha$, and there exists $C(\alpha, \beta, R, \ell^*) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{4,\alpha}^{u(1/2)}, \mathcal{S}_{\cdot,\cdot,\ell^*}^{u}[R]\right) \leq C(\alpha, \beta, R, \ell^*)/\sqrt{L}$$
,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\cdot,\ell^{*}}[R]\right) \leq C(\alpha,\beta,R,\ell^{*})/\sqrt{L}.$

Comments. Proposition 30 and Corollary 29 mean that the tests $\phi_{4,\alpha}^{u(1)}$ and $\phi_{3/4,\alpha}^{u(2)}$ are minimax. Together with the ones obtained for [Alt^u.2], the two above results finally mean that, as when the baseline intensity is known, adaptation with respect to the change location can be achieved with a minimax separation rate of the parametric order, that is without any additional price to pay (possibly except multiplicative constants) as soon as the only change length is known.

3.3 Minimax detection of a transitory change with known location

We consider the problem of testing the null hypothesis (H_0) " $\lambda \in S_0^u[R]$ " versus alternative hypotheses where the location of the change from the baseline intensity is known, with adaptation to the change length, and with or without adaptation to the height. As in Section 2.4, we see that adaptation to the length can be done without any incidence on the minimax separation rate order, while adaptation to both height and length leads to a cost factor of order $\sqrt{\log \log L}$.

3.3.1 Known change height

Let us first investigate the problem of testing (H_0) versus (H_1) " $\lambda \in \mathcal{S}^u_{\delta^*, \tau^*, \dots}[R]$ ", where for R > 0, δ^* in $(-R, R) \setminus \{0\}$ and τ^* in (0, 1),

$$\begin{bmatrix} \mathbf{Alt}^{\mathbf{u}}.\mathbf{5} \end{bmatrix} \mathcal{S}^{u}_{\delta^{*},\tau^{*},\cdots}[R] = \{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (-\delta^{*} \lor 0, (R-\delta^{*}) \land R], \ \exists \ell \in (0,1-\tau^{*}), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta^{*} \mathbb{1}_{(\tau^{*},\tau^{*}+\ell]}(t) \} .$$
(53)

As in Section 2.4, the most intricate point here is the construction of a test achieving the minimax separation rate over $S^u_{\delta^*,\tau^*,\dots}[R]$ which will be proved to be of the parametric order $1/\sqrt{L}$, and therefore necessarily taking the knowledge of the change height δ^* into account. The test we propose is largely inspired from the aggregated test $\phi_{5,\alpha}$ defined by (20), where the test statistic is slightly adapted to compensate for the lack of the baseline intensity knowledge. Since the critical value can not be taken as a quantile of the test statistic, whose distribution under the null hypothesis is not free from the unknown baseline intensity anymore, we use the same conditioning trick as in the above subsections.

Proposition 31 (Minimax lower bound for [Alt^u.5]). Let α, β in (0, 1) with $\alpha + \beta < 1$, $R > 0, \delta^*$ in $(-R, R) \setminus \{0\}, \tau^*$ in (0, 1). For $L > ((R - \delta^*) \wedge R) \log C_{\alpha,\beta} / (2\delta^{*2}\tau^*(1 - \tau^*))$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\delta^{*},\tau^{*},\cdots}[R]\right) \geq \sqrt{\left(\left(R-\delta^{*}\right)\wedge R\right)\log C_{\alpha,\beta}/\left(2L\right)}, \text{ with } C_{\alpha,\beta} = 1 + 4(1-\alpha-\beta)^{2}.$$

Let us now introduce the test

$$\phi_{5,\alpha}^{u}(N) = \mathbb{1}_{\left\{\sup_{\ell \in (0,1-\tau^{*})} S_{\delta^{*},\tau^{*},\tau^{*}+\ell}^{\prime}(N) > s_{N_{1},\delta^{*},\tau^{*},L}^{\prime+}(1-\alpha)\right\}},$$
(54)

where $S'_{\delta^*,\tau_1,\tau_2}(N)$ is the statistic defined for $0 \le \tau_1 < \tau_2 \le 1$ by

$$S_{\delta^*,\tau_1,\tau_2}'(N) = \operatorname{sgn}(\delta^*) \Big(N(\tau_1,\tau_2] - (\tau_2 - \tau_1) N_1 \Big) - |\delta^*| L(\tau_2 - \tau_1) (1 - \tau_2 + \tau_1) / 2 , \quad (55)$$

and $s_{n,\delta^*,\tau^*,L}^{'+}(u)$ is the *u*-quantile of $\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}^{'}(N)$ given $N_1 = n$ under (H_0) . The main argument of the following upper bound is a control of the conditional quantile $s_{n,\delta^*,\tau^*,L}^{'+}(1-\alpha)$, provided in Lemma 55, and which is deduced from a refined Bernstein inequality based on some chaining techniques.

Proposition 32 (Minimax upper bound for [Alt^u.5]). Let $L \ge 1$, α, β in $(0, 1), R > 0, \delta^*$ in $(-R, R) \setminus \{0\}$ and τ^* in (0, 1). Let $\phi_{5,\alpha}^u$ be the test of (H_0) versus (H_1) " $\lambda \in \mathcal{S}_{\delta^*, \tau^*, \dots}^u[R]$ " defined by (54). $\phi_{5,\alpha}^u$ is of level α , that is $\sup_{\lambda_0 \in \mathcal{S}_0^u[R]} P_{\lambda_0} \left(\phi_{5,\alpha}^u(N) = 1\right) \le \alpha$. Moreover, there exists a constant $C(\alpha, \beta, R, \delta^*) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{5,\alpha}^{u}, \mathcal{S}_{\delta^{*},\tau^{*},\cdots}^{u}[R]\right) \leq C(\alpha, \beta, R, \delta^{*})/\sqrt{L}$$

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\delta^{*},\tau^{*},\cdots}[R]\right) \leq C(\alpha,\beta,R,\delta^{*})/\sqrt{L}$.

3.3.2 Unknown change height

Now addressing the question of adaptation to the change height and length together, we consider for R > 0 and τ^* in (0, 1) the alternative set

$$\begin{bmatrix} \mathbf{Alt}^{\mathbf{u}}.\mathbf{6} \end{bmatrix} \mathcal{S}^{u}_{\cdot,\tau^{*},\cdots}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (0,R], \ \exists \delta \in (-\lambda_{0}, R-\lambda_{0}] \setminus \{0\}, \\ \exists \ell \in (0,1-\tau^{*}), \ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta \mathbb{1}_{(\tau^{*},\tau^{*}+\ell]}(t) \right\}.$$
(56)

For the problem of testing (H_0) " $\lambda \in \mathcal{S}_0^u[R]$ " versus (H_1) " $\lambda \in \mathcal{S}_{\cdot,\tau^*,\ldots}^u[R]$ ", we obtain the following lower bound.

Proposition 33 (Minimax lower bound for [Alt^u.6]). Let α, β in (0, 1) with $\alpha + \beta < 1/2$, R > 0 and τ^* in (0, 1). There exists $L_0(\alpha, \beta, R, \tau^*) > 0$ such that for $L \ge L_0(\alpha, \beta, R, \tau^*)$,

$$\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\tau^{*},\cdots}[R]\right) \geq \sqrt{R\tau^{*}\log\log L/(2L)}$$
.

Let us assume now that $L \ge 3$. In order to prove that the above lower bound is of sharp order (with respect to L), we construct two aggregated tests: a first one based on a linear statistic and a second one based on a quadratic statistic as in Section 3.2. We thus consider the discrete subset of $(0, 1 - \tau^*)$ of the dyadic form

$$\{\ell_{\tau^*,k} = (1 - \tau^*) 2^{-k}, k \in \{1, \dots, \lfloor \log_2 L \rfloor\}\},\$$

and $u_{\alpha} = \alpha / \lfloor \log_2(L) \rfloor$, which allows to define the two following tests:

$$\phi_{6,\alpha}^{u(1)}(N) = \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(N(\tau^*,\tau^* + \ell_{\tau^*,k}] - \ell_{\tau^*,k}N_1 - \bar{b}_{N_1,\ell_{\tau^*,k}} \left(1 - \frac{u_\alpha}{2}\right) \right) > 0 \right\}} \\ \vee \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(\bar{b}_{N_1,\ell_{\tau^*,k}} \left(\frac{u_\alpha}{2}\right) - N(\tau^*,\tau^* + \ell_{\tau^*,k}] + \ell_{\tau^*,k}N_1 \right) > 0 \right\}}, \quad (57)$$

where $\bar{b}_{n,\tau_2-\tau_1}(u)$ is the *u*-quantile of $N(\tau_1,\tau_2] - (\tau_2 - \tau_1)N_1$ given $N_1 = n$ under (H_0) , that is the *u*-quantile of a recentered binomial distribution with parameters $(n, (\tau_2 - \tau_1))$, whose sharp bound is obtained via Bennett's inequality (see Lemma 56 for details), and

$$\phi_{6,\alpha}^{u(2)}(N) = \mathbb{1}_{\left\{\max_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} \left(T'_{\tau^*,\tau^* + \ell_{\tau^*,k}}(N) - t'_{N_1,\tau^*,\tau^* + \ell_{\tau^*,k}}(1 - u_\alpha)\right) > 0\right\}},$$
(58)

where $T'_{\tau_1,\tau_2}(N)$ is the quadratic statistic defined in (46) and $t'_{n,\tau_1,\tau_2}(u)$ still denotes the *u*-quantile of its conditional distribution given $N_1 = n$ under (H_0) .

Proposition 34 (Minimax upper bound for [Alt^u.6]). Let α, β in $(0,1), R > 0, \tau^*$ in (0,1), and let $\phi_{6,\alpha}^{u(1/2)}$ be one of the tests $\phi_{6,\alpha}^{u(1)}$ and $\phi_{6,\alpha}^{u(2)}$ of (H_0) versus (H_1) " $\lambda \in S_{\cdot,\tau^*,\cdots}^u[R]$ " respectively defined by (57) and (58). Then $\phi_{6,\alpha}^{u(1/2)}$ is of level α , that is $\sup_{\lambda_0 \in S_0^u[R]} P_{\lambda_0}\left(\phi_{6,\alpha}^{u(1/2)}(N) = 1\right) \leq \alpha$. Moreover, there exists $C(\alpha, \beta, R, \tau^*) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{6,\alpha}^{u(1/2)}, \mathcal{S}_{\cdot,\tau^{*},\cdots}^{u}[R]\right) \leq C(\alpha, \beta, R, \tau^{*})\sqrt{\log\log L/L}$$
,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\tau^{*},\cdots}[R]\right) \leq C(\alpha,\beta,R,\tau^{*})\sqrt{\log\log L/L}.$

3.4 Minimax detection of a possibly transitory change with unknown location and length

Let us discuss as final stage the problem of testing the null hypothesis (H_0) " $\lambda \in S_0^u[R]$ " versus alternatives where both location and length of the change from the unknown baseline intensity are not known, distinguishing as in Section 2.5 the transitory change case from the non transitory change particular case.

From the minimax point of view, we will emphasize that regardless if the baseline intensity is known or not, adaptation to both location and length of the change has the same minimax separation rate cost of order $\sqrt{\log L}$ in the transitory change case, and of order $\sqrt{\log \log L}$ at most (possibly cancelled by the change height knowledge) in the non transitory change case.

Since the non transitory change or jump detection problem, that we here study first, can be viewed as perfectly symmetrical to the transitory change with known location detection problem, our study uses tools and arguments that are very similar to the ones used in Section 3.3.

3.4.1 Non transitory change

In order to investigate the problem of detecting a non-transitory change with unknown location, but known height, we introduce for R > 0 and δ^* in $(-R, R) \setminus \{0\}$ the alternative set

$$\begin{bmatrix} \mathbf{Alt^{u}.7} \end{bmatrix} \mathcal{S}^{u}_{\delta^{*},..,1-..}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (-\delta^{*} \lor 0, (R-\delta^{*}) \land R], \ \exists \tau \in (0,1), \\ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta^{*} \mathbb{1}_{(\tau,1]}(t) \right\} .$$
(59)

Considering the problem of testing the null hypothesis (H_0) " $\lambda \in \mathcal{S}_0^u[R]$ " versus the alternative hypothesis (H_1) " $\lambda \in \mathcal{S}_{\delta^*,..,1-..}^u[R]$ ", we obtain the following lower bound.

Proposition 35 (Minimax lower bound for [Alt^u.7]). Let α, β in (0, 1) with $\alpha + \beta < 1$, R > 0 and δ^* in $(-R, R) \setminus \{0\}$. For all $L > 2((R - \delta^*) \wedge R) \log C_{\alpha,\beta} / \delta^{*2}$,

$$\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}^{u}_{\delta^{*},\dots,1-\dots}[R]) \geq \sqrt{((R-\delta^{*})\wedge R)\log C_{\alpha,\beta}/(2L)}, \text{ with } C_{\alpha,\beta} = 1 + 4(1-\alpha-\beta)^{2}.$$

Following the study and the notation of Section 3.3, we define the test

$$\phi_{7,\alpha}^{u}(N) = \mathbb{1}_{\left\{\sup_{\tau \in (0,1)} S'_{\delta^{*},\tau,1}(N) > s'^{+}_{N_{1},\delta^{*},L}(1-\alpha)\right\}}$$
(60)

where $S'_{\delta^*,\tau_1,\tau_2}(N)$ is the statistic defined for $0 \leq \tau_1 < \tau_2 \leq 1$ by (55) and $s'_{n,\delta^*,L}(u)$ is the *u*-quantile of the conditional distribution of $\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)$ given $N_1 = n$ under (H_0) .

Notice that a control of this conditional quantile $s_{n,\delta^*,L}^{'+}(1-\alpha)$, provided in Lemma 57, and deduced from the same chaining trick combined with Bernstein's inequality as in the proof of Lemma 55, is the main argument of the following result.

Proposition 36 (Minimax upper bound for [Alt^u.7]). Let $L \ge 1$, α and β in (0, 1), R > 0and δ^* in $(-R, R) \setminus \{0\}$. Let $\phi_{7,\alpha}^u$ be the test of (H_0) versus (H_1) " $\lambda \in S_{\delta^*, \dots, 1-\dots}^u$ [R]" defined by (60). Then $\phi_{7,\alpha}^u$ is of level α , that is $\sup_{\lambda_0 \in S_0^u[R]} P_{\lambda_0}(\phi_{7,\alpha}^u(N) = 1) \le \alpha$. Moreover, there exists a constant $C(\alpha, \beta, R, \delta^*) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{7,\alpha}^{u}, \mathcal{S}_{\delta^{*}, \dots, 1-\dots}^{u}[R]\right) \leq C(\alpha, \beta, R, \delta^{*})/\sqrt{L}$$
,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\delta^{*},..,1-..}[R]\right) \leq C(\alpha,\beta,R,\delta^{*})/\sqrt{L}$.

To address the question of adaptation to the change height, we introduce the alternative set

$$\begin{bmatrix} \mathbf{Alt^{u}.8} \end{bmatrix} \mathcal{S}^{u}_{\cdot,\cdot\cdot,1-\cdot\cdot}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (0,R], \ \exists \delta \in (-\lambda_{0}, R-\lambda_{0}] \setminus \{0\}, \\ \exists \tau \in (0,1), \ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta \mathbb{1}_{(\tau,1]}(t) \right\},$$
(61)

and we consider the problem of testing (H_0) " $\lambda \in S_0^u[R]$ " versus (H_1) " $\lambda \in S_{\cdot, \cdot, 1- \cdot, 1}^u[R]$ ". As usual, we start with a lower bound for the corresponding minimax separation rate. **Proposition 37** (Minimax lower bound for [Alt^u.8]). Let α, β in (0, 1) with $\alpha + \beta < 1/2$ and R > 0. There exists $L_0(\alpha, \beta, R) > 0$ such that for all $L \ge L_0(\alpha, \beta, R)$,

$$\operatorname{mSR}_{\alpha,\beta}(\mathcal{S}^{u}_{\cdot,\cdot,1-\cdot,\cdot}[R]) \geq \sqrt{R \log \log L/(4L)}$$
.

Let us assume now that $L \ge 3$. In order to prove that the above lower bound is of sharp order (with respect to L), we consider the discrete subset of (0, 1) of the dyadic form

$$\mathcal{D}_L = \left\{ 2^{-k}, \ k \in \{2, ..., \lfloor \log_2(L) \rfloor \} \right\} \cup \left\{ 1 - 2^{-k}, \ k \in \{1, ..., \lfloor \log_2(L) \rfloor \} \right\} ,$$

we set $u_{\alpha} = \alpha/(2\lfloor \log_2(L) \rfloor - 1)$ and we define the two following tests:

$$\phi_{8,\alpha}^{u(1)}(N) = \mathbb{1}_{\left\{\max_{\tau \in \mathcal{D}_L} \left(N(\tau,1] - (1-\tau)N_1 - \bar{b}_{N_1,1-\tau}(1-u_\alpha/2)\right) > 0\right\}} \\ \vee \mathbb{1}_{\left\{\max_{\tau \in \mathcal{D}_L} \left(\bar{b}_{N_1,1-\tau}(u_\alpha/2) - N(\tau,1] + (1-\tau)N_1\right) > 0\right\}}, \quad (62)$$

where $\bar{b}_{n,1-\tau}(u)$ is still the *u*-quantile of $N(\tau, 1] - (1 - \tau)N_1$ given $N_1 = n$ under (H_0) , that is the *u*-quantile of a recentered binomial distribution with parameters $(n, 1 - \tau)$, and

$$\phi_{8,\alpha}^{u(2)}(N) = \mathbb{1}_{\left\{\max_{\tau \in \mathcal{D}_L} \left(T'_{\tau,1}(N) - t'_{N_1,\tau,1}(1 - u_\alpha)\right) > 0\right\}},$$
(63)

where $T'_{\tau_1,\tau_2}(N)$ is the quadratic statistic defined by (46), and $t'_{n,\tau_1,\tau_2}(u)$ denotes the *u*-quantile of its conditional distribution given $N_1 = n$ under (H_0) .

Proposition 38 (Minimax upper bound for [Alt^u.8]). Let α, β in (0, 1), R > 0, and let $\phi_{8,\alpha}^{u(1/2)}$ be one of the tests $\phi_{8,\alpha}^{u(1)}$ and $\phi_{8,\alpha}^{u(2)}$ of (H_0) versus (H_1) " $\lambda \in S^u_{\cdot,\dots,1-\dots}[R]$ " defined by (62) and (63). Then $\phi_{8,\alpha}^{u(1/2)}$ is of level α , that is $\sup_{\lambda_0 \in S^u_0[R]} P_{\lambda_0}\left(\phi_{8,\alpha}^{u(1/2)}(N) = 1\right) \leq \alpha$. Moreover, there exists a constant $C(\alpha, \beta, R) > 0$ such that

$$\operatorname{SR}_{\beta}\left(\phi_{8,\alpha}^{u(1/2)}, \mathcal{S}_{\cdot,\cdot,1-\cdot,\cdot}^{u}[R]\right) \leq C(\alpha, \beta, R)\sqrt{\log\log L/L}$$
,

which entails in particular $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\cdot,1-\cdot,\cdot}[R]\right) \leq C(\alpha,\beta,R)\sqrt{\log\log L/L}.$

3.4.2 Transitory change

Let us investigate now the transitory change detection problem, focusing on adaptation to unknown location and length. As in Section 2.5.2, we will prove that minimax adaptation to these two parameters together has a cost of order as large as $\sqrt{\log L}$, so that adaptation to the height will have no additional cost. We therefore treat the two corresponding alternative sets quasi-simultaneously. For R > 0, δ^* in $(-R, R) \setminus \{0\}$, let

$$\begin{bmatrix} \mathbf{Alt^{u}.9} \end{bmatrix} \mathcal{S}^{u}_{\delta^{*},\dots,\dots}[R] = \left\{ \lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (-\delta^{*} \lor 0, (R-\delta^{*}) \land R], \\ \exists \tau \in (0,1), \ \exists \ell \in (0,1-\tau), \ \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta^{*} \mathbb{1}_{(\tau,\tau+\ell]}(t) \right\}, \quad (64)$$

$$[\mathbf{Alt^{u}.10}] \ \mathcal{S}^{u}_{\cdot,\cdot\cdot,\cdots}[R] = \{\lambda : [0,1] \to (0,R], \ \exists \lambda_{0} \in (0,R], \ \exists \delta \in (-\lambda_{0}, R - \lambda_{0}] \setminus \{0\}, \\ \exists \tau \in (0,1), \ \exists \ell \in (0,1-\tau), \forall t \in [0,1] \quad \lambda(t) = \lambda_{0} + \delta \mathbb{1}_{(\tau,\tau+\ell]}(t)\} \ .$$
(65)

As usual, we begin by giving lower bounds for the minimax separation rates over these two alternative sets, noticing that the case where the change height is known can be straightforwardly extended (as a simple corollary then) to the general one, where all three parameters, location, length and height of the change are unknown. **Proposition 39** (Minimax lower bound for [Alt^u.9]). Let α, β in (0,1) with $\alpha + \beta < 1$, R > 0, δ^* in $(-R, R) \setminus \{0\}$. There exists $L_0(\alpha, \beta, R, \delta^*) > 0$ such that for $L \ge L_0(\alpha, \beta, R, \delta^*)$,

$$\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\delta^{*},\dots,\dots}[R]\right) \geq \sqrt{\left((R-\delta^{*})\wedge R\right)\log L/\left(4L\right)}$$

Since $S^{u}_{,..,..}[R]$ includes $S^{u}_{R/2,..,..}[R]$, Proposition 39 directly leads to the following corollary, whose proof is omitted for simplicity.

Corollary 40 (Minimax lower bound for [Alt^u.10]). Let α, β in (0,1) with $\alpha + \beta < 1$ and R > 0. There exists $L_0(\alpha, \beta, R) > 0$ such that for all $L \ge L_0(\alpha, \beta, R)$,

$$\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\cdot,\cdot,\ldots}[R]\right) \geq \sqrt{R\log L/(8L)}$$

In order to prove that the above lower bounds are sharp, we then construct two minimax adaptive tests, based on an aggregation principle. In order to customise the tests developed in Section 2.5.2 to the lack of knowledge of the baseline intensity, we consider the linear statistic $N(\tau_1, \tau_2] - (\tau_2 - \tau_1)N_1$ and the quadratic statistic $T'_{\tau_1,\tau_2}(N)$ defined by (46), combined with the conditional trick already used in the above studies through the *u*-quantiles $\bar{b}_{n,\tau_2-\tau_1}(u)$ and $t'_{n,\tau_1,\tau_2}(u)$ of the conditional distributions of these statistics given $N_1 = n$ under (H_0) . Introducing $M_L = \lceil L/\log L \rceil$,

$$\mathcal{K}_{L}^{(1)} = \{ (k, k'), k \in \{0, \dots, \lceil L \rceil - 1\}, k' \in \{1, \dots, \lceil L \rceil - k\} \},
\mathcal{K}_{L}^{(2)} = \{ (k, k'), k \in \{0, \dots, M_{L} - 1\}, k' \in \{1, \dots, M_{L} - k\} \} \setminus \{ (0, M_{L}) \},$$

and the corrected levels $u_{\alpha}^{(1)} = 2\alpha/(\lceil L \rceil(\lceil L \rceil + 1))$ and $u_{\alpha}^{(2)} = 2\alpha/(M_L(M_L + 1) - 2)$, we can thus propose the two following tests:

Proposition 41 (Minimax upper bound for [Alt^u.9] and [Alt^u.10]). Let α, β in (0, 1), R > 0 and δ^* in $(-R, R) \setminus \{0\}$. Let $\phi_{9/10,\alpha}^{u(1/2)}$ be one of the tests $\phi_{9/10,\alpha}^{u(1)}$ and $\phi_{9/10,\alpha}^{u(2)}$ defined by (66) and (67). Then $\phi_{9/10,\alpha}^{u(1/2)}$ is of level α for the problems of testing (H₀) versus (H_1) " $\lambda \in \mathcal{S}^u_{\delta^*,\dots,\dots}[R]$ or (H_1) " $\lambda \in \mathcal{S}^u_{\cdot,\dots,\dots}[R]$ ", that is $\sup_{\lambda_0 \in \mathcal{S}^u_0[R]} P_{\lambda_0}\left(\phi_{9/10,\alpha}^{u(1/2)}(N) = 1\right) \leq \alpha$. Moreover, there exist $C(\alpha, \beta, R, \delta^*) > 0$ and $C(\alpha, \beta, R) > 0$ such that

$$\begin{aligned} \operatorname{SR}_{\beta}\left(\phi_{9/10,\alpha}^{u(1/2)}, \mathcal{S}_{\delta^{*},\dots,m}^{u}[R]\right) &\leq C(\alpha,\beta,R,\delta^{*})\sqrt{\log L/L} ,\\ \operatorname{SR}_{\beta}\left(\phi_{9/10,\alpha}^{u(1/2)}, \mathcal{S}_{\dots,m}^{u}[R]\right) &\leq C(\alpha,\beta,R)\sqrt{\log L/L} . \end{aligned}$$

These upper bounds entail both $\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\delta^{*},\dots,\dots}[R]\right) \leq C(\alpha,\beta,R,\delta^{*})\sqrt{\log L/L} \text{ and } \operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}^{u}_{\cdot,\dots,\dots}[R]\right) \leq C(\alpha,\beta,R)\sqrt{\log L/L}.$

3.5 Adjustment of individual levels for aggregated tests

As in Section 2.6, we discuss here the possibility of adjusting the individual levels of the single tests involved in our aggregated tests to make them more powerful. Most of the tests introduced in the present section are based on aggregation principles coupled with conditional tricks. Among them, we can again distinguish aggregated tests of the form $\phi_{agg1,\alpha}^{u}(N) = \mathbb{1}_{\{\sup_{\theta \in \Theta} S_{\theta}(N) > s_{N_{1}}^{+}(1-\alpha)\}}$ or $\mathbb{1}_{\{\sup_{\theta \in \Theta} S_{\theta}(N) > s_{N_{1}}^{+}(1-\alpha/2)\}} \vee \mathbb{1}_{\{\inf_{\theta \in \Theta} S_{\theta}(N) < s_{N_{1}}^{-}(\alpha/2)\}}$, where $s_{n}^{+}(u)$ and $s_{n}^{-}(u)$ respectively denote the *u*-conditional quantiles of $\sup_{\theta \in \Theta} S_{\theta}(N)$ and $\inf_{\theta \in \Theta} S_{\theta}(N)$ given $N_{1} = n$ under (H_{0}) (concerning $\phi_{3,\alpha}^{u(1)-}$, $\phi_{3,\alpha}^{u(1)+}$, $\phi_{4,\alpha}^{u(1)}$, $\phi_{5,\alpha}^{u}$ and $\phi_{7,\alpha}^{u}$), from aggregated tests of the form $\phi_{agg2,\alpha}^{u}(N) = \mathbb{1}_{\{\sup_{\theta \in \Theta} (S_{\theta}(N) - s_{N_{1},\theta}(1-u_{\alpha}/2)) > 0\}} \vee \mathbb{1}_{\{\sup_{\theta \in \Theta} (s_{N_{1},\theta}(u_{\alpha}/2) - S_{\theta}(N)) > 0\}}$, $s_{n,\theta}(u)$ being the *u*-conditional quantile of $S_{\theta}(N)$ given $N_{1} = n$ under (H_{0}) and $u_{\alpha} = \alpha/|\Theta|$ (as $\phi_{3/4,\alpha}^{u(2)}$, $\phi_{6,\alpha}^{u(1)}$, $\phi_{8,\alpha}^{u(1)}$, $\phi_{8,\alpha}^{u(2)}$, $\phi_{8,\alpha}^{u(1)}$, $\phi_{8,\alpha}^{u(2)}$, $\phi_{9/10,\alpha}^{u(2)}$).

Notice that for any λ_0 in $\mathcal{S}_0^u[R]$, when $N \sim P_{\lambda_0}$, the distribution of $S_\theta(N)$ given $N_1 = n$ is free from λ_0 . Moreover, similarly to (37), a better choice than u_α can be made for levels of the single tests involved in aggregated tests of the form $\phi_{agg2,\alpha}(N)$, namely $u'_{n,\alpha}$ equal to

$$\begin{cases} \sup\left\{u\in(0,1),\ \sup_{\lambda_{0}\in\mathcal{S}_{0}^{u}[R]}P_{\lambda_{0}}\left(\sup_{\theta\in\Theta}\left(S_{\theta}(N)-s_{n,\theta}(1-u)\right)>0\ \middle|N_{1}=n\right)\leq\alpha\right\}\\ \text{or }\sup\left\{u\in(0,1),\ \sup_{\lambda_{0}\in\mathcal{S}_{0}^{u}[R]}P_{\lambda_{0}}\left(\sup_{\theta\in\Theta}\left(S_{\theta}(N)-s_{n,\theta}(1-u/2)\right)\right.\\ \left.\left.\left.\left(s_{n,\theta}(u/2)-S_{\theta}(N)\right)>0\ \middle|N_{1}=n\right)\leq\alpha\right\}\right\}. \end{cases}$$

$$(68)$$

Since for all n in $\mathbb{N} \setminus \{0\}$ $u_{\alpha} \leq u'_{n,\alpha}$, by definition, $s_{n,\theta}(1 - u'_{n,\alpha}) \leq s_{n,\theta}(1 - u_{\alpha})$, $s_{n,\theta}(1 - u'_{n,\alpha}/2) \leq s_{n,\theta}(1 - u_{\alpha}/2)$ and $s_{n,\theta}(u_{\alpha}/2) \leq s_{n,\theta}(u'_{n,\alpha}/2)$ therefore all the above tests of type $\phi^u_{agg2,\alpha}$ but with u_{α} replaced by $u'_{N_{1},\alpha}$, that we can denote by $\phi'^u_{agg2,\alpha}$, satisfy the same minimax properties as $\phi^u_{agg2,\alpha}$. Our simulation study presented in Section 4 focuses on the practical performances of these adjusted aggregated tests $\phi'^u_{agg2,\alpha}$.

3.6 Summary and discussion

As in Section 2.7, we present a summary of the results stated above. Recall (*c.f.* (46) and (55)) that for $0 \le \tau_1 < \tau_2 \le 1$,

$$T_{\tau_1,\tau_2}'(N) = \frac{1}{L^2(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)} \Big((N(\tau_1, \tau_2] - (\tau_2 - \tau_1) N(0, 1])^2 \\ + (\tau_2 - \tau_1) (N(\tau_1, \tau_2] - (\tau_2 - \tau_1) N(0, 1]) - (1 - (\tau_2 - \tau_1)) N(\tau_1, \tau_2] \Big) ,$$

 $S'_{\delta^*,\tau_1,\tau_2}(N) = \operatorname{sgn}(\delta^*) \left(N(\tau_1,\tau_2] - (\tau_2 - \tau_1)N_1 \right) - |\delta^*| L(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)/2, \text{ and that } \bar{b}_{n,\tau_2-\tau_1}(u) \text{ and } t'_{n,\tau_1,\tau_2}(u) \text{ respectively stand for the } u \text{-quantiles of } N(\tau_1,\tau_2] - (\tau_2 - \tau_1)N_1 \text{ and } T'_{\tau_1,\tau_2}(N) \text{ given } N_1 = n \text{ under } (H_0).$

Non transitor	y change	or jump detection
Alternative set	$\mathrm{mSR}_{\alpha,\beta}$	Test statistics
$\mathcal{S}^u_{\delta^*,\tau^*,1-\tau^*}[R]$	-	$N(\tau^*, 1]$
$\frac{\mathcal{S}^{u}_{\delta^{*},\tau^{*},1-\tau^{*}}[R]}{\mathcal{S}^{u}_{\cdot,\tau^{*},1-\tau^{*}}[R]}$		$\frac{N(\tau, \tau)}{N(\tau^*, 1]}$
		$\frac{T'_{\tau^*,1}(N)}{\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)}$
$\mathcal{S}^u_{\delta^*,\cdots,1-\cdots}[R]$	$L^{-1/2}$	$\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)$
$\mathcal{S}^{u}_{\cdot,\cdot,1-\cdots}[R]$	$\sqrt{\frac{\log \log L}{L}}$	$\max_{\tau \in \mathcal{D}_L} \left(N(\tau, 1] - (1 - \tau) N_1 - \bar{b}_{N_1, 1 - \tau} \left(1 - u_{\alpha}/2 \right) \right)$
		$\vee \left(\bar{b}_{N_1, 1-\tau} \left(u_{\alpha}/2 \right) - N(\tau, 1] + (1-\tau)N_1 \right)$
		$\max_{\tau \in \mathcal{D}_{I}} \left(T'_{\tau 1}(N) - t'_{N_{1} \tau 1} (1 - u_{\alpha}) \right)^{'}$
		$\mathcal{D}_L = \left\{ 2^{-k}, \ k \in \{2, \dots, \lfloor \log_2(L) \rfloor \} \right\} \cup \left\{ 1 - 2^{-k}, \ k \in \{1, \dots, \lfloor \log_2(L) \rfloor \} \right\}$
Transitory cha	ange or bu	ump detection
Alternative set	$\mathrm{mSR}_{\alpha,\beta}$	Test statistics
$\frac{\mathcal{S}^{u}_{\delta^{*},\tau^{*},\ell^{*}}[R]}{\mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R]}$	$-L^{-1/2}$	$N(\tau^*, \tau^* + \ell^*]$
$\mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R]$	$L^{-1/2}$	$N(\tau^*,\tau^*+\ell^*]$
		$T'_{\tau^*,\tau^*+\ell^*}(N)$
$\mathcal{S}^{u}_{\delta^{*},\cdots,\ell^{*}}[R]$	-	$\max_{\tau \in [0, 1-\ell^* \land (1/2)]} N(\tau, \tau + \ell^* \land (1/2)],$
		$\min_{\tau \in [0, 1-\ell^* \land (1/2)]} N(\tau, \tau + \ell^* \land (1/2)]$
		$\max_{k \in \{0, \dots, \lceil (1-\ell^*)M \rceil - 1\}} \left(T'_{\frac{k}{M}, \frac{k}{M} + \ell^*}(N) - t'_{N_1, \frac{k}{M}, \frac{k}{M} + \ell^*}(1 - u_{\alpha}) \right)$
		$M = \left\lceil 2/(\ell^*(1-\ell^*)) \right\rceil$
$\mathcal{S}^u_{\cdot,\cdot\cdot,\ell^*}[R]$	$L^{-1/2}$	$\max_{\tau \in [0, 1-\ell^* \land (1/2)]} N(\tau, \tau + \ell^* \land (1/2)],$
		$\min_{\tau \in [0, 1-\ell^* \land (1/2)]} N(\tau, \tau + \ell^* \land (1/2)]$
		$\max_{k \in \{0,\dots,\lceil (1-\ell^*)M\rceil-1\}} \left(T'_{\frac{k}{M},\frac{k}{M}+\ell^*}(N) - t'_{N_1,\frac{k}{M},\frac{k}{M}+\ell^*}(1-u_{\alpha}) \right)$
$\mathcal{S}^{u}_{\delta^{*},\tau^{*},\cdots}[R]$	$L^{-1/2}$	$\sup_{\ell \in (0, 1 - \tau^*)} S'_{\delta^*, \tau^*, \tau^* + \ell}(N)$
$\mathcal{S}^{u}_{\cdot,\tau^{*},\dots}[R]$	$\sqrt{\frac{\log \log L}{L}}$	$\max_{k \in \{1, \dots, \lfloor \log_2 L \rfloor\}} \left(N\left(\tau^*, \tau^* + \frac{1-\tau^*}{2^k}\right) - \frac{1-\tau^*}{2^k} N_1 - \bar{b}_{N_1, \frac{1-\tau^*}{2^k}} \left(1 - u_{\alpha}/2\right) \right)$
		$ \vee \left(\bar{b}_{N_1, \frac{1-\tau^*}{2^k}} \left(u_{\alpha}/2 \right) - N \left(\tau^*, \tau^* + \frac{1-\tau^*}{2^k} \right] + \frac{1-\tau^*}{2^k} N_1 \right) $
		$ \max_{k \in \{1, \dots, \lfloor \log_2 L \rfloor\}} \left(T'_{\tau^*, \tau^* + \frac{1 - \tau^*}{2k}}(N) - t'_{N_1, \tau^*, \tau^* + \frac{1 - \tau^*}{2k}}(1 - u_{\alpha}) \right) $ $ \max_{k \in \{0, \dots, \lceil L \rceil - 1\}, k' \in \{1, \dots, \lceil L \rceil - k\}} \left(N\left(\frac{k}{\lceil L \rceil}, \frac{k + k'}{\lceil L \rceil}\right) - \frac{k'}{\lceil L \rceil}N_1 - \bar{b}_{N_1, \frac{k'}{\lceil L \rceil}}\left(1 - \frac{u_{\alpha}^{(1)}}{2}\right) \right) $ $ \vee \left(\bar{b}_{N_1, \frac{k'}{\lceil L \rceil}}\left(\frac{u_{\alpha}^{(1)}}{2}\right) - N\left(\frac{k}{\lceil L \rceil}, \frac{k + k'}{\lceil L \rceil}\right) + \frac{k'}{\lceil L \rceil}N_1 \right) $
$\mathcal{S}^u_{\delta^*,\dots,\dots}[R]$	$\sqrt{\frac{\log L}{L}}$	$\max_{k \in \{0,\dots,\lceil L\rceil-1\}, k' \in \{1,\dots,\lceil L\rceil-k\}} \left(N\left(\frac{k}{\lceil L\rceil}, \frac{k+k'}{\lceil L\rceil}\right) - \frac{k'}{\lceil L\rceil} N_1 - \bar{b}_{N_1,\frac{k'}{\lceil L\rceil}} \left(1 - \frac{u_{\alpha}^{(1)}}{2}\right) \right)$
$\mathcal{S}^{u}_{\cdot,\cdot,\cdot}[R]$	v	$\bigvee \left(\overline{b}_{N} \xrightarrow{k'} \left(\frac{u_{\alpha}^{(1)}}{2} \right) - N \left(\frac{k}{k+k'} \xrightarrow{k+k'}{k+k'} \right) + \frac{k'}{k'} N_1 \right)$
- •,••,••• [- •]		$\left(\begin{bmatrix} N_1, \frac{\kappa}{ L } \\ \hline \end{pmatrix} \right) = \left(\begin{bmatrix} L \\ 1 \end{bmatrix}, \begin{bmatrix} L \\ L \end{bmatrix} \right) = \left(\begin{bmatrix} L \\ 1 \end{bmatrix}, \begin{bmatrix} L \\ 1 \end{bmatrix} \right)$
		$\max_{\substack{k \in \{0,\dots,M_L-1\}, k' \in \{1,\dots,M_L-k\}\\(k,k') \neq (0,M_L)}} \left(T'_{\frac{k}{M_L},\frac{k+k'}{M_L}}(N) - t'_{N_1,\frac{k}{M_L},\frac{k+k'}{M_L}}\left(1 - u_{\alpha}^{(2)}\right) \right)$
		$M_L = \lceil L/\log L \rceil$

As compared with the above overview in Section 2.7, this one enables to see that the minimax separation rates do not suffer from the lack of knowledge of the baseline distribution: they indeed remain of the same order as in the problem of detecting a change from a given intensity, with the same phase transitions. Of course, these results are obtained at the price of a more important complexity of the test statistics, whether they are of linear or quadratic nature. This, combined with the need to use conditional quantiles instead of direct quantiles as critical values, brought in more technical arguments in the proofs. It can be furthermore noticed that up to our knowledge, except in the work of Verzelen et al. [105] for the jump detection problem, this specific case of an unknown

baseline distribution is in general not treated in the basic Gaussian model, where the only presence of a signal (that is a bump or jump from zero-mean) is tested.

4 Simulation study

We study in this section the performance of our minimax adaptive tests from an experimental point of view, by giving estimations of their size and their power for various distributions of the observed Poisson process, characterised by a jump or a bump in its intensity. Motivated by some applications in epidemiology and in cybersecurity, we check the feasibility of our new change-points detection procedures in practice and compare them with existing procedures.

We focus here on the most general problems investigated in this article of detecting a change (a jump or a bump) in the intensity λ when the change location and height are unknown. The baseline intensity of λ , denoted by λ_0 , is taken equal to 1 on [0, 1] in all the sequel. Recall that this baseline intensity can be considered as a known parameter of the testing problem as in Section 2 or as an unknown parameter as in Section 3.

For several piecewise constant intensities λ with respect to the measure $d\Lambda(t) = Ldt$, where we have chosen L = 50, we take a level of test $\alpha = 0.05$.

We compare the estimated powers of our procedures with more classical conditional tests previously studied in practice by other authors. For instance Cohen and Sackrowitz [23], and Bain, Engelhardt and Wright [7] considered six well-known tests in the context of detecting increasing intensities of a Poisson process. They showed that two of these six tests, namely the so-called Laplace and Z tests (respectively studied first by Cox [24] and Crow [25]) are more efficient from a practical point of view.

The Laplace test, denoted by (La) is based on the statistic $\sum_{i=1}^{N_1} X_i - q_{N_1}^{(La)}(1-\alpha)$, where (X_1, \ldots, X_{N_1}) are the points of the Poisson process N, and for all n in \mathbb{N} , $q_n^{(La)}(1-\alpha)$ is the $(1-\alpha)$ -quantile of the sum of n independent random variables uniformly distributed on [0, 1]. The Z test, denoted by (Z), is based on the statistic $2\sum_{i=1}^{N_1} \log(X_i) + q_{N_1}^{(Z)}(\alpha)$, where for every n in \mathbb{N} , $q_n^{(Z)}(\alpha)$ is the α -quantile of the chi-square distribution with 2n degrees of freedom. Note that these tests were especially designed to test homogeneity versus an increasing trend, with rejection of homogeneity when they take positive values. Therefore, we have had to adapt them to fit our context. More precisely, we decided to reject the null hypotheses (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " or (H_0) " $\lambda \in S_0^u[R]$ " when $\mathcal{T}_{\alpha}^{(La)}(N) > 0$, where

$$\mathcal{T}_{\alpha}^{(La)}(N) = \left(\sum_{i=1}^{N_1} X_i - q_{N_1}^{(La)}(1 - \alpha/2)\right) \vee \left(q_{N_1}^{(La)}(\alpha/2) - \sum_{i=1}^{N_1} X_i\right) ,$$

for the Laplace test (La), and when $\mathcal{T}_{\alpha}^{(Z)}(N) > 0$, where

$$\mathcal{T}_{\alpha}^{(Z)}(N) = \left(2\sum_{i=1}^{N_1} \log(X_i) + q_{N_1}^{(Z)}(\alpha/2)\right) \vee \left(-2\sum_{i=1}^{N_1} \log(X_i) - q_{N_1}^{(Z)}(1-\alpha/2)\right) ,$$

for the Z test (Z). Notice that a generalised version of the Laplace and the Z tests have been studied by Peña [85] and Agustin and Peña [1]. A simulation study has been

performed for these generalised procedures, but we have found that they have very similar estimated powers to the more classical Laplace and Z tests for the considered alternatives. The results are therefore omitted in this section.

4.1 Detection of an abrupt change from a known baseline intensity

We first consider the case where λ_0 is a known parameter, referring to the theoretical results of Section 2. The minimax adaptive tests we introduced to detect a change with unknown parameters from such a known intensity are based on two kinds of statistics. The first statistic is simply $N(\tau_1, \tau_2]$, and is therefore of linear nature, while the second statistic, of quadratic nature, is defined by

$$T_{\tau_1,\tau_2}(N) = \frac{1}{L^2(\tau_2 - \tau_1)} \left(N\left(\tau_1, \tau_2\right)^2 - \left(1 + 2\lambda_0 L(\tau_2 - \tau_1)\right) N\left(\tau_1, \tau_2\right] + \lambda_0^2 L^2(\tau_2 - \tau_1)^2 \right)$$

4.1.1 Detection of a non transitory change or jump

Let us recall that our tests are based on an aggregation principle which involves a scanning of the above linear and quadratic statistics over a discrete subset of possible values for the change location on (0, 1). The subset introduced in Section 2.5.1 is of the dyadic form

$$\Theta_d = \left\{ 1 - 2^{-k}, \ k \in \{1, \dots, 5\} \right\}$$

Considering the alternative [Alt.8], the test statistic of our first procedure denoted by $(CP1(\Theta_d))$ is thus

$$\mathcal{T}_{\lambda_{0},\alpha}^{(1)}(N) = \max_{\tau \in \Theta_{d}} \left(\left(N(\tau, 1] - p_{\lambda_{0}(1-\tau)L} \left(1 - u_{\alpha}^{(1)}/2 \right) \right) \vee \left(p_{\lambda_{0}(1-\tau)L} \left(u_{\alpha}^{(1)}/2 \right) - N(\tau, 1] \right) \right) ,$$

where $p_{\lambda_0(1-\tau)L}(u)$ is the *u*-quantile of a Poisson distribution with parameter $\lambda_0(1-\tau)L$ and $u_{\alpha}^{(1)}$ is defined as in (37) by

$$u_{\alpha}^{(1)} = \sup \left\{ u \in (0,1), \ P_{\lambda_0} \Big(\max_{\tau \in \Theta_d} \Big(\left(N(\tau,1] - p_{\lambda_0(1-\tau)L} \left(1 - u/2 \right) \right) \\ \vee \left(p_{\lambda_0(1-\tau)L} \left(u/2 \right) - N(\tau,1] \right) \Big) > 0 \Big) \le \alpha \right\} , \quad (69)$$

while the test statistic of our second procedure denoted by $(CP2(\Theta_d))$ is

$$\mathcal{T}_{\lambda_0,\alpha}^{(2)}\left(N\right) = \max_{\tau \in \Theta_d} \left(T_{\tau,1}(N) - t_{\lambda_0,\tau,1}\left(1 - u_\alpha^{(2)}\right)\right)$$

where $t_{\lambda_0,\tau_1,\tau_2}(1-u)$ is the (1-u)-quantile of $T_{\tau_1,\tau_2}(N)$ under (H_0) and

$$u_{\alpha}^{(2)} = \sup\left\{ u \in (0,1), \ P_{\lambda_0}\left(\max_{\tau \in \Theta_d} \left(T_{\tau,1}(N) - t_{\lambda_0,\tau,1}\left(1-u\right) \right) > 0 \right) \le \alpha \right\} \quad .$$
(70)

The null hypothesis (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " is rejected when $\mathcal{T}^{(1)}_{\lambda_0,\alpha}(N) > 0$ for $(CP1(\Theta_d))$, or when $\mathcal{T}^{(2)}_{\lambda_0,\alpha}(N) > 0$ for $(CP2(\Theta_d))$.

Noticing that the tests we use for the bump detection problem (see the following subsection) and the jump detection procedure studied in [105] are based on regular grids of possible values of change location instead of the dyadic subset Θ_d , we also consider the same tests but replacing Θ_d by a regular grid, with a cardinality close to the cardinality of Θ_d , namely

$$\Theta_r = \left\{ \frac{2k+1}{10}, \ k \in \{0, \dots, 4\} \right\}$$

The corresponding tests are then respectively denoted by $(CP1(\Theta_r))$ and $(CP2(\Theta_r))$.

For all τ in Θ_d or Θ_r , we have estimated the quantities $u_{\alpha}^{(1)}$, $u_{\alpha}^{(2)}$ and $t_{\lambda_0,\tau,1}(1-u_{\alpha}^{(2)})$ by classical Monte Carlo methods based on the simulation of 200 000 independent copies of a Poisson random variable with parameter $\lambda_0(1-\tau)L$ or $T_{\tau,1}(N)$ under (H_0) . The approximations of $u_{\alpha}^{(1)}$ and $u_{\alpha}^{(2)}$ were obtained by dichotomy, such that the probabilities or estimated probabilities occurring in (69) and (70) are less than α , but as close to α as possible.

4.1.2 Detection of a transitory change or bump

Let us consider the discrete set :

$$\Theta = \left\{ \left(\frac{k}{50}, \frac{k+k'}{50} \right), \ k \in \{0, \dots, 49\}, \ k' \in \{1, \dots, 50-k\} \right\} .$$

Considering the alternative [Alt.10], the test statistic of our first procedure (TC1) is

$$\mathcal{T}_{\lambda_{0},\alpha}^{(3)}(N) = \max_{(\tau,\tau')\in\Theta} \left(\left(N(\tau,\tau'] - p_{\lambda_{0}(\tau'-\tau)L} \left(1 - u_{\alpha}^{(3)}/2 \right) \right) \vee \left(p_{\lambda_{0}(\tau'-\tau)L} \left(u_{\alpha}^{(3)}/2 \right) - N(\tau,\tau'] \right) \right)$$

where $p_{\lambda_0(\tau'-\tau)L}(u)$ is the *u*-quantile of a Poisson distribution with parameter $\lambda_0(\tau'-\tau)L$ and $u_{\alpha}^{(3)}$ is defined, again as in (37) by

$$u_{\alpha}^{(3)} = \sup\left\{ u \in (0,1), \ P_{\lambda_0} \Big(\max_{(\tau,\tau')\in\Theta} \Big(\left(N(\tau,\tau'] - p_{\lambda_0(\tau'-\tau)L} \left(1 - u/2 \right) \right) \\ \vee \left(p_{\lambda_0(\tau'-\tau)L} \left(u/2 \right) - N(\tau,\tau'] \right) \Big) > 0 \Big) \le \alpha \right\} , \quad (71)$$

while the test statistic of our second procedure (TC2) is

$$\mathcal{T}_{\lambda_{0},\alpha}^{(4)}\left(N\right) = \max_{(\tau,\tau')\in\Theta} \left(T_{\tau,\tau'}(N) - t_{\lambda_{0},\tau,\tau'}\left(1 - u_{\alpha}^{(4)}\right)\right)$$

where $u_{\alpha}^{(4)}$ is defined by

$$u_{\alpha}^{(4)} = \sup\left\{ u \in (0,1), \ P_{\lambda_0}\left(\max_{(\tau,\tau')\in\Theta} \left(T_{\tau,\tau'}(N) - t_{\lambda_0,\tau,\tau'}\left(1 - u_{\alpha}^{(4)}\right)\right) > 0\right) \le \alpha \right\} \quad .$$
(72)

The null hypothesis (H_0) " $\lambda \in \mathcal{S}_0[\lambda_0] = \{\lambda_0\}$ " is rejected when $\mathcal{T}^{(3)}_{\lambda_0,\alpha}(N) > 0$ for (TC1), or when $\mathcal{T}^{(4)}_{\lambda_0,\alpha}(N) > 0$ for (TC2).

As above, for all (τ, τ') in Θ , the quantities $u_{\alpha}^{(3)}$, $u_{\alpha}^{(4)}$ and $t_{\lambda_0,\tau,\tau'}(1-u_{\alpha}^{(4)})$ have been estimated by Monte Carlo methods based on the simulation of 200 000 independent copies of a Poisson random variable with parameter $\lambda_0(\tau'-\tau)L$ or $T_{\tau,\tau'}(N)$ under (H_0) . The approximations of $u_{\alpha}^{(3)}$ and $u_{\alpha}^{(4)}$ have been obtained by dichotomy.

4.1.3 Simulation results

We compare the tests (La) and (Z) with $(CP1(\Theta_d))$, $(CP2(\Theta_d))$, $(CP1(\Theta_r))$ and $(CP2(\Theta_r))$ when addressing the jump detection problem (described in 4.1.1), and with (TC1) and (TC2) when addressing the bump detection problem (described in 4.1.2).

Estimated sizes We first study the size of each test via simulation of 10 000 independent homogeneous Poisson processes of intensity $\lambda_0 = 1$ w.r.t. Λ on [0, 1]. The probabilities of first kind error of all the considered tests were simply estimated by the number of rejections divided by 10 000. The results are given in Table 1.

Table 1:	Estimated	sizes
La	Z	

La	Z
0.049	0.049
$\mathtt{CP1}(\Theta_d)$	$\mathtt{CP2}(\Theta_d)$
0.046	0.048
$\mathtt{CP1}(\Theta_r)$	$\mathtt{CP2}(\Theta_r)$
$\begin{array}{c} \texttt{CP1}(\Theta_r) \\ 0.049 \end{array}$	$\begin{array}{c} \texttt{CP2}(\Theta_r) \\ 0.048 \end{array}$
(. ,	

Notice that the estimated sizes of our tests always remain below the target 0.05, as expected from the definitions of $u_{\alpha}^{(1)}$, $u_{\alpha}^{(2)}$, $u_{\alpha}^{(3)}$ and $u_{\alpha}^{(4)}$. It is in particular interesting to see that the Monte Carlo estimation, which is calibrated according to a balance precision/running time, does not affect here the first kind error rate control property.

Estimated powers For both testing problems, we study the estimated power of each test under various alternatives.

Let us start with the jump detection problem. We consider alternative intensities $\lambda_{\tau,\delta}$ defined for all t in [0, 1] by

$$\lambda_{\tau,\delta}(t) = 1 + \delta \mathbb{1}_{(\tau,1]}(t) \quad , \tag{73}$$

where $\delta \in \{-0.8, -0.6, -0.4, -0.2, 0.4, 0.8, 1.2, 1.6, 2\}$, and $\tau = 0.5$ (Table 2), $\tau = 0.9$ (Table 3) and $\tau = 0.95$ (Table 4). For each alternative, 1 000 independent inhomogeneous Poisson processes with intensity $\lambda_{\tau,\delta}$ w.r.t. Λ on [0, 1] have been simulated. The power of the considered tests has then been simply estimated by the number of rejections divided by 1 000, leading to the results gathered in Tables 2-4.

Let us now turn to the bump detection problem. We have considered alternative intensities $\lambda_{\tau,\ell,\delta}$ defined for all t in [0, 1] by

$$\lambda_{\tau,\ell,\delta}(t) = 1 + \delta \mathbb{1}_{(\tau,\tau+\ell]}(t) \quad , \tag{74}$$

where $\tau = 0.2, \ \delta \in \{-0.8, -0.6, -0.4, -0.2, 0.4, 0.8, 1.2, 1.6, 2\}$ and $\ell = 0.1$ (Table 5), $\ell = 0.2$ (Table 6) and $\ell = 0.5$ (Table 7).

For each alternative, we have simulated 1 000 independent inhomogeneous Poisson processes with intensity $\lambda_{\tau,\ell,\delta}$ w.r.t. Λ on [0, 1]. The powers have been estimated for each test by the number of rejections divided by 1 000, and the results are provided in Tables 5-7.

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.92	0.58	0.27	0.09	0.19	0.56	0.84	0.96	0.99
Z	0.62	0.34	0.15	0.07	0.16	0.44	0.70	0.89	0.97
$\mathtt{CP1}(\Theta_d)$	1	0.93	0.44	0.13	0.40	0.90	1	1	1
$\mathtt{CP2}(\Theta_d)$	1	0.87	0.35	0.09	0.40	0.90	1	1	1
$CP1(\Theta_r)$	1	0.91	0.43	0.14	0.42	0.90	1	1	1
$\mathtt{CP2}(\Theta_r)$	1	0.89	0.39	0.11	0.45	0.91	1	1	1

Table 2: Estimated probability of detecting a jump with $\tau=0.5$

Table 3: Estimated probability of detecting a jump with $\tau = 0.9$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.13	0.10	0.07	0.06	0.08	0.15	0.24	0.39	0.51
Z	0.07	0.06	0.04	0.06	0.06	0.09	0.12	0.20	0.29
$\mathtt{CP1}(\Theta_d)$	0.18	0.11	0.05	0.04	0.13	0.31	0.53	0.74	0.86
$\mathtt{CP2}(\Theta_d)$	0.17	0.09	0.04	0.04	0.15	0.33	0.56	0.78	0.88
$\mathtt{CP1}(\Theta_r)$	0.44	0.20	0.09	0.06	0.10	0.25	0.49	0.68	0.84
$\mathtt{CP2}(\Theta_r)$	0.09	0.08	0.04	0.04	0.13	0.33	0.59	0.79	0.89

Table 4: Estimated probability of detecting a jump with $\tau = 0.95$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.05	0.06	0.06	0.05	0.07	0.07	0.12	0.17	0.23
Z	0.05	0.05	0.05	0.04	0.06	0.05	0.08	0.11	0.13
$\mathtt{CP1}(\Theta_d)$	0.06	0.05	0.04	0.04	0.07	0.18	0.29	0.43	0.55
$\mathtt{CP2}(\Theta_d)$	0.05	0.04	0.04	0.04	0.09	0.23	0.35	0.50	0.63
$\mathtt{CP1}(\Theta_r)$	0.10	0.08	0.06	0.06	0.07	0.11	0.17	0.27	0.37
$\mathtt{CP2}(\Theta_r)$	0.04	0.04	0.04	0.05	0.08	0.15	0.24	0.36	0.47

Table 5: Estimated probability of detecting a bump with $\ell=0.1$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.08	0.06	0.06	0.05	0.06	0.07	0.11	0.14	0.19
Z	0.07	0.06	0.05	0.05	0.05	0.05	0.06	0.04	0.05
TC1	0.13	0.08	0.06	0.04	0.08	0.17	0.28	0.48	0.71
TC2	0.06	0.05	0.04	0.04	0.09	0.20	0.31	0.54	0.74

Table 6: Estimated probability of detecting a bump with $\ell = 0.2$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.14	0.10	0.07	0.05	0.06	0.08	0.17	0.26	0.37
Z	0.08	0.07	0.06	0.05	0.04	0.04	0.03	0.05	0.04
TC1	0.52	0.21	0.10	0.06	0.12	0.34	0.65	0.88	0.96
TC2	0.18	0.07	0.05	0.05	0.14	0.38	0.71	0.90	0.97

Table 7: Estimated probability of detecting a bump with $\ell = 0.5$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.18	0.10	0.07	0.07	0.04	0.04	0.06	0.07	0.09
Z	0.15	0.08	0.07	0.07	0.04	0.03	0.04	0.06	0.05
TC1	0.99	0.74	0.27	0.08	0.28	0.79	0.98	1	1
TC2	0.98	0.56	0.15	0.05	0.34	0.83	0.99	1	1

Comments

- 1. It first arises that our procedures have estimated powers significantly larger than the Laplace and the Z tests for both testing problems corresponding to alternatives [Alt.8] and [Alt.10] in most cases. The lower performances of the Laplace and the Z tests may be due to the fact that their construction does not take the knowledge of λ_0 into account. Moreover, among our testing procedures, it is to note that both procedures based on the linear statistics (CP1) and (TC1) are mostly more powerful than (CP2) and (TC2) in the case of negative change heights whereas the procedures based on the quadratic statistics (CP2) and (TC2) are slightly more powerful when positive change heights occur. More precisely, the better performances of the linear statistics to detect negative jumps mainly concern the bump detection and the better performances of the quadratic statistics to detect positive jumps mainly concern the jump detection when the change-point is close to the observation interval endpoint 1.
- 2. The comparison of the estimated powers of the different testing procedures confirm the intuition that detecting a bump is harder than detecting a jump. Moreover, the performances of each test are very different according to the sign of the change height. In both jump and bump detection problems, it is easier to detect a change with large negative jumps than a change with large positive jumps, except for the cases where $\ell = 0.1$ in the bump detection problem and $\tau \ge 0.9$ in the jump detection problem. Indeed, in these cases, the estimated powers are close to the size of the tests for small negative change heights. Note that this capability of our tests to better detect a jump or a bump with negative height than with positive height can be explained by the fact that the significant parameter to evaluate the detectability of a (transitory or not) change is not the change height itself but the ratio between the minimum and the maximum values of the intensity. In other words, this means that it is easier to detect an intensity increasing from 1 to 2 than

an intensity increasing from 100 to 101 whereas in both cases, the jump height is equal to one. In the same way, it is hence easier to detect an intensity decreasing from 1 to 0.8 than an intensity increasing from 1 to 1.2.

- 3. By comparing the estimated powers of our jump detection procedures based on the dyadic and regular sets Θ_d and Θ_r , one can take note that using the dyadic set is as expected more relevant when the jump is close to the observation interval endpoint, that is the most difficult to detect.
- 4. Finally, for the bump detection problem, we have to notice that complementary experiments showed that the estimated powers of (TC1) and (TC2) are equivalent for a same value of change length whatever the values of the change location, which is not always true for the procedures (La) and (Z): for a same change length, these procedures are more powerful when the change location is close to 1.

4.2 Detection of an abrupt change from an unknown baseline intensity

We now consider the case where λ_0 is an unknown parameter, referring to theoretical results in Section 3. The minimax adaptive tests we introduced to detect a change with unknown parameters from such an unknown intensity are still based on two kinds of statistics. The first statistic, of linear nature, is $N(\tau_1, \tau_2] - (\tau_2 - \tau_1)N(0, 1]$, while the second statistic, of quadratic nature, is defined by

$$T'_{\tau_1,\tau_2}(N) = \frac{1}{L^2 (\tau_2 - \tau_1) (1 - \tau_2 + \tau_1)} \Big[(N(\tau_1, \tau_2] - (\tau_2 - \tau_1) N(0, 1])^2 + (\tau_2 - \tau_1) (N(\tau_1, \tau_2] - (\tau_2 - \tau_1) N(0, 1]) - (1 - \tau_2 + \tau_1) N(\tau_1, \tau_2] \Big] .$$

4.2.1 Detection of a non transitory change or jump

As above, the aggregation approach we used to construct our new tests to detect a jump from an unknown baseline intensity consists in scanning these linear and quadratic statistics over a discrete subset of possible values for the change location on (0, 1). The subset introduced in Section 3.4.1 is of the dyadic form

$$\Theta_d^u = \left\{ 2^{-k}, \ k \in \{2, \dots, 5\} \right\} \cup \left\{ 1 - 2^{-k}, \ k \in \{1, \dots, 5\} \right\}$$
.

Considering the alternative [Alt^u.8], the test statistic of our first procedure denoted by $(CP1^u(\Theta_d^u))$ is thus

$$\mathcal{T}_{\cdot,\alpha}^{(1)}(N) = \max_{\tau \in \Theta_d^u} \left(\left(N(\tau, 1] - (1 - \tau) N(0, 1] - \bar{b}_{N_1, 1 - \tau} \left(1 - u_{N_1, \alpha}^{(1)} / 2 \right) \right) \\ \vee \left(\bar{b}_{N_1, 1 - \tau} \left(u_{N_1, \alpha}^{(1)} / 2 \right) - N(\tau, 1] + (1 - \tau) N(0, 1] \right) \right) ,$$

where $\bar{b}_{n,1-\tau}(u)$ is the *u*-quantile of a recentered binomial distribution with parameters

 $(n, 1-\tau)$ and $u_{n,\alpha}^{(1)}$ is defined for all n in \mathbb{N} as in (68) by

$$u_{n,\alpha}^{(1)} = \sup\left\{ u \in (0,1), \sup_{\lambda_0 \in \mathcal{S}_0^u[R]} P_{\lambda_0} \left(\max_{\tau \in \Theta_d^u} \left(\left(N(\tau,1] - (1-\tau)N(0,1] - \bar{b}_{n,1-\tau} \left(1 - u/2 \right) \right) \right) \right) \right\} \\ \times \left(\bar{b}_{n,1-\tau} \left(u/2 \right) - N(\tau,1] + (1-\tau)N(0,1] \right) > 0 \left| N_1 = n \right| \le \alpha \right\}.$$
 (75)

The test statistic of our second procedure denoted by $(CP2^u(\Theta_d^u))$ is

$$\mathcal{T}_{,\alpha}^{(2)}(N) = \max_{\tau \in \Theta_d^u} \left(T_{\tau,1}'(N) - t_{N_1,\tau,1}'\left(1 - u_{N_1,\alpha}^{(2)}\right) \right)$$

where $t'_{n,\tau_1,\tau_2}(u)$ is the *u*-quantile of the conditional distribution of $T'_{\tau_1,\tau_2}(N)$ given $N_1 = n$ under (H_0) and $u^{(2)}_{n,\alpha}$ is defined for all n in \mathbb{N} by

$$u_{n,\alpha}^{(2)} = \sup\left\{ u \in (0,1), \sup_{\lambda_0 \in \mathcal{S}_0^u[R]} P_{\lambda_0} \left(\max_{\tau \in \Theta_d^u} \left(T'_{\tau,1}(N) - t'_{N_1,\tau,1} \left(1 - u \right) \right) > 0 \ \Big| N_1 = n \right) \le \alpha \right\}$$
(76)

Then, the null hypothesis (H_0) " $\lambda \in \mathcal{S}_0^u[R]$ " is rejected when $\mathcal{T}_{,\alpha}^{(1)}(N) > 0$ for $(\operatorname{CP1}^u(\Theta_d^u))$, and when $\mathcal{T}_{,\alpha}^{(2)}(N) > 0$ for $(\operatorname{CP2}^u(\Theta_d^u))$.

As in the known baseline case, we have also considered the same tests, but replacing the dyadic set Θ_d^u by the regular set

$$\Theta_r^u = \left\{ \frac{k}{10}, \ k \in \{1, \dots, 9\} \right\}$$

The corresponding testing procedures are then denoted by $(\text{CP1}^u(\Theta_r^u))$ and $(\text{CP2}^u(\Theta_r^u))$ The quantities $u_{n,\alpha}^{(1)}$, $u_{n,\alpha}^{(2)}$ and $t'_{n,\tau,1}(1-u_{n,\alpha}^{(2)})$ have been estimated by Monte Carlo methods based on the simulation of 200 000 samples of n independent copies of a recentered binomial random variable with parameters $(n, 1-\tau)$ and of an uniform random variable on [0, 1]. These samples were used to approximate the conditional probabilities occurring in (75) and (76). The approximations of $u_{n,\alpha}^{(1)}$ and $u_{n,\alpha}^{(2)}$ were obtained by dichotomy.

4.2.2 Detection of a transitory change or bump

Let us consider the discrete sets

$$\Theta_1 = \left\{ \left(\frac{k}{50}, \frac{k+k'}{50}\right), \ k \in \{0, \dots, 49\}, \ k' \in \{1, \dots, 50-k\} \right\} ,$$

and

$$\Theta_2 = \left\{ \left(\frac{k}{13}, \frac{k+k'}{13}\right), \ k \in \{0, \dots, 12\}, \ k' \in \{1, \dots, 13-k\}, \ (k,k') \neq (0,13)\} \right\} .$$

Considering the alternative $[Alt^u.10]$, the test statistic of our first procedure denoted by $(TC1^u)$ is

$$\mathcal{T}_{,\alpha}^{(3)}(N) = \max_{(\tau,\tau')\in\Theta_1} \left(\left(N(\tau,\tau'] - (\tau'-\tau)N(0,1] - \bar{b}_{N_1,\tau'-\tau} \left(1 - u_{N_1,\alpha}^{(3)}/2 \right) \right) \\ \vee \left(\bar{b}_{N_1,\tau'-\tau} \left(u_{N_1,\alpha}^{(3)}/2 \right) - N(\tau,\tau'] + (\tau'-\tau)N(0,1] \right) \right) ,$$

where $\bar{b}_{n,\tau'-\tau}(u)$ is the *u*-quantile of a recentered binomial distribution with parameters $(n, \tau' - \tau)$ and $u_{n,\alpha}^{(3)}$ is defined for all n in \mathbb{N} by

$$u_{n,\alpha}^{(3)} = \sup\left\{u \in (0,1), \sup_{\lambda_0 \in \mathcal{S}_0^u[R]} P_{\lambda_0}\left(\max_{(\tau,\tau') \in \Theta_1} \left(\left(N(\tau,\tau'] - (\tau'-\tau)N(0,1] - \bar{b}_{n,\tau'-\tau}(1-u/2)\right) \right) \right) \right\} \\ \times \left(\bar{b}_{n,\tau'-\tau}(u/2) - N(\tau,\tau'] + (\tau'-\tau)N(0,1]\right) > 0 \left|N_1 = n\right| \le \alpha \right\}.$$
(77)

while the test statistic of our second procedure denoted by $(TC2^u)$ is

$$\mathcal{T}^{(4)}_{\cdot,\alpha}(N) = \max_{(\tau,\tau')\in\Theta_2} \left(T'_{\tau,\tau'}(N) - t'_{N_1,\tau,\tau'}\left(1 - u^{(4)}_{N_1,\alpha}\right) \right) \ ,$$

where $u_{n,\alpha}^{(4)}$ is defined for all n in \mathbb{N} by

$$u_{n,\alpha}^{(4)} = \sup\left\{ u \in (0,1), \sup_{\lambda_0 \in \mathcal{S}_0^u[R]} P_{\lambda_0} \left(\max_{(\tau,\tau') \in \Theta_2} \left(T'_{\tau,\tau'}(N) - t'_{N_1,\tau,\tau'}(1-u) \right) > 0 \ \Big| N_1 = n \right) \le \alpha \right\}$$
(78)

The null hypothesis (H_0) " $\lambda \in \mathcal{S}_0^u[R]$ " is rejected when $\mathcal{T}_{,\alpha}^{(3)}(N) > 0$ for $(\mathsf{TC1}^u)$, and when $\mathcal{T}_{,\alpha}^{(4)}(N) > 0$ for $(\mathsf{TC2}^u)$.

The quantities $u_{n,\alpha}^{(3)}$, $u_{n,\alpha}^{(4)}$ and $t'_{n,\tau,\tau'}(1-u_{n,\alpha}^{(4)})$ have been estimated by Monte Carlo methods based on the simulation of 200 000 independent copies of a recentered binomial random variable with parameter $(n, \tau' - \tau)$ and 200 000 independent copies of $T'_{\tau,\tau'}(N)$ given $N_1 = n$ under (H_0) obtained from the simulation of 200 000 samples of n i.i.d. random variables uniformly distributed on [0, 1]. These samples have been used to approximate the conditional probabilities occurring in (77) and (78). The approximations of $u_{n,\alpha}^{(3)}$ and $u_{n,\alpha}^{(4)}$ have been obtained by dichotomy.

4.2.3 Simulation results

We compare the tests (La) and (Z) with $(CP1^u(\Theta_d^u))$, $(CP2^u(\Theta_d^u))$, $(CP1^u(\Theta_r^u))$ and $(CP2^u(\Theta_r^u))$ when addressing the jump detection problem (described in Section 4.2.1), and with $(TC1^u)$ and $(TC2^u)$ when addressing the bump detection problem (described in Section 4.2.2).

Estimated sizes We first study the size of each test by simulating 10 000 independent homogeneous Poisson processes of intensity $\lambda_0 = 1$ w.r.t. Λ on [0, 1]. The probabilities of first kind error of all the considered tests have been estimated by the number of rejections divided by 10 000. The results are given in Table 8.

Notice again that the estimated sizes of our tests always remain below the target 0.05, as expected from the definitions of $u_{n,\alpha}^{(1)}$, $u_{n,\alpha}^{(2)}$, $u_{n,\alpha}^{(3)}$ and $u_{n,\alpha}^{(4)}$: the Monte Carlo estimation procedure does not affect this first kind error rate control property.

Estimated powers For both testing problems, we study the estimated power of each test under various alternatives.

For the jump detection problem, we consider the same alternative intensities $\lambda_{\tau,\delta}$ as in the known baseline intensity case, defined for all t in [0, 1] by (73), but with τ varying in $\{0.05, 0.1, 0.5, 0.9, 0.95\}$.

La	Z
0.050	0.049
$(\mathtt{CP1}^u(\Theta^u_d))$	$\left({\rm CP2}^u(\Theta^u_d) ight)$
0.047	0.046
$(\operatorname{CP1}^u(\Theta_r^u))$	$(\mathtt{CP2}^u(\Theta^u_r))$
0.046	0.047
$TC1^u$	$TC2^u$
0.049	0.049

Table 8: Estimated sizes

For each alternative, we have simulated 1 000 independent inhomogeneous Poisson processes with intensity $\lambda_{\tau,\delta}$ w.r.t. Λ on [0, 1], and the powers have been estimated for each test by the mean number of rejections. The results are gathered in Tables 9-13.

Concerning the bump detection problem, we have considered the same alternative intensities $\lambda_{\tau,\ell,\delta}$ as in the known baseline intensity case. For each alternative, we have simulated 1 000 independent inhomogeneous Poisson processes with intensity $\lambda_{\tau,\ell,\delta}$ w.r.t. Λ on [0, 1]. The powers have been estimated for each test by the mean number of rejections, giving the results presented in Tables 14-16.

Table 9: Estimated probability of detecting a jump with $\tau = 0.05$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.19	0.10	0.06	0.05	0.05	0.06	0.07	0.09	0.10
Z	0.36	0.18	0.09	0.06	0.05	0.09	0.13	0.17	0.24
$\mathtt{CP1}^u(\Theta_d^u)$	0.34	0.16	0.09	0.06	0.05	0.08	0.09	0.14	0.20
$\mathtt{CP2}^u(\Theta^u_d)$	0.34	0.17	0.10	0.06	0.05	0.06	0.07	0.11	0.16
$\mathtt{CP1}^u(\Theta^u_r)$	0.22	0.10	0.06	0.05	0.06	0.07	0.07	0.10	0.14
$\mathtt{CP2}^u(\Theta^u_r)$	0.23	0.11	0.07	0.05	0.06	0.06	0.07	0.09	0.13

Table 10: Estimated probability of detecting a jump with $\tau = 0.1$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.41	0.17	0.08	0.06	0.07	0.09	0.13	0.19	0.24
Z	0.54	0.26	0.12	0.07	0.08	0.14	0.27	0.39	0.54
$\mathtt{CP1}^u(\Theta^u_d)$	0.54	0.25	0.11	0.07	0.06	0.11	0.20	0.35	0.49
$\mathtt{CP2}^u(\Theta^u_d)$	0.55	0.27	0.12	0.07	0.05	0.09	0.17	0.28	0.43
$\mathtt{CP1}^u(\Theta^u_r)$	0.53	0.24	0.10	0.06	0.07	0.13	0.24	0.42	0.60
$\mathtt{CP2}^u(\Theta^u_r)$	0.54	0.27	0.11	0.07	0.06	0.11	0.21	0.39	0.56

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.91	0.60	0.28	0.09	0.19	0.55	0.85	0.96	0.99
Z	0.62	0.35	0.17	0.08	0.16	0.45	0.72	0.88	0.97
${ m CP1}^u(\Theta^u_d)$	0.92	0.55	0.22	0.07	0.15	0.49	0.83	0.96	1
$\mathtt{CP2}^u(\Theta^u_d)$	0.90	0.51	0.20	0.07	0.14	0.46	0.81	0.95	0.99
$\mathtt{CP1}^u(\Theta^u_r)$	0.93	0.58	0.25	0.09	0.16	0.53	0.85	0.97	1
${\rm CP2}^u(\Theta^u_r)$	0.91	0.56	0.23	0.08	0.16	0.51	0.84	0.96	1

Table 11: Estimated probability of detecting a jump with $\tau = 0.5$

Table 12: Estimated probability of detecting a jump with $\tau = 0.9$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.13	0.10	0.07	0.05	0.08	0.15	0.24	0.37	0.53
Z	0.08	0.08	0.06	0.04	0.06	0.10	0.13	0.20	0.27
$\mathtt{CP1}^u(\Theta_d^u)$	0.17	0.10	0.07	0.05	0.09	0.22	0.36	0.57	0.73
$\mathtt{CP2}^u(\Theta^u_d)$	0.14	0.09	0.05	0.04	0.10	0.24	0.39	0.60	0.75
$\mathtt{CP1}^u(\Theta^u_r)$	0.22	0.13	0.07	0.05	0.08	0.20	0.36	0.58	0.75
${\rm CP2}^u(\Theta_r^u)$	0.15	0.09	0.05	0.04	0.09	0.22	0.40	0.61	0.79

Table 13: Estimated probability of detecting a jump with $\tau = 0.95$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.07	0.06	0.06	0.06	0.06	0.08	0.12	0.16	0.23
Z	0.05	0.05	0.06	0.05	0.06	0.06	0.08	0.10	0.12
$\mathtt{CP1}^u(\Theta^u_d)$	0.06	0.05	0.05	0.04	0.09	0.14	0.22	0.31	0.46
$\mathtt{CP2}^u(\Theta^u_d)$	0.05	0.04	0.05	0.04	0.10	0.15	0.25	0.35	0.50
$\mathtt{CP1}^u(\Theta^u_r)$	0.08	0.05	0.07	0.05	0.06	0.08	0.13	0.20	0.29
${\rm CP2}^u(\Theta_r^u)$	0.06	0.05	0.06	0.05	0.07	0.09	0.15	0.22	0.32

Table 14: Estimated probability of detecting a bump with $\ell=0.1$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.08	0.06	0.07	0.05	0.06	0.08	0.10	0.13	0.19
Z	0.08	0.06	0.07	0.06	0.05	0.04	0.04	0.05	0.06
$TC1^u$	0.12	0.09	0.07	0.05	0.06	0.14	0.26	0.37	0.59
$TC2^u$	0.11	0.07	0.07	0.05	0.06	0.13	0.25	0.35	0.56

Comments

1. Considering the single change-point or jump detection problem, it first arises that

Table 15: Estimated probability of detecting a bump with $\ell = 0.2$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.14	0.10	0.08	0.06	0.07	0.11	0.18	0.29	0.39
Z	0.09	0.09	0.07	0.05	0.05	0.04	0.04	0.05	0.05
$TC1^u$	0.42	0.19	0.09	0.06	0.09	0.24	0.47	0.71	0.87
$TC2^u$	0.29	0.17	0.08	0.06	0.10	0.24	0.46	0.70	0.86

Table 16: Estimated probability of detecting a bump with $\ell = 0.5$

$\delta =$	-0.8	-0.6	-0.4	-0.2	0.4	0.8	1.2	1.6	2
La	0.15	0.12	0.07	0.06	0.04	0.04	0.05	0.07	0.09
Z	0.15	0.11	0.07	0.06	0.04	0.05	0.04	0.04	0.05
$TC1^u$	0.83	0.45	0.17	0.08	0.12	0.37	0.67	0.89	0.98
$TC2^u$	0.83	0.44	0.17	0.08	0.13	0.39	0.68	0.89	0.98

among the (La) and (Z) procedures, neither is preferable to use: the Laplace and Z tests can have very low powers depending on when the change occurs. One can notice that their performances are significantly smaller than the ones of our procedures $(CP1^u)$ and $(CP2^u)$ when the jump occurs near to the endpoint 1, while the estimated powers remain comparable in the other cases. Moreover, it is worthwhile to note again that the jump detection problem in a Poisson process is not a symmetric problem. Indeed, it is easier to detect large negative jumps occurring close to 0 than close to 1, and easier to detect large positive jumps occurring close to 1 than close to 0.

- 2. Considering the transitory change or bump detection problem, our procedures have estimated powers significantly larger in all cases than the Laplace and Z tests. Moreover, we have to mention that complementary experiments (omitted in this study) showed that the estimated powers of $(TC1^u)$ and $(TC2^u)$ are equivalent for a same value of change length whatever the change location, which is not true for the procedures (La) and (Z).
- 3. Among our testing procedures, the estimated powers are quite similar in most cases. Nevertheless, it is to note that the procedures $(CP1^u)$ based on the linear statistics, are slightly more powerful than the procedures $(CP2^u)$ based on the quadratic statistics, for some positive jumps occurring near 0 and for some negative jumps occurring near 1, whereas the procedures $(CP2^u)$ are slightly more powerful for some positive jumps occurring near 1. As expected, the aggregated tests based on dyadic sets are significantly more efficient that the ones based on regular sets when the change occurs near 0 or 1. For the bump detection problem, the main difference in the estimated powers concerns the case $\ell = 0.2$ where the better performance of the linear statistics based test $(TC1^u)$ as compared with the quadratic statistics based test $(TC2^u)$ to detect large negative jumps is pronounced.

4. The comparison of the simulated powers of the different testing procedures confirm again the intuition that detecting a bump is harder than detecting a jump. The simulation study also highlights that it is substantially easier to detect a jump or a bump with negative change height than with positive change height probably for the same reasons as in the above known baseline intensity case.

5 Proofs of the main results

Notation

As explained in the introduction, the main tools to prove our nonasymptotic minimax separation rates upper bounds are exponential inequalities. Many of these exponential inequalities involve the function g defined for every x > -1 by

$$g(x) = (1+x)\log(1+x) - x \quad . \tag{79}$$

The inverse function of g restricted to $]0, +\infty[$ is denoted by g^{-1} and is defined on $]0, +\infty[$. For any x > 0, one has:

$$g^{-1}(x) \le 2x/3 + \sqrt{2x}$$
 (80)

The inverse function of g restricted to]-1,0[is denoted by $g_{|]-1,0[}^{-1}$ and is defined on]0,1[. It is easy to see that $g(-x) \ge x^2/2 \ge g(x)$ for all x in]0,1[, which allows to prove that

$$g_{|]-1,0[}^{-1}(x) \ge -\sqrt{2x} \ge -g^{-1}(x)$$
 (81)

5.1 Proof of Proposition 2

The first statement of Proposition 2 directly results from the Neyman-Pearson fundamental lemma and Girsanov's lemma recalled in Lemma 1.

Assume that $\delta^* > 0$ and notice that the assumption (7) leads to

$$\delta^* \ell^* L \ge \sqrt{\frac{(\lambda_0 + \delta^*)\ell^* L}{\beta}} + \sqrt{\frac{\lambda_0 \ell^* L}{\alpha}} .$$
(82)

From (82), the quantile bound (280) and the Bienayme-Chebyshev inequality, we obtain

$$P_{\lambda}(\phi_{1,\alpha}^{+}(N) = 0) \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] \leq \sqrt{\frac{\lambda_{0}\ell^{*}L}{\alpha}} + \lambda_{0}\ell^{*}L\right)$$
$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] \leq (\lambda_{0}^{*}+\delta^{*})\ell^{*}L - \sqrt{\frac{(\lambda_{0}+\delta^{*})\ell^{*}L}{\beta}}\right)$$
$$\leq \beta .$$

Assume now that $-\lambda_0^* < \delta^* < 0$ and notice that the assumption (7) leads to

$$\delta^* \ell^* L \le -\sqrt{\frac{(\lambda_0 + \delta^*)\ell^* L}{\beta}} - \sqrt{\frac{\lambda_0 \ell^* L}{\alpha}} . \tag{83}$$

As above, using (83), (280) and the Bienayme-Chebyshev inequality, we obtain

$$P_{\lambda}(\phi_{1,\alpha}^{-}(N) = 0) \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] \geq -\sqrt{\frac{\lambda_{0}\ell^{*}L}{\alpha}} + \lambda_{0}\ell^{*}L\right)$$
$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] \geq (\lambda_{0}+\delta^{*})\ell^{*}L + \sqrt{\frac{(\lambda_{0}+\delta^{*})\ell^{*}L}{\beta}}\right)$$
$$\leq \beta .$$

5.2 Proof of Proposition 3

Let $C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2$. Let us introduce for r > 0 the Poisson intensity λ_r defined by

$$\lambda_r(t) = \lambda_0 + \frac{r}{\sqrt{\ell^*}} \mathbb{1}_{(\tau^*, \tau^* + \ell^*]}(t) \text{ for all } t \text{ in } [0, 1]$$

Notice that λ_r belongs to $(\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0])_r = \{\lambda \in \mathcal{S}_{,\tau^*,\ell^*}[\lambda_0], d_2(\lambda,\mathcal{S}_0[\lambda_0]) \ge r\}$, as defined by Lemma 42. We get from Lemma 1 and Lemma 44 that

$$E_{\lambda_0}\left[\left(\frac{dP_{\lambda_r}}{dP_{\lambda_0}}\right)^2(N)\right] = \exp\left(\frac{r^2L}{\lambda_0}\right) .$$

Choosing $r = (\lambda_0 \log C_{\alpha,\beta}/L)^{1/2}$ then leads to $E_{\lambda_0} \left[(dP_{\lambda_r}/dP_{\lambda_0})^2 (N) \right] = C_{\alpha,\beta}$, and thanks to Lemma 43, we conclude that $\rho_{\alpha} \left((\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0])_r \right) \geq \beta$ and $\mathrm{mSR}_{\alpha,\beta} \left(\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0] \right) \geq r$.

5.3 Proof of Proposition 4

The first statement of the proposition is straightforward.

As for the second kind error study, let us consider first $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*, \tau^* + \ell^*]}$ in $\mathcal{S}_{,\tau^*,\ell^*}[\lambda_0]$ with $\delta > 0$. From the quantile bound (280), one deduces that

$$P_{\lambda}\left(\phi_{2,\alpha}^{(1)}=0\right) \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] \leq \sqrt{\frac{\lambda_{0}\ell^{*}L}{\alpha_{1}}} + \lambda_{0}\ell^{*}L\right) ,$$
$$= P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] - (\lambda_{0}+\delta)\,\ell^{*}L \leq -\delta\ell^{*}L + \sqrt{\frac{\lambda_{0}\ell^{*}L}{\alpha_{1}}}\right) .$$

It remains to find a condition on $d_2(\lambda, S_0[\lambda_0])$ which will guarantee that

$$-\delta\ell^*L + \sqrt{\frac{\lambda_0\ell^*L}{\alpha_1}} \le -\sqrt{\frac{(\lambda_0+\delta)\ell^*L}{\beta}} , \qquad (84)$$

so that $P_{\lambda}(\phi_{2,\alpha}^{(1)}=0) \leq \beta$ thanks to the Bienayme-Chebyshev inequality. Let us assume for instance that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge 2\sqrt{\frac{\lambda_0}{L}} \left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_1}}\right) + \frac{1}{\beta\sqrt{\ell^*L}}$$

Since $d_2(\lambda, \mathcal{S}_0[\lambda_0]) = \delta \sqrt{\ell^*}$, this implies

$$\delta\sqrt{\ell^*} \ge 2\sqrt{\frac{\lambda_0}{L}}\left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_1}}\right) + \frac{1}{\beta\sqrt{\ell^*}L}$$
,

whereby

$$\delta\sqrt{\ell^*} - \left(\frac{\delta\sqrt{\ell^*}}{2} + \frac{1}{2\beta\sqrt{\ell^*}L}\right) \ge \sqrt{\frac{\lambda_0}{L}} \left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_1}}\right) \quad .$$

Using the basic inequality $\sqrt{ab} \leq (a+b)/2$ then leads to

$$\delta\sqrt{\ell^*} - \sqrt{\frac{\delta}{\beta L}} \ge \sqrt{\frac{\lambda_0}{L}} \left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_1}}\right) ,$$

and (84) conveniently follows.

Let us consider now $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*, \tau^* + \ell^*]}$ in $\mathcal{S}_{, \tau^*, \ell^*}[\lambda_0]$ with δ in $(-\lambda_0, 0)$. From the quantile bound (280) again, one deduces that

$$P_{\lambda}\left(\phi_{2,\alpha}^{(1)}=0\right) \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] \geq -\sqrt{\frac{\lambda_{0}\ell^{*}L}{\alpha_{2}}}+\lambda_{0}\ell^{*}L\right)$$
$$=P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]-(\lambda_{0}+\delta)\,\ell^{*}L \geq -\delta\ell^{*}L-\sqrt{\frac{\lambda_{0}\ell^{*}L}{\alpha_{2}}}\right).$$

As above, it remains to find a condition on $d_2(\lambda, S_0[\lambda_0])$ which will guarantee that

$$-\delta\ell^*L - \sqrt{\frac{\lambda_0\ell^*L}{\alpha_2}} \ge \sqrt{\frac{(\lambda_0+\delta)\ell^*L}{\beta}} , \qquad (85)$$

•

so that $P_{\lambda}(\phi_{2,\alpha}^{(1)}=0) \leq \beta$. Since $d_2(\lambda, \mathcal{S}_0[\lambda_0]) = -\delta\sqrt{\ell^*}$ and $\delta < 0$, the following condition suffices

$$d_2\left(\lambda, \mathcal{S}_0[\lambda_0]\right) \ge \sqrt{\frac{\lambda_0}{L}} \left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_2}}\right)$$

Taking $C(\alpha, \beta, \lambda_0, \ell^*) = \max\left(2\sqrt{\lambda_0}\left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_1}}\right) + \frac{1}{\beta\sqrt{\ell^*}}, \sqrt{\lambda_0}\left(\frac{1}{\sqrt{\beta}} + \frac{1}{\sqrt{\alpha_2}}\right)\right)$ finally allows to conclude the proof.

5.4 Proof of Proposition 5

The first statement of the proposition is straightforward. Let us now consider $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*, \tau^* + \ell^*]}$ in $\mathcal{S}_{,\tau^*, \ell^*}[\lambda_0]$. Since $T_{\tau^*, \tau^* + \ell^*}(N)$ is centered under $(H_0), t_{\lambda_0, \tau^*, \tau^* + \ell^*}(1 - \alpha) \leq \sqrt{\operatorname{Var}_{\lambda_0}(T_{\tau^*, \tau^* + \ell^*}(N))/\alpha}$. From the variance computation of Lemma 47 under (H_0) , we derive the upper bound $t_{\lambda_0, \tau^*, \tau^* + \ell^*}(1 - \alpha) \leq (\lambda_0/L)\sqrt{2/\alpha}$. Moreover, still using Lemma 47 but under (H_1) now, one can see that $E_{\lambda}[T_{\tau^*, \tau^* + \ell^*}(N)] = \delta^2 \ell^*$ (recall that $T_{\tau^*, \tau^* + \ell^*}(N)$ is an unbiased estimator of $d_2^2(\lambda, \mathcal{S}_0[\lambda_0] = \delta^2 \ell^*)$, and

$$\operatorname{Var}_{\lambda}(T_{\tau^*,\tau^*+\ell^*}(N)) = \frac{4(\lambda_0+\delta)\delta^2\ell^*}{L} + \frac{2(\lambda_0+\delta)^2}{L^2}$$

Therefore,

$$P_{\lambda}(\phi_{2,\alpha}^{(2)} = 0) = P_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) \le t_{\lambda_0,\tau^*,\tau^* + \ell^*}(1 - \alpha) \right) ,$$

$$\leq P_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) \le \frac{\lambda_0}{L} \sqrt{\frac{2}{\alpha}} \right) ,$$

$$\leq P_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) - \delta^2 \ell^* \le \frac{\lambda_0}{L} \sqrt{\frac{2}{\alpha}} - \delta^2 \ell^* \right) .$$

Assume now that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{C(\alpha, \beta, \lambda_0, \ell^*)}{\sqrt{L}}$$

with

$$C(\alpha,\beta,\lambda_0,\ell^*) = \max\left(\sqrt{3\lambda_0\left(\sqrt{\frac{2}{\alpha}} + \sqrt{\frac{2}{\beta}}\right)} , \frac{6\sqrt{\lambda_0}}{\sqrt{\beta}} + \frac{3\sqrt{2}}{\sqrt{\beta\ell^*L}} , \frac{36}{\beta\sqrt{\ell^*L}}\right) .$$

This implies

$$\delta^2 \ell^* \ge 3 \max\left(\frac{\lambda_0}{L} \left(\sqrt{\frac{2}{\alpha}} + \sqrt{\frac{2}{\beta}}\right), \ |\delta| \sqrt{\ell^*} \left(\frac{2}{\sqrt{L}} \sqrt{\frac{\lambda_0}{\beta}} + \frac{1}{L} \sqrt{\frac{2}{\beta\ell^*}}\right), \ \frac{2|\delta|^{3/2} \sqrt{\ell^*}}{\sqrt{\beta L}}\right)$$

and then

$$\delta^2 \ell^* \geq \frac{\lambda_0}{L} \sqrt{\frac{2}{\alpha}} + 2\sqrt{\frac{\lambda_0}{\beta L}} |\delta| \sqrt{\ell^*} + \sqrt{\frac{2}{\beta}} \left(\frac{\lambda_0}{L} + \frac{1}{L\sqrt{\ell^*}} |\delta| \sqrt{\ell^*}\right) + \frac{2|\delta|^{3/2} \sqrt{\ell^*}}{\sqrt{\beta L}} ,$$

hence, using $\sqrt{\lambda_0 + \delta} \le \sqrt{\lambda_0} + \sqrt{|\delta|}$,

$$\delta^{2}\ell^{*} \geq \frac{\lambda_{0}}{L}\sqrt{\frac{2}{\alpha}} + 2\sqrt{\frac{\lambda_{0}+\delta}{L\beta}}|\delta|\sqrt{\ell^{*}} + \sqrt{\frac{2}{\beta}}\frac{\lambda_{0}+\delta}{L} \quad .$$

$$(86)$$

Therefore,

$$\begin{aligned} P_{\lambda}(\phi_{2,\alpha}^{(2)} &= 0) \leq P_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) - \delta^2 \ell^* \leq \frac{\lambda_0}{L} \sqrt{\frac{2}{\alpha}} - \delta^2 \ell^* \right) \\ &\leq P_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) - \delta^2 \ell^* \leq -\frac{1}{\sqrt{\beta}} \sqrt{\frac{4(\lambda_0 + \delta)\delta^2 \ell^*}{L} + \frac{2(\lambda_0 + \delta)^2}{L^2}} \right) \\ &\leq P_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) - E_{\lambda} \left[T_{\tau^*,\tau^* + \ell^*}(N) \right] \leq -\sqrt{\frac{\operatorname{Var}_{\lambda} \left(T_{\tau^*,\tau^* + \ell^*}(N) \right)}{\beta}} \right) \\ &\leq \beta \end{aligned}$$

This concludes the proof.

5.5 Proof of Proposition 6

Let us first give a short proof for the tests $\phi_{3,\alpha}^{(1)+}$ and $\phi_{3,\alpha}^{(1)-}$. Start by remarking that the first kind error rates control of both tests is straightforward. Since $g^{-1}(x) \leq 2x/3 + \sqrt{2x}$ for all x > 0 (see (80)), Lemma 49 leads to

$$p_{\lambda_0,\ell^*}^+(1-\alpha) \le \lambda_0 \ell^* L + 2\sqrt{2\lambda_0 \log(2/\alpha) L} + 4\log(2/\alpha)/3 , \qquad (87)$$

and

$$p_{\lambda_0,\ell^*}^-(\alpha) \ge \lambda_0 \ell^* L - 2\sqrt{2\lambda_0 \log\left(2/\alpha\right)L} - 4\log\left(2/\alpha\right)/3 . \tag{88}$$

Let us consider $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau, \tau+\ell^*]}$ in $\mathcal{S}_{\delta^*, \dots, \ell^*}$ and assume that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{1}{\sqrt{L}} \left(2\sqrt{\frac{2\lambda_0 \log\left(2/\alpha\right)}{\ell^*}} + \sqrt{\frac{\lambda_0 + \delta^*}{\beta}} \right) + \frac{4\log\left(2/\alpha\right)}{3L\sqrt{\ell^*}} \quad (89)$$

If $\delta^* > 0$, the condition (89) yields

$$\delta^* \ell^* L \ge 2\sqrt{2\lambda_0 \log\left(2/\alpha\right)L} + \frac{4}{3}\log\left(2/\alpha\right) + \sqrt{\frac{(\lambda_0 + \delta^*)\ell^* L}{\beta}} , \qquad (90)$$

which entails

$$P_{\lambda}\left(\max_{t\in[0,1-\ell^*]} N(t,t+\ell^*] \le p_{\lambda_0,\ell^*}^+(1-\alpha)\right)$$
$$\le P_{\lambda}\left(\max_{t\in[0,1-\ell^*]} N(t,t+\ell^*] \le (\lambda_0+\delta^*)\ell^*L - \sqrt{\frac{(\lambda_0+\delta^*)\ell^*L}{\beta}}\right)$$
$$\le P_{\lambda}\left(N(\tau,\tau+\ell^*] - (\lambda_0+\delta^*)L\ell^* \le -\sqrt{\frac{(\lambda_0+\delta^*)\ell^*L}{\beta}}\right)$$
$$\le \beta \quad \text{with the Bionexme Chebyshev inequality}$$

 $\leq \beta$ with the Bienayme-Chebyshev inequality .

This concludes the proof for $\phi_{3,\alpha}^{(1)+}$. If δ^* belongs to $(-\lambda_0, 0)$, the condition (89) yields

$$-\delta^* \ell^* L \ge 2\sqrt{2\lambda_0 \log\left(2/\alpha\right)L} + \frac{4}{3}\log\left(2/\alpha\right) + \sqrt{\frac{(\lambda_0 + \delta^*)\ell^* L}{\beta}} \quad . \tag{91}$$

We get then as above, with (88), (91) and the Bienayme-Chebyshev inequality,

$$\begin{split} P_{\lambda} \Biggl(\min_{t \in [0, 1-\ell^*]} N(t, t+\ell^*] \geq p_{\lambda_0, \ell^*}^-(\alpha) \Biggr) \\ &\leq P_{\lambda} \left(\min_{t \in [0, 1-\ell^*]} N(t, t+\ell^*] \geq \lambda_0 \ell^* L - 2\sqrt{2\lambda_0 \log\left(2/\alpha\right) L} - \frac{4}{3} \log\left(2/\alpha\right) \Biggr) \\ &\leq P_{\lambda} \left(\min_{t \in [0, 1-\ell^*]} N(t, t+\ell^*] \geq (\lambda_0 + \delta^*) \ell^* L + \sqrt{\frac{(\lambda_0 + \delta^*) \ell^* L}{\beta}} \right) \\ &\leq P_{\lambda} \left(N(\tau, \tau+\ell^*] - (\lambda_0 + \delta^*) \ell^* L \geq \sqrt{\frac{(\lambda_0 + \delta^*) \ell^* L}{\beta}} \right) \\ &\leq \beta \ . \end{split}$$

This concludes the proof for $\phi_{3,\alpha}^{(1)-}$.

Now, let us turn to the test $\phi_{3/4,\alpha}^{(2)}$. As above, start by remarking that the first kind error rate control of this test straightforwardly follows from a basic union bound:

$$P_{\lambda_0}\left(\phi_{3/4,\alpha}^{(2)}(N)=1\right) \leq \sum_{k=0}^{\lceil (1-\ell^*)M\rceil-1} P_{\lambda_0}\left(T_{\frac{k}{M},\frac{k}{M}+\ell^*}(N) > t_{\frac{k}{M},\frac{k}{M}+\ell^*}\left(1-u_{\alpha}\right)\right)$$
$$\leq \sum_{k=0}^{\lceil (1-\ell^*)M\rceil-1} \frac{\alpha}{\lceil (1-\ell^*)M\rceil}$$
$$\leq \alpha \quad .$$

Let λ in $\mathcal{S}_{\delta^*, \dots, \ell^*}$ such that $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau, \tau + \ell^*]}$ with τ in $[0, 1 - \ell^*]$, and assume that the following holds:

$$d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right) \geq \frac{2}{\sqrt{L}} \max\left(8\sqrt{\frac{\lambda_{0} + |\delta^{*}|}{\beta}}, \left(\frac{4\sqrt{2}(\lambda_{0} + |\delta^{*}|)}{\sqrt{\beta}} + 8\lambda_{0}\left(\frac{2}{3}\frac{\log\left(3/u_{\alpha}\right)}{\sqrt{\lambda_{0}\ell^{*}L}} + \sqrt{2\log\left(3/u_{\alpha}\right)}\right)^{2}\right)^{\frac{1}{2}}\right). \quad (92)$$

This entails

$$d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{8\lambda_0}{L} \left(\frac{2}{3} \frac{\log\left(3/u_\alpha\right)}{\sqrt{\lambda_0 \ell^* L}} + \sqrt{2\log\left(3/u_\alpha\right)}\right)^2 + \frac{4\sqrt{2}(\lambda_0 + |\delta^*|)}{\sqrt{\beta}L} + \frac{8d_2(\lambda, \mathcal{S}_0[\lambda_0])}{\sqrt{L}} \sqrt{\frac{\lambda_0 + |\delta^*|}{\beta}} \right)$$
(93)

Noticing that

$$P_{\lambda}\left(\phi_{3/4,\alpha}^{(2)}(N)=0\right) \leq \min_{k \in \{0,\dots,\lceil (1-\ell^*)M\rceil-1\}} P_{\lambda}\left(T_{\frac{k}{M},\frac{k}{M}+\ell^*}(N) \leq t_{\frac{k}{M},\frac{k}{M}+\ell^*}(1-u_{\alpha})\right) ,$$

we only need to exhibit some k_{τ} in $\{0, ..., \lceil (1 - \ell^*)M \rceil - 1\}$ satisfying

$$P_{\lambda}\left(T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) \leq t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha})\right) \leq \beta \quad .$$

We set $k_{\tau} = \lfloor \tau M \rfloor$. Since $0 < \tau < 1 - \ell^*$, k_{τ} actually belongs to $\{0, ..., \lceil (1 - \ell^*)M \rceil - 1\}$, and since $M = \lceil 2/\ell^* \rceil$, $k_{\tau}/M \leq \tau < k_{\tau}/M + \ell^*/2$. Therefore, using Lemma 47 equation (282), we get on the one hand

$$E_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] = \delta^{*2}\frac{(\ell^{*}+k_{\tau}/M-\tau)^{2}}{\ell^{*}} \ge \frac{\delta^{*2}\ell^{*}}{4} ,$$

that is

$$E_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] \geq \frac{d_{2}^{2}(\lambda,\mathcal{S}_{0}[\lambda_{0}])}{4} , \qquad (94)$$

and on the other hand with Lemma 47 equation (283),

$$\operatorname{Var}_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] = \frac{4\delta^{*2}\left(\lambda_{0}\ell^{*}+\delta^{*}(k_{\tau}/M+\ell^{*}-\tau)\right)}{L}\frac{(k_{\tau}/M+\ell^{*}-\tau)^{2}}{\ell^{*2}} + \frac{2}{L^{2}}\frac{(\lambda_{0}\ell^{*}+\delta^{*}(k_{\tau}/M+\ell^{*}-\tau))^{2}}{\ell^{*2}} \leq \frac{4(\lambda_{0}+|\delta^{*}|)}{L}d_{2}^{2}(\lambda,\mathcal{S}_{0}[\lambda_{0}]) + \frac{2(\lambda_{0}+|\delta^{*}|)^{2}}{L^{2}}.$$
(95)

Moreover, Lemma 48 entails

$$t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha}) \leq 2\lambda_{0}^{2}\ell^{*}\left(g^{-1}\left(\frac{\log\left(3/u_{\alpha}\right)}{\lambda_{0}\ell^{*}L}\right)\right)^{2},$$

with $g^{-1}(x) \le 2x/3 + \sqrt{2x}$ (see (80)), which implies, with (93), (94) and (95) that

$$t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha}) \leq E_{\lambda} \left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) \right] - \sqrt{\operatorname{Var}_{\lambda} \left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) \right] / \beta}$$

We simply conclude the proof for $\phi^{(2)}_{3/4,\alpha}$ with

$$P_{\lambda}\left(T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) \leq t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha})\right)$$

$$\leq P_{\lambda}\left(T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) - E_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] \leq -\sqrt{\operatorname{Var}_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right]/\beta}\right)$$

$$\leq \beta \quad .$$

5.6 Proof of Proposition 8

The control of the first kind error rates of the two tests $\phi_{4,\alpha}^{(1)}$ and $\phi_{3/4,\alpha}^{(2)}$ is straightforward using simple union bounds.

(i) Control of the second kind error rate of $\phi_{4,\alpha}^{(1)}$. Recall from the proof of Proposition 6, equations (87) and (88), that

$$\begin{cases}
p_{\lambda_{0},\ell^{*}}^{+}(1-\alpha/2) \leq \lambda_{0}\ell^{*}L + 2\sqrt{2\lambda_{0}\log(4/\alpha)L} + 4\log(4/\alpha)/3, \\
p_{\lambda_{0},\ell^{*}}^{-}(\alpha/2) \geq \lambda_{0}\ell^{*}L - 2\sqrt{2\lambda_{0}\log(4/\alpha)L} - 4\log(4/\alpha)/3.
\end{cases}$$
(96)

Let us first set λ in $\mathcal{S}_{,.,\ell^*}$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell^*]}$ with $\delta > 0$ or δ in $(-\lambda_0, 0)$, τ in $(0, 1 - \ell^*)$, and

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{2\sqrt{\lambda_0}}{\sqrt{L}} \left(\frac{1}{\sqrt{\beta}} + 2\sqrt{\frac{2\log(4/\alpha)}{\ell^*}}\right) + \frac{1}{\sqrt{\ell^*}L} \left(\frac{1}{\beta} + \frac{8}{3}\log(4/\alpha)\right) \quad . \tag{97}$$

The condition (97) entails

$$|\delta|\sqrt{\ell^*} \ge \frac{|\delta|\sqrt{\ell^*}}{2} + \frac{\sqrt{\lambda_0}}{\sqrt{L}} \left(\frac{1}{\sqrt{\beta}} + 2\sqrt{\frac{2\log(4/\alpha)}{\ell^*}}\right) + \frac{1}{\sqrt{\ell^*}L} \left(\frac{1}{2\beta} + \frac{4}{3}\log(4/\alpha)\right) ,$$

and therefore, using the inequalities $\sqrt{ab} \leq (a+b)/2$ and $\sqrt{a+b} \leq \sqrt{a} + \sqrt{b}$ for every $a, b \geq 0$,

$$\begin{aligned} |\delta|\sqrt{\ell^*} &\geq \sqrt{\frac{|\delta|}{\beta L}} + \frac{\sqrt{\lambda_0}}{\sqrt{L}} \left(\frac{1}{\sqrt{\beta}} + 2\sqrt{\frac{2\log(4/\alpha)}{\ell^*}}\right) + \frac{4}{3} \frac{\log(4/\alpha)}{\sqrt{\ell^*}L} \\ &\geq \sqrt{\frac{|\delta| + \lambda_0}{\beta L}} + 2\sqrt{\frac{2\log(4/\alpha)\lambda_0}{\ell^*L}} + \frac{4}{3} \frac{\log(4/\alpha)}{\sqrt{\ell^*}L} . \end{aligned}$$
(98)

Then, assuming that $\delta > 0$, we conclude with the following inequalities:

$$\begin{aligned} P_{\lambda}\left(\phi_{4,\alpha}^{(1)}(N)=0\right) &\leq P_{\lambda}\left(\phi_{3,\alpha/2}^{(1)+}(N)=0\right) \\ &\leq P_{\lambda}\left(\max_{t\in[0,1-\ell^{*}]}N(t,t+\ell^{*}]\leq p_{\lambda_{0},\ell^{*}}^{+}(1-\alpha/2)\right) \\ &\leq P_{\lambda}\left(N(\tau,\tau+\ell^{*}]\leq\lambda_{0}\ell^{*}L+2\sqrt{2\lambda_{0}\log\left(4/\alpha\right)L}+4\log\left(4/\alpha\right)/3\right) \text{ with (96)} \\ &\leq P_{\lambda}\left(N(\tau,\tau+\ell^{*}]-(\lambda_{0}^{*}+\delta)\ell^{*}L\leq-\sqrt{\frac{(\lambda_{0}+\delta)\ell^{*}L}{\beta}}\right) \text{ with (98)} \\ &\leq \beta \text{ with the Bienayme-Chebyshev inequality }. \end{aligned}$$

Assuming now that δ is in $(-\lambda_0, 0)$,

$$P_{\lambda}\left(\phi_{4,\alpha}^{(1)}(N)=0\right) \leq P_{\lambda}\left(\phi_{3,\alpha/2}^{(1)-}(N)=0\right)$$

$$\leq P_{\lambda}\left(\min_{t\in[0,1-\ell^{*}]}N(t,t+\ell^{*}]\geq p_{\lambda_{0},\ell^{*}}^{-}(\alpha/2)\right)$$

$$\leq P_{\lambda}\left(N(\tau,\tau+\ell^{*}]\geq\lambda_{0}\ell^{*}L-2\sqrt{2\lambda_{0}\log\left(4/\alpha\right)L}-4\log\left(4/\alpha\right)/3\right) \text{ with (96)}$$

$$\leq P_{\lambda}\left(N(\tau,\tau+\ell^{*}]-(\lambda_{0}+\delta)\ell^{*}L\geq\sqrt{\frac{(\lambda_{0}+|\delta|)\ell^{*}L}{\beta}}\right) \text{ with (98)}$$

$$\leq \beta .$$

(ii) Control of the second kind error rate of $\phi_{3/4,\alpha}^{(2)}$.

Let us now set λ in $\mathcal{S}_{,..,\ell^*}$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell^*]}$ with $\delta > 0$ or δ in $(-\lambda_0, 0), \tau$ in $(0, 1 - \ell^*)$, and

$$d_{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) \geq 2 \max\left(12\frac{\sqrt{\lambda_{0}}}{\sqrt{\beta L}} + \frac{6\sqrt{2}}{\sqrt{\beta \ell^{*}L}}, \frac{288}{\beta\sqrt{\ell^{*}L}}\right), \frac{\sqrt{3\lambda_{0}(4\log(3/u_{\alpha}) + \sqrt{2/\beta})}}{\sqrt{L}} + \frac{2\sqrt{2}\log(3/u_{\alpha})}{\sqrt{3\ell^{*}L}} + \frac{2\sqrt{2}(2\lambda_{0})^{1/4}\log^{3/4}(3/u_{\alpha})}{\ell^{*1/4}L^{3/4}}\right).$$
(99)

Notice that (99) entails that

$$d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \ge 3 \max\left(d_2(\lambda, \mathcal{S}_0[\lambda_0]) \left(\frac{8\sqrt{\lambda_0}}{\sqrt{\beta L}} + \frac{4\sqrt{2}}{\sqrt{\beta \ell^* L}}\right), \quad d_2^{3/2}(\lambda, \mathcal{S}_0[\lambda_0]) \frac{8}{\ell^{*1/4}\sqrt{\beta L}}, \\ \frac{16\lambda_0 \log\left(3/u_\alpha\right)}{L} + \frac{32\log^2\left(3/u_\alpha\right)}{9\ell^* L^2} + \frac{32\sqrt{2\lambda_0}\log^{3/2}\left(3/u_\alpha\right)}{3\sqrt{\ell^* L^{3/2}}} + \frac{4\lambda_0\sqrt{2}}{\sqrt{\beta L}}\right),$$

and therefore

$$\begin{aligned} d_{2}^{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) &\geq d_{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) \left(\frac{8\sqrt{\lambda_{0}}}{\sqrt{\beta L}} + \frac{4\sqrt{2}}{\sqrt{\beta \ell^{*}L}} \right) + d_{2}^{3/2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) \frac{8}{\ell^{*1/4}\sqrt{\beta L}} + \frac{16\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{32\log^{2}\left(3/u_{\alpha}\right)}{9\ell^{*}L^{2}} + \frac{32\sqrt{2\lambda_{0}}\log^{3/2}\left(3/u_{\alpha}\right)}{3\sqrt{\ell^{*}L^{3/2}}} + \frac{4\lambda_{0}\sqrt{2}}{\sqrt{\beta L}} . \end{aligned}$$

Since $d_2(\lambda, \mathcal{S}_0[\lambda_0]) = |\delta|\sqrt{\ell^*}$ and $\sqrt{\lambda_0 + |\delta|} \le \sqrt{\lambda_0} + \sqrt{|\delta|}$, this implies that

$$\frac{d_2^2(\lambda, \mathcal{S}_0[\lambda_0])}{4} \ge \frac{2\sqrt{\lambda_0 + |\delta|}}{\sqrt{\beta L}} d_2(\lambda, \mathcal{S}_0[\lambda_0]) + \frac{\sqrt{2}(|\delta| + \lambda_0)}{\sqrt{\beta L}} + \frac{4\lambda_0 \log(3/u_\alpha)}{L} + \frac{8\log^2(3/u_\alpha)}{9\ell^* L^2} + \frac{8\sqrt{2\lambda_0}\log^{3/2}(3/u_\alpha)}{3\sqrt{\ell^*} L^{3/2}} .$$
(100)

Let us now prove that $P_{\lambda}\left(\phi_{3/4,\alpha}^{(2)}(N)=0\right) \leq \beta$. From the definition (18), we notice that

$$P_{\lambda}\left(\phi_{3/4,\alpha}^{(2)}(N)=0\right) \leq \min_{k \in \{0,\dots,\lceil (1-\ell^{*})M\rceil-1\}} P_{\lambda}\left(T_{\frac{k}{M},\frac{k}{M}+\ell^{*}}(N)\leq t_{\frac{k}{M},\frac{k}{M}+\ell^{*}}\left(1-u_{\alpha}\right)\right) ,$$

so that we only need to exhibit some k_{τ} in $\{0, ..., \lceil (1-\ell^*)M \rceil - 1\}$ such that

$$P_{\lambda}\left(T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) \leq t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha})\right) \leq \beta$$

As in the proof of Proposition 6, we choose $k_{\tau} = \lfloor \tau M \rfloor$, which leads (see (94) and (95)) to

$$E_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] \geq \frac{d_{2}^{2}(\lambda,\mathcal{S}_{0}[\lambda_{0}])}{4} , \qquad (101)$$

and

$$\operatorname{Var}_{\lambda}\left[T_{\frac{k\tau}{M},\frac{k\tau}{M}+\ell^{*}}(N)\right] \leq \frac{4(\lambda_{0}+|\delta|)}{L}d_{2}^{2}(\lambda,\mathcal{S}_{0}[\lambda_{0}]) + \frac{2(\lambda_{0}+|\delta|)^{2}}{L^{2}} \quad (102)$$

From (100), (101) and (102), we derive that

$$E_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] \geq \sqrt{\operatorname{Var}_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right]/\beta} + \frac{4\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{8\log^{2}\left(3/u_{\alpha}\right)}{9\ell^{*}L^{2}} + \frac{8\sqrt{2\lambda_{0}}\log^{3/2}\left(3/u_{\alpha}\right)}{3\sqrt{\ell^{*}}L^{3/2}} .$$

The conclusion then basically follows from Lemma 48, supplemented by the upper bound (80), which allows to see that

$$E_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^*}(N)\right] \geq \sqrt{\operatorname{Var}_{\lambda}\left[T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^*}(N)\right]/\beta} + t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^*}(1-u_{\alpha}) ,$$

and the Bienayme-Chebyshev inequality, which entails

$$P_{\lambda}\left(T_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N) \leq t_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha})\right) \leq \beta .$$

5.7 Proof of Proposition 9

Let $C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2$, $r = (\lambda_0 \log C_{\alpha,\beta}/L)^{1/2}$ and λ_r defined for all t in (0,1) by $\lambda_r(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau^* \tau^* + r^2/\delta^*)}(t)$.

Notice that for all $L \geq \lambda_0 \log C_{\alpha,\beta}/(\delta^{*2}(1-\tau^*))$, $r^2/\delta^{*2} \leq 1-\tau^*$ and λ_r belongs to $(\mathcal{S}_{\delta^*,\tau^*},...[\lambda_0])_r$ in the notation of Lemma 42. We get now from Lemma 1 and Lemma 44

$$E_{\lambda_0}\left[\left(\frac{dP_{\lambda_r}}{dP_0}\right)^2(N)\right] = \exp\left(\frac{r^2L}{\lambda_0}\right) = C_{\alpha,\beta}.$$

Lemmas 43 and 42 then entail $\rho_{\alpha}\left(\left(\mathcal{S}_{\delta^{*},\tau^{*},\dots}[\lambda_{0}]\right)_{r}\right) \geq \beta$ and $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^{*},\tau^{*},\dots}[\lambda_{0}]\right) \geq r$.

5.8 Proof of Proposition 10

The first kind error rate control is straightforward. As for the second kind error rate control, let $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau^*, \tau^* + \ell]}$ belonging to $S_{\delta^*, \tau^*, \dots}[\lambda_0]$ with ℓ in $(0, 1 - \tau^*)$ and satisfying

$$d_{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) \geq \frac{2}{\sqrt{L}} \max\left(2\sqrt{\frac{\lambda_{0}+\delta^{*}}{\beta}}, \sqrt{\delta^{*}s^{+}_{\lambda_{0},\frac{\delta^{*}}{2}}(1-\alpha)}\mathbb{1}_{\{\delta^{*}>0\}} + \sqrt{\frac{|\delta^{*}|\log\left(1/\alpha\right)}{\log\left(\lambda_{0}/\left(\lambda_{0}-|\delta^{*}|/2\right)\right)}}\mathbb{1}_{\{-\lambda_{0}<\delta^{*}<0\}}\right).$$
(103)

Assume that $\delta^* > 0$ and recall that $s^+_{\lambda_0,\delta^*/2}(1-\alpha)$ defined in Lemma 50 is a constant which does not depend on L. The assumption (103) implies

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{2}{\sqrt{L}} \max\left(2\sqrt{\frac{\lambda_0 + \delta^*}{\beta}}, \sqrt{\delta^* s^+_{\lambda_0, \frac{\delta^*}{2}}(1-\alpha)}\right) ,$$

which yields

$$\delta^* \ell \ge 4 \max\left(\sqrt{\frac{(\lambda_0 + \delta^*)\ell}{\beta L}}, \frac{s^+_{\lambda_0, \frac{\delta^*}{2}}(1 - \alpha)}{L}\right) ,$$

hence

$$\frac{\delta^*}{2}\ell L \ge \sqrt{\frac{(\lambda_0 + \delta^*)\ell L}{\beta}} + s^+_{\lambda_0,\frac{\delta^*}{2}} (1 - \alpha) \quad . \tag{104}$$

We get then

$$P_{\lambda}(\phi_{5,\alpha}(N) = 0) \le P_{\lambda}\left(\sup_{\ell' \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell'}(N) \le s^+_{\lambda_0,\delta^*,\tau^*,L}(1-\alpha)\right) ,$$

where

$$S_{\delta^*,\tau^*,\tau^*+\ell'}(N) = \operatorname{sgn}(\delta^*) \Big(N(\tau^*,\tau^*+\ell'] - \lambda_0 L\ell' \Big) - |\delta^*| L\ell'/2 ,$$

as defined by (21) and $s^+_{\lambda_0,\delta^*,\tau^*,L}(u)$ is the *u*-quantile of $\sup_{\ell' \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell'}(N)$ under (H_0) . From the quantile upper bound (286), we deduce

$$P_{\lambda}\left(\phi_{5,\alpha}(N)=0\right) \leq P_{\lambda}\left(\sup_{\ell'\in(0,1-\tau^{*})} S_{\delta^{*},\tau^{*},\tau^{*}+\ell'}(N) \leq s_{\lambda_{0},\frac{\delta^{*}}{2}}^{+}\left(1-\alpha\right)\right)$$

$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell]-\left(\lambda_{0}+\frac{\delta^{*}}{2}\right)\ell L \leq s_{\lambda_{0},\frac{\delta^{*}}{2}}^{+}\left(1-\alpha\right)-\frac{\delta^{*}}{2}\ell L\right)$$

$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell]-(\lambda_{0}+\delta^{*})\ell L \leq s_{\lambda_{0},\frac{\delta^{*}}{2}}^{+}\left(1-\alpha\right)-\frac{\delta^{*}}{2}\ell L\right)$$

$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell]-(\lambda_{0}+\delta^{*})\ell L \leq -\sqrt{\frac{(\lambda_{0}+\delta^{*})\ell L}{\beta}}\right) \text{ with } (104)$$

$$\leq \beta \ ,$$

with a last line simply following from the Bienayme-Chebyshev inequality. Assume now that δ^* is in $(-\lambda_0, 0)$. The assumption (103) implies

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{2}{\sqrt{L}} \max\left(2\sqrt{\frac{\lambda_0 + \delta^*}{\beta}}, \sqrt{\frac{|\delta^*|\log(1/\alpha)|}{\log(\lambda_0/(\lambda_0 - |\delta^*|/2))}}\right)$$

which yields

$$|\delta^*|\ell \ge 4 \max\left(\sqrt{\frac{(\lambda_0 + \delta^*)\ell}{\beta L}}, \frac{\log\left(1/\alpha\right)}{L \log\left(\lambda_0/\left(\lambda_0 - |\delta^*|/2\right)\right)}\right).$$

Hence

$$\frac{|\delta^*|}{2}\ell L \ge \sqrt{\frac{(\lambda_0 + \delta^*)\ell L}{\beta}} + \frac{\log\left(1/\alpha\right)}{\log\left(\lambda_0/\left(\lambda_0 - |\delta^*|/2\right)\right)} ,$$

and then

$$\left(\lambda_0 - \frac{|\delta^*|}{2}\right)\ell L - \frac{\log\left(1/\alpha\right)}{\log\left(\lambda_0/\left(\lambda_0 - |\delta^*|/2\right)\right)} - (\lambda_0 + \delta^*)\ell L \ge \sqrt{\frac{(\lambda_0 + \delta^*)\ell L}{\beta}} \quad . \tag{105}$$

We conclude with the following inequalities:

$$P_{\lambda}\left(\phi_{5,\alpha}(N)=0\right)$$

$$\leq P_{\lambda}\left(\sup_{\ell'\in(0,1-\tau^{*})}S_{\delta^{*},\tau^{*},\tau^{*}+\ell'}(N)\leq s_{\lambda_{0},\delta^{*},\tau^{*},L}^{+}(1-\alpha)\right)$$

$$\leq P_{\lambda}\left(\left(\lambda_{0}-\frac{|\delta^{*}|}{2}\right)\ell L-N(\tau^{*},\tau^{*}+\ell]\leq \frac{\log\left(1/\alpha\right)}{\log\left(\lambda_{0}/\left(\lambda_{0}-|\delta^{*}|/2\right)\right)}\right) \text{ with (286)}$$

$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell]\geq\left(\lambda_{0}-\frac{|\delta^{*}|}{2}\right)\ell L-\frac{\log\left(1/\alpha\right)}{\log\left(\lambda_{0}/\left(\lambda_{0}-|\delta^{*}|/2\right)\right)}\right)$$

$$\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell]-(\lambda_{0}+\delta^{*})\ell L\geq\sqrt{\frac{(\lambda_{0}+\delta^{*})\ell L}{\beta}}\right) \text{ with (105)}$$

 $\leq \beta$ with the Bienayme-Chebyshev inequality .

5.9 Proof of Lemma 11

Let $\lambda_0 > 0$, τ^* in (0,1), and ϕ_{α} a level- α test of (H_0) " $\lambda \in \mathcal{S}_0[\lambda_0] = {\lambda_0}$ " versus (H_1) " $\lambda \in \mathcal{S}_{,\tau^*,\dots}[\lambda_0]$ ", with

$$\mathcal{S}_{,\tau^*,\dots}[\lambda_0] = \left\{ \lambda \colon \exists \delta \in (-\lambda_0, +\infty) \setminus \{0\}, \exists \ell \in (0, 1 - \tau^*), \lambda(t) = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^* + \ell]}(t) \right\} ,$$

as defined by (22).

Let r > 0 and λ in $\mathcal{S}_{,\tau^*,\dots}[\lambda_0]$ satisfying $d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge r$. We compute

$$P_{\lambda} \left(\phi_{\alpha}(N) = 0 \right) = 1 - P_{\lambda} \left(\phi_{\alpha}(N) = 1 \right) + P_{\lambda_{0}} \left(\phi_{\alpha}(N) = 1 \right) - P_{\lambda_{0}} \left(\phi_{\alpha}(N) = 1 \right)$$

$$\geq 1 - \alpha - |P_{\lambda_{0}} \left(\phi_{\alpha}(N) = 1 \right) - P_{\lambda} \left(\phi_{\alpha}(N) = 1 \right) |$$

$$\geq 1 - \alpha - V \left(P_{\lambda}, P_{\lambda_{0}} \right) ,$$

where $V(P_{\lambda}, P_{\lambda_0})$ is the total variation distance between the probability measures P_{λ} and P_{λ_0} . Then, using the Pinsker inequality (see for example Lemma 2.5 in [104]),

$$P_{\lambda}(\phi_{\alpha}(N)=0) \ge 1 - \alpha - \sqrt{\frac{K(P_{\lambda}, P_{\lambda_0})}{2}} ,$$

where $K(P_{\lambda}, P_{\lambda_0})$ is the Kullback divergence between the probability measures P_{λ} and P_{λ_0} . We deduce from Lemma 42 that if there exists λ in $\mathcal{S}_{,\tau^*,\dots}[\lambda_0]$ such that $d_2(\lambda, \mathcal{S}_0[\lambda_0]) \geq r$ satisfying $1 - \alpha - \sqrt{K(P_{\lambda}, P_{\lambda_0})/2} \geq \beta$, then $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{,\tau^*,\dots}[\lambda_0]) \geq r$. Let us introduce for all ℓ in $(0, 1 - \tau^*)$, $\lambda_r = \lambda_0 + r\ell^{-1/2}\mathbb{1}_{(\tau^*,\tau^*+\ell]}$ in $\mathcal{S}_{,\tau^*,\dots}[\lambda_0]$ which satisfies $d_2(\lambda_r, \mathcal{S}_0[\lambda_0]) = r$. Then, Lemma 1 entails

$$K(P_{\lambda_r}, P_{\lambda_0}) = \int \log\left(\frac{dP_{\lambda_r}}{dP_{\lambda_0}}\right) dP_{\lambda_r} = \log\left(1 + \frac{r}{\lambda_0\sqrt{\ell}}\right) \left(\lambda_0 + \frac{r}{\sqrt{\ell}}\right) \ell L - Lr\sqrt{\ell} .$$

Hence choosing ℓ close enough to 0 – which is allowed as long as λ_r is not constrained to be upper bounded by some given constant, $K(P_{\lambda_r}, P_{\lambda_0}) \leq 2(1 - \alpha - \beta)^2$. This entails

$$\operatorname{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\cdot,\tau^*,\ldots}[\lambda_0]\right) \ge r$$

for every r > 0 and allows to conclude that $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{\cdot,\tau^*,\ldots}[\lambda_0]) = +\infty$.

5.10 Proof of Proposition 12

Assume that $L \geq 3$ and $\alpha + \beta < 1/2$. Let $C'_{\alpha,\beta} = 4(1-\alpha-\beta)^2$, $K_{\alpha,\beta,L} = \lceil (\log_2 L)/C'_{\alpha,\beta} \rceil$, and for k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, $\lambda_k = \lambda_0 + \delta_k \mathbb{1}_{(\tau^*,\tau^*+\ell_k]}$ with $\ell_k = (1-\tau^*)/2^k$ and $\delta_k = (\lambda_0 \log \log L/(\ell_k L))^{1/2}$. Then $d_2(\lambda_k, \mathcal{S}_0[\lambda_0]) = \sqrt{\lambda_0 \log \log L/L}$ for all k in $\{1, \ldots, K_{\alpha,\beta,L}\}$ and assuming that

$$\frac{\log \log L}{L^{1-1/C'_{\alpha,\beta}}} \le (R - \lambda_0)^2 \frac{1 - \tau^*}{2\lambda_0} \quad , \tag{106}$$

 λ_k belongs to $\mathcal{S}_{,\tau^*,\dots}[\lambda_0, R]$. Recall that for any k in $\{1, \dots, K_{\alpha,\beta,L}\}$ P_{λ_k} denotes the distribution of a Poisson process with intensity λ_k with respect to the measure Λ , and consider κ , a random variable with uniform distribution on $\{1, \dots, K_{\alpha,\beta,L}\}$, which allows to define the probability distribution μ of λ_{κ} . From Lemma 43, we know that it is enough to prove $E_{\lambda_0}[(dP_{\mu}/dP_{\lambda_0})^2] \leq 1 + C'_{\alpha,\beta}$ to conclude that $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{,\tau^*,\dots}[\lambda_0,R]) \geq \sqrt{\lambda_0 \log \log L/L}$.

By definition, $(dP_{\mu}/dP_{\lambda_0})(N) = E_{\kappa} [(dP_{\lambda_{\kappa}}/dP_{\lambda_0})(N)]$ (where E_{κ} denotes the expectation w.r.t. the uniform variable κ) and therefore

$$\frac{dP_{\mu}}{dP_{\lambda_0}}(N) = \frac{1}{K_{\alpha,\beta,L}} \sum_{k=1}^{K_{\alpha,\beta,L}} \exp\left(\log\left(1 + \frac{\delta_k}{\lambda_0}\right) N(\tau^*, \tau^* + \ell_k] - L\ell_k \delta_k\right)$$

Since $\ell_{k'} < \ell_k$ for all k' > k,

$$\left(\frac{dP_{\mu}}{dP_{\lambda_0}}(N)\right)^2 = \frac{1}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}} \exp\left(2\log\left(1+\frac{\delta_k}{\lambda_0}\right)N(\tau^*,\tau^*+\ell_k] - 2L\ell_k\delta_k\right) + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}-1} \sum_{k'=k+1}^{K_{\alpha,\beta,L}} \exp\left(\left(\log\left(1+\frac{\delta_k}{\lambda_0}\right) + \log\left(1+\frac{\delta_{k'}}{\lambda_0}\right)\right)N(\tau^*,\tau^*+\ell_{k'}] + \log\left(1+\frac{\delta_k}{\lambda_0}\right)N(\tau^*+\ell_{k'},\tau^*+\ell_k] - L\ell_k\delta_k - L\ell_{k'}\delta_{k'}\right) .$$

Recall that under (H_0) , N is a homogeneous Poisson process with intensity λ_0 with respect to the measure Λ , so

$$E_{\lambda_0} \left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}} \right)^2 \right] = \frac{1}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}} \exp\left(\frac{L\ell_k \delta_k^2}{\lambda_0} \right) + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}-1} \sum_{k'=k+1}^{K_{\alpha,\beta,L}} \exp\left(\frac{L\ell_{k'} \delta_k \delta_{k'}}{\lambda_0} \right) \quad .$$

Then

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] = \frac{\log L}{K_{\alpha,\beta,L}} + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}-1} \sum_{k'=k+1}^{K_{\alpha,\beta,L}} \exp\left(2^{\frac{k-k'}{2}} (\log\log L)\right) \quad , \tag{107}$$

which entails

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] = \frac{\log L}{K_{\alpha,\beta,L}} + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{l=1}^{K_{\alpha,\beta,L}-1} \left(K_{\alpha,\beta,L} - l\right) \exp\left(2^{-\frac{l}{2}}(\log\log L)\right)$$
$$\leq C'_{\alpha,\beta} \log 2 + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{l=1}^{K_{\alpha,\beta,L}-1} \left(K_{\alpha,\beta,L} - l\right) \exp\left(2^{-\frac{l}{2}}(\log\log L)\right) \quad .$$

Now taking η such that $0 < \eta < 1 - 1/\sqrt{2}$,

$$E_{\lambda_{0}}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_{0}}}\right)^{2}\right] \leq C_{\alpha,\beta}^{\prime}\log 2 + \frac{2}{K_{\alpha,\beta,L}^{2}}\sum_{l=1}^{K_{\alpha,\beta,L}-1} (K_{\alpha,\beta,L}-l)\exp\left(2^{-\frac{l}{2}}(\log\log L)\right)$$
$$= C_{\alpha,\beta}^{\prime}\log 2 + \frac{2}{K_{\alpha,\beta,L}^{2}}\sum_{l=1}^{\lfloor(\log L)^{\eta}\rfloor} (K_{\alpha,\beta,L}-l)\exp\left(2^{-\frac{l}{2}}(\log\log L)\right)$$
$$+ \frac{2}{K_{\alpha,\beta,L}^{2}}\sum_{l=\lfloor(\log L)^{\eta}\rfloor+1}^{K_{\alpha,\beta,L}-1} (K_{\alpha,\beta,L}-l)\exp\left(2^{-\frac{l}{2}}(\log\log L)\right)$$
$$\leq C_{\alpha,\beta}^{\prime}\log 2 + \frac{2C_{\alpha,\beta}^{\prime}\log 2}{\log L}(\log L)^{\eta+\frac{1}{\sqrt{2}}} + \exp\left(\frac{\log\log L}{2^{(\log L)^{\eta}/2}}\right) \quad . \tag{108}$$

If we assume now that

$$\exp\left(\frac{\log\log L}{2^{(\log L)^{\eta/2}}}\right) + \frac{2C'_{\alpha,\beta}\log 2}{\left(\log L\right)^{1-\eta-\frac{1}{\sqrt{2}}}} \le 1 + (1-\log 2)C'_{\alpha,\beta} \quad , \tag{109}$$

we finally obtain the expected result, i.e.

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] \le 1 + C'_{\alpha,\beta}$$

To end the proof, it remains to notice that there exists $L_0(\alpha, \beta, \lambda_0, R) \ge 3$ such that for all $L \ge L_0(\alpha, \beta, \lambda_0, R)$, both assumptions (106) and (109) hold.

5.11 Proof of Proposition 13

The control of the first kind error rates of the two tests $\phi_{6,\alpha}^{(1)}$ and $\phi_{6,\alpha}^{(2)}$ is straightforward using simple union bounds.

(i) Control of the second kind error rate of $\phi_{6,\alpha}^{(1)}$. Let λ in $\mathcal{S}_{,\tau^*,\dots}[\lambda_0, R]$ be such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^*+\ell]}$, with δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$, ℓ in $(0, 1 - \tau^*)$, and such that

$$d_2\left(\lambda, \mathcal{S}_0[\lambda_0]\right) \ge \sqrt{2} \max\left(2\sqrt{\frac{R\log\left(2/u_\alpha\right)}{3L}} , \ 2\sqrt{\frac{2\lambda_0\log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{R}{\beta L}} , \ \frac{R}{\sqrt{L}}\right).$$
(110)

Let us prove that $P_{\lambda}\left(\phi_{6,\alpha}^{(1)}(N)=0\right) \leq \beta$. Assume first that δ belongs to $(0, R-\lambda_0]$. Noticing that

$$P_{\lambda}\left(\phi_{6,\alpha}^{(1)}(N)=0\right) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(N(\tau^*,\tau^*+\ell_{\tau^*,k}] \leq p_{\lambda_0\ell_{\tau^*,k}L}\left(1-u_{\alpha}/2\right)\right) ,$$

one can see that it is enough to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(N(\tau^*,\tau^*+\ell_{\tau^*,k}] \le p_{\lambda_0\ell_{\tau^*,k}L}\left(1-u_{\alpha}/2\right)\right) \le \beta .$$

We get from (110) that $d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \geq 2R^2/L$ which entails $\ell \geq 2/L$, so $(1-\tau^*)2^{-\lfloor \log_2 L \rfloor} \leq 2(1-\tau^*)/L < 2/L \leq \ell$ and $\ell < (1-\tau^*)2^{-1+1}$. Therefore, one can find k_{τ^*} in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying $(1-\tau^*)2^{-k_{\tau^*}} \leq \ell < (1-\tau^*)2^{-k_{\tau^*}+1}$. Consider now $\ell_{\tau^*} = (1-\tau^*)2^{-k_{\tau^*}} = \ell_{\tau^*,k_{\tau^*}}$ such that $\ell/2 < \ell_{\tau^*} \leq \ell$. We get

$$\delta^2 \ell_{\tau^*} > d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0] \right) / 2 \quad . \tag{111}$$

Moreover, we deduce from (110) that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \sqrt{2} \max\left(2\sqrt{\frac{\delta \log\left(2/u_\alpha\right)}{3L}}, 2\sqrt{\frac{2\lambda_0 \log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{\lambda_0 + \delta}{\beta L}}\right) ,$$

and then with (111),

$$\delta\sqrt{\ell_{\tau^*}} \ge \max\left(2\sqrt{\frac{\delta\log\left(2/u_\alpha\right)}{3L}} , \ 2\sqrt{\frac{2\lambda_0\log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{\lambda_0+\delta}{\beta L}}\right) .$$

This entails in particular $\delta \ell_{\tau^*} \ge 4 \log \left(2/u_\alpha \right) / (3L)$ as well as

$$\delta \ell_{\tau^*} \ge 2\sqrt{\ell_{\tau^*}} \left(\sqrt{\frac{2\lambda_0 \log\left(2/u_\alpha\right)}{L}} + \sqrt{\frac{\lambda_0 + \delta}{\beta L}} \right)$$

Therefore,

$$\delta \ell_{\tau^*} \ge 2 \max\left(\frac{2\log\left(2/u_\alpha\right)}{3L} , \sqrt{\ell_{\tau^*}} \left(\sqrt{\frac{2\lambda_0 \log\left(2/u_\alpha\right)}{L}} + \sqrt{\frac{\lambda_0 + \delta}{\beta L}}\right)\right) ,$$

hence

$$\delta\ell_{\tau^*}L \ge \frac{2}{3}\log\left(2/u_\alpha\right) + \sqrt{2\lambda_0\ell_{\tau^*}L\log\left(2/u_\alpha\right)} + \sqrt{\frac{(\lambda_0+\delta)\ell_{\tau^*}L}{\beta}} \quad . \tag{112}$$

On the one hand, since $\ell_{\tau^*} \leq \ell$, Lemma 44 gives $E_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*}]] = \operatorname{Var}_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*}]] = (\lambda_0 + \delta)\ell_{\tau^*}L$. On the other hand, Lemma 46 gives

$$p_{\lambda_0 \ell_{\tau^*} L} \left(1 - u_{\alpha}/2 \right) \le \lambda_0 \ell_{\tau^*} L + \lambda_0 \ell_{\tau^*} L g^{-1} \left(\frac{\log \left(2/u_{\alpha} \right)}{\lambda_0 L \ell_{\tau^*}} \right) ,$$

with $g^{-1}(x) \leq 2x/3 + \sqrt{2x}$ for all x > 0 (see (80)), which leads to

$$p_{\lambda_0 \ell_{\tau^*} L} \left(1 - u_{\alpha}/2 \right) \le \lambda_0 \ell_{\tau^*} L + \frac{2}{3} \log \left(2/u_{\alpha} \right) + \sqrt{2\lambda_0 \ell_{\tau^*} L \log \left(2/u_{\alpha} \right)} \ .$$

Combined with (112), these computations yield

$$E_{\lambda}\left[N(\tau^*,\tau^*+\ell_{\tau^*}]\right] \ge p_{\lambda_0\ell_{\tau^*L}}\left(1-u_{\alpha}/2\right) + \sqrt{\operatorname{Var}_{\lambda}\left[N(\tau^*,\tau^*+\ell_{\tau^*}]\right]/\beta} \quad . \tag{113}$$

We conclude with (113) and the Bienayme-Chebyshev inequality successively:

$$P_{\lambda}\Big(N(\tau^*,\tau^*+\ell_{\tau^*}] \leq p_{\lambda_0\ell_{\tau^*}L}\left(1-u_{\alpha}/2\right)\Big)$$

$$\leq P_{\lambda}\left(N(\tau^*,\tau^*+\ell_{\tau^*}]-E_{\lambda}\left[N(\tau^*,\tau^*+\ell_{\tau^*}]\right] \leq -\sqrt{\operatorname{Var}_{\lambda}\left[N(\tau^*,\tau^*+\ell_{\tau^*}]\right]/\beta}\right)$$

$$\leq \beta \ .$$

Assume now that δ belongs to $(-\lambda_0, 0)$ and notice that

$$P_{\lambda}\left(\phi_{6,\alpha}^{(1)}(N) = 0\right) \le \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(N(\tau^*,\tau^* + \ell_{\tau^*,k}] \ge p_{\lambda_0 \ell_{\tau^*,k}L}(u_{\alpha}/2)\right)$$

Lemma 46 with (80) gives

$$p_{\lambda_0\ell_{\tau^*}L}(u_{\alpha}/2) \ge \lambda_0\ell_{\tau^*}L - \frac{2}{3}\log\left(2/u_{\alpha}\right) - \sqrt{2\lambda_0\ell_{\tau^*}L\log\left(2/u_{\alpha}\right)} \ .$$

The same choice of k_{τ^*} and $\ell_{\tau^*} = \ell_{\tau^*,k_{\tau^*}}$ as in the above case where $\delta \in (0, R - \lambda_0]$ entails

$$|\delta|\ell_{\tau^*}L \ge \lambda_0 \ell_{\tau^*}L - p_{\lambda_0 \ell_{\tau^*}L} \left(u_\alpha/2 \right) + \sqrt{\frac{(\lambda_0 + \delta)\ell_{\tau^*}L}{\beta}} , \qquad (114)$$

and since $E_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*}]] = \operatorname{Var}_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*}]] = (\lambda_0 + \delta)\ell_{\tau^*}L = \lambda_0\ell_{\tau^*}L - |\delta|\ell_{\tau^*}L$, we obtain in the same way

$$P_{\lambda}\Big(N(\tau^*,\tau^*+\ell_{\tau^*}] \ge p_{\lambda_0\ell_{\tau^*}L}(u_{\alpha}/2)\Big)$$

$$\le P_{\lambda}\left(N(\tau^*,\tau^*+\ell_{\tau^*}] - E_{\lambda}[N(\tau^*,\tau^*+\ell_{\tau^*}]] \ge \sqrt{\operatorname{Var}_{\lambda}[N(\tau^*,\tau^*+\ell_{\tau^*}]]/\beta}\Big)$$

$$\le \beta .$$

Finally (110) leads in both cases to $P_{\lambda}\left(\phi_{6,\alpha}^{(1)}(N)=0\right) \leq \beta$, which allows to conclude that

$$\operatorname{SR}_{\beta}\left(\phi_{6,\alpha}^{(1)}, \mathcal{S}_{,\tau^{*},\dots}[\lambda_{0}, R]\right) \leq \sqrt{2} \max\left(2\sqrt{\frac{R\log\left(2\lfloor \log_{2} L\rfloor/\alpha\right)}{3L}}, \frac{2\sqrt{\frac{2\lambda_{0}\log\left(2\lfloor \log_{2} L\rfloor/\alpha\right)}{L}} + 2\sqrt{\frac{R}{\beta L}}, \frac{R}{\sqrt{L}}\right).$$

(ii) Control of the second kind error rate of $\phi_{6,\alpha}^{(2)}$. Let λ in $\mathcal{S}_{,\tau^*},...[\lambda_0, R]$ be such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^*+\ell]}$, with δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}, \ell$ in $(0, 1 - \tau^*)$, and such that

$$d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right) \geq \max\left(4\sqrt{\frac{2\lambda_{0}\log\left(3/u_{\alpha}\right)}{L}} + 2\sqrt{2\sqrt{\frac{2}{\beta}\frac{R}{L}}}, 2\sqrt{\frac{2\sqrt{2}R\log\left(3/u_{\alpha}\right)}{3L}}, \frac{4\left(\frac{2}{3}\right)^{1/3}\lambda_{0}^{1/6}R^{1/3}\sqrt{\frac{\log\left(3/u_{\alpha}\right)}{L}}, 16\sqrt{\frac{R}{\beta L}}, \frac{\sqrt{2}R}{\sqrt{L}}\right).$$
(115)

Let us prove that $P_{\lambda}\left(\phi_{6,\alpha}^{(2)}(N)=0\right) \leq \beta$. Noticing that

$$P_{\lambda}\left(\phi_{6,\alpha}^{(2)}(N)=0\right) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(T_{\tau^*,\tau^*+\ell_{\tau^*,k}}(N) \leq t_{\lambda_0,\tau^*,\tau^*+\ell_{\tau^*,k}}\left(1-u_{\alpha}\right)\right) ,$$

one can see that it is enough to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(T_{\tau^*,\tau^*+\ell_{\tau^*,k}}(N) \le t_{\lambda_0,\tau^*,\tau^*+\ell_{\tau^*,k}}\left(1-u_{\alpha}\right)\right) \le \beta \quad .$$

From (115), we deduce that $d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \geq 2R^2/L$ which entails $\ell \geq 2/L$. Therefore, as in the above part (i) of the proof, let k_{τ^*} in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ be such that $(1 - \tau^*)2^{-k_{\tau^*}} \leq \ell < (1 - \tau^*)2^{-k_{\tau^*}+1}$, and consider $\ell_{\tau^*} = (1 - \tau^*)2^{-k_{\tau^*}} = \ell_{\tau^*,k_{\tau^*}}$. Then $\ell/2 < \ell_{\tau^*} \leq \ell$ and

$$\delta^2 \ell_{\tau^*} > d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0] \right) / 2 \quad . \tag{116}$$

Moreover, we get from (115)

$$d_2\left(\lambda, \mathcal{S}_0[\lambda_0]\right) \ge \max\left(4\sqrt{\frac{2\lambda_0 \log\left(3/u_\alpha\right)}{L}} + 2\sqrt{2\sqrt{\frac{2}{\beta}\frac{R}{L}}}, 2\sqrt{\frac{2\sqrt{2}R\log\left(3/u_\alpha\right)}{3L}}, \frac{4\left(\frac{2}{3}\right)^{1/3}\lambda_0^{1/6}R^{1/3}\sqrt{\frac{\log\left(3/u_\alpha\right)}{L}}, 16\sqrt{\frac{R}{\beta L}}\right).$$

On the one hand, this entails in particular $d_2^4(\lambda, \mathcal{S}_0[\lambda_0]) \ge 128R^2 \log^2(3/u_\alpha)/(9L^2)$, and then, with (116), $d_2^4(\lambda, \mathcal{S}_0[\lambda_0]) \ge 64d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \log^2(3/u_\alpha)/(9L^2\ell_{\tau^*})$. On the other hand, this yields

$$d_2^3(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{64}{3} d_2(\lambda, \mathcal{S}_0[\lambda_0]) \sqrt{\frac{2\lambda_0 \log^3\left(3/u_\alpha\right)}{L^3 \ell_{\tau^*}}} ,$$

using the same arguments. Therefore

$$d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \max\left(32\frac{\lambda_0 \log\left(3/u_\alpha\right)}{L} + 8\sqrt{\frac{2}{\beta}}\frac{R}{L}, \frac{64 \log^2\left(3/u_\alpha\right)}{9L^2\ell_{\tau^*}}, \frac{64}{3}\sqrt{\frac{2\lambda_0 \log^3\left(3/u_\alpha\right)}{L^3\ell_{\tau^*}}}, 16d_2(\lambda, \mathcal{S}_0[\lambda_0])\sqrt{\frac{R}{\beta L}}\right).$$

Hence

$$\frac{d_2^2\left(\lambda, \mathcal{S}_0[\lambda_0]\right)}{2} \ge \frac{4\lambda_0 \log\left(3/u_\alpha\right)}{L} + \sqrt{\frac{2}{\beta}} \frac{R}{L} + \frac{8\log^2\left(3/u_\alpha\right)}{9L^2\ell_{\tau^*}} + \frac{8}{3}\sqrt{\frac{2\lambda_0 \log^3\left(3/u_\alpha\right)}{L^3\ell_{\tau^*}}} + 2d_2(\lambda, \mathcal{S}_0[\lambda_0])\sqrt{\frac{R}{\beta L}} \quad (117)$$

Since $\ell_{\tau^*} \leq \ell$, Lemma 47 gives $E_{\lambda}[T_{\tau^*,\tau^*+\ell_{\tau^*}}(N)] = \delta^2 \ell_{\tau^*}$ and

$$\operatorname{Var}_{\lambda}\left[T_{\tau^*,\tau^*+\ell_{\tau^*}}(N)\right] = \frac{4\delta^2(\lambda_0+\delta)\ell_{\tau^*}}{L} + \frac{2(\lambda_0+\delta)^2}{L^2} \ .$$

This leads with (116) to

$$E_{\lambda}\left[T_{\tau^*,\tau^*+\ell_{\tau^*}}(N)\right] \ge \frac{d_2^2\left(\lambda,\mathcal{S}_0[\lambda_0]\right)}{2} \quad , \tag{118}$$

and

$$\operatorname{Var}_{\lambda}\left[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*}}}(N)\right] \leq \frac{4d_{2}^{2}\left(\lambda,\mathcal{S}_{0}[\lambda_{0}]\right)R}{L} + \frac{2R^{2}}{L^{2}} \quad (119)$$

With (118) and (119), the inequality (117) yields

$$E_{\lambda}\left[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*}}}(N)\right] \geq \frac{4\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{8\log^{2}\left(3/u_{\alpha}\right)}{9L^{2}\ell_{\tau^{*}}} + \frac{8}{3}\sqrt{\frac{2\lambda_{0}\log^{3}\left(3/u_{\alpha}\right)}{L^{3}\ell_{\tau^{*}}}} + \sqrt{\frac{\operatorname{Var}_{\lambda}\left[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*}}}(N)\right]}{\beta}} \cdot (120)$$

Moreover, Lemma 48 gives

$$t_{\lambda_0,\tau^*,\tau^*+\ell_{\tau^*}} \left(1 - u_{\alpha}\right) \le 2\lambda_0^2 \ell_{\tau^*} \left(g^{-1} \left(\frac{\log(3/u_{\alpha})}{\lambda_0 \ell_{\tau^*} L}\right)\right)^2 ,$$

where $g^{-1}(x) \le 2x/3 + \sqrt{2x}$ for all x > 0 (see (80)), and then

$$t_{\lambda_{0},\tau^{*},\tau^{*}+\ell_{\tau^{*}}}(1-u_{\alpha}) \leq \frac{4\lambda_{0}\log(3/u_{\alpha})}{L} + \frac{8\log^{2}(3/u_{\alpha})}{9L^{2}\ell_{\tau^{*}}} + \frac{8}{3}\sqrt{\frac{2\lambda_{0}\log^{3}(3/u_{\alpha})}{L^{3}\ell_{\tau^{*}}}} \\ \leq E_{\lambda}[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*}}}(N)] - \sqrt{\frac{\operatorname{Var}_{\lambda}[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*}}}(N)]}{\beta}}.$$

We obtain with the Bienayme-Chebyshev inequality

$$P_{\lambda}\Big(T_{\tau^*,\tau^*+\ell_{\tau^*}}(N) \leq t_{\lambda_0,\tau^*,\tau^*+\ell_{\tau^*}}(1-u_{\alpha})\Big)$$
$$\leq P_{\lambda}\left(T_{\tau^*,\tau^*+\ell_{\tau^*}}(N) \leq E_{\lambda}\left[T_{\tau^*,\tau^*+\ell_{\tau^*}}(N)\right] - \sqrt{\frac{\operatorname{Var}_{\lambda}\left[T_{\tau^*,\tau^*+\ell_{\tau^*}}(N)\right]}{\beta}}\right)$$
$$\leq \beta ,$$

which entails $P_{\lambda}\left(\phi_{6,\alpha}^{(2)}(N)=0\right) \leq \beta$. This finally allows to conclude that

$$\operatorname{SR}_{\beta}\left(\phi_{6,\alpha}^{(2)}, \mathcal{S}_{,\tau^{*},\dots}[\lambda_{0},R]\right) \leq \max\left(4\sqrt{\frac{2\lambda_{0}\log\left(3\lfloor\log_{2}L\rfloor/\alpha\right)}{L}} + 2\sqrt{2\sqrt{\frac{2}{\beta}\frac{R}{L}}}, \frac{2\sqrt{2\sqrt{2R}\log\left(3\lfloor\log_{2}L\rfloor/\alpha\right)}}{3L}, 4\left(\frac{2}{3}\right)^{1/3}\lambda_{0}^{1/6}R^{1/3}\sqrt{\frac{\log\left(3\lfloor\log_{2}L\rfloor/\alpha\right)}{L}}, 16\sqrt{\frac{R}{\beta L}}, \frac{\sqrt{2R}}{\sqrt{L}}\right)$$

5.12 Proof of Proposition 14

Let $C_{\alpha,\beta} = 1 + 4(1 - \alpha - \beta)^2$, $r = (\lambda_0 \log C_{\alpha,\beta}/L)^{1/2}$ and λ_r defined for all t in (0,1) by

$$\lambda_r(t) = \lambda_0 + \delta^* \mathbb{1}_{(1-r^2/\delta^{*2},1]}(t)$$

Notice that for all $L \ge \lambda_0 \log C_{\alpha,\beta}/\delta^{*2}$, we have $r \le |\delta^*|$ and λ_r belongs to $(\mathcal{S}_{\delta^*,..,1-..}[\lambda_0])_r$ in the notation of Lemma 42. We get now from Lemma 1 and Lemma 44

$$E_{\lambda_0}\left[\left(\frac{dP_{\lambda_r}}{dP_0}\right)^2(N)\right] = \exp\left(\frac{r^2L}{\lambda_0}\right) = C_{\alpha,\beta}.$$

Lemmas 43 and 42 then entail $\rho_{\alpha}\left(\left(\mathcal{S}_{\delta^{*},..,1-..}[\lambda_{0}]\right)_{r}\right) \geq \beta$ and $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}_{\delta^{*},..,1-..}[\lambda_{0}]\right) \geq r$.

5.13 Proof of Proposition 15

The first kind error rate control is straightforward. As for the second kind error rate control, let $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau,1]}$ belonging to $\mathcal{S}_{\delta^*,\dots,1-\dots}[\lambda_0]$ with τ in (0,1) and satisfying

$$d_{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) \geq \frac{2}{\sqrt{L}} \max\left(2\sqrt{\frac{\lambda_{0}+\delta^{*}}{\beta}}, \sqrt{\delta^{*}s^{+}_{\lambda_{0},\frac{\delta^{*}}{2}}(1-\alpha)}\mathbb{1}_{\{\delta^{*}>0\}} + \sqrt{\frac{|\delta^{*}|\log\left(1/\alpha\right)}{\log\left(\lambda_{0}/\left(\lambda_{0}-|\delta^{*}|/2\right)\right)}}\mathbb{1}_{\{-\lambda_{0}<\delta^{*}<0\}}\right).$$
(121)

Then the proof essentially follows the same line as the one of Proposition 10 just replacing ℓ by $(1 - \tau)$.

Assume that $\delta^* > 0$ and recall that $s^+_{\lambda_0, \delta^*/2} (1 - \alpha)$ defined in Lemma 50 is a constant which does not depend on L. The assumption (121) implies that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{2}{\sqrt{L}} \max\left(2\sqrt{\frac{\lambda_0 + \delta^*}{\beta}}, \sqrt{\delta^* s^+_{\lambda_0, \frac{\delta^*}{2}}(1-\alpha)}\right) ,$$

which entails

$$\frac{\delta^*}{2}(1-\tau)L \ge \sqrt{\frac{(\lambda_0 + \delta^*)(1-\tau)L}{\beta}} + s^+_{\lambda_0,\frac{\delta^*}{2}}(1-\alpha) \quad .$$
(122)

Then we get from the quantile upper bound (287)

$$P_{\lambda}\left(\phi_{7,\alpha}(N)=0\right) \leq P_{\lambda}\left(\sup_{\tau'\in(0,1)} S_{\delta^{*},\tau',1}(N) \leq s_{\lambda_{0},\frac{\delta^{*}}{2}}^{+}\left(1-\alpha\right)\right)$$

$$\leq P_{\lambda}\left(N(\tau,1]-\left(\lambda_{0}+\frac{\delta^{*}}{2}\right)(1-\tau)L \leq s_{\lambda_{0},\frac{\delta^{*}}{2}}^{+}\left(1-\alpha\right)-\frac{\delta^{*}}{2}(1-\tau)L\right)$$

$$\leq P_{\lambda}\left(N(\tau,1]-(\lambda_{0}+\delta^{*})(1-\tau)L \leq -\sqrt{\frac{(\lambda_{0}+\delta^{*})(1-\tau)L}{\beta}}\right) \text{ with (122)}$$

$$\leq \beta .$$

Assume now that δ^* is in $(-\lambda_0, 0)$. The assumption (121) implies that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{2}{\sqrt{L}} \max\left(2\sqrt{\frac{\lambda_0 + \delta^*}{\beta}}, \sqrt{\frac{|\delta^*|\log(1/\alpha)|}{\log(\lambda_0/(\lambda_0 - |\delta^*|/2))}}\right),$$

which entails

$$\frac{|\delta^*|}{2}(1-\tau)L \ge \sqrt{\frac{(\lambda_0+\delta^*)(1-\tau)L}{\beta}} + \frac{\log(1/\alpha)}{\log(\lambda_0/(\lambda_0-|\delta^*|/2))} ,$$

and then

$$\left(\lambda_0 - \frac{|\delta^*|}{2}\right)(1-\tau)L - \frac{\log\left(1/\alpha\right)}{\log\left(\lambda_0/\left(\lambda_0 - |\delta^*|/2\right)\right)} - (\lambda_0 + \delta^*)(1-\tau)L \ge \sqrt{\frac{(\lambda_0 + \delta^*)(1-\tau)L}{\beta}}$$
(123)

We conclude with the following inequalities, deduced from (287), (123) and the Bienayme-Chebyshev inequality successively:

$$P_{\lambda}\left(\phi_{7,\alpha}(N)=0\right) \leq P_{\lambda}\left(\sup_{\tau'\in(0,1)} S_{\delta^{*},\tau',1}(N) \leq s^{+}_{\lambda_{0},\delta^{*},\tau^{*},L}(1-\alpha)\right)$$
$$\leq P_{\lambda}\left(\left(\lambda_{0}-\frac{|\delta^{*}|}{2}\right)(1-\tau)L - N(\tau,1] \leq \frac{\log\left(1/\alpha\right)}{\log\left(\lambda_{0}/\left(\lambda_{0}-|\delta^{*}|/2\right)\right)}\right)$$
$$\leq P_{\lambda}\left(N(\tau,1]-(\lambda_{0}+\delta^{*})(1-\tau)L \geq \sqrt{\frac{(\lambda_{0}+\delta^{*})(1-\tau)L}{\beta}}\right)$$
$$\leq \beta .$$

Coming back to the formulation of assumption (121), we therefore can take in the statement of Proposition 15

$$C(\alpha, \beta, \lambda_0, \delta^*) = 2 \max \left(2\sqrt{\frac{\lambda_0 + \delta^*}{\beta}}, \sqrt{\delta^* s^+_{\lambda_0, \frac{\delta^*}{2}} (1 - \alpha)} \mathbb{1}_{\{\delta^* > 0\}} + \sqrt{\frac{|\delta^*| \log(1/\alpha)}{\log(\lambda_0/(\lambda_0 - |\delta^*|/2))}} \mathbb{1}_{\{-\lambda_0 < \delta^* < 0\}} \right).$$

5.14 Proof of Lemma 16

Let $\lambda_0 > 0$ and ϕ_{α} a level- α test of the null hypothesis (H_0) " $\lambda \in S_0[\lambda_0] = \{\lambda_0\}$ " versus the alternative (H_1) " $\lambda \in S_{\dots,1-\dots}[\lambda_0]$ ", with $S_{\dots,1-\dots}[\lambda_0]$ defined by (28). Let us fix some r > 0. As in the proof of Lemma 11, we can argue that if there exists λ in $S_{\dots,1-\dots}[\lambda_0]$ such that $d_2(\lambda, S_0[\lambda_0]) \geq r$ satisfying $1 - \alpha - \sqrt{K(P_\lambda, P_{\lambda_0})/2} \geq \beta$, then $\mathrm{mSR}_{\alpha,\beta}(S_{\dots,1-\dots}[\lambda_0]) \geq r$.

Let us introduce for all τ in (0, 1), $\lambda_r = \lambda_0 + r(1-\tau)^{-1/2} \mathbb{1}_{(\tau,1]}$ in $\mathcal{S}_{,\dots,1-\dots}[\lambda_0]$ which satisfies $d_2(\lambda_r, \mathcal{S}_0[\lambda_0]) = r$. The end of the proof follows the same line as the one of Lemma 11, noticing that

$$K(P_{\lambda_r}, P_{\lambda_0}) = \log\left(1 + \frac{r}{\lambda_0\sqrt{1-\tau}}\right)\left(\lambda_0 + \frac{r}{\sqrt{1-\tau}}\right)(1-\tau)L - Lr\sqrt{1-\tau} ,$$

and choosing τ close enough to 1 in order to get $K(P_{\lambda_r}, P_{\lambda_0}) \leq 2(1 - \alpha - \beta)^2$, which yields $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{\cdot,\cdot,1-\cdot\cdot}[\lambda_0]) \geq r$ for all r > 0 and allows to conclude.

5.15 Proof of Proposition 17

Assume that $L \geq 3$ and $\alpha + \beta < 1/2$. As in the proof of Proposition 12, we consider $C'_{\alpha,\beta} = 4(1 - \alpha - \beta)^2$, $K_{\alpha,\beta,L} = \lceil (\log_2 L)/C'_{\alpha,\beta} \rceil$ and for k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, $\lambda_k = \lambda_0 + \delta_k \mathbb{1}_{(\tau_k,1]}$ with $\tau_k = 1 - 2^{-k}$ and $\delta_k = (2^k \lambda_0 \log \log L/L)^{1/2}$. Then, for every k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, $d_2(\lambda_k, \mathcal{S}_0[\lambda_0]) = \sqrt{\lambda_0 \log \log L/L}$, and assuming that

$$\frac{\log \log L}{L^{1-1/C'_{\alpha,\beta}}} \le \frac{(R-\lambda_0)^2}{2\lambda_0} \quad , \tag{124}$$

 λ_k belongs to $\mathcal{S}_{\cdot,\cdot,1-\cdots}[\lambda_0, R]$. The proof then essentially follows the same arguments as the proof of Proposition 12. Thus, considering a random variable κ with uniform distribution on $\{1, \ldots, K_{\alpha,\beta,L}\}$ and the probability distribution μ of λ_{κ} , we aim at proving that $E_{\lambda_0}[(dP_{\mu}/dP_{\lambda_0})^2] \leq 1 + C'_{\alpha,\beta}$, with P_{μ} defined as in Lemma 43, in order to conclude that $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}_{\cdot,\cdots,1-\cdots}[\lambda_0, R]) \geq \sqrt{\lambda_0 \log \log L/L}$. By definition,

$$\frac{dP_{\mu}}{dP_{\lambda_0}}(N) = \frac{1}{K_{\alpha,\beta,L}} \sum_{k=1}^{K_{\alpha,\beta,L}} \exp\left(\log\left(1 + \frac{\delta_k}{\lambda_0}\right) N(\tau_k, 1] - L(1 - \tau_k)\delta_k\right) \quad .$$

Since $\tau_{k'} > \tau_k$ for all k' > k,

$$\left(\frac{dP_{\mu}}{dP_{\lambda_0}}(N)\right)^2 = \frac{1}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}} \exp\left(2\log\left(1+\frac{\delta_k}{\lambda_0}\right)N(\tau_k,1] - 2L(1-\tau_k)\delta_k\right) + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}-1} \sum_{k'=k+1}^{K_{\alpha,\beta,L}} \exp\left(\left(\log\left(1+\frac{\delta_k}{\lambda_0}\right) + \log\left(1+\frac{\delta_{k'}}{\lambda_0}\right)\right)N(\tau_{k'},1] + \log\left(1+\frac{\delta_k}{\lambda_0}\right)N(\tau_k,\tau_{k'}] - L(1-\tau_k)\delta_k - L(1-\tau_{k'})\delta_{k'}\right) ,$$

hence

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] = \frac{1}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}} \exp\left(\frac{L(1-\tau_k)\delta_k^2}{\lambda_0}\right) + \frac{2}{K_{\alpha,\beta,L}^2} \sum_{k=1}^{K_{\alpha,\beta,L}-1} \sum_{k'=k+1}^{K_{\alpha,\beta,L}} \exp\left(\frac{L(1-\tau_{k'})\delta_k\delta_{k'}}{\lambda_0}\right)$$

We then obtain the same expression of $E_{\lambda_0} \left[(dP_{\mu}/dP_{\lambda_0})^2 \right]$ as in Equation (107), and the proof ends exactly as the one of Proposition 12, just replacing (106) by (124) in the final argument.

5.16 Proof of Proposition 18

This proof is very similar to the one of Proposition 13. For the sake of completeness, we nevertheless detail it below.

The control of the first kind error rates of the two tests $\phi_{8,\alpha}^{(1)}$ and $\phi_{8,\alpha}^{(2)}$ is straightforward using simple union bounds.

(i) Control of the second kind error rate of $\phi_{8.\alpha}^{(1)}$.

Let λ in $\mathcal{S}_{,..,1-..}[\lambda_0, R]$ be such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,1]}$, with τ in (0,1), δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$, and such that

$$d_2\left(\lambda, \mathcal{S}_0[\lambda_0]\right) \ge \sqrt{2} \max\left(2\sqrt{\frac{R\log\left(2/u_\alpha\right)}{3L}} , \ 2\sqrt{\frac{2\lambda_0\log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{R}{\beta L}} , \ \frac{R}{\sqrt{L}}\right),\tag{125}$$

as in (110).

We prove here that $P_{\lambda}\left(\phi_{8,\alpha}^{(1)}(N)=0\right) \leq \beta$, assuming first that δ belongs to $(0, R-\lambda_0]$. Noticing that

$$P_{\lambda}\left(\phi_{8,\alpha}^{(1)}(N)=0\right) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(N(\tau_k,1] \leq p_{\lambda_0(1-\tau_k)L}\left(1-u_{\alpha}/2\right)\right) ,$$

one can see that it is enough to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(N(\tau_k, 1] \le p_{\lambda_0(1-\tau_k)L}\left(1-u_{\alpha}/2\right)\right) \le \beta .$$

Let $k_{\tau} = \lfloor -\log_2(1-\tau) \rfloor + 1$. Since $0 < 1-\tau < 1$, $k_{\tau} \ge 1$. Moreover, from (125), we obtain $d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \ge 2R^2/L$ which entails $(1-\tau) \ge 2/L$ and $k_{\tau} \le \lfloor \log_2(L/2) \rfloor + 1 \le \lfloor \log_2 L \rfloor$. Consider now $\tau_{k_{\tau}} = 1 - 2^{-k_{\tau}}$, which satisfies $(1-\tau)/2 \le 1 - \tau_{k_{\tau}} < 1 - \tau$ as well as

$$\delta^2(1-\tau_{k_\tau}) \ge d_2^2\left(\lambda, \mathcal{S}_0[\lambda_0]\right)/2 \quad . \tag{126}$$

We get from (125)

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \sqrt{2} \max\left(2\sqrt{\frac{\delta \log\left(2/u_\alpha\right)}{3L}}, \ 2\sqrt{\frac{2\lambda_0 \log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{\lambda_0 + \delta}{\beta L}}\right) ,$$

which gives with (126)

$$\delta\sqrt{1-\tau_{k_{\tau}}} \ge \max\left(2\sqrt{\frac{\delta\log\left(2/u_{\alpha}\right)}{3L}}, \ 2\sqrt{\frac{2\lambda_{0}\log\left(2/u_{\alpha}\right)}{L}} + 2\sqrt{\frac{\lambda_{0}+\delta}{\beta L}}\right) .$$

This entails in particular $\delta(1 - \tau_{k_{\tau}}) \ge (4/3) \log(2/u_{\alpha})/L$ and also

$$\delta\left(1-\tau_{k_{\tau}}\right) \geq 2\sqrt{1-\tau_{k_{\tau}}} \left(\sqrt{\frac{2\lambda_0 \log\left(2/u_{\alpha}\right)}{L}} + \sqrt{\frac{\lambda_0+\delta}{\beta L}}\right)$$

Hence,

$$\delta(1 - \tau_{k_{\tau}}) L \ge \frac{2}{3} \log(2/u_{\alpha}) + \sqrt{2\lambda_0 (1 - \tau_{k_{\tau}}) L \log(2/u_{\alpha})} + \sqrt{\frac{(\lambda_0 + \delta) (1 - \tau_{k_{\tau}}) L}{\beta}} .$$
(127)

On the one hand, since $\tau_{k_{\tau}} > \tau$, Lemma 44 gives $E_{\lambda}[N(\tau_{k_{\tau}}, 1]] = \operatorname{Var}_{\lambda}[N(\tau_{k_{\tau}}, 1]] = (\lambda_0 + \delta)(1 - \tau_{k_{\tau}})L$. On the other hand, Lemma 46 with the inequality (80) give

$$p_{\lambda_0(1-\tau_{k_\tau})L}\left(1-u_{\alpha}/2\right) \le \lambda_0(1-\tau_{k_\tau})L + \frac{2}{3}\log\left(2/u_{\alpha}\right) + \sqrt{2\lambda_0\left(1-\tau_{k_\tau}\right)L\log\left(2/u_{\alpha}\right)} \ .$$

Combined with (127), these computations yield

$$E_{\lambda}\left[N(\tau_{k_{\tau}},1]\right] \ge p_{\lambda_0(1-\tau_{k_{\tau}})L}\left(1-u_{\alpha}/2\right) + \sqrt{\operatorname{Var}_{\lambda}\left[N(\tau_{k_{\tau}},1]\right]/\beta} \quad .$$
(128)

The Bienayme-Chebyshev then leads to

$$P_{\lambda}\Big(N(\tau_{k_{\tau}},1] \leq p_{\lambda_0(1-\tau_{k_{\tau}})L}(1-u_{\alpha}/2)\Big) \leq \beta .$$

Assume now that δ belongs to $(-\lambda_0, 0)$ and notice that

$$P_{\lambda}\left(\phi_{8,\alpha}^{(1)}(N)=0\right) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(N(\tau_k,1] \geq p_{\lambda_0(1-\tau_k)L}\left(u_{\alpha}/2\right)\right)$$

Lemma 46 with (80) gives

$$p_{\lambda_0(1-\tau_k)L}(u_{\alpha}/2) \ge \lambda_0(1-\tau_k)L - \frac{2}{3}\log(2/u_{\alpha}) - \sqrt{2\lambda_0(1-\tau_k)L\log(2/u_{\alpha})} .$$

The same choice of k_{τ} as in the above case where $\delta \in (0, R - \lambda_0]$ thus entails

$$|\delta| (1 - \tau_{k_{\tau}}) L \ge \lambda_0 (1 - \tau_{k_{\tau}}) L - p_{\lambda_0 (1 - \tau_{k_{\tau}}) L} (u_{\alpha}/2) + \sqrt{\frac{(\lambda_0 + \delta) (1 - \tau_{k_{\tau}}) L}{\beta}} , \quad (129)$$

and since $E_{\lambda}[N(\tau_{k_{\tau}}, 1]] = \operatorname{Var}_{\lambda}[N(\tau_{k_{\tau}}, 1]] = (\lambda_0 + \delta)(1 - \tau_{k_{\tau}})L = \lambda_0(1 - \tau_{k_{\tau}})L - |\delta|(1 - \tau_{k_{\tau}})L$, we obtain in the same way

$$P_{\lambda}\Big(N(\tau_{k_{\tau}},1] \ge p_{\lambda_0(1-\tau_{k_{\tau}})L}(u_{\alpha}/2)\Big) \le P_{\lambda}\left(N(\tau_{k_{\tau}},1] - E_{\lambda}\left[N(\tau_{k_{\tau}},1]\right] \ge \sqrt{\operatorname{Var}_{\lambda}\left[N(\tau_{k_{\tau}},1]\right]/\beta}\right) \le \beta .$$

Finally, (125) leads in both cases to $P_{\lambda}\left(\phi_{8,\alpha}^{(1)}(N)=0\right) \leq \beta$, which allows to conclude that

$$\operatorname{SR}_{\beta}\left(\phi_{8,\alpha}^{(1)}, \mathcal{S}_{\cdot,\cdot,1-\cdot\cdot}[\lambda_{0}, R]\right) \leq \sqrt{2} \max\left(2\sqrt{\frac{R\log\left(2\lfloor \log_{2} L\rfloor/\alpha\right)}{3L}}, \frac{2\sqrt{\frac{2\lambda_{0}\log\left(2\lfloor \log_{2} L\rfloor/\alpha\right)}{L}} + 2\sqrt{\frac{R}{\beta L}}, \frac{R}{\sqrt{L}}\right).$$

(ii) Control of the second kind error rate of $\phi_{8,\alpha}^{(2)}$. Let λ in $\mathcal{S}_{\dots,1-\dots}[\lambda_0, R]$ be such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,1]}$, with τ in (0,1), δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$, and such that

$$d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right) \geq \max\left(4\sqrt{\frac{2\lambda_{0}\log\left(3/u_{\alpha}\right)}{L}} + 2\sqrt{2\sqrt{\frac{2}{\beta}\frac{R}{L}}}, 2\sqrt{\frac{2\sqrt{2}R\log\left(3/u_{\alpha}\right)}{3L}}, 4\left(\frac{2}{3}\right)^{1/3}\lambda_{0}^{1/6}R^{1/3}\sqrt{\frac{\log\left(3/u_{\alpha}\right)}{L}}, 16\sqrt{\frac{R}{\beta L}}, \frac{\sqrt{2}R}{\sqrt{L}}\right), \quad (130)$$

as in (115).

Let us prove that this implies that $P_{\lambda}\left(\phi_{8,\alpha}^{(2)}(N)=0\right) \leq \beta$. Notice that

$$P_{\lambda}\left(\phi_{8,\alpha}^{(2)}(N)=0\right) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(T_{\tau_k,1}(N) \leq t_{\lambda_0,\tau_k,1}\left(1-u_\alpha\right)\right) ,$$

to see that one only needs to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(T_{\tau_{k},1}(N) \leq t_{\lambda_{0},\tau_{k},1}\left(1-u_{\alpha}\right)\right) \leq \beta ,$$

to obtain the expected result.

As in the above part (i) of the proof, let $k_{\tau} = \lfloor -\log_2(1-\tau) \rfloor + 1$ and $\tau_{k_{\tau}} = 1 - 2^{-k_{\tau}}$. From (130) which in particular entails $d_2^2(\lambda, S_0[\lambda_0]) \ge 2R^2/L$, we get that k_{τ} actually belongs to $\{1, \ldots, \lfloor \log_2 L \rfloor\}$. Furthermore, by definition,

$$\delta^2 \left(1 - \tau_{k_\tau} \right) \ge d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0] \right) / 2 \quad . \tag{131}$$

Now, we also deduce from (130) that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \max\left(4\sqrt{\frac{2\lambda_0 \log\left(3/u_\alpha\right)}{L}} + 2\sqrt{2\sqrt{\frac{2}{\beta}\frac{R}{L}}}, \sqrt{\frac{\sqrt{2R}\log\left(3/u_\alpha\right)}{3L}}, \frac{4\left(\frac{2}{3}\right)^{1/3}\lambda_0^{1/6}R^{1/3}\sqrt{\frac{\log\left(3/u_\alpha\right)}{L}}, 16\sqrt{\frac{R}{\beta L}}\right).$$

This entails on the one hand that $d_2^4(\lambda, \mathcal{S}_0[\lambda_0]) \geq 128R^2 \log^2(3/u_\alpha)/(9L^2)$, and with (131), $d_2^4(\lambda, \mathcal{S}_0[\lambda_0]) \geq 64d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \log^2(3/u_\alpha)/(9L^2(1-\tau_{k_\tau}))$. On the other hand, we deduce that

$$d_2^3(\lambda, \mathcal{S}_0[\lambda_0]) \ge \frac{64}{3} d_2(\lambda, \mathcal{S}_0[\lambda_0]) \sqrt{\frac{2\lambda_0 \log^3 (3/u_\alpha)}{L^3 (1 - \tau_{k_\tau})}}$$

Therefore

$$d_2^2\left(\lambda, \mathcal{S}_0[\lambda_0]\right) \ge \max\left(32\frac{\lambda_0 \log\left(3/u_\alpha\right)}{L} + 8\sqrt{\frac{2}{\beta}}\frac{R}{L}, \frac{64 \log^2\left(3/u_\alpha\right)}{9L^2\left(1-\tau_{k_\tau}\right)}, \frac{64}{3}\sqrt{\frac{2\lambda_0 \log^3\left(3/u_\alpha\right)}{L^3\left(1-\tau_{k_\tau}\right)}}, \frac{64}{3}\sqrt{\frac{2\lambda_0 \log^3\left(3/u_\alpha\right)}{L^3\left(1-\tau_{k_\tau}\right)}}}, \frac{64}{3}\sqrt{\frac{2\lambda_0 \log^3\left(3/u_\alpha\right)}}{L^3\left(1-\tau_{k_\tau}\right)}}, \frac{64}{3}\sqrt{\frac{2\lambda_0 \log^3\left(3/u$$

Hence,

$$\frac{d_2^2(\lambda, \mathcal{S}_0[\lambda_0])}{2} \ge \frac{4\lambda_0 \log(3/u_\alpha)}{L} + \sqrt{\frac{2}{\beta}} \frac{R}{L} + \frac{8\log^2(3/u_\alpha)}{9L^2(1-\tau_{k_\tau})} + \frac{8}{3} \sqrt{\frac{2\lambda_0 \log^3(3/u_\alpha)}{L^3(1-\tau_{k_\tau})}} + 2d_2(\lambda, \mathcal{S}_0[\lambda_0]) \sqrt{\frac{R}{\beta L}} \quad (132)$$

Since $\tau_{k_{\tau}} > \tau$, Lemma 47 gives that $E_{\lambda} \left[T_{\tau_{k_{\tau}},1}(N) \right] = \delta^2 \left(1 - \tau_{k_{\tau}} \right)$ and

$$\operatorname{Var}_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right] = \frac{4\delta^{2}(\lambda_{0}+\delta)\left(1-\tau_{k_{\tau}}\right)}{L} + \frac{2(\lambda_{0}+\delta)^{2}}{L^{2}}$$

From (131), we get

$$E_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right] \ge \frac{d_2^2\left(\lambda, \mathcal{S}_0[\lambda_0]\right)}{2} \quad . \tag{133}$$

Moreover,

$$\operatorname{Var}_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right] \leq \frac{4d_{2}^{2}\left(\lambda,\mathcal{S}_{0}[\lambda_{0}]\right)R}{L} + \frac{2R^{2}}{L^{2}} \quad (134)$$

With (133) and (134), the inequality (132) yields

$$E_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right] \geq \frac{4\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{8\log^{2}\left(3/u_{\alpha}\right)}{9L^{2}\left(1-\tau_{k_{\tau}}\right)} + \frac{8}{3}\sqrt{\frac{2\lambda_{0}\log^{3}\left(3/u_{\alpha}\right)}{L^{3}\left(1-\tau_{k_{\tau}}\right)}} + \sqrt{\frac{\operatorname{Var}_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right]}{\beta}}$$

Furthermore, Lemma 48 gives

$$t_{\lambda_{0},\tau_{k_{\tau}},1}(1-u_{\alpha}) \leq \frac{4\lambda_{0}\log(3/u_{\alpha})}{L} + \frac{8\log^{2}(3/u_{\alpha})}{9L^{2}(1-\tau_{k_{\tau}})} + \frac{8}{3}\sqrt{\frac{2\lambda_{0}\log^{3}(3/u_{\alpha})}{L^{3}(1-\tau_{k_{\tau}})}} \\ \leq E_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right] - \sqrt{\frac{\operatorname{Var}_{\lambda}\left[T_{\tau_{k_{\tau}},1}(N)\right]}{\beta}},$$

and the Bienayme-Chebyshev leads to

$$P_{\lambda}\Big(T_{\tau_{k_{\tau}},1}(N) \le t_{\lambda_0,\tau_{k_{\tau}},1}(1-u_{\alpha})\Big) \le \beta .$$

This entails the expected result $P_{\lambda}\left(\phi_{8,\alpha}^{(2)}(N)=0\right) \leq \beta$, and finally allows to conclude that

$$\operatorname{SR}_{\beta}\left(\phi_{8,\alpha}^{(2)}, \mathcal{S}_{\cdot,\cdot,1-\cdot\cdot}[\lambda_{0}, R]\right) \leq \max\left(4\sqrt{\frac{2\lambda_{0}\log\left(3\lfloor\log_{2}L\rfloor/\alpha\right)}{L}} + 2\sqrt{2\sqrt{\frac{2}{\beta}\frac{R}{L}}}, \frac{2\sqrt{2\sqrt{2}R\log\left(3\lfloor\log_{2}L\rfloor/\alpha\right)}}{3L}, 4\left(\frac{2}{3}\right)^{1/3}\lambda_{0}^{1/6}R^{1/3}\sqrt{\frac{\log\left(3\lfloor\log_{2}L\rfloor/\alpha\right)}{L}}, 16\sqrt{\frac{R}{\beta L}}, \frac{\sqrt{2}R}{\sqrt{L}}\right).$$

5.17 Proof of Proposition 19

Let $L \ge 2$. For all k in $\{1, \ldots, \lceil L^{3/4} \rceil\}$, let us define $\lambda_k(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau_k, \tau_k + \ell]}(t)$ with $\tau_k = k/L$, and $\ell = \lambda_0 \log L/(2\delta^{*2}L)$. Then λ_k belongs to $\mathcal{S}_{\delta^*, \ldots, \ldots}[\lambda_0]$ for all k in $\{1, \ldots, \lceil L^{3/4} \rceil\}$ as soon as

$$\frac{\left[L^{3/4}\right]}{L} + \frac{\lambda_0 \log L}{2\delta^{*2}L} < 1 \quad , \tag{135}$$

and it satisfies

$$d_2^2(\lambda_k, \mathcal{S}_0[\lambda_0]) = \lambda_0 \log L/(2L)$$

Considering a random variable J uniformly distributed on $\{1, \ldots, \lceil L^{3/4} \rceil\}$ and the distribution μ of λ_J and using Lemma 43, one can see that it is enough to prove that $E_{\lambda_0}[(dP_{\mu}/dP_{\lambda_0})^2] \leq 1 + 4(1 - \alpha - \beta)^2$ to obtain the expected lower bound. By definition, $(dP_{\mu}/dP_{\lambda_0})(N) = E_J[(dP_{\lambda_J}/dP_{\lambda_0})(N)]$, therefore

$$\frac{dP_{\mu}}{dP_{\lambda_0}}(N) = \frac{1}{\lceil L^{3/4} \rceil} \sum_{k=1}^{\lceil L^{3/4} \rceil} \exp\left(\log\left(1 + \frac{\delta^*}{\lambda_0}\right) N(\tau_k, \tau_k + \ell] - L\delta^*\ell\right)$$

We then expand the square as

$$\begin{pmatrix} \frac{dP_{\mu}}{dP_{\lambda_0}} \end{pmatrix}^2 (N) = \frac{1}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil} \exp\left(2\log\left(1 + \frac{\delta^*}{\lambda_0}\right) N(\tau_k, \tau_k + \ell] - 2L\delta^*\ell\right)$$

$$+ \frac{2}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil - 1} \sum_{k'=k+1}^{\lceil L^{3/4} \rceil} \exp\left(\log\left(1 + \frac{\delta^*}{\lambda_0}\right) N(\tau_k, \tau_k + \ell] \right)$$

$$+ \log\left(1 + \frac{\delta^*}{\lambda_0}\right) N(\tau_{k'}, \tau_{k'} + \ell] - 2L\delta^*\ell \right) .$$

For k in $\{1, \ldots, \lceil L^{3/4} \rceil - 1\}$, setting $K_0(k) = \max(k' \in \{k+1, \ldots, \lceil L^{3/4} \rceil\} : \tau_{k'} < \tau_k + \ell)$, we may write

$$\begin{split} \left(\frac{dP_{\mu}}{dP_{\lambda_{0}}}\right)^{2}(N) = & \frac{1}{\lceil L^{3/4} \rceil^{2}} \sum_{k=1}^{\lceil L^{3/4} \rceil^{2}} \exp\left(2\log\left(1 + \frac{\delta^{*}}{\lambda_{0}}\right) N(\tau_{k}, \tau_{k} + \ell] - 2L\delta^{*}\ell\right) \\ &+ \frac{2}{\lceil L^{3/4} \rceil^{2}} \sum_{k=1}^{\lceil L^{3/4} \rceil^{-1}} \sum_{k'=K_{0}(k)+1}^{\lceil L^{3/4} \rceil} \exp\left(\log\left(1 + \frac{\delta^{*}}{\lambda_{0}}\right) N(\tau_{k}, \tau_{k} + \ell] \right) \\ &+ \log\left(1 + \frac{\delta^{*}}{\lambda_{0}}\right) N(\tau_{k'}, \tau_{k'} + \ell] - 2L\delta^{*}\ell\right) \\ &+ \frac{2}{\lceil L^{3/4} \rceil^{2}} \sum_{k=1}^{\lceil L^{3/4} \rceil^{-1}} \sum_{k'=k+1}^{K_{0}(k)} \exp\left(\log\left(1 + \frac{\delta^{*}}{\lambda_{0}}\right) (N(\tau_{k}, \tau_{k'}] + N(\tau_{k'}, \tau_{k} + \ell]) \\ &+ \log\left(1 + \frac{\delta^{*}}{\lambda_{0}}\right) (N(\tau_{k'}, \tau_{k} + \ell] + N(\tau_{k} + \ell, \tau_{k'} + \ell]) - 2L\delta^{*}\ell\right) \end{split}$$

Under (H_0) , N is a homogeneous Poisson process with intensity λ_0 with respect to the measure Λ . Thus

$$\begin{split} E_{\lambda_0} \left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}} \right)^2 (N) \right] &= \frac{\sqrt{L}}{\lceil L^{3/4} \rceil} + \frac{2}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil - 1} (\lceil L^{3/4} \rceil - K_0(k)) \\ &+ \frac{2}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil - 1} \sum_{k'=k+1}^{K_0(k)} \exp\left(\frac{\delta^{*2}L}{\lambda_0} (\tau_k - \tau_{k'} + \ell) \right) \\ &= \frac{\sqrt{L}}{\lceil L^{3/4} \rceil} + \frac{2}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil - 1} (\lceil L^{3/4} \rceil - K_0(k)) \\ &+ \frac{2\sqrt{L}}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil - 1} \sum_{k'=k+1}^{K_0(k)} \exp\left(-\frac{\delta^{*2}}{\lambda_0} (k' - k) \right) \\ &\leq \frac{\sqrt{L}}{\lceil L^{3/4} \rceil} + \frac{2}{\lceil L^{3/4} \rceil^2} \sum_{k=1}^{\lceil L^{3/4} \rceil - 1} (\lceil L^{3/4} \rceil - K_0(k)) + \frac{2\sqrt{L}}{\lceil L^{3/4} \rceil} \left(e^{\delta^{*2}/\lambda_0} - 1 \right)^{-1} \end{split}$$

Since $K_0(k) \ge k+1$ for all k in $\{1, \ldots, \lceil L^{3/4} \rceil - 1\}$, notice that

$$\sum_{k=1}^{\lceil L^{3/4}\rceil - 1} (\lceil L^{3/4}\rceil - K_0(k)) \le \sum_{k=1}^{\lceil L^{3/4}\rceil - 2} k \le \frac{\lceil L^{3/4}\rceil^2}{2} ,$$

hence

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2(N)\right] \le 1 + \frac{\sqrt{L}}{\left\lceil L^{3/4} \right\rceil} \frac{e^{\delta^{*2}/\lambda_0} + 1}{e^{\delta^{*2}/\lambda_0} - 1}$$

Finally, assuming that

$$\frac{\sqrt{L}}{\lceil L^{3/4} \rceil} \frac{e^{\delta^{*2}/\lambda_0} + 1}{e^{\delta^{*2}/\lambda_0} - 1} \le 4(1 - \alpha - \beta)^2 \quad , \tag{136}$$

we get

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2(N)\right] \le 1 + 4(1 - \alpha - \beta)^2 .$$

Noticing that there exists $L_0(\alpha, \beta, \lambda_0, \delta^*) \ge 2$ such that for all $L \ge L_0(\alpha, \beta, \lambda_0, \delta^*)$, both assumptions (135) and (136) hold then allows to end the proof.

5.18 Proof of Proposition 22

The control of the first kind error rates of the two tests $\phi_{9/10,\alpha}^{(1)}$ and $\phi_{9/10,\alpha}^{(2)}$ is straightforward using simple union bounds.

(i) Control of the second kind error rate of $\phi_{9/10,\alpha}^{(1)}$.

Let λ in $\mathcal{S}_{,..,..}[\lambda_0, R]$. We may fix δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$, τ in (0, 1), ℓ in $(0, 1 - \tau)$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau, \tau + \ell]}$. We assume by now that

$$d_2(\lambda, \mathcal{S}_0[\lambda_0]) \ge \sqrt{3} \max\left(\sqrt{\frac{4R\log\left(2/u_\alpha\right)}{3L}}, 2\sqrt{\frac{2\lambda_0\log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{R}{\beta L}}, \frac{R}{\sqrt{L}}\right),$$
(137)

and we prove the inequality $P_{\lambda}\left(\phi_{9/10,\alpha}^{(1)}(N)=0\right) \leq \beta$. Assume first that δ belongs to $(0, R-\lambda_0]$. Noticing that

$$P_{\lambda}\left(\phi_{9/10,\alpha}^{(1)}(N)=0\right)$$

$$\leq \inf_{k\in\{0,\dots,\lceil L\rceil-1\}}\inf_{k'\in\{1,\dots,\lceil L\rceil-k\}}P_{\lambda}\left(N(k/\lceil L\rceil,(k+k')/\lceil L\rceil]\leq p_{\lambda_0k'L/\lceil L\rceil}(1-u_{\alpha}/2)\right),$$

one can see that it is enough to exhibit some k_0 in $\{0, \ldots, \lceil L \rceil - 1\}$ and k'_0 in $\{1, \ldots, \lceil L \rceil - k_0\}$ satisfying

$$P_{\lambda}\left(N(k_0/\lceil L\rceil, (k_0+k'_0)/\lceil L\rceil] \le p_{\lambda_0k'_0L/\lceil L\rceil}(1-u_{\alpha}/2)\right) \le \beta .$$

We get from (137) that

$$d_2^2(\lambda, \mathcal{S}_0[\lambda_0]) \ge 3R^2/L > 3\delta^2/\lceil L \rceil , \qquad (138)$$

which entails

$$\ell > 3/\lceil L \rceil$$
 and $\tau < 1 - 3/\lceil L \rceil$. (139)

We therefore can define $k_0 = \min(k \in \{0, \dots, \lceil L \rceil - 1\}, \tau \leq k/\lceil L \rceil)$ and $k'_0 = \max(k' \in \{1, \dots, \lceil L \rceil - k_0\}, (k_0 + k')/\lceil L \rceil \leq \tau + \ell)$, so that $\tau \leq k_0/\lceil L \rceil < (k_0 + k'_0)/\lceil L \rceil \leq \tau + \ell$. Since by definition $k_0/\lceil L \rceil - \tau < 1/\lceil L \rceil$ and $\tau + \ell - (k_0 + k'_0)/\lceil L \rceil < 1/\lceil L \rceil$, notice that

$$\frac{k_0'}{\lceil L \rceil} = \ell - \left(\left(\frac{k_0}{\lceil L \rceil} - \tau \right) + \left(\tau + \ell - \frac{k_0 + k_0'}{\lceil L \rceil} \right) \right) > \ell - \frac{2}{\lceil L \rceil}$$

This, combined with (139) and the expression of $d_2^2(\lambda, S_0[\lambda_0]) = \delta^2 \ell$, implies that

$$\delta^2 \frac{k_0'}{\lceil L \rceil} > d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0]\right) - \frac{2\delta^2}{\lceil L \rceil} > \frac{d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0]\right)}{3} , \qquad (140)$$

which yields with (137)

$$\delta \sqrt{\frac{k_0'}{\lceil L \rceil}} > \max\left(\sqrt{\frac{4\delta \log\left(2/u_\alpha\right)\right)}{3L}} , \ 2\sqrt{\frac{2\lambda_0 \log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{\lambda_0 + \delta}{\beta L}}\right) . \tag{141}$$

We then deduce on the one hand

$$\frac{\delta k_0'}{\lceil L \rceil} > \frac{4 \log \left(2/u_\alpha \right)}{3L} \ ,$$

and on the other hand

$$\frac{\delta k_0'}{\lceil L \rceil} \ge \sqrt{\frac{k_0'}{\lceil L \rceil}} \left(2\sqrt{\frac{2\lambda_0 \log\left(2/u_\alpha\right)}{L}} + 2\sqrt{\frac{\lambda_0 + \delta}{\beta L}} \right) ,$$

which together can be synthesized in

$$\frac{\delta k_0'}{\lceil L \rceil} > 2 \max \left(\frac{2 \log \left(2/u_\alpha \right)}{3L} \ , \ \sqrt{\frac{k_0'}{\lceil L \rceil}} \left(\sqrt{\frac{2\lambda_0 \log \left(2/u_\alpha \right)}{L}} + \sqrt{\frac{\lambda_0 + \delta}{\beta L}} \right) \right) \ .$$

Hence

$$\frac{\delta k_0' L}{\lceil L \rceil} > \frac{2 \log (2/u_\alpha)}{3} + \sqrt{\frac{2 \log (2/u_\alpha) \lambda_0 k_0' L}{\lceil L \rceil}} + \sqrt{\frac{(\lambda_0 + \delta) k_0' L}{\beta \lceil L \rceil}} .$$
(142)

•

From Lemma 44, we easily deduce that $E_{\lambda}[N(k_0/\lceil L \rceil, (k_0 + k'_0)/\lceil L \rceil]) = (\lambda_0 + \delta)k'_0L/\lceil L \rceil,$ and from Lemma 46

$$p_{\lambda_0 k_0' L/\lceil L\rceil} \left(1 - u_\alpha/2\right) \le \frac{\lambda_0 k_0' L}{\lceil L\rceil} + \frac{\lambda_0 k_0' L}{\lceil L\rceil} g^{-1} \left(\frac{\log\left(2/u_\alpha\right) \lceil L\rceil}{\lambda_0 k_0' L}\right)$$

Using the upper bound (80), this leads to

$$p_{\lambda_0 k_0' L/\lceil L\rceil} \left(1 - u_\alpha/2\right) \le \frac{\lambda_0 k_0' L}{\lceil L\rceil} + \frac{2\log\left(2/u_\alpha\right)}{3} + \sqrt{\frac{2\log\left(2/u_\alpha\right)\lambda_0 k_0' L}{\lceil L\rceil}} \quad (143)$$

The inequality (142) therefore entails

$$E_{\lambda}\left[N(k_{0}/\lceil L\rceil, (k_{0}+k_{0}')/\lceil L\rceil]\right] > p_{\lambda_{0}k_{0}'L/\lceil L\rceil}\left(1-u_{\alpha}/2\right) + \sqrt{\operatorname{Var}_{\lambda}\left(N(k_{0}/\lceil L\rceil, (k_{0}+k_{0}')/\lceil L\rceil]\right)/\beta} \quad (144)$$

We conclude with (144) and the Bienayme-Chebyshev inequality:

$$P_{\lambda}\left(N(k_{0}/\lceil L\rceil, (k_{0}+k_{0}')/\lceil L\rceil] \leq p_{\lambda_{0}k_{0}'L/\lceil L\rceil} (1-u_{\alpha}/2)\right)$$

$$\leq P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L\rceil}, \frac{k_{0}+k_{0}'}{\lceil L\rceil}\right) - E_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil}, \frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)\right] < -\sqrt{\operatorname{Var}_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L\rceil}, \frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)\right)/\beta}\right)$$

$$\leq \beta \quad .$$

Assume now that δ belongs to $(-\lambda_0, 0)$ and notice that we have here

$$P_{\lambda}\left(\phi_{9/10,\alpha}^{(1)}(N)=0\right)$$

$$\leq \inf_{k\in\{0,\dots,\lceil L\rceil-1\}}\inf_{k'\in\{1,\dots,\lceil L\rceil-k\}}P_{\lambda}\left(N(k/\lceil L\rceil,(k+k')/\lceil L\rceil]\geq p_{\lambda_0k'L/\lceil L\rceil}(u_{\alpha}/2)\right).$$

Lemma 46 and (80) give

$$p_{\lambda_0 k' L/\lceil L\rceil} \left(u_{\alpha}/2 \right) \geq \frac{\lambda_0 k'_0 L}{\lceil L\rceil} - \frac{2\log\left(2/u_{\alpha}\right)}{3} - \sqrt{\frac{2\log\left(2/u_{\alpha}\right)\lambda_0 k'_0 L}{\lceil L\rceil}}$$

The same choice of k_0 and k'_0 as in the previous case yields $E_{\lambda} \left[N(k_0/\lceil L \rceil, (k_0 + k'_0)/\lceil L \rceil) \right] =$ Var_{λ} $\left(N(k_0/\lceil L \rceil, (k_0 + k'_0)/\lceil L \rceil) \right) = (\lambda_0 + \delta)k'_0L/\lceil L \rceil = (\lambda_0 - |\delta|)k'_0L/\lceil L \rceil$, so

$$P_{\lambda}\left(N(k_0/\lceil L\rceil, (k_0+k'_0)/\lceil L\rceil] \ge p_{\lambda_0k'_0L/\lceil L\rceil}(u_{\alpha}/2)\right) \le \beta ,$$

in a similar way, notably replacing δ by $|\delta|$ (except when it is involved in $\lambda_0 + \delta$) in the rest of the proof.

Coming back to the assumption (137) and the definition of u_{α} , one can finally claim that

$$\operatorname{SR}_{\beta}\left(\phi_{9/10,\alpha}^{(1)}, \mathcal{S}_{\cdot, \cdots, \cdots}[\lambda_{0}, R]\right) \leq \sqrt{3} \max\left(\sqrt{\frac{4R \log\left(\lceil L \rceil (\lceil L \rceil + 1)/\alpha\right)}{3L}}, \frac{2\sqrt{\frac{2\lambda_{0} \log\left(\lceil L \rceil (\lceil L \rceil + 1)/\alpha\right)}{L}} + 2\sqrt{\frac{R}{\beta L}}, \frac{R}{\sqrt{L}}\right),$$

which ends the proof for the test $\phi_{9/10,\alpha}^{(1)}$.

(ii) Control of the second kind error rate of $\phi_{9/10,\alpha}^{(2)}$.

Let λ in $\mathcal{S}_{,..,..}[\lambda_0, R]$. There exist δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}, \tau$ in $(0, 1), \ell$ in $(0, 1 - \tau)$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau, \tau + \ell]}$. Let us assume now that

$$d_{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) \geq \max\left(R\sqrt{\frac{3}{L}}, 4\sqrt{\frac{3\lambda_{0}\log(3/u_{\alpha})}{L}} + 2\sqrt{\frac{3\sqrt{2}R}{\sqrt{\beta}L}}, 2\sqrt{\frac{\sqrt{2}R\log(3/u_{\alpha})}{L}}, 32^{1/3}(6\lambda_{0})^{1/6}R^{1/3}\sqrt{\frac{\log(3/u_{\alpha})}{L}}, 24\sqrt{\frac{R}{\beta L}}\right), (145)$$

and prove that it entails $P_{\lambda}\left(\phi_{9/10,\alpha}^{(2)}(N)=0\right) \leq \beta$. As in (*i*), we begin by noticing that

$$P_{\lambda}\left(\phi_{9/10,\alpha}^{(2)}(N)=0\right) \leq \inf_{k \in \{0,\dots,\lceil L \rceil-1\}} \inf_{k' \in \{1,\dots,\lceil L \rceil-k\}} P_{\lambda}\left(T_{\frac{k}{\lceil L \rceil},\frac{k+k'}{\lceil L \rceil}}(N) \leq t_{\lambda_{0},\frac{k}{\lceil L \rceil},\frac{k+k'}{\lceil L \rceil}}(1-u_{\alpha})\right) ,$$

so it suffices to find k_0 in $\{0, \ldots, \lceil L \rceil - 1\}$ and k'_0 in $\{1, \ldots, \lceil L \rceil - k_0\}$ satisfying

$$P_{\lambda}\left(T_{\frac{k_{0}}{\lceil L \rceil},\frac{k_{0}+k_{0}'}{\lceil L \rceil}}(N) \leq t_{\lambda_{0},\frac{k_{0}}{\lceil L \rceil},\frac{k_{0}+k_{0}'}{\lceil L \rceil}}(1-u_{\alpha})\right) \leq \beta$$

to obtain $P_{\lambda}\left(\phi_{9/10,\alpha}^{(2)}(N)=0\right) \leq \beta$. We get from (145) that $d_2^2\left(\lambda, \mathcal{S}_0[\lambda_0]\right) \geq 3R^2/L > 3\delta^2/\lceil L\rceil$ which entails

 $\ell > 3/\lceil L \rceil$ and $\tau < 1 - 3/\lceil L \rceil$. (146)

We therefore can define $k_0 = \min(k \in \{0, \dots, \lceil L \rceil - 1\}, \tau \leq k/\lceil L \rceil)$ and $k'_0 = \max(k' \in \{1, \dots, \lceil L \rceil - k_0\}, (k_0 + k')/\lceil L \rceil \leq \tau + \ell)$, so that $\tau \leq k_0/\lceil L \rceil < (k_0 + k'_0)/\lceil L \rceil \leq \tau + \ell$.

As in (i) above, starting from the remark that $k_0/\lceil L \rceil - \tau < 1/\lceil L \rceil$ and $\tau + \ell - (k_0 + k'_0)/\lceil L \rceil < 1/\lceil L \rceil$, which implies $k'_0/\lceil L \rceil > \ell - 2/\lceil L \rceil$, we obtain, combined with (146) and the expression of $d_2^2(\lambda, S_0[\lambda_0]) = \delta^2 \ell$:

$$\delta^2 \frac{k'_0}{\lceil L \rceil} > d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0]\right) - \frac{2\delta^2}{\lceil L \rceil} > \frac{d_2^2 \left(\lambda, \mathcal{S}_0[\lambda_0]\right)}{3} \quad . \tag{147}$$

Moreover, we get from (145)

$$d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right) > \max\left(4\sqrt{\frac{3\lambda_{0}\log\left(3/u_{\alpha}\right)}{L}} + 2\sqrt{\frac{3\sqrt{2}(\lambda_{0}+\delta)}{\sqrt{\beta}L}}, 2\sqrt{\frac{\sqrt{2}|\delta|\log\left(3/u_{\alpha}\right)}{L}}, 32^{1/3}(6\lambda_{0})^{1/6}|\delta|^{1/3}\sqrt{\frac{\log\left(3/u_{\alpha}\right)}{L}}, 24\sqrt{\frac{\lambda_{0}+\delta}{\beta L}}\right),$$

which entails

$$d_{2}^{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) > \max\left(\frac{48\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{12\sqrt{2}(\lambda_{0}+\delta)}{\sqrt{\beta}L}, \frac{32\delta^{2}\log^{2}\left(3/u_{\alpha}\right)}{L^{2}d_{2}^{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}])}, \frac{32\sqrt{6\lambda_{0}}|\delta|\log^{3/2}\left(3/u_{\alpha}\right)}{L^{3/2}d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right)}, \frac{24\sqrt{\lambda_{0}+\delta}d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right)}{\sqrt{\beta}L}\right).$$

Then with (147),

$$d_{2}^{2}(\lambda, \mathcal{S}_{0}[\lambda_{0}]) > \max\left(\frac{48\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{12\sqrt{2}(\lambda_{0}+\delta)}{\sqrt{\beta}L}, \frac{32\log^{2}\left(3/u_{\alpha}\right)\left\lceil L\right\rceil}{3k_{0}^{\prime}L^{2}}, \frac{32\log^{3/2}\left(3/u_{\alpha}\right)}{L^{3/2}}\sqrt{\frac{2\lambda_{0}\left\lceil L\right\rceil}{k_{0}^{\prime}}}, \frac{24\sqrt{\lambda_{0}+\delta}d_{2}\left(\lambda, \mathcal{S}_{0}[\lambda_{0}]\right)}{\sqrt{\beta}L}\right),$$

hence

$$\frac{d_2^2\left(\lambda, \mathcal{S}_0[\lambda_0]\right)}{3} > \frac{4\lambda_0 \log\left(3/u_\alpha\right)}{L} + \frac{\sqrt{2}(\lambda_0 + \delta)}{\sqrt{\beta}L} + \frac{8\log^2\left(3/u_\alpha\right)\left\lceil L\right\rceil}{9k'_0 L^2} + \frac{8\log^{3/2}\left(3/u_\alpha\right)}{3L^{3/2}}\sqrt{\frac{2\lambda_0\left\lceil L\right\rceil}{k'_0}} + \frac{2\sqrt{\lambda_0 + \delta}d_2\left(\lambda, \mathcal{S}_0[\lambda_0]\right)}{\sqrt{\beta}L}\right) . \quad (148)$$

Using Lemma 47, we compute

$$E_{\lambda}\left[T_{\frac{k_{0}}{\lceil L \rceil},\frac{k_{0}+k_{0}'}{\lceil L \rceil}}(N)\right] = \delta^{2}\frac{k_{0}'}{\lceil L \rceil} > \frac{d_{2}^{2}\left(\lambda,\mathcal{S}_{0}[\lambda_{0}]\right)}{3} \quad \text{with (147)} \quad ,$$

and

$$\operatorname{Var}_{\lambda}\left(T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N)\right) = \frac{4(\lambda_{0}+\delta)\delta^{2}}{L}\frac{k_{0}'}{|L|} + \frac{2(\lambda_{0}+\delta)^{2}}{L^{2}}$$
$$\leq \frac{4(\lambda_{0}+\delta)d_{2}^{2}(\lambda,\mathcal{S}_{0}[\lambda_{0}])}{L} + \frac{2(\lambda_{0}+\delta)^{2}}{L^{2}}$$

•

These computations combined with (148) leads to

$$E_{\lambda}\left[T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N)\right] > \frac{4\lambda_{0}\log\left(3/u_{\alpha}\right)}{L} + \frac{8\log^{2}\left(3/u_{\alpha}\right)\left[L\right]}{9k_{0}'L^{2}} + \frac{8\log^{3/2}\left(3/u_{\alpha}\right)}{3L^{3/2}}\sqrt{\frac{2\lambda_{0}\left[L\right]}{k_{0}'}} + \sqrt{\operatorname{Var}_{\lambda}\left(T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N)\right)/\beta} \quad (149)$$

Furthermore, Lemma 48 gives

$$t_{\lambda_0,\frac{k_0}{\lceil L \rceil},\frac{k_0+k'_0}{\lceil L \rceil}}(1-u_\alpha) \le \frac{2\lambda_0^2 k'_0}{\lceil L \rceil} \left(g^{-1} \left(\frac{\log\left(3/u_\alpha\right) \lceil L \rceil}{\lambda_0 k'_0 L}\right)\right)^2 ,$$

where $g^{-1}(x) \leq 2x/3 + \sqrt{2x}$ for all x > 0 (see (80)), and then

$$t_{\lambda_{0},\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(1-u_{\alpha}) \leq \frac{8\log^{2}\left(3/u_{\alpha}\right)\left[L\right]}{9k_{0}'L^{2}} + \frac{8\log^{3/2}\left(3/u_{\alpha}\right)}{3L^{3/2}}\sqrt{\frac{2\lambda_{0}\left[L\right]}{k_{0}'} + \frac{4\lambda_{0}\log\left(3/u_{\alpha}\right)}{L}} (150)$$

It follows from (149) and (150) that

$$E_{\lambda}\left[T_{\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}}(N)\right] > t_{\lambda_{0},\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}}(1-u_{\alpha}) + \sqrt{\operatorname{Var}_{\lambda}\left(T_{\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}}(N)\right)}/\beta \ ,$$

whereby

$$\begin{split} P_{\lambda}\left(T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N) \leq t_{\lambda_{0},\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}\left(1-u_{\alpha}\right)\right) \\ \leq P_{\lambda}\left(T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N) \leq E_{\lambda}\left[T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N)\right] - \sqrt{\operatorname{Var}_{\lambda}\left(T_{\frac{k_{0}}{|L|},\frac{k_{0}+k_{0}'}{|L|}}(N)\right)/\beta}\right) \end{split}$$

The Bienayme-Chebyshev inequality allows to conclude that

$$P_{\lambda}\left(T_{\frac{k_{0}}{\lceil L \rceil},\frac{k_{0}+k_{0}'}{\lceil L \rceil}}(N) \leq t_{\lambda_{0},\frac{k_{0}}{\lceil L \rceil},\frac{k_{0}+k_{0}'}{\lceil L \rceil}}(1-u_{\alpha})\right) \leq \beta .$$

Coming back to the assumption (145) and the definition of u_{α} , one can finally claim that

$$\operatorname{SR}_{\beta}\left(\phi_{9/10,\alpha}^{(2)}, \mathcal{S}_{\cdot,\cdot,\cdot,\cdot}[\lambda_{0}, R]\right) \leq \max\left(4\sqrt{\frac{3\lambda_{0}\log\left(3\lceil L\rceil(\lceil L\rceil + 1)/(2\alpha)\right)}{L}} + 2\sqrt{\frac{3\sqrt{2}R}{\sqrt{\beta}L}}, \frac{3\sqrt{2}R}{\sqrt{\beta}L}\right)$$

$$2\sqrt{\frac{\sqrt{2}R\log\left(3\lceil L\rceil(\lceil L\rceil + 1)/2\alpha\right)}{L}}, 24\sqrt{\frac{R}{\beta L}}, R\sqrt{\frac{3}{L}}, 32^{1/3}(6\lambda_{0})^{1/6}R^{1/3}\sqrt{\frac{\log\left(3\lceil L\rceil(\lceil L\rceil + 1)/2\alpha\right)}{L}}\right)$$

,

which ends the proof for the test $\phi_{9/10,\alpha}^{(2)}$.

5.19 Proof of Proposition 25

By definition of $b_{n,\ell^*}(u)$ as the *u*-quantile of a binomial distribution with parameters (n, ℓ^*) , the Bienayme-Chebyshev inequality easily gives

$$b_{n,\ell^*}(1-\alpha) \le n\ell^* + \sqrt{n\ell^*(1-\ell^*)/\alpha}$$

for all n in \mathbb{N} . It also gives for every n in \mathbb{N} and every $\varepsilon > 0$, $b_{n,\ell^*}(\alpha) > n\ell^* - \sqrt{n\ell^*(1-\ell^*)/(\alpha-\varepsilon)}$. Therefore, letting ε tending to 0,

$$b_{n,\ell^*}(\alpha) \ge n\ell^* - \sqrt{n\ell^*(1-\ell^*)/\alpha} \ .$$

(i) Assume first that $0 < \delta^* < R$ and let λ in $\mathcal{S}^u_{\delta^*, \tau^*, \ell^*}[R]$. Setting $I(\lambda) = \int_0^1 \lambda(t) dt \leq R$, the assumption (42) leads to

$$\delta^* \sqrt{\ell^* (1-\ell^*)} \ge \frac{1}{\sqrt{L}} \left(\sqrt{\frac{2(\lambda_0 + \delta^*)}{\beta(1-\ell^*)}} + 2\sqrt{\frac{I(\lambda)\ell^*}{\beta(1-\ell^*)}} + \sqrt{\frac{1}{\alpha} \left(I(\lambda) + 2\sqrt{\frac{I(\lambda)}{\beta L}} \right)} \right)$$

Hence

$$\delta^* \ell^* (1-\ell^*) L \ge \sqrt{\frac{2(\lambda_0+\delta^*)\ell^* L}{\beta}} + 2\ell^* \sqrt{\frac{I(\lambda)L}{\beta}} + \sqrt{\frac{\ell^* (1-\ell^*)}{\alpha}} \left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}} \right) ,$$

and since $I(\lambda) = \lambda_0 + \delta^* \ell^*$,

$$\left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}}\right)\ell^* + \sqrt{\frac{\ell^*(1-\ell^*)}{\alpha}\left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}}\right)} - (\lambda_0 + \delta^*)\ell^*L \le -\sqrt{\frac{2(\lambda_0 + \delta^*)\ell^*L}{\beta}} \quad (151)$$

We get then the following inequalities

$$\begin{split} &P_{\lambda}\left(\phi_{1,\alpha}^{u,+}(N)=0\right)\\ &=P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] < b_{N_{1},\ell^{*}}(1-\alpha)\right)\\ &=P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] < b_{N_{1},\ell^{*}}(1-\alpha) \ , \ |N_{1}-I(\lambda)L| \leq 2\sqrt{I(\lambda)L/\beta}\right)\\ &+P_{\lambda}\left(N_{1} < I(\lambda)L - 2\sqrt{I(\lambda)L/\beta}\right) + P_{\lambda}\left(N_{1} > I(\lambda)L + 2\sqrt{I(\lambda)L/\beta}\right)\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] < b_{N_{1},\ell^{*}}(1-\alpha) \ , \ |N_{1}-I(\lambda)L| \leq 2\sqrt{I(\lambda)L/\beta}\right) + \frac{\beta}{2}\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] < N_{1}\ell^{*} + \sqrt{\frac{N_{1}\ell^{*}(1-\ell^{*})}{\alpha}} \ , \ |N_{1}-I(\lambda)L| \leq 2\sqrt{I(\lambda)L/\beta}\right) + \frac{\beta}{2}\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] < \left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}}\right)\ell^{*} + \sqrt{\frac{\ell^{*}(1-\ell^{*})}{\alpha}}\left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}}\right)\right) + \frac{\beta}{2}\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] - (\lambda_{0}+\delta^{*})\ell^{*}L < -\sqrt{\frac{2(\lambda_{0}+\delta^{*})\ell^{*}L}{\beta}}\right) + \frac{\beta}{2} \ (\text{thanks to (151)})\\ &\leq \beta \ (\text{Bienayme-Chebyshev}) \ . \end{split}$$

This concludes the proof of (i). (ii) Assume now that $-R < \delta^* < 0$ and let λ in $\mathcal{S}^{u}_{\delta^*,\tau^*,\ell^*}[R]$. The assumption (42) entails

$$|\delta^*|\sqrt{\ell^*(1-\ell^*)} \ge \frac{1}{\sqrt{L}} \left(\sqrt{\frac{2(\lambda_0+\delta^*)}{\beta(1-\ell^*)}} + 2\sqrt{\frac{I(\lambda)\ell^*}{\beta(1-\ell^*)}} + \sqrt{\frac{1}{\alpha} \left(I(\lambda) + 2\sqrt{\frac{I(\lambda)}{\beta L}}\right)} \right) .$$

Hence

$$-\delta^*\ell^*(1-\ell^*)L \ge \sqrt{\frac{2(\lambda_0+\delta^*)\ell^*L}{\beta}} + 2\ell^*\sqrt{\frac{I(\lambda)L}{\beta}} + \sqrt{\frac{\ell^*(1-\ell^*)}{\alpha}}\left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}}\right) ,$$

and then

$$\left(I(\lambda)L - 2\sqrt{\frac{I(\lambda)L}{\beta}}\right)\ell^* - \sqrt{\frac{\ell^*(1-\ell^*)}{\alpha}\left(I(\lambda)L + 2\sqrt{\frac{I(\lambda)L}{\beta}}\right)} - (\lambda_0 + \delta^*)\ell^*L \ge \sqrt{\frac{2(\lambda_0 + \delta^*)\ell^*L}{\beta}} \quad . \tag{152}$$

We get then

$$\begin{split} &P_{\lambda}\left(\phi_{1,\alpha}^{u,-}(N)=0\right)\\ &=P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]>b_{N_{1},\ell^{*}}(\alpha)\right)\\ &=P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]>b_{N_{1},\ell^{*}}(\alpha)\ ,\ |N_{1}-I(\lambda)L|\leq 2\sqrt{I(\lambda)L/\beta}\right)\\ &+P_{\lambda}\left(N_{1}< I(\lambda)L-2\sqrt{I(\lambda)L/\beta}\right)+P_{\lambda}\left(N_{1}>I(\lambda)L+2\sqrt{I(\lambda)L/\beta}\right)\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]>b_{N_{1},\ell^{*}}(\alpha)\ ,\ |N_{1}-I(\lambda)L|\leq 2\sqrt{I(\lambda)L/\beta}\right)+\frac{\beta}{2}\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]>N_{1}\ell^{*}-\sqrt{\frac{N_{1}\ell^{*}(1-\ell^{*})}{\alpha}},\ |N_{1}-I(\lambda)L|\leq 2\sqrt{I(\lambda)L/\beta}\right)+\frac{\beta}{2}\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]>\left(I(\lambda)L-2\sqrt{\frac{I(\lambda)L}{\beta}}\right)\ell^{*}-\sqrt{\frac{\ell^{*}(1-\ell^{*})}{\alpha}}\left(I(\lambda)L+2\sqrt{\frac{I(\lambda)L}{\beta}}\right)\right)+\frac{\beta}{2}\\ &\leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]-(\lambda_{0}+\delta^{*})\ell^{*}L>\sqrt{\frac{2(\lambda_{0}+\delta^{*})\ell^{*}L}{\beta}}\right)+\frac{\beta}{2}\ (\text{thanks to (152)})\\ &\leq \beta\ (\text{Bienayme-Chebyshev})\ . \end{split}$$

This concludes the proof of (ii).

5.20 Proof of Proposition 26

Let us set $\lambda_0 = R/2$ and introduce for r > 0 the Poisson intensity λ_r defined for all t in [0, 1] by

$$\lambda_r(t) = \lambda_0 + \frac{r}{\sqrt{\ell^*(1-\ell^*)}} \mathbb{1}_{(\tau^*,\tau^*+\ell^*]}(t) \ .$$

Notice that when $0 < r/\sqrt{\ell^*(1-\ell^*)} \le R/2$, λ_r belongs to

$$(\mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R])_{r} = \{\lambda \in \mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R], \ d_{2}(\lambda,\mathcal{S}_{0}[R]) \geq r\} \ ,$$

as defined in Lemma 42. We get from Lemma 1 and Lemma 44 that

$$E_{\lambda_0}\left[\left(\frac{dP_{\lambda}}{dP_{\lambda_0}}\right)^2(N)\right] = \exp\left(\frac{r^2L}{\lambda_0(1-\ell^*)}\right)$$

Choosing $r = (\lambda_0(1-\ell^*)\log C_{\alpha,\beta}/L)^{1/2}$ then leads to $E_{\lambda_0}\left[(dP_{\lambda}/dP_{\lambda_0})^2(N)\right] = C_{\alpha,\beta}$. For $L \ge 1$ such that $2\log C_{\alpha,\beta}/(\ell^*L) \le R$, we obtain $0 < r/\sqrt{\ell^*(1-\ell^*)} \le R/2$ whereby λ_r belongs to $(\mathcal{S}^u_{\cdot,\tau^*,\ell^*}[R])_r$ and Lemma 43 allows us to conclude that $\rho\left(\left(\mathcal{S}^u_{\cdot,\tau^*,\ell^*}[R]\right)_r\right) \ge \beta$ and $\mathrm{mSR}_{\alpha,\beta}\left(\mathcal{S}^u_{\cdot,\tau^*,\ell^*}[R]\right) \ge r$.

5.21 Proof of Proposition 27

The first statement of Proposition 27 is straightforward, just noticing that for every λ_0 in $\mathcal{S}_0^u[R]$

$$E_{\lambda_0}\left[\phi_{2,\alpha}^{u(1/2)}(N)\right] = E_{\lambda_0}\left[E_{\lambda_0}\left[\phi_{2,\alpha}^{u(1/2)}(N)\middle|N_1\right]\right] \le \alpha .$$

Let us assume that $\lambda \in \mathcal{S}^{u}_{\cdot,\tau^{*},\ell^{*}}[R]$, that is there exists λ_{0} in (0, R) and δ in $(-\lambda_{0}, R - \lambda_{0}] \setminus \{0\}$ satisfying $\lambda(t) = \lambda_{0} + \delta \mathbb{1}_{(\tau^{*},\tau^{*}+\ell^{*}]}(t)$ for all t in [0, 1]. Let us first consider the test $\phi^{u(1)}_{2,\alpha}(N)$ and assume

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge \frac{1}{\sqrt{L}} \left(\sqrt{\frac{2R}{\beta(1-\ell^*)}} + 2\sqrt{\frac{R\ell^*}{\beta(1-\ell^*)}} + \sqrt{\frac{1}{\alpha_1 \wedge \alpha_2} \left(R + 2\sqrt{\frac{R}{\beta L}}\right)} \right).$$
(153)

We may write $\phi_{2,\alpha}^{u(1)}(N) = \phi_{1,\alpha_2}^{u,-}(N) \lor \phi_{1,\alpha_1}^{u,+}(N)$ by definition of the tests $\phi_{1,\alpha}^{u,-}$ and $\phi_{1,\alpha}^{u,+}$ in (40). We therefore obtain $P_{\lambda}\left(\phi_{2,\alpha}^{u(1)}(N)=0\right) = P_{\lambda}\left(\phi_{1,\alpha_2}^{u,-}(N)=0, \ \phi_{1,\alpha_1}^{u,+}(N)=0\right)$. From the assumption (153) and the same computations as in the proof of Proposition 25, we get

$$P_{\lambda}\left(\phi_{1,\alpha_{2}}^{u,-}(N)=0\right) \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}]>b_{N_{1},\ell^{*}}(\alpha_{2})\right) \leq \beta ,$$

when $-\lambda_0 < \delta < 0$ and

$$P_{\lambda}\left(\phi_{1,\alpha_{1}}^{u,+}(N)=0\right) \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell^{*}] < b_{N_{1},\ell^{*}}(1-\alpha_{1})\right) \leq \beta ,$$

when $0 < \delta \leq R - \lambda_0$.

The result of Proposition 27 for the test $\phi_{2,\alpha}^{u(1)}(N)$ follows with

$$C(\alpha,\beta,R,\ell^*) = \sqrt{\frac{2R}{\beta(1-\ell^*)}} + 2\sqrt{\frac{R\ell^*}{\beta(1-\ell^*)}} + \sqrt{\frac{1}{\alpha_1 \wedge \alpha_2}} \left(R + 2\sqrt{\frac{R}{\beta L}}\right) .$$

Let us consider then the test $\phi_{2,\alpha}^{u(2)}(N)$. We get from Lemma 53

$$t'_{n,\tau^*,\tau^*+\ell^*}(1-\alpha) \le \frac{C}{L^2} \left(5n \log\left(\frac{2.77}{\alpha}\right) + 3 \max\left(\frac{1-\ell^*}{\ell^*}, \frac{\ell^*}{1-\ell^*}\right) \log^2\left(\frac{2.77}{\alpha}\right) \right) .$$

Now, the Bienayme-Chebyshev inequality and the bound $\int_0^1 \lambda(x) L dx \leq RL$ give

$$P_{\lambda}\left(N_1 \ge RL + \sqrt{\frac{2RL}{\beta}}\right) \le \frac{\beta}{2}$$

This yields $P_{\lambda}\left(t'_{N_1,\tau^*,\tau^*+\ell^*}(1-\alpha) \ge C'(\alpha,\beta,R,\ell^*,L)\right) \le \beta/2$ with

$$C'(\alpha,\beta,R,\ell^*,L) = C\left(5R\frac{\log\left(\frac{2.77}{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(\frac{2.77}{\alpha}\right)}{L^{3/2}} + 3\max\left(\frac{1-\ell^*}{\ell^*}, \frac{\ell^*}{1-\ell^*}\right)\frac{\log^2\left(\frac{2.77}{\alpha}\right)}{L^2}\right)$$

Noticing that

$$P_{\lambda}\left(\phi_{2,\alpha}^{u(2)}(N) = 0\right) \le P_{\lambda}\left(T'_{\tau^{*},\tau^{*}+\ell^{*}}(N) \le C'(\alpha,\beta,R,\ell^{*},L)\right) + P_{\lambda}\left(t'_{N_{1},\tau^{*},\tau^{*}+\ell^{*}}(1-\alpha) > C'(\alpha,\beta,R,\ell^{*},L)\right) ,$$

this enables to write

$$P_{\lambda}\left(\phi_{2,\alpha}^{u(2)}(N) = 0\right) \le P_{\lambda}\left(T'_{\tau^*,\tau^*+\ell^*}(N) \le C'(\alpha,\beta,R,\ell^*,L)\right) + \frac{\beta}{2} .$$
(154)

Assume now

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge \frac{1}{\sqrt{L}} \max\left(4\sqrt{\frac{2R}{\beta}} , \left(\frac{4R}{\sqrt{\beta}} + 2LC'(\alpha, \beta, R, \ell^*, L)\right)^{1/2}\right) , \qquad (155)$$

which ensures

$$|\delta|\sqrt{\ell^*(1-\ell^*)} \ge \frac{1}{\sqrt{L}} \max\left(4\sqrt{\frac{2(\lambda_0+\delta(1-\ell^*))}{\beta}}, \left(\frac{4(\lambda_0+\delta(1-\ell^*))}{\sqrt{\beta}}+2LC'(\alpha,\beta,R,\ell^*,L)\right)^{1/2}\right)$$

We get then

$$\delta^{2}\ell^{*}(1-\ell^{*}) \geq 2 \max\left(2\sqrt{\frac{2(\lambda_{0}+\delta(1-\ell^{*}))}{\beta L}}|\delta|\sqrt{\ell^{*}(1-\ell^{*})}, \frac{2(\lambda_{0}+\delta(1-\ell^{*}))}{\sqrt{\beta}L} + C'(\alpha,\beta,R,\ell^{*},L)\right),$$

and using the simple facts that $a + b \le 2 \max(a, b)$ and $\sqrt{a + b} \le \sqrt{a} + \sqrt{b}$ for all $a, b \ge 0$,

$$\delta^{2}\ell^{*}(1-\ell^{*}) \geq \sqrt{\frac{2}{\beta} \left(\frac{4(\lambda_{0}+\delta(1-\ell^{*}))}{L} \delta^{2}\ell^{*}(1-\ell^{*}) + \frac{2(\lambda_{0}+\delta(1-\ell^{*}))^{2}}{L^{2}}\right)} + C'(\alpha,\beta,R,\ell^{*},L)$$
(156)

Furthermore, Lemma 52 gives $E_{\lambda}[T'_{\tau^*,\tau^*+\ell^*}(N)] = \delta^2 \ell^*(1-\ell^*)$ and

$$\operatorname{Var}_{\lambda}\left(T_{\tau^{*},\tau^{*}+\ell^{*}}^{\prime}(N)\right) = \frac{2(\lambda_{0}+\delta(1-\ell^{*}))^{2}}{L^{2}} + \frac{4(\lambda_{0}+\delta(1-\ell^{*}))}{L}\delta^{2}\ell^{*}(1-\ell^{*})$$

so (156) leads to

$$E_{\lambda}[T'_{\tau^*,\tau^*+\ell^*}(N)] \ge \sqrt{2\operatorname{Var}_{\lambda}\left(T'_{\tau^*,\tau^*+\ell^*}(N)\right)/\beta} + C'(\alpha,\beta,R,\ell^*,L) \quad .$$

Combined with (154), this inequality entails

$$P_{\lambda}\left(\phi_{2,\alpha}^{u(2)}(N)=0\right) \le P_{\lambda}\left(T'_{\tau^{*},\tau^{*}+\ell^{*}}(N)-E_{\lambda}[T'_{\tau^{*},\tau^{*}+\ell^{*}}(N)] \le \sqrt{2\mathrm{Var}_{\lambda}\left(T'_{\tau^{*},\tau^{*}+\ell^{*}}(N)\right)/\beta}\right) + \frac{\beta}{2} ,$$

and the proof ends with the Bienayme-Chebyshev inequality, thus giving

$$P_{\lambda}\left(\phi_{2,\alpha}^{u(2)}(N)=0\right) \leq \beta$$
.

The result of Proposition 27 for the test $\phi_{2,\alpha}^{u(2)}(N)$ then follows with

$$C(\alpha, \beta, R, \ell^*) = \max\left(4\sqrt{\frac{2R}{\beta}}, \left(\frac{4R}{\sqrt{\beta}} + 2C\left(5R\log\left(\frac{2.77}{\alpha}\right) + 5\sqrt{\frac{2R}{\beta}}\log\left(\frac{2.77}{\alpha}\right) + 3\max\left(\frac{\ell^*}{1-\ell^*}, \frac{1-\ell^*}{\ell^*}\right)\log^2\left(\frac{2.77}{\alpha}\right)\right)\right)^{1/2}\right),$$

where the constant C is defined in Lemma 53.

5.22 Proof of Proposition 28

Start by remarking that the control of the first kind error rates of the three tests $\phi_{3,\alpha}^{u(1)+}$, $\phi_{3,\alpha}^{u(1)-}$ and $\phi_{3/4,\alpha}^{u(2)}$ is straightforward, considering the same conditioning trick as in the beginning of the proof of Proposition 27 above.

Let us first address the statement of Proposition 28 for $\phi_{3,\alpha}^{u(1)+}$. Let $L \geq 1$ and let us consider $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau,\tau+\ell^*]}$ in $\mathcal{S}^u_{\delta^*,\cdot,\cdot,\ell^*}[R]$ with $\delta^* > 0$. Notice that

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)+}(N)=0\right) = P_{\lambda}\left(\max_{t\in[0,1-\ell^{*}\wedge(1/2)]}N(t,t+\ell^{*}\wedge(1/2)] \le b_{N_{1},\ell^{*}\wedge(1/2)}^{+}(1-\alpha)\right)$$
$$\le P_{\lambda}\left(N(\tau,\tau+\ell^{*}\wedge(1/2)] \le b_{N_{1},\ell^{*}\wedge(1/2)}^{+}(1-\alpha)\right).$$

One deduces from Lemma 54 that for every n in $\mathbb{N} \setminus \{0\}$,

$$b_{n,\ell^* \wedge (1/2)}^+(1-\alpha) \le (\ell^* \wedge (1/2))n + \frac{n}{2}g^{-1}\left(\frac{32}{n}\log\left(\frac{320}{\alpha}\right)\right) ,$$

with g defined by (79), and then from the inequality $g^{-1}(x) \leq 2x/3 + \sqrt{2x}$ for all x > 0 (see (80)),

$$b_{n,\ell^* \wedge (1/2)}^+(1-\alpha) \le (\ell^* \wedge (1/2))n + 4\sqrt{n\log\left(\frac{320}{\alpha}\right)} + \frac{32}{3}\log\left(\frac{320}{\alpha}\right)$$

Since $b_{0,\ell^* \wedge (1/2)}^+(1-\alpha) = 0$, the above control holds in fact for every n in \mathbb{N} , whereby

$$b_{N_1,\ell^*\wedge(1/2)}^+(1-\alpha) \le (\ell^*\wedge(1/2))N_1 + 4\sqrt{N_1\log\left(\frac{320}{\alpha}\right)} + \frac{32}{3}\log\left(\frac{320}{\alpha}\right)$$

Setting $I_{\lambda,L}(\beta) = I(\lambda)L + \sqrt{\frac{2I(\lambda)L}{\beta}}$ with $I(\lambda) = \int_0^1 \lambda(t)dt$, and

$$Q(\alpha, \beta, \ell, L) = \ell I_{\lambda, L}(\beta) + 4 \sqrt{I_{\lambda, L}(\beta) \log\left(\frac{320}{\alpha}\right) + \frac{32}{3} \log\left(\frac{320}{\alpha}\right)} \quad , \tag{157}$$

we obtain for ℓ in (0, 1/2]

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)+}(N)=0\right) \leq P_{\lambda}\left(N(\tau,\tau+\ell^*\wedge(1/2))\right) \leq Q(\alpha,\beta,\ell^*\wedge(1/2),L) + P_{\lambda}\left(N_1 > I_{\lambda,L}(\beta)\right)$$

Therefore

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)+}(N) = 0\right) \le P_{\lambda}\left(N(\tau, \tau + \ell^* \land (1/2)] \le Q(\alpha, \beta, \ell^* \land (1/2), L)\right) + \frac{\beta}{2} , \quad (158)$$

with

$$Q(\alpha, \beta, \ell^* \wedge (1/2), L) \leq \left(\ell^* \wedge \frac{1}{2}\right) (\lambda_0 + \delta^* \ell^*) L + \left(\ell^* \wedge \frac{1}{2}\right) \sqrt{\frac{2RL}{\beta}} + 4\sqrt{\left(RL + \sqrt{\frac{2RL}{\beta}}\right) \log\left(\frac{320}{\alpha}\right)} + \frac{32}{3} \log\left(\frac{320}{\alpha}\right) , \quad (159)$$

since $I(\lambda) = \lambda_0 + \delta^* \ell^* \leq R$. Let us now assume that

$$d_{2}\left(\lambda, \mathcal{S}_{0}^{u}[R]\right) = \delta^{*}\sqrt{\ell^{*}(1-\ell^{*})} \geq \sqrt{\frac{\ell^{*}}{(1-\ell^{*})L}} \left(\sqrt{\frac{2R}{\beta}} + \sqrt{\frac{2R}{\beta\left(\ell^{*}\wedge\frac{1}{2}\right)}}\right) + \frac{4}{\ell^{*}\wedge\frac{1}{2}}\sqrt{\left(R + \sqrt{\frac{2R}{\beta L}}\right)\log\left(\frac{320}{\alpha}\right)} + \frac{32\log\left(320/\alpha\right)}{3\left(\ell^{*}\wedge\frac{1}{2}\right)\sqrt{L}}\right) . \quad (160)$$

Then

$$\delta^* (1 - \ell^*) \left(\ell^* \wedge \frac{1}{2}\right) L \ge \left(\ell^* \wedge \frac{1}{2}\right) \sqrt{\frac{2RL}{\beta}} + \sqrt{\frac{2RL}{\beta}} \left(\ell^* \wedge \frac{1}{2}\right) + 4\sqrt{\left(RL + \sqrt{\frac{2RL}{\beta}}\right) \log\left(\frac{320}{\alpha}\right)} + \frac{32}{3} \log\left(\frac{320}{\alpha}\right) \quad .$$

With (159) and using $R \ge \lambda_0 + \delta^*$, this implies

$$\delta^*(1-\ell^*)\left(\ell^*\wedge\frac{1}{2}\right)L \ge Q(\alpha,\beta,\ell^*\wedge(1/2),L) + \sqrt{\frac{2\left(\lambda_0+\delta^*\right)L}{\beta}\left(\ell^*\wedge\frac{1}{2}\right)} - \left(\ell^*\wedge\frac{1}{2}\right)(\lambda_0+\delta^*\ell^*)L ,$$

and

$$Q(\alpha,\beta,\ell^*\wedge(1/2),L) \le \left(\ell^*\wedge\frac{1}{2}\right)(\lambda_0+\delta^*)L - \sqrt{\frac{2(\lambda_0+\delta^*)L}{\beta}\left(\ell^*\wedge\frac{1}{2}\right)}$$

By simply using the exact computation of $E_{\lambda}[N(\tau, \tau + \ell^* \wedge (1/2))]$ and $\operatorname{Var}_{\lambda}[N(\tau, \tau + \ell^* \wedge (1/2))]$ which both equal $(\lambda_0 + \delta^*) (\ell^* \wedge (1/2)) L$, one can notice that this is equivalent to

$$Q(\alpha,\beta,\ell^*\wedge(1/2),L) \le E_{\lambda}[N(\tau,\tau+\ell^*\wedge(1/2)]] - \sqrt{\frac{2\operatorname{Var}_{\lambda}[N(\tau,\tau+\ell^*\wedge(1/2)]]}{\beta}} \quad .$$

Coming back to (158), one finally deduces from the Bienayme-Chebyshev inequality that

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)+}(N)=0\right) \leq P_{\lambda}\left(N(\tau,\tau+\ell^*\wedge(1/2))\right) \leq Q(\alpha,\beta,\ell^*\wedge(1/2),L)\right) + \frac{\beta}{2} \leq \beta .$$

Let us now address the statement of Proposition 28 for $\phi_{3,\alpha}^{u(1)-}$. Let $L \geq 1$ and let us consider again $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau,\tau+\ell^*]}$ in $\mathcal{S}^u_{\delta^*,\dots,\ell^*}[R]$, but with $-\lambda_0 < \delta^* < 0$ here. Notice first that

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)-}(N)=0\right) = P_{\lambda}\left(\min_{t\in[0,1-\ell^*\wedge(1/2)]} N(t,t+\ell^*\wedge(1/2)] \ge b_{N_1,\ell^*\wedge(1/2)}^-(\alpha)\right)$$
$$\le P_{\lambda}\left(N(\tau,\tau+\ell^*\wedge(1/2)] \ge b_{N_1,\ell^*\wedge(1/2)}^-(\alpha)\right) .$$

From Lemma 54, one deduces that

$$b_{N_1,\ell^*\wedge(1/2)}^-(\alpha) \ge N_1(\ell^*\wedge(1/2)) - 4\sqrt{2N_1\log\left(\frac{320}{\alpha}\right)}$$

Setting for ℓ in (0, 1/2],

$$Q'(\alpha,\beta,\ell,L) = \ell I(\lambda)L - 2\ell \sqrt{\frac{I(\lambda)L}{\beta}} - 4\sqrt{I(\lambda)L} + 2\sqrt{\frac{I(\lambda)L}{\beta}} \sqrt{2\log\left(\frac{320}{\alpha}\right)} , \quad (161)$$

with $I(\lambda) = \int_0^1 \lambda(t) dt$ as above, this entails

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)-}(N)=0\right) \leq P_{\lambda}\left(N(\tau,\tau+\ell^*\wedge(1/2))\right) \geq Q'(\alpha,\beta,\ell^*\wedge(1/2),L)\right) + P_{\lambda}\left(N_1 \not\in \left[I(\lambda)L - 2\sqrt{I(\lambda)L/\beta};I(\lambda)L + 2\sqrt{I(\lambda)L/\beta}\right]\right)$$

The Bienayme-Chebyshev inequality therefore leads to

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)-}(N)=0\right) \le P_{\lambda}\left(N(\tau,\tau+\ell^*\wedge(1/2)] \ge Q'(\alpha,\beta,\ell^*\wedge(1/2),L)\right) + \frac{\beta}{2} , \quad (162)$$

with

$$Q'(\alpha, \beta, \ell^* \wedge (1/2), L) \ge \left(\ell^* \wedge \frac{1}{2}\right) (\lambda_0 + \delta^* \ell^*) L - 2\left(\ell^* \wedge \frac{1}{2}\right) \sqrt{\frac{RL}{\beta}} - 4\sqrt{RL + 2\sqrt{\frac{RL}{\beta}}} \sqrt{2\log\left(\frac{320}{\alpha}\right)} . \quad (163)$$

since $I(\lambda) = \lambda_0 + \delta^* \ell^*$ belongs to (0, R]. Let us furthermore assume that

$$d_2\left(\lambda, \mathcal{S}_0^u[R]\right) \ge \sqrt{\frac{\ell^*}{(1-\ell^*)L}} \left(2\sqrt{\frac{R}{\beta}} + \sqrt{\frac{2R}{\beta\left(\ell^* \wedge \frac{1}{2}\right)}} + \frac{4}{\ell^* \wedge \frac{1}{2}}\sqrt{R} + 2\sqrt{\frac{R}{\beta L}}\sqrt{2\log\left(\frac{320}{\alpha}\right)}\right)$$
(164)

Then

$$\begin{split} |\delta^*|(1-\ell^*)\left(\ell^*\wedge\frac{1}{2}\right)L &\geq \sqrt{\frac{2R\left(\ell^*\wedge(1/2)\right)L}{\beta}} + 2\left(\ell^*\wedge\frac{1}{2}\right)\sqrt{\frac{RL}{\beta}} \\ &+ 4\sqrt{RL + 2\sqrt{\frac{RL}{\beta}}}\sqrt{2\log\left(\frac{320}{\alpha}\right)} \ . \end{split}$$

With (163) and the bound $R \ge \lambda_0 + \delta^*$, this yields

$$|\delta^*|(1-\ell^*)\left(\ell^*\wedge\frac{1}{2}\right)L \ge \sqrt{\frac{2(\lambda_0+\delta^*)\left(\ell^*\wedge(1/2)\right)L}{\beta}} + \left(\ell^*\wedge\frac{1}{2}\right)(\lambda_0+\delta^*\ell^*)L - Q'(\alpha,\beta,\ell^*\wedge(1/2),L)$$

From $E_{\lambda}[N(\tau, \tau + \ell^* \land (1/2)]] = \operatorname{Var}_{\lambda}[N(\tau, \tau + \ell^* \land (1/2)]] = (\lambda_0 + \delta^*) (\ell^* \land (1/2)) L$, we then deduce that

$$Q'(\alpha,\beta,\ell^* \wedge (1/2),L) \geq E_{\lambda}[N(\tau,\tau+\ell^* \wedge (1/2)]] + \sqrt{\frac{2\operatorname{Var}_{\lambda}[N(\tau,\tau+\ell^* \wedge (1/2)]]}{\beta}}$$

Inserting this inequality in (162) and using the Bienayme-Chebyshev inequality again, we finally obtain

$$P_{\lambda}\left(\phi_{3,\alpha}^{u(1)-}(N)=0\right) \leq P_{\lambda}\left(N(\tau,\tau+\ell^*\wedge(1/2))\right) \geq Q'(\alpha,\beta,\ell^*\wedge(1/2),L)\right) + \frac{\beta}{2} \leq \beta$$

This concludes the proof for the test $\phi_{3,\alpha}^{u(1)-}$.

Let us finally turn to the test $\phi_{3/4,\alpha}^{u(2)}$. Let $L \ge 1$ and let us consider $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau, \tau + \ell^*]}$ in $\mathcal{S}^u_{\delta^*, \dots, \ell^*}[R]$ satisfying

$$d_{2}(\lambda, \mathcal{S}_{0}^{u}[R]) \geq 2 \max\left(\frac{1}{\sqrt{L}}\left(\sqrt{10CR\log\left(\frac{2.77}{u_{\alpha}}\right)} + 2\sqrt{\frac{R}{\sqrt{\beta}}}\right) + \frac{1}{L^{3/4}}\sqrt{10C\sqrt{\frac{2R}{\beta}}\log\left(\frac{2.77}{u_{\alpha}}\right)} + \frac{1}{L}\sqrt{6C\max\left(\frac{\ell^{*}}{1-\ell^{*}}, \frac{1-\ell^{*}}{\ell^{*}}\right)}\log\left(\frac{2.77}{u_{\alpha}}\right), 8\sqrt{\frac{2R}{\beta L}}\right), \quad (165)$$

C being the constant defined in Lemma 53. In order to prove $P_{\lambda}(\phi_{3/4,\alpha}^{u(2)}(N) = 0) \leq \beta$, noticing first that

$$P_{\lambda}(\phi_{3/4,\alpha}^{u(2)}(N) = 0) \leq \inf_{k \in \{0,\dots,\lceil (1-\ell^*)M\rceil - 1\}} P_{\lambda}\left(T'_{\frac{k}{M},\frac{k}{M}+\ell^*}(N) \leq t'_{N_1,\frac{k}{M},\frac{k}{M}+\ell^*}(1-u_{\alpha})\right)$$

we only need to exhibit some k_{τ} in $\{0, ..., \lceil (1 - \ell^*)M \rceil - 1\}$ such that

$$P_{\lambda}\left(T'_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^*}(N) \le t'_{N_1\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^*}(1-u_{\alpha})\right) \le \beta$$

Let

$$C'(u_{\alpha},\beta,R,\ell^{*},L) = C\left(5R\frac{\log\left(\frac{2.77}{u_{\alpha}}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(\frac{2.77}{u_{\alpha}}\right)}{L^{3/2}} + 3\max\left(\frac{1-\ell^{*}}{\ell^{*}}, \frac{\ell^{*}}{1-\ell^{*}}\right)\frac{\log^{2}\left(\frac{2.77}{u_{\alpha}}\right)}{L^{2}}\right)$$

Since Lemma 53 with the Bienayme-Chebyshev inequality together entail

$$P_{\lambda}\left(t'_{N_{1}\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(1-u_{\alpha})\geq C'(u_{\alpha},\beta,R,\ell^{*},L)\right)\leq\frac{\beta}{2},$$

 k_{τ} only needs to satisfy

$$P_{\lambda}\left(T'_{\frac{k\tau}{M},\frac{k\tau}{M}+\ell^*}(N) \le C'(u_{\alpha},\beta,R,\ell^*,L)\right) \le \frac{\beta}{2} \quad . \tag{166}$$

Using now the simple facts that $(a+b)^2 \ge a^2 + b^2$ and $a+b \le 2\max(a,b)$ for all $a, b \ge 0$, (165) entails

$$d_2^2(\lambda, \mathcal{S}_0^u[R]) \ge 4C \left(5R \frac{\log\left(\frac{2.77}{u_\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}} \frac{\log\left(\frac{2.77}{u_\alpha}\right)}{L^{3/2}} + 3\max\left(\frac{\ell^*}{1-\ell^*}, \frac{1-\ell^*}{\ell^*}\right) \frac{\log^2\left(\frac{2.77}{u_\alpha}\right)}{L^2} \right) + \frac{8R}{L\sqrt{\beta}} + \frac{8\sqrt{2R}}{\sqrt{L\beta}} |\delta^*| \sqrt{\ell^*(1-\ell^*)} .$$

Further using $\sqrt{a+b} \leq \sqrt{a} + \sqrt{b}$ for all $a, b \geq 0$, by definition of $C'(u_{\alpha}, \beta, R, \ell^*, L)$, this yields

$$\frac{\delta^{*2}\ell^*(1-\ell^*)}{4} \ge C'(u_{\alpha},\beta,R,\ell^*,L) + \sqrt{\frac{4R^2}{L^2\beta} + \frac{8R\delta^{*2}\ell^*(1-\ell^*)}{L\beta}} \quad . \tag{167}$$

,

Let us set $k_{\tau} = \lfloor \tau M \rfloor$. Since $0 < \tau < 1 - \ell^*$, k_{τ} actually belongs to $\{0, ..., \lceil (1 - \ell^*)M \rceil - 1\}$, and since $M = \lceil 2/(\ell^*(1 - \ell^*)) \rceil$, $k_{\tau}/M \leq \tau < k_{\tau}/M + (\ell^*(1 - \ell^*)/2)$. Therefore, since we get in particular $k_{\tau}/M \leq \tau < k_{\tau}/M + \ell^*$, using Lemma 52 (equations (288) and (289)) with $x = \lambda_0 k_{\tau}/M$, $y = \lambda_0 \ell^* + \delta(k_{\tau}/M + \ell^* - \tau)$ and $z = \lambda_0 (1 - \ell^* - k_{\tau}/M) + \delta(\tau - k_{\tau}/M)$, we get on the one hand

$$E_{\lambda}\left[T'_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] = \delta^{*2}\frac{\left(\ell^{*}(1-\ell^{*})+k_{\tau}/M-\tau\right)^{2}}{\ell^{*}(1-\ell^{*})} \ge \frac{\delta^{*2}\ell^{*}(1-\ell^{*})}{4}$$

and on the other hand

$$\operatorname{Var}_{\lambda} \left[T_{\frac{k\tau}{M}, \frac{k\tau}{M} + \ell^{*}}^{\prime}(N) \right] = \frac{2}{L^{2}} \left(\lambda_{0} + \delta^{*} \left(1 - \ell^{*} + \left(\tau - \frac{k_{\tau}}{M} \right) \frac{2\ell^{*} - 1}{\ell^{*}(1 - \ell^{*})} \right) \right)^{2} + \frac{4}{L} \delta^{*2} \frac{1 - \ell^{*}}{\ell^{*}} \left(\ell^{*} - \frac{\tau - k_{\tau}/M}{1 - \ell^{*}} \right)^{2} \left(\lambda_{0} + \delta^{*} \left(1 - \ell^{*} + \left(\tau - \frac{k_{\tau}}{M} \right) \frac{2\ell^{*} - 1}{\ell^{*}(1 - \ell^{*})} \right) \right) \quad .$$

Using again the fact that $\tau - k_{\tau}/M < (\ell^*(1-\ell^*)/2)$, we obtain

$$0 \le 1 - \ell^* + \left(\tau - \frac{k_\tau}{M}\right) \frac{2\ell^* - 1}{\ell^*(1 - \ell^*)} \le 1$$

for all ℓ^* in (0, 1), leading to

$$\begin{aligned} \operatorname{Var}_{\lambda}\left[T'_{\frac{k\tau}{M},\frac{k\tau}{M}+\ell^{*}}(N)\right] &\leq \left(\frac{2}{L^{2}}\left(\lambda_{0}+\delta^{*}\right)^{2}+\frac{4}{L}\left(\lambda_{0}+\delta^{*}\right)\delta^{*2}\ell^{*}(1-\ell^{*})\right)\mathbb{1}_{\delta^{*}>0} \\ &+ \left(\frac{2}{L^{2}}\lambda_{0}^{2}+\frac{4}{L}\lambda_{0}\delta^{*2}\ell^{*}(1-\ell^{*})\right)\mathbb{1}_{\delta^{*}<0} \end{aligned}$$

whereby

$$\operatorname{Var}_{\lambda}\left[T'_{\frac{k_{\tau}}{M},\frac{k_{\tau}}{M}+\ell^{*}}(N)\right] \leq \frac{2R^{2}}{L^{2}} + \frac{4R}{L}\delta^{*2}\ell^{*}(1-\ell^{*}) \quad .$$

Combined with these computations, (167) leads to

$$E_{\lambda}\left[T'_{\frac{k\tau}{M},\frac{k\tau}{M}+\ell^{*}}(N)\right] \geq C'(u_{\alpha},\beta,R,\ell^{*},L) + \sqrt{\frac{2\operatorname{Var}_{\lambda}\left[T'_{\frac{k\tau}{M},\frac{k\tau}{M}+\ell^{*}}(N)\right]}{\beta}}$$

The Bienayme-Chebyshev inequality finally allows to obtain (166), which ends the proof.

5.23 Proof of Proposition 30

As for the other Bonferroni type aggregated tests, the control of the first kind error rates of the two tests $\phi_{4,\alpha}^{u(1)}$ and $\phi_{3/4,\alpha}^{u(2)}$ is straightforward using simple union bounds.

(i) Control of the second kind error rate of $\phi_{4,\alpha}^{u(1)}$.

Let us first set λ in $\mathcal{S}^{u}_{\cdot, \cdot, \ell^{*}}[R]$ such that $\lambda = \lambda_{0} + \delta \mathbb{1}_{(\tau, \tau + \ell^{*}]}$ with τ in $(0, 1 - \ell^{*})$, λ_{0} in (0, R], δ in $(0, R - \lambda_{0}]$ or δ in $(-\lambda_{0}, 0)$, and

$$d_{2}(\lambda, \mathcal{S}_{0}^{u}[R]) \geq \sqrt{\frac{\ell^{*}}{(1-\ell^{*})L}} \left(2\sqrt{\frac{R}{\beta}} + \sqrt{\frac{2R}{\beta\left(\ell^{*} \wedge \frac{1}{2}\right)}} + \frac{4}{\ell^{*} \wedge \frac{1}{2}} \sqrt{2\left(R + 2\sqrt{\frac{R}{\beta L}}\right) \log\left(\frac{640}{\alpha}\right)} + \frac{32\log\left(640/\alpha\right)}{3\left(\ell^{*} \wedge \frac{1}{2}\right)\sqrt{L}} \right) .$$

This condition ensures that (160) and (164) both hold, but with α replaced by $\alpha/2$. Then, it suffices to notice that if $\delta \in (0, R - \lambda_0]$,

$$P_{\lambda}\left(\phi_{4,\alpha}^{u(1)}(N)=0\right) \le P_{\lambda}\left(\phi_{3,\alpha/2}^{(1)+}(N)=0\right)$$

and if δ in $(-\lambda_0, 0)$,

$$P_{\lambda}\left(\phi_{4,\alpha}^{u(1)}(N)=0\right) \le P_{\lambda}\left(\phi_{3,\alpha/2}^{(1)-}(N)=0\right)$$
.

Since (160) and (164) hold with $\alpha/2$ instead of α , by using exactly the same arguments as in the proof of Proposition 28, we obtain $P_{\lambda}\left(\phi_{3,\alpha/2}^{(1)+}(N)=0\right) \leq \beta$ when $\delta \in (0, R-\lambda_0]$ on the one hand, and $P_{\lambda}\left(\phi_{3,\alpha/2}^{(1)-}(N)=0\right) \leq \beta$ when δ in $(-\lambda_0, 0)$ on the other hand. In any case, whatever the value of δ in $(-\lambda_0, R-\lambda_0] \setminus \{0\}$, one has

$$P_{\lambda}\left(\phi_{4,\alpha}^{u(1)}(N)=0\right) \leq \beta$$
.

(ii) Control of the second kind error rate of $\phi_{3/4,\alpha}^{u(2)}$.

Let us set now λ in $\mathcal{S}^{u}_{\cdot, \cdot, \ell^{*}}[R]$ such that $\lambda = \lambda_{0} + \delta \mathbb{1}_{(\tau, \tau + \ell^{*}]}$ with τ in $(0, 1 - \ell^{*})$, λ_{0} in (0, R], δ in $(-\lambda_{0}, R - \lambda_{0}] \setminus \{0\}$ as above, but with

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge 2 \max\left(\frac{1}{\sqrt{L}} \left(\sqrt{10CR \log\left(\frac{2.77}{u_\alpha}\right)} + 2\sqrt{\frac{R}{\sqrt{\beta}}}\right) + \frac{1}{L^{3/4}} \sqrt{10C\sqrt{\frac{2R}{\beta}} \log\left(\frac{2.77}{u_\alpha}\right)} + \frac{1}{L} \sqrt{6C \max\left(\frac{\ell^*}{1-\ell^*}, \frac{1-\ell^*}{\ell^*}\right)} \log\left(\frac{2.77}{u_\alpha}\right) , 8\sqrt{\frac{2R}{\beta L}}\right) ,$$

so that (165) holds.

Following the same arguments as in the proof of Proposition 28 (with the only change of δ instead of δ^*), we prove that

$$P_{\lambda}\left(\phi_{3/4,\alpha}^{u(2)}(N)=0\right) \leq \beta .$$

This ends the proof, just taking for instance $C(\alpha, \beta, R, \ell^*)$ as the maximum between

$$\sqrt{\frac{\ell^*}{(1-\ell^*)}} \left(2\sqrt{\frac{R}{\beta}} + \sqrt{\frac{2R}{\beta\left(\ell^* \wedge \frac{1}{2}\right)}} + \frac{4}{\ell^* \wedge \frac{1}{2}} \sqrt{2\left(R + 2\sqrt{\frac{R}{\beta}}\right)\log\left(\frac{640}{\alpha}\right)} + \frac{32\log\left(640/\alpha\right)}{3\left(\ell^* \wedge \frac{1}{2}\right)} \right) = \frac{1}{2} \left(\frac{1}{2} + \frac{1}{$$

and

$$2 \max\left(\left(\sqrt{10CR\log\left(\frac{2.77}{u_{\alpha}}\right)} + 2\sqrt{\frac{R}{\sqrt{\beta}}}\right) + \sqrt{10C\sqrt{\frac{2R}{\beta}}\log\left(\frac{2.77}{u_{\alpha}}\right)} + \sqrt{6C\max\left(\frac{\ell^*}{1-\ell^*}, \frac{1-\ell^*}{\ell^*}\right)\log\left(\frac{2.77}{u_{\alpha}}\right)} + 8\sqrt{\frac{2R}{\beta}}\right).$$

5.24 Proof of Proposition 31

Assume that

$$L > \frac{\left((R - \delta^*) \wedge R\right) \log C_{\alpha,\beta}}{2\delta^{*2}\tau^*(1 - \tau^*)} , \qquad (168)$$

and set

$$r = \sqrt{\frac{((R - \delta^*) \wedge R) \log C_{\alpha,\beta}}{2L}} \quad . \tag{169}$$

The assumption (168) ensures

$$r^2 < \delta^{*2} \tau^* (1 - \tau^*) \le \delta^{*2} / 4$$
, (170)

which enables us to define λ_r for t in (0, 1) by

$$\lambda_r(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau^*, \tau^* + \ell_r]}(t) \text{ with } \lambda_0 = \left((R - \delta^*) \wedge R \right) \text{ and } \ell_r = \frac{1}{2} \left(1 - \frac{\sqrt{\delta^{*2} - 4r^2}}{|\delta^*|} \right) .$$

First, ℓ_r belongs to $(0, 1 - \tau^*)$ for all τ^* in (0, 1). Indeed, if $\tau^* \leq 1/2$ the result is straightforward by definition of ℓ_r and if $\tau^* > 1/2$ the result follows from (170).

Moreover, the definition of ℓ_r implies $\delta^{*2}\ell_r(1-\ell_r) = r^2$ and ensures that λ_r belongs to $(\mathcal{S}^u_{\delta^*,\tau^*,\dots}[R])_r$. Furthermore, we get from (168) that $L > (2\lambda_0 \log C_{\alpha,\beta})/\delta^{*2}$ and then $1 - \lambda_0 \log C_{\alpha,\beta}/(L\delta^{*2}) > 1/2$, which leads to

$$r^{2} < \frac{\lambda_{0} \log C_{\alpha,\beta}}{L} \left(1 - \frac{\lambda_{0} \log C_{\alpha,\beta}}{L\delta^{*2}} \right) = \frac{\delta^{*2}}{4} \left(1 - \left(1 - \frac{2\lambda_{0} \log C_{\alpha,\beta}}{L\delta^{*2}} \right)^{2} \right) ,$$

hence $\delta^{*2}\ell_r < \lambda_0 \log C_{\alpha,\beta}/L$. We then obtain from Lemma 1 and Lemma 44

$$E_{\lambda_0}\left[\left(\frac{dP_{\lambda_r}}{dP_{\lambda_0}}\right)^2(N)\right] = \exp\left(\frac{L\delta^{*2}\ell_r}{\lambda_0}\right) < C_{\alpha,\beta} \ .$$

Lemmas 42 and 43 then entail $\rho\left((\mathcal{S}^{u}_{\delta^{*},\tau^{*},\dots}[R])_{r}\right) \geq \beta$ and $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}^{u}_{\delta^{*},\tau^{*},\dots}[R]) \geq r$.

5.25 Proof of Proposition 32

The control of the first kind error rate is straightforward, and even more strong by using the same conditioning trick as in the proof of Proposition 27: in fact, for every λ_0 in $\mathcal{S}_0^u[R]$ and n in \mathbb{N} , $E_{\lambda_0}\left[\phi_{5,\alpha}^u(N)\middle|N_1=n\right] = P_{\lambda_0}\left(\phi_{5,\alpha}^u(N)=1\middle|N_1=n\right) \leq \alpha$, so

$$\forall \lambda_0 \in \mathcal{S}_0^u[R], \ P_{\lambda_0}\left(\phi_{5,\alpha}^u(N) = 1\right) = E_{\lambda_0}\left[P_{\lambda_0}\left(\phi_{5,\alpha}^u(N) \middle| N_1\right)\right] \le \alpha \ .$$

Let us turn to the control of the second kind error rate of $\phi_{5,\alpha}^u$. Set λ in $\mathcal{S}^u_{\delta^*,\tau^*,\dots}[R]$ such that $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau^*,\tau^*+\ell]}$ with λ_0 in $(-\delta^* \vee 0, (R-\delta^*) \wedge R]$ and ℓ in $(0, 1-\tau^*)$, and satisfying

$$d_2(\lambda, \mathcal{S}_0[R]) \ge \frac{2}{\sqrt{L}} \max\left(\sqrt{|\delta^*|Q(2R, \delta^*, \alpha)}, \ 2\sqrt{\frac{2R}{\beta}}, \ \frac{|\delta^*|}{2\sqrt{2\beta R}}\right) , \qquad (171)$$

where $Q(2R, \delta^*, \alpha)$ is the quantile upper bound defined by Lemma 55, and which does not depend on *L*. The condition (171) ensures that $|\delta^*|\sqrt{\ell(1-\ell)} \ge |\delta^*|/\sqrt{2\beta RL}$, that is, using the fact that $\ell(1-\ell) \le 1/4$,

$$L \ge \frac{2}{\beta R} \quad . \tag{172}$$

Setting $I(\lambda) = \int_0^1 \lambda(t) dt$ as in the proof of Proposition 28, with $I(\lambda) \leq R$ (and therefore obviously $2R - I(\lambda) \geq R$), since (172) also entails $L \geq 2R/(\beta R^2)$ we obtain

$$L \ge \frac{2I(\lambda)}{(2R - I(\lambda))^2\beta} \quad . \tag{173}$$

Notice now that (173) and the Bienayme-Chebyshev inequality yield

$$P_{\lambda}(N_1 > 2RL) = P_{\lambda} \left(N_1 - I(\lambda)L > L(2R - I(\lambda)) \right)$$
$$\leq P_{\lambda} \left(N_1 - I(\lambda)L > \sqrt{\frac{2I(\lambda)L}{\beta}} \right) \leq \frac{\beta}{2} .$$

This leads to

$$P_{\lambda}\left(\phi_{5,\alpha}^{u}(N)=0\right) \leq P_{\lambda}\left(\sup_{\ell'\in(0,1-\tau^{*})}S_{\delta^{*},\tau^{*},\tau^{*}+\ell'}(N)\leq s_{N_{1},\delta^{*},\tau^{*},L}^{\prime+}(1-\alpha), \ N_{1}\leq 2RL\right) + \frac{\beta}{2} ,$$

and Lemma 55 allows to write that

$$P_{\lambda}\left(\phi_{5,\alpha}^{u}(N)=0\right) \leq P_{\lambda}\left(\sup_{\ell'\in(0,1-\tau^{*})}S_{\delta^{*},\tau^{*},\tau^{*}+\ell'}^{\prime}(N)\leq Q(2R,\delta^{*},\alpha)\right) + \frac{\beta}{2}$$
$$\leq P_{\lambda}\left(S_{\delta^{*},\tau^{*},\tau^{*}+\ell}^{\prime}(N)\leq Q(2R,\delta^{*},\alpha)\right) + \frac{\beta}{2}.$$

Moreover, the assumption (171) implies

$$|\delta^*|\sqrt{\ell(1-\ell)} \ge \frac{2}{\sqrt{L}} \max\left(\sqrt{|\delta^*|Q(2R,\delta^*,\alpha)}, 2\sqrt{\frac{2R}{\beta}}\right) ,$$

which entails

$$\frac{|\delta^*|}{2}\ell(1-\ell)L \ge 2\max\left(Q(2R,\delta^*,\alpha), \sqrt{\frac{2R\ell(1-\ell)L}{\beta}}\right) ,$$

hence

$$\frac{|\delta^*|}{2}\ell(1-\ell)L \ge 2\max\left(Q(2R,\delta^*,\alpha), \sqrt{\frac{2(\lambda_0+\delta^*(1-\ell))\ell(1-\ell)L}{\beta}}\right).$$

Noticing that $E_{\lambda}[S'_{\delta^*,\tau^*,\tau^*+\ell}(N)] = |\delta^*|\ell(1-\ell)L/2$ and $\operatorname{Var}_{\lambda}(S'_{\delta^*,\tau^*,\tau^*+\ell}(N)) = (\lambda_0 + \delta^*(1-\ell))\ell(1-\ell)L$, we get

$$E_{\lambda}[S'_{\delta^*,\tau^*,\tau^*+\ell}(N)] \ge Q(2R,\delta^*,\alpha) + \sqrt{\frac{2\operatorname{Var}(S'_{\delta^*,\tau^*,\tau^*+\ell}(N))}{\beta}} \quad . \tag{174}$$

Therefore,

$$P_{\lambda}\left(\phi_{5,\alpha}^{u}(N)=0\right) \leq P_{\lambda}\left(S_{\delta^{*},\tau^{*},\tau^{*}+\ell}(N)\leq Q(2R,\delta^{*},\alpha)\right) + \frac{\beta}{2}$$
$$\leq P_{\lambda}\left(S_{\delta^{*},\tau^{*},\tau^{*}+\ell}(N) - E_{\lambda}[S_{\delta^{*},\tau^{*},\tau^{*}+\ell}(N)] \leq -\sqrt{\frac{2\mathrm{Var}(S_{\delta^{*},\tau^{*},\tau^{*}+\ell}(N))}{\beta}}\right) + \frac{\beta}{2}$$
$$\leq \beta \ ,$$

with a last line simply following from the Bienayme-Chebyshev inequality.

5.26 Proof of Proposition 33

Assume that $L \geq 3$ and $\alpha + \beta < 1/2$, and set $\lambda_0 = R/2$. As in the proof of Proposition 12, we consider $C'_{\alpha,\beta} = 4(1 - \alpha - \beta)^2$, $K_{\alpha,\beta,L} = \lceil (\log_2 L)/C'_{\alpha,\beta} \rceil$, and for k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, $\lambda_k = \lambda_0 + \delta_k \mathbb{1}_{(\tau^*,\tau^*+\ell_k]}$ with $\ell_k = (1 - \tau^*)/2^k$ and $\delta_k = (\lambda_0 \log \log L/(\ell_k L))^{1/2}$. Then, for all k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, noticing that $\ell_k < 1 - \tau^*$, we get $d_2(\lambda_k, \mathcal{S}_0[R]) > \sqrt{\lambda_0 \tau^* \log \log L/L}$. Furthermore, assuming that

$$\frac{\log \log L}{L^{1+1/C'_{\alpha,\beta}}} \le \frac{R(1-\tau^*)}{4} , \qquad (175)$$

one obtains that λ_k belongs to $\mathcal{S}^u_{;\tau^*,...}[R]$.

Recall also that for all k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, P_{λ_k} denotes the distribution of a Poisson process with intensity λ_k with respect to the measure Λ , and consider κ , a random variable with uniform distribution on $\{1, \ldots, K_{\alpha,\beta,L}\}$, which allows to define the probability distribution μ of λ_{κ} . From Lemma 43, we know that it is enough to prove $E_{\lambda_0}[(dP_{\mu}/dP_{\lambda_0})^2] \leq 1 + C'_{\alpha,\beta}$ to conclude that $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}^u_{,\tau^*,\ldots}[R]) \geq \sqrt{R\tau^*\log\log L/(2L)}$. The same calculation as in the proof of Proposition 12 (see (108)) gives for η such that $0 < \eta < 1 - 1/\sqrt{2}$,

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] \le C'_{\alpha,\beta}\log 2 + \frac{2C'_{\alpha,\beta}\log 2}{\log L}\left(\log L\right)^{\eta+\frac{1}{\sqrt{2}}} + \exp\left(\frac{\log\log L}{2^{(\log L)^{\eta/2}}}\right) \quad .$$

If we assume now that

$$\exp\left(\frac{\log\log L}{2^{(\log L)^{\eta/2}}}\right) + \frac{2C'_{\alpha,\beta}\log 2}{(\log L)^{1-\eta-\frac{1}{\sqrt{2}}}} \le 1 + (1-\log 2)C'_{\alpha,\beta} \quad , \tag{176}$$

we finally obtain the expected result

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] \le 1 + C'_{\alpha,\beta}$$

To end the proof, it remains to notice that there exists $L_0(\alpha, \beta, R, \tau^*) \geq 3$ such that for all $L \geq L_0(\alpha, \beta, R, \tau^*)$, both assumptions (175) and (176) hold.

5.27 Proof of Proposition 34

As for all our Bonferroni type aggregated tests, the control of the first kind error rates of the two tests $\phi_{6,\alpha}^{u(1)}$ and $\phi_{6,\alpha}^{u(2)}$ is straightforward using simple union bounds and the conditioning trick of the above proofs for upper bounds.

(i) Control of the second kind error rate of $\phi_{6,\alpha}^{u(1)}$

Let λ in $\mathcal{S}^{u}_{,\tau^*,\dots}[R]$ be such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau^*,\tau^*+\ell]}$ with λ_0 in (0, R], δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$ and ℓ in $(0, 1 - \tau^*)$ and assume that

$$d_{2}(\lambda, \mathcal{S}_{0}^{u}[R]) \geq 2 \max\left(\sqrt{\frac{R}{3}} \sqrt{\frac{1+\tau^{*}}{\tau^{*}}} \sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}}, \frac{1+\tau^{*}}{\tau^{*}} \left(\sqrt{\frac{2\log\left(2/u_{\alpha}\right)}{L}} \sqrt{R} + \sqrt{\frac{2R}{\beta L}} + \sqrt{\frac{2R}{\beta L}}\right), \frac{\sqrt{1-\tau^{*}R}}{\sqrt{2L}}\right) . \quad (177)$$

Let us prove the inequality $P_{\lambda}(\phi_{6,\alpha}^{u(1)}(N) = 0) \leq \beta$. Applying Lemma 44, we get for all k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ and $\ell_{\tau^*,k} = (1 - \tau^*) 2^{-k}$

$$E_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*, k}] - \ell_{\tau^*, k} N_1] = \delta(\ell_{\tau^*, k} \wedge \ell)(1 - \ell_{\tau^*, k} \vee \ell)L \quad , \tag{178}$$

and

$$\operatorname{Var}_{\lambda}\left[N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k}]-\ell_{\tau^{*},k}N_{1}\right] = \begin{cases} (\lambda_{0}(1-\ell_{\tau^{*},k})+\delta(1-2\ell_{\tau^{*},k}+\ell_{\tau^{*},k}\ell))\ell_{\tau^{*},k}L & \text{if } \ell_{\tau^{*},k} \leq \ell \\ (\lambda_{0}\ell_{\tau^{*},k}+\delta\ell(1-\ell_{\tau^{*},k}))(1-\ell_{\tau^{*},k})L & \text{if } \ell_{\tau^{*},k} \geq \ell \end{cases}$$
(179)

Assume first that δ belongs to $(0, R - \lambda_0]$. Noticing that

$$P_{\lambda}(\phi_{6,\alpha}^{u(1)}(N) = 0) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(N(\tau^*, \tau^* + \ell_{\tau^*,k}] - \ell_{\tau^*,k}N_1 \leq \bar{b}_{N_1,\ell_{\tau^*,k}}\left(1 - u_{\alpha}/2\right)\right) ,$$

one can see it is enough to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(N(\tau^*,\tau^*+\ell_{\tau^*,k}]-\ell_{\tau^*,k}N_1\leq \bar{b}_{N_1,\ell_{\tau^*,k}}\left(1-u_{\alpha}/2\right)\right)\leq\beta \ .$$

Under the condition (177), we have $d_2^2(\lambda, \mathcal{S}_0^u[R]) \geq 2R^2(1-\tau^*)/L$ which ensures the inequality $\ell(1-\ell) > 2(1-\tau^*)/L > (1-\tau^*)2^{-\lfloor \log_2 L \rfloor}$ and then

$$1 - \tau^* > \ell > \frac{1 - \tau^*}{2^{\lfloor \log_2 L \rfloor}} .$$
 (180)

From (180), we deduce the existence of k_{ℓ} in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying $(1 - \tau^*)2^{-k_{\ell}} \leq \ell < (1 - \tau^*)2^{-k_{\ell}+1}$, that is

$$\ell_{\tau^*,k_{\ell}} \le \ell < \ell_{\tau^*,k_{\ell}-1} \quad . \tag{181}$$

•

Then

$$\frac{\ell_{\tau^*,k_\ell}}{1-\ell_{\tau^*,k_\ell}} \le \frac{\ell}{1-\ell} < \frac{\ell_{\tau^*,k_\ell-1}}{1-\ell_{\tau^*,k_\ell-1}} \ ,$$

and

$$\frac{\ell_{\tau^*,k_{\ell}}}{1-\ell_{\tau^*,k_{\ell}}} = \underbrace{\frac{\ell_{\tau^*,k_{\ell}-1}}{1-\ell_{\tau^*,k_{\ell}-1}}}_{>\frac{\ell}{1-\ell}} \frac{1-\ell_{\tau^*,k_{\ell}-1}}{1-\ell_{\tau^*,k_{\ell}}} \underbrace{\frac{\ell_{\tau^*,k_{\ell}}}{\ell_{\tau^*,k_{\ell}-1}}}_{=\frac{1}{2}} > \frac{\ell}{2(1-\ell)} \frac{1-\ell_{\tau^*,k_{\ell}-1}}{1-\ell_{\tau^*,k_{\ell}}}$$

But

$$\frac{1-\ell_{\tau^*,k_\ell-1}}{1-\ell_{\tau^*,k_\ell}} = 1 - \frac{1-\tau^*}{2^{k_\ell} - (1-\tau^*)} \ge 2\frac{\tau^*}{1+\tau^*} \ ,$$

so we finally obtain

$$\frac{\ell_{\tau^*,k_{\ell}}}{1-\ell_{\tau^*,k_{\ell}}} > \frac{\tau^*}{1+\tau^*} \frac{\ell}{1-\ell} \quad . \tag{182}$$

The condition (177) then gives on the one hand

$$\delta\sqrt{\ell(1-\ell)} \ge \frac{2}{\sqrt{3}}\sqrt{R}\sqrt{\frac{1+\tau^*}{\tau^*}}\sqrt{\frac{\log\left(2/u_\alpha\right)}{L}} ,$$

and using the fact that $\delta < R$

$$\delta \ell (1-\ell) \frac{\tau^*}{1+\tau^*} \ge \frac{4 \log (2/u_{\alpha})}{3L} ,$$

which entails with (181) and (182)

$$\delta \ell_{\tau^*,k_\ell}(1-\ell) \ge \frac{4\log(2/u_\alpha)}{3L}$$
 (183)

On the other hand, (177) yields

$$\delta\sqrt{\ell(1-\ell)} \ge 2\frac{1+\tau^*}{\tau^*} \left(\sqrt{\frac{2\log\left(2/u_\alpha\right)}{L}}\sqrt{R+\sqrt{\frac{2R}{\beta L}}} + \sqrt{\frac{2R}{\beta L}}\right) ,$$

which entails with (181) and (182)

$$\delta(1-\ell)\sqrt{\frac{\ell_{\tau^*,k_\ell}}{1-\ell_{\tau^*,k_\ell}}} \ge 2\left(\sqrt{\frac{2\log\left(2/u_\alpha\right)}{L}}\sqrt{R+\sqrt{\frac{2R}{\beta L}}} + \sqrt{\frac{2R}{\beta L}}\right) \quad,$$

whereby

$$\delta\ell_{\tau^*,k_\ell}(1-\ell) \ge 2\sqrt{\ell_{\tau^*,k_\ell}(1-\ell_{\tau^*,k_\ell})} \left(\sqrt{\frac{2\log\left(2/u_\alpha\right)}{L}}\sqrt{R+\sqrt{\frac{2R}{\beta L}}} + \sqrt{\frac{2R}{\beta L}}\right) \quad . \tag{184}$$

Thus, with (183) and (184), the condition (177) leads to

$$\delta\ell_{\tau^*,k_\ell}(1-\ell) \ge \max\left(\frac{4\log\left(2/u_\alpha\right)}{3L} , \ 2\sqrt{\ell_{\tau^*,k_\ell}(1-\ell_{\tau^*,k_\ell})} \left(\sqrt{\frac{2\log\left(2/u_\alpha\right)}{L}}\sqrt{R+\sqrt{\frac{2R}{\beta L}}} + \sqrt{\frac{2R}{\beta L}}\right)\right)$$

Using the fact that $2 \max(a, b) \ge a + b$ for all $a, b \ge 0$, we get

$$\delta\ell_{\tau^*,k_\ell}(1-\ell)L \ge \frac{2}{3}\log\left(\frac{2}{u_\alpha}\right) + \sqrt{\ell_{\tau^*,k_\ell}(1-\ell_{\tau^*,k_\ell})} \left(\sqrt{2\log\left(\frac{2}{u_\alpha}\right)}\sqrt{LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}} + \sqrt{\frac{2RL}{\beta}}}\right) , \quad (185)$$

where we recall that $I(\lambda)$ stands for $\int_0^1 \lambda(t) dt$. Moreover, with (181), the equations (178) and (179) lead to $E_{\lambda} \left[N(\tau^*, \tau^* + \ell_{\tau^*, k_{\ell}}] - \ell_{\tau^*, k_{\ell}} N_1 \right] = \delta \ell_{\tau^*, k_{\ell}} (1 - \ell) L$ and $\operatorname{Var}_{\lambda} \left[N(\tau^*, \tau^* + \ell_{\tau^*, k_{\ell}}] - \ell_{\tau^*, k_{\ell}} N_1 \right] = (\lambda_0 (1 - \ell_{\tau^*, k}) + \delta (1 - 2\ell_{\tau^*, k} + \ell_{\tau^*, k_{\ell}})) \ell_{\tau^*, k} L \leq \ell_{\tau^*, k_{\ell}} (1 - \ell_{\tau^*, k_{\ell}}) RL$. So (185) entails

$$E_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*, k_{\ell}}] - \ell_{\tau^*, k_{\ell}} N_1]$$

$$\geq \frac{2}{3} \log\left(\frac{2}{u_{\alpha}}\right) + \sqrt{\ell_{\tau^*, k_{\ell}}(1 - \ell_{\tau^*, k_{\ell}})} \sqrt{2 \log\left(\frac{2}{u_{\alpha}}\right)} \sqrt{LI(\lambda)} + \sqrt{\frac{2LI(\lambda)}{\beta}}$$

$$+ \sqrt{2 \operatorname{Var}_{\lambda} \left[N(\tau^*, \tau^* + \ell_{\tau^*, k_{\ell}}] - \ell_{\tau^*, k_{\ell}} N_1\right]/\beta} \quad . \quad (186)$$

The total probability formula then ensures that

$$\begin{aligned} P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}]-\ell_{\tau^{*},k_{\ell}}N_{1} \leq \bar{b}_{N_{1},\ell_{\tau^{*},k_{\ell}}}\left(1-u_{\alpha}/2\right)\right) \\ \leq P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}]-\ell_{\tau^{*},k_{\ell}}N_{1} \leq E_{\lambda}[N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}]-\ell_{\tau^{*},k_{\ell}}N_{1}] \\ -\sqrt{2}\mathrm{Var}_{\lambda}\left[N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}]-\ell_{\tau^{*},k_{\ell}}N_{1}\right]/\beta}\right) + P_{\lambda}\left(\bar{b}_{N_{1},\ell_{\tau^{*},k_{\ell}}}\left(1-u_{\alpha}/2\right) > \frac{2}{3}\log\left(\frac{2}{u_{\alpha}}\right) \\ +\sqrt{\ell_{\tau^{*},k_{\ell}}(1-\ell_{\tau^{*},k_{\ell}})}\sqrt{2\log\left(\frac{2}{u_{\alpha}}\right)}\sqrt{LI(\lambda)}+\sqrt{\frac{2LI(\lambda)}{\beta}}\right) \,. \end{aligned}$$

From the Bienayme-Chebyshev inequality, we deduce on the one hand that the first right hand side term is upper bounded by $\beta/2$, and on the other hand that

$$P_{\lambda}\left(N_1 \ge LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}\right) \le \frac{\beta}{2}$$
.

This last inequality, combined with Lemma 56, which follows from a simple application of Bennett's inequality, leads to

$$P_{\lambda}\left(\bar{b}_{N_{1},\ell_{\tau^{*},k_{\ell}}}\left(1-u_{\alpha}/2\right) > \frac{2}{3}\log\left(\frac{2}{u_{\alpha}}\right) + \sqrt{\ell_{\tau^{*},k_{\ell}}(1-\ell_{\tau^{*},k_{\ell}})}\sqrt{2\log\left(\frac{2}{u_{\alpha}}\right)}\sqrt{LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}}\right) \le \frac{\beta}{2}$$

We therefore conclude that

$$P_{\lambda}\left(N(\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}]-\ell_{\tau^*,k_{\ell}}N_1\leq \bar{b}_{N_1,\ell_{\tau^*,k_{\ell}}}\left(1-u_{\alpha}/2\right)\right)\leq\beta \ ,$$

so, as expected, $P_{\lambda}(\phi_{6,\alpha}^{u(1)}(N) = 0) \leq \beta$.

Assume now that δ belongs to $(-\lambda_0, 0)$ and notice that

$$P_{\lambda}(\phi_{6,\alpha}^{u(1)}(N) = 0) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(N(\tau^*, \tau^* + \ell_{\tau^*,k}] - \ell_{\tau^*,k}N_1 \geq \bar{b}_{N_1,\ell_{\tau^*,k}}\left(u_{\alpha}/2\right)\right)$$

One can see it is enough to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k}]-\ell_{\tau^{*},k}N_{1}\geq\bar{b}_{N_{1},\ell_{\tau^{*},k}}\left(u_{\alpha}/2\right)\right)\leq\beta$$

The same choice of k_{ℓ} as above leads, with (178) and (179), to $E_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*, k_{\ell}}] - \ell_{\tau^*, k_{\ell}}N_1] = -|\delta|\ell_{\tau^*, k_{\ell}}(1-\ell)L$ and $\operatorname{Var}_{\lambda}[N(\tau^*, \tau^* + \ell_{\tau^*, k_{\ell}}] - \ell_{\tau^*, k_{\ell}}N_1] \leq \ell_{\tau^*, k_{\ell}}(1-\ell_{\tau^*, k_{\ell}})\lambda_0L \leq \ell_{\tau^*, k_{\ell}}(1-\ell_{\tau^*, k_{\ell}})RL$. With very similar arguments and calculations (mainly replacing δ by $|\delta|$ and using the lower bound for $\bar{b}_{N_1, \ell_{\tau^*, k_{\ell}}}(u_{\alpha}/2)$ of Lemma 56) as above, we also conclude that condition (177) implies

$$P_{\lambda}\left(N(\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}]-\ell_{\tau^{*},k_{\ell}}N_{1}\geq\bar{b}_{N_{1},\ell_{\tau^{*},k_{\ell}}}(u_{\alpha}/2)\right)\leq\beta ,$$

and $P_{\lambda}(\phi_{6,\alpha}^{u(1)}(N) = 0) \leq \beta$. Since $\log (2/u_{\alpha}) = \log(2\lfloor \log_2 L \rfloor / \alpha)$ and $L \geq 3$, there exists $C(\alpha, \beta, R, \tau^*) > 0$ such that

$$2 \max\left(\sqrt{\frac{R}{3}}\sqrt{\frac{1+\tau^*}{\tau^*}}\sqrt{\frac{\log\left(2/u_\alpha\right)}{L}} , \frac{1+\tau^*}{\tau^*}\left(\sqrt{\frac{2\log\left(2/u_\alpha\right)}{L}}\sqrt{R}+\sqrt{\frac{2R}{\beta L}}+\sqrt{\frac{2R}{\beta L}}\right), \frac{\sqrt{1-\tau^*}R}{\sqrt{2L}}\right) \le C(\alpha,\beta,R,\tau^*)\sqrt{\frac{\log\log L}{L}} ,$$

which allows to conclude the proof.

(ii) Control of the second kind error rate of $\phi_{6,\alpha}^{u(2)}$.

Let λ in $\mathcal{S}^{u}_{;\tau^{*},\cdots}[R]$ such that $\lambda = \lambda_{0} + \delta \mathbb{1}_{(\tau^{*},\tau^{*}+\ell]}$ with λ_{0} in (0,R], δ in $(-\lambda_{0}, R - \lambda_{0}] \setminus \{0\}$ and ℓ in $(0, 1 - \tau^{*})$ and assume that

$$d_{2}(\lambda, \mathcal{S}_{0}^{u}[R]) \geq \sqrt{\frac{1+\tau^{*}}{\tau^{*}}} \max\left(\sqrt{2C}\left(\sqrt{5R}\sqrt{\frac{\log\left(2.77/u_{\alpha}\right)}{L}} + \sqrt{5}\left(\frac{2R}{\beta}\right)^{1/4}\frac{\sqrt{\log\left(2.77/u_{\alpha}\right)}}{L^{3/4}}\right) + \frac{\sqrt{3}\log\left(2.77/u_{\alpha}\right)}{2\sqrt{2L}\log\log L}\right) + \frac{2\sqrt{R}}{\sqrt{L\sqrt{\beta}}}, \ 4\sqrt{\frac{1+\tau^{*}}{\tau^{*}}}\sqrt{\frac{2R}{L\beta}}, \ 4R\sqrt{\frac{\tau^{*}}{1+\tau^{*}}}\sqrt{\frac{\log\log L}{L}}\right), \ (187)$$

where C is the constant defined in Lemma 53, and let us prove $P_{\lambda}(\phi_{6,\alpha}^{u(2)}(N) = 0) \leq \beta$. Noticing that

$$P_{\lambda}(\phi_{6,\alpha}^{u(2)}(N) = 0) \leq \inf_{k \in \{1,\dots,\lfloor \log_2 L \rfloor\}} P_{\lambda}\left(T'_{\tau^*,\tau^* + \ell_{\tau^*,k}}(N) \leq t'_{N_1,\tau^*,\tau^* + \ell_{\tau^*,k}}\left(1 - u_{\alpha}\right)\right) ,$$

one only needs to exhibit some k in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying

$$P_{\lambda}\left(T'_{\tau^{*},\tau^{*}+\ell_{\tau^{*},k}}(N) \leq t'_{N_{1},\tau^{*},\tau^{*}+\ell_{\tau^{*},k}}(1-u_{\alpha})\right) \leq \beta ,$$

in order to prove the result.

Under the condition (187), we first obtain $d_2^2(\lambda, \mathcal{S}_0^u[R]) \ge 16R^2(\log \log L)/L$ which ensures $\ell(1-\ell) > 16(\log \log L)/L$ as well as

$$\ell > \frac{16\log\log L}{L} \text{ and } 1 - \ell > \frac{16\log\log L}{L}$$
 (188)

Moreover, since $16(\log \log L)/L > 2^{-\lfloor \log_2 L \rfloor} > (1 - \tau^*)2^{-\lfloor \log_2 L \rfloor}$, we obtain

$$1 - \tau^* > \ell > \frac{1 - \tau^*}{2^{\lfloor \log_2 L \rfloor}}$$
 (189)

From (189), as in the above part (i), we deduce that there exists k_{ℓ} in $\{1, \ldots, \lfloor \log_2 L \rfloor\}$ satisfying $(1 - \tau^*)2^{-k_{\ell}} \leq \ell < (1 - \tau^*)2^{-k_{\ell}+1}$, that is

$$\ell_{\tau^*,k_{\ell}} \le \ell < \ell_{\tau^*,k_{\ell}-1} \quad . \tag{190}$$

As above again, this leads to the same inequality as (182), i.e.

$$\frac{\ell_{\tau^*,k_{\ell}}}{1-\ell_{\tau^*,k_{\ell}}} > \frac{\tau^*}{1+\tau^*} \frac{\ell}{1-\ell} \quad . \tag{191}$$

Applying Lemma 52, we get with (190)

$$E_{\lambda}[T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N)] = \delta^2 (1-\ell)^2 \frac{\ell_{\tau^*,k_{\ell}}}{1-\ell_{\tau^*,k_{\ell}}} , \qquad (192)$$

and

$$\operatorname{Var}_{\lambda}\left[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}}'(N)\right] = \frac{2}{L^{2}} \left(\lambda_{0}+\delta\left(1-(1-\ell)\frac{\ell_{\tau^{*},k_{\ell}}}{1-\ell_{\tau^{*},k_{\ell}}}\right)\right)^{2} + \frac{4}{L}\delta^{2}(1-\ell)^{2}\frac{\ell_{\tau^{*},k_{\ell}}}{1-\ell_{\tau^{*},k_{\ell}}} \left(\lambda_{0}+\delta\left(1-(1-\ell)\frac{\ell_{\tau^{*},k_{\ell}}}{1-\ell_{\tau^{*},k_{\ell}}}\right)\right) \quad . \tag{193}$$

Using (191),

$$E_{\lambda}[T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N)] > \delta^2 \ell (1-\ell) \frac{\tau^*}{1+\tau^*} , \qquad (194)$$

and with (190) we obtain

$$\operatorname{Var}_{\lambda}\left[T_{\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}}'(N)\right] \leq \frac{2R^{2}}{L^{2}} + \frac{4R}{L}\delta^{2}\ell(1-\ell) \quad .$$
(195)

On the one hand, using $a^2 + b^2 \le (a + b)^2$ for $a, b \ge 0$, the condition (187) ensures that

$$\delta^{2}\ell(1-\ell) \geq 2\frac{1+\tau^{*}}{\tau^{*}} \left(C\left(5R\frac{\log\left(2.77/u_{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(2.77/u_{\alpha}\right)}{L^{3/2}} + \frac{3\log^{2}\left(2.77/u_{\alpha}\right)}{8L\log\log L} \right) + \frac{2R}{L\sqrt{\beta}} \right) + \frac{2R}{L\sqrt{\beta}} \right) + \frac{1}{2} \left(\frac{1}{2} + \frac{1$$

and on the other hand

$$\delta^2 \ell(1-\ell) \ge 2 \frac{1+\tau^*}{\tau^*} \left(2\sqrt{\frac{2R}{L\beta}} |\delta| \sqrt{\ell(1-\ell)} \right) ,$$

hence

$$\begin{split} \delta^2 \ell(1-\ell) &\geq 2 \frac{1+\tau^*}{\tau^*} \max\left(C \left(5R \frac{\log\left(2.77/u_{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}} \frac{\log\left(2.77/u_{\alpha}\right)}{L^{3/2}} + \frac{3\log^2\left(2.77/u_{\alpha}\right)}{8L\log\log L} \right) \right. \\ &+ \frac{2R}{L\sqrt{\beta}} \;,\; 2\sqrt{\frac{2R}{L\beta}} |\delta| \sqrt{\ell(1-\ell)} \right) \;. \end{split}$$

Finally, with the inequality $a + b \le 2 \max(a, b)$, we get

$$\begin{split} \delta^2 \ell(1-\ell) \frac{\tau^*}{1+\tau^*} &\geq C \left(5R \frac{\log\left(2.77/u_\alpha\right)}{L} + 5\sqrt{\frac{2R}{\beta}} \frac{\log\left(2.77/u_\alpha\right)}{L^{3/2}} + \frac{3\log^2\left(2.77/u_\alpha\right)}{8L\log\log L} \right) + \frac{2R}{L\sqrt{\beta}} \\ &+ 2\sqrt{\frac{2R}{L\beta}} |\delta| \sqrt{\ell(1-\ell)} \ , \end{split}$$

that is, with (194), (195) and using $\sqrt{a+b} \le \sqrt{a} + \sqrt{b}$,

$$E_{\lambda}[T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N)] \ge C\left(5R\frac{\log\left(2.77/u_{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(2.77/u_{\alpha}\right)}{L^{3/2}} + \frac{3\log^2\left(2.77/u_{\alpha}\right)}{8L\log\log L}\right) + \sqrt{2\mathrm{Var}_{\lambda}\left[T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N)\right]/\beta} \quad (196)$$

Furthermore, (188) and (190) lead to

$$\frac{1}{\ell_{\tau^*,k_\ell}} < \frac{L}{8\log\log L} \text{ and } \frac{1}{1-\ell_{\tau^*,k_\ell}} < \frac{L}{16\log\log L}$$

and therefore

$$\max\left(\frac{1-\ell_{\tau^*,k_\ell}}{\ell_{\tau^*,k_\ell}}, \frac{\ell_{\tau^*,k_\ell}}{1-\ell_{\tau^*,k_\ell}}\right) < \frac{L}{8\log\log L} .$$
(197)

We set now

$$\begin{aligned} Q(\alpha, \beta, L, R, \ell_{\tau^*, k_{\ell}}) &= C \Biggl(5R \frac{\log\left(2.77/u_{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}} \frac{\log\left(2.77/u_{\alpha}\right)}{L^{3/2}} \\ &+ 3 \max\left(\frac{\ell_{\tau^*, k_{\ell}}}{1 - \ell_{\tau^*, k_{\ell}}}, \frac{1 - \ell_{\tau^*, k_{\ell}}}{\ell_{\tau^*, k_{\ell}}}\right) \frac{\log^2\left(2.77/u_{\alpha}\right)}{L^2} \Biggr) \ , \end{aligned}$$

and with (197), the condition (196) ensures that

$$E_{\lambda}[T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N)] \ge Q(\alpha,\beta,L,R,k_{\ell}) + \sqrt{2\operatorname{Var}_{\lambda}\left[T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N)\right]/\beta} \quad .$$
(198)

The Bienayme-Chebyshev inequality leads to

$$P_{\lambda}\left(N_1 \ge LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}\right) \le \frac{\beta}{2}$$
,

and combined with Lemma 53 and the fact that $I(\lambda) \leq R$, we obtain

$$P_{\lambda}\left(t'_{N_{1},\tau^{*},\tau^{*}+\ell_{\tau^{*},k_{\ell}}}(1-u_{\alpha}) \geq Q(\alpha,\beta,L,R,\ell_{\tau^{*},k_{\ell}})\right) \leq \frac{\beta}{2} \quad .$$
(199)

As a consequence,

$$P_{\lambda}(T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N) \le t'_{N_1,\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(1-u_{\alpha})) \le P_{\lambda}\left(T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N) < Q(\alpha,\beta,L,R,\ell_{\tau^*,k_{\ell}})\right) + \frac{\beta}{2}$$

whereby we finally obtain with (198) and the Bienayme-Chebyshev inequality

$$P_{\lambda}(T'_{\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(N) \le t'_{N_1,\tau^*,\tau^*+\ell_{\tau^*,k_{\ell}}}(1-u_{\alpha})) \le \beta$$

The proof is ended by noticing that there exists $C(\alpha, \beta, R, \tau^*) > 0$ such that

$$\sqrt{\frac{1+\tau^{*}}{\tau^{*}}} \max\left(\sqrt{2C} \left(\sqrt{5R} \sqrt{\frac{\log\left(2.77/u_{\alpha}\right)}{L}} + \sqrt{5} \left(\frac{2R}{\beta}\right)^{1/4} \frac{\sqrt{\log\left(2.77/u_{\alpha}\right)}}{L^{3/4}} + \frac{\sqrt{3}\log\left(2.77/u_{\alpha}\right)}{2\sqrt{2L\log\log L}}\right) + \frac{2\sqrt{R}}{\beta^{1/4}} \frac{1}{\sqrt{L}}, \ 4\sqrt{\frac{1+\tau^{*}}{\tau^{*}}} \sqrt{\frac{2R}{\beta}} \frac{1}{\sqrt{L}}, \ 4R\sqrt{\frac{\tau^{*}}{1+\tau^{*}}} \sqrt{\frac{\log\log L}{L}}\right) \le C(\alpha, \beta, R, \tau^{*}) \sqrt{\frac{\log\log L}{L}} ,$$

as $\log (2.77/u_{\alpha}) = \log(2.77 \lfloor \log_2 L \rfloor / \alpha)$ and $L \ge 3$.

5.28 Proof of Proposition 35

Assume that

$$L > \frac{2((R - \delta^*) \wedge R) \log C_{\alpha,\beta}}{\delta^{*2}} \quad , \tag{200}$$

and set

$$r = \sqrt{\frac{((R - \delta^*) \wedge R) \log C_{\alpha,\beta}}{2L}} .$$
(201)

The assumption (200) entails

$$r^2 < \frac{\delta^{*2}}{4} \quad , \tag{202}$$

which enables us to define λ_r by $\lambda_r(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau_r, 1]}(t)$ for t in (0, 1), with $\lambda_0 = (R - \delta^*) \wedge R$ and

$$\tau_r = \frac{1}{2} \left(1 + \frac{\sqrt{\delta^{*2} - 4r^2}}{|\delta^*|} \right)$$

Thanks to (202), τ_r belongs to (0, 1) and satisfies $\delta^{*2}\tau_r(1-\tau_r) = r^2$, that is λ_r belongs to $\left(\mathcal{S}^u_{\delta^*,\dots,1-\dots}[R]\right)_r$. Moreover, the inequality $1 - \lambda_0 \log C_{\alpha,\beta}/(L\delta^{*2}) > 1/2$ comes from (200) and yields

$$r^{2} < \frac{\lambda_{0} \log C_{\alpha,\beta}}{L} \left(1 - \frac{\lambda_{0} \log C_{\alpha,\beta}}{L\delta^{*2}} \right)$$
$$= \frac{\delta^{*2}}{4} \left(1 - \left(1 - \frac{2\lambda_{0} \log C_{\alpha,\beta}}{L\delta^{*2}} \right)^{2} \right)$$

hence $\delta^{*2}(1-\tau_r) < \lambda_0 \log C_{\alpha,\beta}/L$. We get now from Lemma 1 and Lemma 44

$$E_{\lambda_0}\left[\left(\frac{dP_{\lambda}}{dP_{\lambda_0}}\right)^2(N)\right] = \exp\left(\frac{L(1-\tau_r)\delta^{*2}}{\lambda_0}\right) < C_{\alpha,\beta} .$$

Lemmas 42 and 43 then entail $\rho\left((\mathcal{S}^{u}_{\delta^{*},..,1-..}[R])_{r}\right) \geq \beta$ and $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}^{u}_{\delta^{*},..,1-..}[R]) \geq r$.

5.29 Proof of Proposition 36

The control of the first kind error rate is straightforward, and even more strong by using the same conditioning trick as in the proof of Proposition 32: in fact, for every λ_0 in $\mathcal{S}_0^u[R]$ and n in \mathbb{N} , $E_{\lambda_0} \left[\phi_{7,\alpha}^u(N) \middle| N_1 = n \right] = P_{\lambda_0} \left(\phi_{7,\alpha}^u(N) = 1 \middle| N_1 = n \right) \leq \alpha$. Let us turn to the control of the second kind error rate of $\phi_{7,\alpha}^u$. Set λ in $\mathcal{S}_{\delta^*,\dots,1-\dots}^u[R]$ such that $\lambda = \lambda_0 + \delta^* \mathbb{1}_{(\tau,1]}$ with λ_0 in $(-\delta^* \vee 0, (R - \delta^*) \wedge R]$ and τ in (0, 1), and satisfying

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge \frac{2}{\sqrt{L}} \max\left(\sqrt{|\delta^*|Q(2R, \delta^*, \alpha)}, \ 2\sqrt{\frac{2R}{\beta}}, \ \frac{|\delta^*|}{2\sqrt{2\beta R}}\right) \quad , \tag{203}$$

where $Q(2R, \delta^*, \alpha)$ is the constant, not depending on L, defined in Lemma 55 and used in Lemma 57. The condition (203) ensures that $|\delta^*|\sqrt{\tau(1-\tau)} \ge |\delta^*|/\sqrt{2\beta RL}$, and therefore, using the fact that $\tau(1-\tau) \le 1/4$,

$$L \ge \frac{2}{\beta R} \quad . \tag{204}$$

,

Setting $I(\lambda) = \int_0^1 \lambda(t) dt$, we obtain with $I(\lambda) \leq R$ (and therefore obviously $2R - I(\lambda) \geq R$) and the fact that (204) entails $L \geq 2R/(\beta R^2)$

$$L \ge \frac{2I(\lambda)}{(2R - I(\lambda))^2\beta} \quad . \tag{205}$$

Notice now that (205) and the Bienayme-Chebyshev inequality yield

$$P_{\lambda}(N_1 > 2RL) = P_{\lambda} \left(N_1 - I(\lambda)L > L(2R - I(\lambda)) \right)$$
$$\leq P_{\lambda} \left(N_1 - I(\lambda)L > \sqrt{\frac{2I(\lambda)L}{\beta}} \right) \leq \frac{\beta}{2} .$$

This leads to

$$P_{\lambda}\left(\phi_{7,\alpha}^{u}(N)=0\right) \leq P_{\lambda}\left(\sup_{\tau'\in(0,1)}S_{\delta^{*},\tau',1}'(N)\leq s_{N_{1},\delta^{*},L}'(1-\alpha), N_{1}\leq 2RL\right) + \frac{\beta}{2}$$
$$\leq P_{\lambda}\left(\sup_{\tau'\in(0,1)}S_{\delta^{*},\tau',1}'(N)\leq Q(2R,\delta^{*},\alpha)\right) + \frac{\beta}{2} \text{ with Lemma 57}$$
$$\leq P_{\lambda}\left(S_{\delta^{*},\tau,1}'(N)\leq Q(2R,\delta^{*},\alpha)\right) + \frac{\beta}{2}.$$

Moreover, the assumption (203) implies

$$|\delta^*|\sqrt{\tau(1-\tau)} \ge \frac{2}{\sqrt{L}} \max\left(\sqrt{|\delta^*|Q(2R,\delta^*,\alpha)}, 2\sqrt{\frac{2R}{\beta}}\right) ,$$

which entails

$$\frac{|\delta^*|}{2}\tau(1-\tau)L \ge 2\max\left(Q(2R,\delta^*,\alpha), \sqrt{\frac{2R\tau(1-\tau)L}{\beta}}\right) ,$$

whereby

$$\frac{|\delta^*|}{2}\tau(1-\tau)L \ge 2\max\left(Q(2R,\delta^*,\alpha), \sqrt{\frac{2(\lambda_0+\delta^*\tau)\tau(1-\tau)L}{\beta}}\right) .$$

Noticing that $E_{\lambda}[S'_{\delta^*,\tau,1}(N)] = |\delta^*|\tau(1-\tau)L/2$ and $\operatorname{Var}(S'_{\delta^*,\tau,1}(N)) = (\lambda_0 + \delta^*\tau)\tau(1-\tau)L$, this leads to

$$E_{\lambda}[S'_{\delta^*,\tau,1}(N)] \ge Q(2R,\delta^*,\alpha) + \sqrt{\frac{2\operatorname{Var}(S'_{\delta^*,\tau,1}(N))}{\beta}} \quad . \tag{206}$$

Therefore,

$$P_{\lambda}\left(\phi_{7,\alpha}^{u}(N)=0\right) \leq P_{\lambda}\left(S_{\delta^{*},\tau,1}^{\prime}(N)-E_{\lambda}[S_{\delta^{*},\tau,1}^{\prime}(N)] \leq -\sqrt{\frac{2\operatorname{Var}(S_{\delta^{*},\tau,1}^{\prime}(N))}{\beta}}\right) + \frac{\beta}{2} \quad \text{with} \ (206)$$
$$\leq \beta \ ,$$

with a last line simply deduced from the Bienayme-Chebyshev inequality. Setting

$$C(\alpha, \beta, R, \delta^*) = 2 \max\left(\sqrt{|\delta^*|Q(2R, \delta^*, \alpha)}, \ 2\sqrt{\frac{2R}{\beta}}, \ \frac{|\delta^*|}{2\sqrt{2\beta R}}\right) ,$$

allows to conclude the proof.

5.30 Proof of Proposition 37

Assume that $L \geq 3$ and $\alpha + \beta < 1/2$. We consider $C'_{\alpha,\beta} = 4(1 - \alpha - \beta)^2$, $K_{\alpha,\beta,L} = \lceil (\log_2 L)/C'_{\alpha,\beta} \rceil$, $\lambda_0 = R/2$ and for k in $\{1, \ldots, K_{\alpha,\beta,L}\}$, $\lambda_k = \lambda_0 + \delta_k \mathbb{1}_{(\tau_k,1]}$, with $\tau_k = 1 - 2^{-k}$

and $\delta_k = (2^k \lambda_0 \log \log L/L)^{1/2}$. Then, for every k in $\{1, \ldots, K_{\alpha,\beta,L}\}, d_2(\lambda_k, \mathcal{S}^u_0[R]) \geq \sqrt{R \log \log L/(4L)}$, and λ_k belongs to $\mathcal{S}^u_{\cdot, \cdot, 1- \cdot, \cdot}[R]$ assuming

$$\frac{\log \log L}{L^{1-1/C'_{\alpha,\beta}}} \le \frac{R}{4} \quad . \tag{207}$$

The proof then essentially follows the same arguments as the proof of Proposition 17. Considering a random variable κ with uniform distribution on $\{1, \ldots, K_{\alpha,\beta,L}\}$ and the probability distribution μ of λ_{κ} , we aim at proving that $E_{\lambda_0}[(dP_{\mu}/dP_{\lambda_0})^2] \leq 1 + C'_{\alpha,\beta}$, with P_{μ} defined as in Lemma 43, in order to conclude that $\mathrm{mSR}_{\alpha,\beta}(\mathcal{S}^u_{\cdot,\cdot,1-\cdots}[R]) \geq \sqrt{R\log\log L/(4L)}$. The same calculation as in the proof of Proposition 12 and Proposition 17 gives for η such that $0 < \eta < 1 - 1/\sqrt{2}$,

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] \le C'_{\alpha,\beta}\log 2 + \frac{2C'_{\alpha,\beta}\log 2}{\log L}\left(\log L\right)^{\eta+\frac{1}{\sqrt{2}}} + \exp\left(\frac{\log\log L}{2^{(\log L)^{\eta/2}}}\right) \quad .$$

If we assume now that

$$\exp\left(\frac{\log\log L}{2^{(\log L)^{\eta/2}}}\right) + \frac{2C'_{\alpha,\beta}\log 2}{(\log L)^{1-\eta-\frac{1}{\sqrt{2}}}} \le 1 + (1-\log 2)C'_{\alpha,\beta} \quad , \tag{208}$$

we finally obtain the expected result, that is

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2\right] \le 1 + C'_{\alpha,\beta}$$

To end the proof, it remains to notice that there exists $L_0(\alpha, \beta, R) \ge 3$ such that for all $L \ge L_0(\alpha, \beta, R)$, both assumptions (207) and (208) hold.

5.31 Proof of Proposition 38

As for all our Bonferroni type aggregated tests, the control of the first kind error rates of the two tests $\phi_{8,\alpha}^{u(1)}$ and $\phi_{8,\alpha}^{u(2)}$ is straightforward using union bounds and the conditioning trick of the above proofs for upper bounds. Let us now turn to the second kind error rates.

(i) Control of the second kind error rate of $\phi_{8,\alpha}^{u(1)}$.

Let λ in $\mathcal{S}^{u}_{\cdot,\dots,1-\dots}[R]$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,1]}$, with λ_0 in (0, R], τ in (0, 1), δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$, and assume that

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge \max\left(\sqrt{\frac{\log\left(2/u_\alpha\right)}{L}} 2\sqrt{R}, \sqrt{\frac{\log\left(2/u_\alpha\right)}{L}} 2\sqrt{6}\sqrt{R} + \sqrt{\frac{2R}{\beta L}} + \frac{2}{\sqrt{L}}\sqrt{\frac{6R}{\beta}}, \sqrt{\frac{2}{L}}R\right)$$

$$(209)$$

Let us prove that under this assumption, $P_{\lambda}(\phi_{8,\alpha}^{u(1)}(N) = 0) \leq \beta$. Applying Lemma 44, we get for all τ' in \mathcal{D}_L

$$E_{\lambda}[N(\tau',1] - (1-\tau')N_1] = \delta(\tau' \wedge \tau)(1-\tau' \vee \tau)L \quad , \tag{210}$$

and

$$\operatorname{Var}_{\lambda} \left[N(\tau', 1] - (1 - \tau') N_1 \right] = \begin{cases} (\lambda_0 (1 - \tau') + \delta \tau' (1 - \tau)) \tau' L & \text{if } \tau' \leq \tau \\ (\lambda_0 \tau' + \delta (\tau' - \tau + \tau' \tau)) (1 - \tau') L & \text{if } \tau' \geq \tau \end{cases}$$
(211)

Assume first that δ belongs to $(0, R - \lambda_0]$. Noticing that

$$P_{\lambda}(\phi_{8,\alpha}^{u(1)}(N) = 0) \le \inf_{\tau' \in \mathcal{D}_L} P_{\lambda}\left(N(\tau', 1] - (1 - \tau')N_1 \le \bar{b}_{N_1, 1 - \tau'}(1 - u_{\alpha}/2)\right)$$

one can see that it is enough to exhibit some τ' in \mathcal{D}_L such that

$$P_{\lambda}\left(N(\tau',1] - (1-\tau')N_1 \le \bar{b}_{N_1,1-\tau'}(1-u_{\alpha}/2)\right) \le \beta$$

Under the condition (209), we have $d_2^2(\lambda, \mathcal{S}_0^u[R]) \geq 2R^2/L$ which entails

$$\tau(1-\tau) > \frac{2}{L}$$
 (212)

On the one hand, if τ belongs to (0, 1/2), the condition (212) implies the inequalities $2^{-1} > \tau > 2/L > 2^{-\lfloor \log_2 L \rfloor}$ and the existence of k_{τ} in $\{2, ..., \lfloor \log_2 L \rfloor\}$ satisfying $2^{-k_{\tau}} \leq \tau < 2^{-k_{\tau}+1}$. We set $\tau_{k_{\tau}} = 2^{-k_{\tau}}$ and then

$$\tau_{k_{\tau}} \le \tau < \tau_{k_{\tau}-1} \quad , \tag{213}$$

whereby

$$\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}} \le \frac{\tau}{1 - \tau} < \frac{\tau_{k_{\tau} - 1}}{1 - \tau_{k_{\tau} - 1}}$$

and therefore

$$\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}} = \underbrace{\frac{\tau_{k_{\tau}-1}}{1-\tau_{k_{\tau}-1}}}_{>\frac{\tau}{1-\tau}} \frac{1-\tau_{k_{\tau}-1}}{1-\tau_{k_{\tau}}} \underbrace{\frac{\tau_{k_{\tau}}}{\tau_{k_{\tau}-1}}}_{=\frac{1}{2}} > \frac{1}{2} \frac{\tau}{1-\tau} \frac{1-\tau_{k_{\tau}-1}}{1-\tau_{k_{\tau}}} .$$

But

$$\frac{1-\tau_{k_{\tau}-1}}{1-\tau_{k_{\tau}}} = 1 - \frac{1}{2^{k_{\tau}}-1} \ge \frac{2}{3} \ ,$$

so we finally obtain

$$\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}} > \frac{\tau}{3(1 - \tau)} \quad . \tag{214}$$

On the other hand, if τ belongs to [1/2, 1), the condition (212) implies the inequalities $1 - 2^{-1} \leq \tau < 1 - 2/L < 1 - 2^{-\lfloor \log_2 L \rfloor}$ and the existence of k_{τ} in $\{1, ..., \lfloor \log_2 L \rfloor - 1\}$ satisfying $1 - 2^{-k_{\tau}} \leq \tau < 1 - 2^{-k_{\tau}-1}$. We set $\tau_{k_{\tau}} = 1 - 2^{-k_{\tau}}$ and we obtain

$$\tau_{k_{\tau}} \le \tau < \tau_{k_{\tau}+1} \quad , \tag{215}$$

whereby

$$\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}} \le \frac{\tau}{1 - \tau} < \frac{\tau_{k_{\tau}+1}}{1 - \tau_{k_{\tau}+1}} \ ,$$

and therefore

$$\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}} = \underbrace{\frac{\tau_{k_{\tau}+1}}{1-\tau_{k_{\tau}+1}}}_{>\frac{\tau}{1-\tau}} \underbrace{\frac{1-\tau_{k_{\tau}+1}}{1-\tau_{k_{\tau}}}}_{=\frac{1}{2}} \frac{\tau_{k_{\tau}}}{\tau_{k_{\tau}+1}} > \frac{1}{2} \frac{\tau}{1-\tau} \frac{\tau_{k_{\tau}}}{\tau_{k_{\tau}+1}}$$

But

$$\frac{\tau_{k_{\tau}}}{\tau_{k_{\tau}+1}} = 1 - \frac{1}{2^{k_{\tau}+1} - 1} \ge \frac{2}{3} \quad .$$

so we finally get

$$\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}} > \frac{\tau}{3(1 - \tau)} \quad . \tag{216}$$

Recall that $I(\lambda) = \int_0^1 \lambda(t) dt \leq R$ and notice that the assumption (209) leads to

$$\delta\sqrt{\tau(1-\tau)} \ge \max\left(\sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}}2\sqrt{R}, \sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}}2\sqrt{6}\sqrt{I(\lambda)} + \sqrt{\frac{2I(\lambda)}{\beta L}} + \frac{2}{\sqrt{L}}\sqrt{\frac{6(\lambda_{0}+\delta\tau_{k_{\tau}})}{\beta}}\right)$$
(217)

In particular, (217) gives on the one hand

$$\delta^2 \tau (1-\tau) \ge 4R \frac{\log \left(2/u_\alpha\right)}{L} \; ,$$

which entails, using (213), (215) and the inequality $\delta < R$

$$\delta \tau (1 - \tau_{k_{\tau}}) \ge 4 \frac{\log \left(2/u_{\alpha}\right)}{L}$$

Then, (214) and (216) ensure that $\tau(1-\tau_{k_{\tau}})/(3(1-\tau)) < \tau_{k_{\tau}}$ and thus

$$\delta \tau_{k_{\tau}}(1-\tau) \ge \frac{4}{3} \frac{\log(2/u_{\alpha})}{L}$$
 (218)

.

On the other hand, it follows from (217) that

$$\delta\sqrt{\tau(1-\tau)} \ge \sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}} 2\sqrt{6}\sqrt{I(\lambda)} + \sqrt{\frac{2I(\lambda)}{\beta L}} + \frac{2}{\sqrt{L}}\sqrt{\frac{6(\lambda_0 + \delta\tau_{k_{\tau}})}{\beta}} ,$$

which entails with (214) and (216)

$$\delta(1-\tau)\sqrt{\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}} \ge \sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}} 2\sqrt{2}\sqrt{I(\lambda)} + \sqrt{\frac{2I(\lambda)}{\beta L}} + \frac{2}{\sqrt{L}}\sqrt{\frac{2(\lambda_{0}+\delta\tau_{k_{\tau}})}{\beta}} ,$$

that is

$$\delta\tau_{k_{\tau}}(1-\tau)L \ge 2\sqrt{2\tau_{k_{\tau}}(1-\tau_{k_{\tau}})}\sqrt{L\log\left(2/u_{\alpha}\right)}\sqrt{I(\lambda)} + \sqrt{\frac{2I(\lambda)}{\beta L}} + 2\sqrt{\frac{2\tau_{k_{\tau}}(1-\tau_{k_{\tau}})(\lambda_{0}+\delta\tau_{k_{\tau}})L}{\beta}}$$
(219)

Thereby, with (218) and (219),

$$\delta \tau_{k_{\tau}}(1-\tau)L \ge 2 \max\left(\frac{2}{3}\log\left(\frac{2}{u_{\alpha}}\right), \sqrt{\tau_{k_{\tau}}(1-\tau_{k_{\tau}})}\sqrt{2\log\left(\frac{2}{u_{\alpha}}\right)}\sqrt{LI(\lambda)} + \sqrt{\frac{2LI(\lambda)}{\beta}} + \sqrt{\frac{2\tau_{k_{\tau}}(1-\tau_{k_{\tau}})(\lambda_{0}+\delta\tau_{k_{\tau}})L}{\beta}}\right),$$

hence

$$\delta \tau_{k_{\tau}}(1-\tau)L \ge Q(\alpha,\beta,L,\tau_{k_{\tau}}) + \sqrt{\frac{2\tau_{k_{\tau}}(1-\tau_{k_{\tau}})(\lambda_0+\delta\tau_{k_{\tau}})L}{\beta}} , \qquad (220)$$

with

$$Q(\alpha,\beta,L,\tau_{k_{\tau}}) = \frac{2}{3}\log\left(\frac{2}{u_{\alpha}}\right) + \sqrt{\tau_{k_{\tau}}(1-\tau_{k_{\tau}})}\sqrt{2\log\left(\frac{2}{u_{\alpha}}\right)}\sqrt{LI(\lambda)} + \sqrt{\frac{2LI(\lambda)}{\beta}}$$

Furthermore, with (213) and (215), the expressions of $E_{\lambda}[N(\tau', 1] - (1 - \tau')N_1]$ and $\operatorname{Var}_{\lambda}[N(\tau', 1] - (1 - \tau')N_1]$ given in (210) and (211) entail

$$\begin{cases} E_{\lambda}[N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1}] = \delta \tau_{k_{\tau}}(1 - \tau)L ,\\ \operatorname{Var}_{\lambda}[N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1}] \leq (\lambda_{0} + \delta \tau_{k_{\tau}})\tau_{k_{\tau}}(1 - \tau_{k_{\tau}})L . \end{cases}$$
(221)

The Bienayme-Chebyshev inequality leads to

$$P_{\lambda}\left(N_1 \ge LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}\right) \le \frac{\beta}{2}$$
,

and combined with Lemma 56, we get that

$$P_{\lambda}\left(\bar{b}_{N_1,1-\tau_{k_{\tau}}}(1-u_{\alpha}/2) \ge Q(\alpha,\beta,L,\tau_{k_{\tau}})\right) \le \frac{\beta}{2} \quad .$$

$$(222)$$

.

We conclude with the following inequalities:

$$P_{\lambda}(N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1} \leq s'_{N_{1},\tau_{k_{\tau}},1}(1 - u_{\alpha})) \\ \leq P_{\lambda}(N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1} < Q(\alpha, \beta, L, \tau_{k_{\tau}})) + \frac{\beta}{2} \text{ with (222)} \\ \leq P_{\lambda}\Biggl(N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1} - E_{\lambda}[N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1}] \\ < -\sqrt{\frac{2\operatorname{Var}_{\lambda}[N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_{1}]}{\beta}}\Biggr) + \frac{\beta}{2} \text{ with (220) and (221)}$$

 $\leq \beta$ with the Bienayme-Chebyshev inequality again.

Assume now that δ belongs to $(-\lambda_0,0)$ and notice that we have also

$$P_{\lambda}(\phi_{8,\alpha}^{u(1)}(N) = 0) \le \inf_{\tau' \in \mathcal{D}_L} P_{\lambda}\left(N(\tau', 1] - (1 - \tau')N_1 \ge \bar{b}_{N_1, 1 - \tau'}(u_{\alpha}/2)\right)$$

The same choice of $\tau_{k_{\tau}}$ as above leads, with (210) and (211), to $E_{\lambda}[N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_1] = -|\delta|\tau_{k_{\tau}}(1 - \tau)L$ and $\operatorname{Var}_{\lambda}[N(\tau_{k_{\tau}}, 1] - (1 - \tau_{k_{\tau}})N_1] \leq \lambda_0 \tau_{k_{\tau}}(1 - \tau_{k_{\tau}})L$ and we obtain, with very similar arguments and calculations (mainly replacing δ by $|\delta|$ and using the lower bound for $\bar{b}_{N_1,1-\tau_{k_{\tau}}}(u_{\alpha}/2)$ deduced from Lemma 56), that the assumption (209) implies

$$P_{\lambda}\left(N(\tau_{k_{\tau}},1]-(1-\tau_{k_{\tau}})N_{1}\geq \bar{b}_{N_{1},1-\tau_{k_{\tau}}}(u_{\alpha}/2)\right)\leq\beta$$
,

so $P_{\lambda}(\phi_{8,\alpha}^{u(1)}(N) = 0) \leq \beta$. Since $\log(2/u_{\alpha}) = \log(2(2\lfloor \log_2 L \rfloor - 1)/\alpha)$ and $L \geq 3$, there exists $C(\alpha, \beta, R) > 0$ such that

$$\max\left(\sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}}2\sqrt{R}, \sqrt{\frac{\log\left(2/u_{\alpha}\right)}{L}}2\sqrt{6}\sqrt{R} + \sqrt{\frac{2R}{\beta L}} + \frac{2}{\sqrt{L}}\sqrt{\frac{6R}{\beta}}, \sqrt{\frac{2}{L}}R\right) \le C(\alpha, \beta, R)\sqrt{\frac{\log\log L}{L}}, \quad (223)$$

which allows to conclude the proof.

(ii) Control of the second kind error rate of $\phi_{8,\alpha}^{u(2)}$.

Let λ in $S^{u}_{:,.,1-..}[R]$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,1]}$ with λ_0 in (0,R], τ in (0,1) and δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$ and assume that

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge \max\left(\sqrt{\frac{30CR\log\left(2.77/u_\alpha\right)}{L}} + \frac{\sqrt{30C\log\left(2.77/u_\alpha\right)}}{L^{3/4}} \left(\frac{2R}{\beta}\right)^{1/4} + \frac{3\sqrt{C}\log\left(2.77/u_\alpha\right)}{2\sqrt{L\log\log L}} + 2\sqrt{\frac{3R}{L\sqrt{\beta}}}, \ 12\sqrt{\frac{2R}{L\beta}}, \ 4R\sqrt{\frac{\log\log L}{L}}\right), \ (224)$$

where C is the constant defined in Lemma 53. Let us prove the inequality $P_{\lambda}(\phi_{8,\alpha}^{u(2)}(N) = 0) \leq \beta$. Notice first that

$$P_{\lambda}(\phi_{8,\alpha}^{u(2)}(N) = 0) \leq \inf_{\tau' \in \mathcal{D}_L} P_{\lambda}\left(T'_{\tau',1}(N) \leq t'_{N_1,\tau',1}(1 - u_{\alpha})\right) ,$$

so one only needs to exhibit some τ' in \mathcal{D}_L satisfying

$$P_{\lambda}\left(T'_{\tau',1}(N) \le t'_{N_1,\tau',1}(1-u_{\alpha})\right) \le \beta ,$$

to obtain the expected result.

Under the assumption (224), we get $d_2^2(\lambda, \mathcal{S}_0^u[R]) \ge 16R^2(\log \log L)/L$ which entails

$$\tau(1-\tau) > \frac{16\log\log L}{L} \quad . \tag{225}$$

Assume first that τ belongs to (0, 1/2). The condition (225) leads to $\tau > 16(\log \log L)/L$ and since $L \ge 3$, we get the inequality $16(\log \log L)/L > 2^{-\lfloor \log_2 L \rfloor}$ and the existence of k_{τ} in $\{2, ..., \lfloor \log_2 L \rfloor\}$ satisfying $2^{-k_{\tau}} \le \tau < 2^{-k_{\tau}+1}$. Setting $\tau_{k_{\tau}} = 2^{-k_{\tau}}$ we can prove as in the above case (i) that

$$\tau_{k_{\tau}} \le \tau < \tau_{k_{\tau}-1} \quad , \tag{226}$$

and

$$\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}} > \frac{\tau}{3(1 - \tau)} \quad . \tag{227}$$

Moreover, since $\tau_{k_{\tau}} < 1/2$ and by definition of k_{τ} ,

$$\max\left(\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}, \frac{1-\tau_{k_{\tau}}}{\tau_{k_{\tau}}}\right) = \frac{1-\tau_{k_{\tau}}}{\tau_{k_{\tau}}} < \frac{2-\tau}{\tau} .$$

Since (225) implies $\tau > 16(\log \log L)/L$, we get

$$\max\left(\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}, \frac{1-\tau_{k_{\tau}}}{\tau_{k_{\tau}}}\right) < \frac{L}{8\log\log L} \quad .$$
(228)

Assume now that τ belongs to [1/2, 1). The condition (225) entails $\tau < 1-16(\log \log L)/L$ and since $L \ge 3$, $1-2^{-1} \le \tau < 1-2^{-\lfloor \log_2 L \rfloor}$. Hence, there exists k_{τ} in $\{1, ..., \lfloor \log_2 L \rfloor - 1\}$ satisfying $1-2^{-k_{\tau}} \le \tau < 1-2^{-k_{\tau}-1}$. We set $\tau_{k_{\tau}} = 1-2^{-k_{\tau}}$ and we obtain as in the above case (i)

$$\tau_{k_{\tau}} \le \tau < \tau_{k_{\tau}+1} \quad , \tag{229}$$

and

$$\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}} > \frac{\tau}{3(1 - \tau)} \quad . \tag{230}$$

Moreover, since $\tau_{k_{\tau}}$ belongs to (1/2, 1), we get using (229)

$$\max\left(\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}, \frac{1-\tau_{k_{\tau}}}{\tau_{k_{\tau}}}\right) = \frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}} \le \frac{\tau}{1-\tau}$$

Since (225) yields $1 - \tau > 16(\log \log L)/L$, we get also

$$\max\left(\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}, \frac{1-\tau_{k_{\tau}}}{\tau_{k_{\tau}}}\right) < \frac{L}{16\log\log L} \quad .$$
(231)

Applying Lemma 52, we obtain with (226) and (229)

$$E_{\lambda}[T'_{\tau_{k_{\tau}},1}(N)] = \delta^2 (1-\tau)^2 \frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}} , \qquad (232)$$

and

$$\operatorname{Var}_{\lambda}\left[T_{\tau_{k_{\tau}},1}'(N)\right] = \frac{4}{L}\delta^{2}(1-\tau)^{2}\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}\left(\lambda_{0}+\delta(1-\tau)\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}\right) + \frac{2}{L^{2}}\left(\lambda_{0}+\delta(1-\tau)\frac{\tau_{k_{\tau}}}{1-\tau_{k_{\tau}}}\right)^{2}$$
(233)

On the one hand, we finally get with (227) and (230),

$$E_{\lambda}[T'_{\tau_{k_{\tau}},1}(N)] > \frac{\delta^2 \tau (1-\tau)}{3} ,$$
 (234)

and on the other hand, using (226) and (229),

$$\operatorname{Var}_{\lambda}\left[T'_{\tau_{k_{\tau}},1}(N)\right] \leq \frac{4\delta^{2}\tau(1-\tau)R}{L} + \frac{2R^{2}}{L^{2}}$$
 (235)

Notice that the assumption (224) entails

$$\begin{split} \delta^2 \tau(1-\tau) &\geq \max\left(\frac{30CR\log\left(2.77/u_{\alpha}\right)}{L} + \frac{30C\log\left(2.77/u_{\alpha}\right)}{L^{3/2}}\sqrt{\frac{2R}{\beta}} + \frac{9C\log^2\left(2.77/u_{\alpha}\right)}{4L\log\log L} + \frac{12R}{L\sqrt{\beta}} , \ 12|\delta|\sqrt{\tau(1-\tau)}\sqrt{\frac{2R}{L\beta}}\right) \,, \end{split}$$

hence

$$\delta^{2}\tau(1-\tau) \geq \frac{15CR\log\left(2.77/u_{\alpha}\right)}{L} + \frac{15C\log\left(2.77/u_{\alpha}\right)}{L^{3/2}}\sqrt{\frac{2R}{\beta}} + \frac{9C\log^{2}\left(2.77/u_{\alpha}\right)}{8L\log\log L} + \frac{6R}{L\sqrt{\beta}} + 6|\delta|\sqrt{\tau(1-\tau)}\sqrt{\frac{2R}{L\beta}} \quad (236)$$

Using (234) and (235), the condition (236) then leads to

$$E_{\lambda}[T'_{\tau_{k_{\tau}},1}(N)] - \sqrt{2\operatorname{Var}_{\lambda}\left[T'_{\tau_{k_{\tau}},1}(N)\right]/\beta} \geq C\left(5R\frac{\log\left(2.77/u_{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(2.77/u_{\alpha}\right)}{L^{3/2}} + \frac{3\log^{2}\left(2.77/u_{\alpha}\right)}{L\log\log L}\right) \quad . \quad (237)$$

Now, if we set

$$Q(\alpha, \beta, L, R, \tau_{k_{\tau}}) = C \left(5R \frac{\log \left(2.77/u_{\alpha}\right)}{L} + 5\sqrt{\frac{2R}{\beta}} \frac{\log \left(2.77/u_{\alpha}\right)}{L^{3/2}} + 3 \max \left(\frac{\tau_{k_{\tau}}}{1 - \tau_{k_{\tau}}}, \frac{1 - \tau_{k_{\tau}}}{\tau_{k_{\tau}}}\right) \frac{\log^2 \left(2.77/u_{\alpha}\right)}{L^2} \right) , \quad (238)$$

combined with (228) and (231), the condition (237) yields

$$E_{\lambda}[T'_{\tau_{k_{\tau}},1}(N)] - \sqrt{2\operatorname{Var}_{\lambda}\left[T'_{\tau_{k_{\tau}},1}(N)\right]}/\beta} \ge Q(\alpha,\beta,L,R,\tau_{k_{\tau}}) \quad .$$
(239)

Furthermore, from the inequality

$$P_{\lambda}\left(N_1 \ge LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}\right) \le \frac{\beta}{2}$$
,

Lemma 53, and the fact that $I(\lambda) \leq R$, we deduce that

$$P_{\lambda}\left(t'_{N_{1},\tau_{k_{\tau}},1}(1-u_{\alpha}) \ge Q(\alpha,\beta,L,R,\tau_{k_{\tau}})\right) \le \frac{\beta}{2} \quad .$$

$$(240)$$

We finally obtain using successively (240), (239) and the Bienayme-Chebyshev inequality

$$P_{\lambda}(T'_{\tau_{k_{\tau}},1}(N) \leq t'_{N_{1},\tau_{k_{\tau}},1}(1-u_{\alpha})) \leq P_{\lambda}\left(T'_{\tau_{k_{\tau}},1}(N) < Q(\alpha,\beta,L,R,\tau_{k_{\tau}})\right) + \frac{\beta}{2} \\ \leq P_{\lambda}\left(T'_{\tau_{k_{\tau}},1}(N) - E_{\lambda}[T'_{\tau_{k_{\tau}},1}(N)] < -\sqrt{2\mathrm{Var}_{\lambda}\left[T'_{\tau_{k_{\tau}},1}(N)\right]}/\beta\right) + \frac{\beta}{2} \\ \leq \beta \ .$$

Since $\log(2/u_{\alpha}) = \log((4\lfloor \log_2 L \rfloor - 2)/\alpha)$ and $L \ge 3$, there exists $C(\alpha, \beta, R) > 0$ such that (224) is implied by

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge C(\alpha, \beta, R) \sqrt{\frac{\log \log L}{L}}$$

which concludes the proof.

5.32 Proof of Proposition 39

Let $L \geq 2$ and set $\lambda_0 = (R - \delta^*) \wedge R$. For all k in $\{1, \ldots, \lceil L^{3/4} \rceil\}$, let us define $\lambda_k(t) = \lambda_0 + \delta^* \mathbb{1}_{(\tau_k, \tau_k + \ell]}(t)$ with $\tau_k = k/L$ and $\ell = \lambda_0 \log L/(2\delta^{*2}L)$. Then, as soon as

$$\frac{\lceil L^{3/4} \rceil}{L} + \frac{((R - \delta^*) \wedge R) \log L}{2{\delta^*}^2 L} < 1 \quad , \tag{241}$$

 λ_k belongs to $\mathcal{S}^u_{\delta^*,\dots,\dots}[R]$ for any k in $\{1,\dots,\lceil L^{3/4}\rceil\}$. If in addition

$$\frac{\log L}{L} \le \frac{\delta^{*2}}{(R - \delta^*) \wedge R} \quad , \tag{242}$$

 λ_k satisfies for all k in $\{1, \ldots, \lceil L^{3/4} \rceil\},\$

$$d_2^2(\lambda_k, \mathcal{S}_0^u[R]) \ge \frac{((R - \delta^*) \wedge R) \log L}{4L}$$

Using Lemma 43 and considering a random variable J uniformly distributed on $\{1, \ldots, \lceil L^{3/4} \rceil\}$ and the distribution μ of λ_J , one can see that it is enough to prove that $E_{\lambda_0}[(dP_{\mu}/dP_{\lambda_0})^2] \leq 1 + 4(1 - \alpha - \beta)^2$ to obtain the expected lower bound. The same calculation as in the proof of Proposition 19 leads to

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2(N)\right] \le 1 + \frac{\sqrt{L}}{\left\lceil L^{3/4} \right\rceil} \frac{e^{\delta^{*2}/\lambda_0} + 1}{e^{\delta^{*2}/\lambda_0} - 1}$$

Therefore, assuming that

$$\frac{\sqrt{L}}{\lceil L^{3/4} \rceil} \frac{e^{\delta^{*2}/(R-\delta^*)\wedge R} + 1}{e^{\delta^{*2}/(R-\delta^*)\wedge R} - 1} \le 4(1-\alpha-\beta)^2 \quad , \tag{243}$$

we get the expected result, that is

$$E_{\lambda_0}\left[\left(\frac{dP_{\mu}}{dP_{\lambda_0}}\right)^2(N)\right] \le 1 + 4(1 - \alpha - \beta)^2 .$$

We then conclude the proof noticing that there exists $L_0(\alpha, \beta, \delta^*, R) \ge 2$ such that the assumptions (241), (242) and (243) hold for all $L \ge L_0(\alpha, \beta, \delta^*, R)$.

5.33 Proof of Proposition 41

As for all our Bonferroni type aggregated tests of Section 3 dedicated to change detection from an unknown baseline intensity, the control of the first kind error rates of the two tests $\phi_{9/10,\alpha}^{u(1)}$ and $\phi_{9/10,\alpha}^{u(2)}$ is easily deduced from union bounds and the conditioning trick of the above proofs for upper bounds. We therefore focus here on the second kind error rates.

(i) Control of the second kind error rate of $\phi_{9/10,\alpha}^{u(1)}$.

Let λ in $\mathcal{S}^{u}_{\cdot,\cdot,\cdot,\cdots}[R]$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau,\tau+\ell]}$, with λ_0 in (0, R], τ in (0, 1), δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$ and ℓ in $(0, 1 - \tau)$. By now, we aim at proving that $P_{\lambda}\left(\phi^{u(1)}_{9/10,\alpha}(N) = 0\right) \leq \beta$ as soon as we assume that

$$d_2(\lambda, \mathcal{S}_0^u[R]) \ge \max\left(2\sqrt{\frac{R\log\left(2/u_\alpha^{(1)}\right)}{L}}, 6\sqrt{\frac{2\log\left(2/u_\alpha^{(1)}\right)}{L}}\sqrt{R} + \sqrt{\frac{2R}{\beta L}} + 6\sqrt{\frac{2R}{\beta L}}, \frac{\sqrt{3R}}{\sqrt{L}}\right)$$
(244)

Let us first consider the case where δ belongs to $(0, R - \lambda_0]$. Noticing that

$$P_{\lambda}\left(\phi_{9/10,\alpha}^{u(1)}(N)=0\right)$$

$$\leq \inf_{k\in\{0,\dots,\lceil L\rceil-1\}}\inf_{k'\in\{1,\dots,\lceil L\rceil-k\}}P_{\lambda}\left(N\left(\frac{k}{\lceil L\rceil},\frac{k+k'}{\lceil L\rceil}\right)-\frac{k'}{\lceil L\rceil}N_{1}\leq\bar{b}_{N_{1},\frac{k'}{\lceil L\rceil}}\left(1-\frac{u_{\alpha}^{(1)}}{2}\right)\right),$$

one can see that it is enough to exhibit some k_0 in $\{0, \ldots, \lceil L \rceil - 1\}$ and k'_0 in $\{1, \ldots, \lceil L \rceil - k_0\}$ such that

$$P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\leq \bar{b}_{N_{1},\frac{k_{0}'}{\lceil L\rceil}}\left(1-\frac{u_{\alpha}^{(1)}}{2}\right)\right)\leq\beta \quad .$$

We get from (244) that $d_2^2(\lambda, \mathcal{S}_0^u[R]) \geq 3R^2/L \geq 3R^2/\lceil L \rceil$ which entails

$$\ell(1-\ell) > 3/\lceil L \rceil \quad . \tag{245}$$

Assume first that $\ell \leq 1/2$. The condition (245) leads to

$$\ell > 3/\lceil L \rceil$$
 and $\tau < 1 - 3/\lceil L \rceil$. (246)

Therefore, we can define $k_0 = \min(k \in \{0, \dots, \lceil L \rceil - 1\}, \tau \leq k/\lceil L \rceil)$ and $k'_0 = \max(k' \in \{1, \dots, \lceil L \rceil - k_0\}, (k_0 + k')/\lceil L \rceil \leq \tau + \ell)$, so that $\tau \leq k_0/\lceil L \rceil < (k_0 + k'_0)/\lceil L \rceil \leq \tau + \ell$. Since by definition $k_0/\lceil L \rceil - \tau < 1/\lceil L \rceil$ and $\tau + \ell - (k_0 + k'_0)/\lceil L \rceil < 1/\lceil L \rceil$, we get that

$$\frac{k_0'}{\lceil L \rceil} = \ell - \left(\left(\frac{k_0}{\lceil L \rceil} - \tau \right) + \left(\tau + \ell - \frac{k_0 + k_0'}{\lceil L \rceil} \right) \right) > \ell - \frac{2}{\lceil L \rceil},$$

and then, combining with (246),

$$\frac{k_0'}{\lceil L \rceil} > \frac{\ell}{3} \quad . \tag{247}$$

Lemma 44 leads to $E_{\lambda} \left[N \left(k_0 / \lceil L \rceil, \left(k_0 + k'_0 \right) / \lceil L \rceil \right] - \left(k'_0 / \lceil L \rceil \right) N_1 \right] = \delta(1 - \ell) L k'_0 / \lceil L \rceil$ and $\operatorname{Var}_{\lambda} \left[N \left(\frac{k_0}{\lceil L \rceil}, \frac{k_0 + k'_0}{\lceil L \rceil} \right) - \frac{k'_0}{\lceil L \rceil} N_1 \right] = \left(\lambda_0 \left(1 - \frac{k'_0}{\lceil L \rceil} \right) + \delta \left(1 - (2 - \ell) \frac{k'_0}{\lceil L \rceil} \right) \right) \frac{k'_0}{\lceil L \rceil} L$.

With (247), we obtain on the one hand

$$E_{\lambda}\left[N\left(\frac{k_0}{\lceil L\rceil}, \frac{k_0 + k'_0}{\lceil L\rceil}\right) - \frac{k'_0}{\lceil L\rceil}N_1\right] > \frac{\delta}{3}\ell(1-\ell)L \quad .$$
(248)

On the other hand, $k'_0/\lceil L \rceil \leq \ell < 1$ yields $0 < 1 - (2 - \ell)k'_0/\lceil L \rceil < 1 - k'_0/\lceil L \rceil$. Moreover, using $k'_0/\lceil L \rceil \leq \ell$ again and the fact that $\ell \leq 1/2$ one obtains

$$\frac{k_0'}{\lceil L \rceil} \left(1 - \frac{k_0'}{\lceil L \rceil} \right) \le \ell (1 - \ell) \quad , \tag{249}$$

and we finally get

$$\operatorname{Var}_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil}, \frac{k_{0} + k_{0}'}{\lceil L\rceil}\right) - \frac{k_{0}'}{\lceil L\rceil}N_{1}\right] \leq (\lambda_{0} + \delta)\ell(1 - \ell)L \quad .$$

$$(250)$$

Recall that $I(\lambda) = \int_0^1 \lambda(t) dt$ and notice that (244) entails

$$\delta\sqrt{\ell(1-\ell)} \ge \max\left(2\sqrt{\frac{\delta\log\left(2/u_{\alpha}^{(1)}\right)}{L}}, \ 6\sqrt{\frac{2\log\left(2/u_{\alpha}^{(1)}\right)}{L}}\sqrt{I(\lambda) + \sqrt{\frac{2I(\lambda)}{\beta L}}} + 6\sqrt{\frac{2(\lambda_0+\delta)}{\beta L}}\right) ,$$

thereby

$$\delta\ell(1-\ell) \ge 2 \max\left(\frac{2\log\left(2/u_{\alpha}^{(1)}\right)}{L} , \ 3\sqrt{\ell(1-\ell)}\left(\sqrt{\frac{2\log\left(2/u_{\alpha}^{(1)}\right)}{L}}\sqrt{I(\lambda) + \sqrt{\frac{2I(\lambda)}{\beta L}}} + \sqrt{\frac{2(\lambda_0+\delta)}{\beta L}}\right)\right)$$

and then

$$\frac{\delta}{3}\ell(1-\ell)L \ge \frac{2}{3}\log\left(\frac{2}{u_{\alpha}^{(1)}}\right) + \sqrt{\ell(1-\ell)}\sqrt{2L\log\left(\frac{2}{u_{\alpha}^{(1)}}\right)}\sqrt{I(\lambda)} + \sqrt{\frac{2I(\lambda)}{\beta L}} + \sqrt{\frac{2\ell(1-\ell)L(\lambda_0+\delta)}{\beta L}}$$
(251)

Thus, with (248) and (250), the inequality (251) ensures that

$$E_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right] \geq \frac{2}{3}\log\left(\frac{2}{u_{\alpha}^{(1)}}\right)+\sqrt{\ell(1-\ell)}\sqrt{2L\log\left(\frac{2}{u_{\alpha}^{(1)}}\right)}\sqrt{I(\lambda)}+\sqrt{\frac{2I(\lambda)}{\beta L}}+\sqrt{\frac{2Var_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right]}{\beta}}.$$
 (252)

Furthermore, we deduce from the inequality

$$P_{\lambda}\left(N_1 \ge LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}\right) \le \frac{\beta}{2}$$
,

combined with Lemma 56 that

$$P_{\lambda}\left(\bar{b}_{N_{1},\frac{k_{0}'}{\lceil L \rceil}}\left(1-\frac{u_{\alpha}^{(1)}}{2}\right) \ge Q(\alpha,\beta,L,k_{0}')\right) \le \frac{\beta}{2} \quad , \tag{253}$$

with

$$Q(\alpha,\beta,L,k_0') = \frac{2}{3}\log\left(\frac{2}{u_{\alpha}^{(1)}}\right) + \sqrt{\frac{k_0'}{\lceil L \rceil}\left(1 - \frac{k_0'}{\lceil L \rceil}\right)}\sqrt{2\log\left(\frac{2}{u_{\alpha}^{(1)}}\right)}\sqrt{LI(\lambda)} + \sqrt{\frac{2LI(\lambda)}{\beta}}$$
(254)

Using (249), the inequality (252) leads to

$$E_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right] \geq Q(\alpha,\beta,L,k_{0}') + \sqrt{2\operatorname{Var}_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right]/\beta} , \quad (255)$$

and we conclude with the following inequalities

$$P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L \rceil}, \frac{k_{0} + k_{0}'}{\lceil L \rceil}\right) - \frac{k_{0}'}{\lceil L \rceil}N_{1} \leq \bar{b}_{N_{1}, \frac{k_{0}'}{\lceil L \rceil}}\left(1 - \frac{u_{\alpha}^{(1)}}{2}\right)\right)$$

$$\leq P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L \rceil}, \frac{k_{0} + k_{0}'}{\lceil L \rceil}\right) - \frac{k_{0}'}{\lceil L \rceil}N_{1} < Q(\alpha, \beta, L, k_{0}')\right)$$

$$+ P_{\lambda}\left(\bar{b}_{N_{1}, \frac{k_{0}'}{\lceil L \rceil}}\left(1 - \frac{u_{\alpha}^{(1)}}{2}\right) \geq Q(\alpha, \beta, L, k_{0}')\right)$$

$$\leq P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L \rceil}, \frac{k_{0} + k_{0}'}{\lceil L \rceil}\right) - \frac{k_{0}'}{\lceil L \rceil}N_{1} < Q(\alpha, \beta, L, k_{0}')\right) + \frac{\beta}{2} \text{ with } (253)$$

$$\leq P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L \rceil}, \frac{k_{0} + k_{0}'}{\lceil L \rceil}\right) - \frac{k_{0}'}{\lceil L \rceil}N_{1} - E_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L \rceil}, \frac{k_{0} + k_{0}'}{\lceil L \rceil}\right) - \frac{k_{0}'}{\lceil L \rceil}N_{1}\right]$$

$$< -\sqrt{2} \text{Var}_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L \rceil}, \frac{k_{0} + k_{0}'}{\lceil L \rceil}\right) - \frac{k_{0}'}{\lceil L \rceil}N_{1}\right] - \frac{k_{0}'}{\lceil L \rceil}N_{1}\right] /\beta} + \frac{\beta}{2} \text{ with } (255)$$

 $\leq \beta~$ with the Bienayme-Chebyshev inequality .

Assume now that $\ell > 1/2$. We define $k_0 = \max(k \in \{0, \dots, \lceil L \rceil - 1\}, \tau \ge k/\lceil L \rceil)$ and $k'_0 = \min(k' \in \{1, \dots, \lceil L \rceil - k_0\}, (k_0 + k')/\lceil L \rceil \ge \tau + \ell)$ so that $k_0/\lceil L \rceil \le \tau < \tau + \ell \le (k_0 + k'_0)/\lceil L \rceil$. Notice that the condition (245) entails

$$1 - \ell > 3/\lceil L \rceil \quad . \tag{256}$$

Since by definition of k_0 and k'_0 , $\tau - k_0/\lceil L \rceil < 1/\lceil L \rceil$ and $(k_0 + k'_0)/\lceil L \rceil - (\tau + \ell) < 1/\lceil L \rceil$, then, with (256),

$$\frac{k'_0}{\lceil L \rceil} = \frac{k_0 + k'_0}{\lceil L \rceil} - (\tau + \ell) + \ell + \tau - \frac{k_0}{\lceil L \rceil} < 1/\lceil L \rceil + (1 - 3/\lceil L \rceil) + 1/\lceil L \rceil \le 1 - 1/\lceil L \rceil < 1$$

In the same way, we can also obtain with (256) that

$$1 - \frac{k_0'}{\lceil L \rceil} = 1 - \ell - \left(\left(\tau - \frac{k_0}{\lceil L \rceil} \right) + \left(\frac{k_0 + k_0'}{\lceil L \rceil} - (\tau + \ell) \right) \right) > 1 - \ell - \frac{2}{\lceil L \rceil} > \frac{1 - \ell}{3} \quad . \tag{257}$$

Furthermore, by definition of k_0 and k'_0 , one has $k'_0/\lceil L \rceil \ge \ell$ and since $\ell > 1/2$,

$$\frac{k_0'}{\lceil L \rceil} \left(1 - \frac{k_0'}{\lceil L \rceil} \right) \le \ell (1 - \ell) \quad . \tag{258}$$

From Lemma 44, we therefore deduce

$$E_{\lambda} \left[N \left(\frac{k_0}{\lceil L \rceil}, \frac{k_0 + k'_0}{\lceil L \rceil} \right] - \frac{k'_0}{\lceil L \rceil} N_1 \right] = \delta \ell \left(1 - \frac{k'_0}{\lceil L \rceil} \right) L$$
$$> \frac{\delta}{3} \ell (1 - \ell) L \quad \text{with} \ (257) \ ,$$

and since $k'_0/\lceil L\rceil \ge \ell$,

$$\operatorname{Var}_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil}, \frac{k_{0} + k_{0}'}{\lceil L\rceil}\right) - \frac{k_{0}'}{\lceil L\rceil}N_{1}\right] = \left(\lambda_{0}\frac{k_{0}'}{\lceil L\rceil} + \delta\ell\left(1 - \frac{k_{0}'}{\lceil L\rceil}\right)\right)\left(1 - \frac{k_{0}'}{\lceil L\rceil}\right)L$$
$$\leq \left(\lambda_{0} + \delta\left(1 - \frac{k_{0}'}{L}\right)\right)\left(1 - \frac{k_{0}'}{L}\right)\frac{k_{0}'}{\lceil L\rceil}L$$
$$\leq (\lambda_{0} + \delta)\ell(1 - \ell)L \quad \text{with } (258) \quad .$$

Thus, the condition (251) also yields

$$E_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right] \geq Q(\alpha,\beta,L,k_{0}') + \sqrt{2\mathrm{Var}_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right]}/\beta ,$$

where $Q(\alpha, \beta, L, k_0')$ is defined by (254) and we conclude that

$$P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\leq\bar{b}_{N_{1},\frac{k_{0}'}{\lceil L\rceil}}\left(1-\frac{u_{\alpha}^{(1)}}{2}\right)\right)\leq\beta \quad,$$

with the same arguments as above.

Let us then treat the case where δ belongs to $(-\lambda_0, 0)$. We start by noticing that

$$P_{\lambda}\left(\phi_{9/10,\alpha}^{(1)}(N)=0\right) \leq \inf_{k \in \{0,\dots,\lceil L\rceil-1\}} \inf_{k' \in \{1,\dots,\lceil L\rceil-k\}} P_{\lambda}\left(N\left(\frac{k}{\lceil L\rceil},\frac{k+k'}{\lceil L\rceil}\right] - \frac{k'}{\lceil L\rceil}N_1 \geq \bar{b}_{N_1,\frac{k'}{\lceil L\rceil}}\left(\frac{u_{\alpha}^{(1)}}{2}\right)\right)$$

The same choice of k_0 and k'_0 as in the case where δ belongs to $(0, R - \lambda_0]$ leads to

$$E_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right]-\frac{k_{0}'}{\lceil L\rceil}N_{1}\right]<-\frac{|\delta|\ell(1-\ell)L}{3}$$

and

$$\operatorname{Var}_{\lambda}\left[N\left(\frac{k_{0}}{\lceil L\rceil}, \frac{k_{0}+k_{0}'}{\lceil L\rceil}\right) - \frac{k_{0}'}{\lceil L\rceil}N_{1}\right] \leq \lambda_{0}\frac{k_{0}'}{\lceil L\rceil}\left(1 - \frac{k_{0}'}{\lceil L\rceil}\right)L \leq R\frac{k_{0}'}{\lceil L\rceil}\left(1 - \frac{k_{0}'}{\lceil L\rceil}\right)L ,$$

and we obtain

$$P_{\lambda}\left(N\left(\frac{k_{0}}{\lceil L\rceil},\frac{k_{0}+k_{0}'}{\lceil L\rceil}\right)-\frac{k_{0}'}{\lceil L\rceil}N_{1}\geq\bar{b}_{N_{1},\frac{k_{0}'}{\lceil L\rceil}}\left(\frac{u_{\alpha}^{(1)}}{2}\right)\right)\leq\beta \quad,$$

following the same lines of proof as above, but with δ replaced by $|\delta|$ except when it is involved in $\lambda_0 + \delta$, and using the lower bound for $\bar{b}_{N_1,\frac{k'_0}{|L|}}\left(u_{\alpha}^{(1)}/2\right)$ deduced from Lemma 56. Coming back to the assumption (244) and the definition of $u_{\alpha}^{(1)}$, one can finally claim that

$$\operatorname{SR}_{\beta}\left(\phi_{9/10,\alpha}^{u(1)}, \mathcal{S}_{\cdot,\cdots,\cdots}^{u}[R]\right) \leq \max\left(2\sqrt{\frac{R\log\left(\lceil L\rceil\left(\lceil L\rceil + 1\right)/\alpha\right)}{L}}, \frac{6\sqrt{2\log\left(\lceil L\rceil\left(\lceil L\rceil + 1\right)/\alpha\right)}}{L}\sqrt{R + \sqrt{\frac{2R}{\beta L}}} + 6\sqrt{\frac{2R}{\beta L}}, \frac{\sqrt{3}R}{\sqrt{L}}\right), \quad (259)$$

which leads to the result stated in Proposition 41 for $\phi_{9/10,\alpha}^{u(1)}$ and the set $\mathcal{S}^{u}_{\cdot,\cdot,\cdot,\cdots}[R]$ of the alternative [Alt^u.10]. Since $\mathcal{S}^{u}_{\delta^{*},\cdot,\cdot,\cdots}[R] \subset \mathcal{S}^{u}_{\cdot,\cdot,\cdot,\cdots}[R]$ for any δ^{*} in $(-R, R) \setminus \{0\}$, the same result holds for $\phi_{9/10,\alpha}^{u(1)}$ and the set $\mathcal{S}^{u}_{\delta^{*},\cdot,\cdot,\cdots}[R]$ of the alternative [Alt^u.9]. Notice that in this case, the constant $C(\alpha, \beta, R)$ involved in the upper bound can be refined, benefiting from the knowledge of δ^{*} , which explains the formulation with $C(\alpha, \beta, R, \delta^{*})$ instead of $C(\alpha, \beta, R)$ in Proposition 41.

(ii) Control of the second kind error rate of $\phi_{9/10,\alpha}^{u(2)}$.

Let λ in $\mathcal{S}^{u}_{\cdot, \cdots, \cdots}[R]$ such that $\lambda = \lambda_0 + \delta \mathbb{1}_{(\tau, \tau+\ell]}$, with λ_0 in (0, R], τ in (0, 1), δ in $(-\lambda_0, R - \lambda_0] \setminus \{0\}$ and ℓ in $(0, 1 - \tau)$. We assume by now that

$$d_{2}(\lambda, \mathcal{S}_{0}^{u}[R]) \geq \max\left(\sqrt{\frac{60CR\log\left(2.77/u_{\alpha}^{(2)}\right)}{L}} + \left(\frac{2R}{\beta}\right)^{1/4} \frac{\sqrt{60C\log\left(2.77/u_{\alpha}^{(2)}\right)}}{L^{3/4}} + 6\sqrt{C}\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{\sqrt{L\log L}} + 2\sqrt{\frac{6R}{L\sqrt{\beta}}}, \ 24\sqrt{\frac{2R}{\beta L}}, \ R\sqrt{\frac{3\log L}{L}}\right), \ (260)$$

where C is the constant defined in Lemma 53. Let us prove that this assumption implies $P_{\lambda}\left(\phi_{9/10,\alpha}^{u(2)}(N)=0\right) \leq \beta$. As above, we invoke that

$$P_{\lambda}\left(\phi_{9/10,\alpha}^{u(2)}(N)=0\right) \leq \inf_{\substack{k \in \{0,\dots,M_{L}-1\}}} \inf_{\substack{k' \in \{1,\dots,M_{L}-k\}\\(k,k') \neq (0,M_{L})}} P_{\lambda}\left(T'_{\frac{k}{M_{L}},\frac{k+k'}{M_{L}}}(N) \leq t'_{N_{1},\frac{k}{M_{L}},\frac{k+k'}{M_{L}}}\left(1-u_{\alpha}^{(2)}\right)\right)$$

to argue that we only need to exhibit some k_0 in $\{0, \ldots, M_L - 1\}$ and k'_0 in $\{1, \ldots, M_L - k_0\}$ such that $(k_0, k'_0) \neq (0, M_L)$ satisfying

$$P_{\lambda}\left(T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N) \leq t'_{N_{1},\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}\left(1-u_{\alpha}^{(2)}\right)\right) \leq \beta \quad .$$

Recalling that $M_L = \lceil L/\log L \rceil$, we get from (260) that $d_2^2(\lambda, \mathcal{S}_0^u[R]) \ge 3R^2 \log L/L \ge 3R^2/M_L$ which entails

$$\ell(1-\ell) > 3/M_L$$
 (261)

Let us first consider the case where $\ell \leq 1/2$. The condition (261) leads to

$$\ell > 3/M_L \quad , \tag{262}$$

and therefore, we can define $k_0 = \min(k \in \{1, \ldots, M_L - 1\}, \tau \leq k/M_L)$ and $k'_0 = \max(k' \in \{1, \ldots, M_L - k_0\}, (k_0 + k')/M_L \leq \tau + \ell)$, so that $\tau \leq k_0/M_L < (k_0 + k'_0)/M_L \leq \tau + \ell$. Since by definition $k_0/M_L - \tau < 1/M_L$ and $\tau + \ell - (k_0 + k'_0)/M_L < 1/M_L$, notice that

$$\frac{k_0'}{M_L} = \ell - \left(\left(\frac{k_0}{M_L} - \tau \right) + \left(\tau + \ell - \frac{k_0 + k_0'}{M_L} \right) \right) > \ell - \frac{2}{M_L} ,$$

bined with (262)

and then, combined with (262),

$$\frac{k_0'}{M_L} > \frac{\ell}{3} \quad . \tag{263}$$

Lemma 52 gives $E_{\lambda}\left[T'_{k_0/M_L,(k_0+k'_0)/M_L}(N)\right] = \delta^2(1-\ell)^2(k'_0/M_L)/(1-k'_0/M_L)$, and with (263) and the facts that $1-\ell \ge 1/2$ and $1/(1-k'_0/M_L) > 1$, we get

$$E_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] > \frac{\delta^{2}}{6}\ell(1-\ell) \quad .$$
(264)

Lemma 52 also gives

$$\operatorname{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] = \frac{2}{L^{2}}\left(\lambda_{0} + \delta \frac{1 - (2 - \ell)k'_{0}/M_{L}}{1 - k'_{0}/M_{L}}\right)^{2} + \frac{4}{L}\delta^{2}(1 - \ell)^{2}\frac{k'_{0}/M_{L}}{1 - k'_{0}/M_{L}}\left(\lambda_{0} + \delta \frac{1 - (2 - \ell)k'_{0}/M_{L}}{1 - k'_{0}/M_{L}}\right) , \quad (265)$$

and since $k'_0/M_L \leq \ell < 1$, notice that $0 < 1 - (2 - \ell)k'_0/M_L < 1 - k'_0/M_L$. Moreover using $k'_0/M_L \leq \ell$ again, one obtains

$$\frac{k_0'/M_L}{1 - k_0'/M_L} \le \frac{\ell}{1 - \ell} \quad . \tag{266}$$

Therefore, using (266),

$$\begin{aligned} \operatorname{Var}_{\lambda} \left[T'_{\frac{k_{0}}{M_{L}}, \frac{k_{0}+k'_{0}}{M_{L}}}(N) \right] &\leq \left(\frac{2}{L^{2}} (\lambda_{0}+\delta)^{2} + \frac{4}{L} (\lambda_{0}+\delta) \delta^{2} \ell (1-\ell) \right) \mathbb{1}_{\delta > 0} \\ &+ \left(\frac{2}{L^{2}} \lambda_{0}^{2} + \frac{4}{L} \lambda_{0} \delta^{2} \ell (1-\ell) \right) \mathbb{1}_{\delta < 0} \end{aligned}$$

hence

$$\operatorname{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] \leq \frac{2R^{2}}{L^{2}} + \frac{4R}{L}\delta^{2}\ell(1-\ell) \quad .$$
(267)

Now, on the one hand, notice that the assumption (260) ensures that

$$\begin{split} |\delta|\sqrt{\ell(1-\ell)} \ge \sqrt{\frac{60CR\log\left(2.77/u_{\alpha}^{(2)}\right)}{L}} + \left(\frac{2R}{\beta}\right)^{1/4} \frac{\sqrt{60C\log\left(2.77/u_{\alpha}^{(2)}\right)}}{L^{3/4}} \\ + 6\sqrt{C(M_L-1)} \frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L} + 2\sqrt{\frac{6R}{L\sqrt{\beta}}} , \end{split}$$

which entails

$$\delta^{2}\ell(1-\ell) \geq 60CR \frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L} + 60C\sqrt{\frac{2R}{\beta}} \frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L^{3/2}} + 36C(M_{L}-1)\frac{\log^{2}\left(2.77/u_{\alpha}^{(2)}\right)}{L^{2}} + \frac{24R}{\sqrt{\beta}L} \quad (268)$$

On the other hand, the same assumption (260) ensures that $|\delta|\sqrt{\ell(1-\ell)} \ge 24\sqrt{2R/(\beta L)}$, which leads to

$$\delta^2 \ell(1-\ell) \ge 24 \sqrt{\frac{2R}{\beta L}} |\delta| \sqrt{\ell(1-\ell)} \quad . \tag{269}$$

Therefore, (268) and (269) imply

$$\delta^{2}\ell(1-\ell) \geq 2 \max\left(C\left(30R\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L} + 30\sqrt{\frac{2R}{\beta}}\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L^{3/2}} + 18(M_{L}-1)\frac{\log^{2}\left(2.77/u_{\alpha}^{(2)}\right)}{L^{2}}\right) + \frac{12R}{\sqrt{\beta}L}, \ 12\sqrt{\frac{2R}{\beta L}}|\delta|\sqrt{\ell(1-\ell)}\right),$$

 \mathbf{SO}

$$\frac{\delta^2}{6}\ell(1-\ell) \ge C\left(5R\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L^{3/2}} + 3(M_L-1)\frac{\log^2\left(2.77/u_{\alpha}^{(2)}\right)}{L^2}\right) + \frac{2R}{\sqrt{\beta}L} + 2\sqrt{\frac{2R}{\beta}L}|\delta|\sqrt{\ell(1-\ell)} \quad (270)$$

Thus, with (264) and (267), the inequality (270) ensures that

$$E_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] \geq C\left(5R\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L} + 5\sqrt{\frac{2R}{\beta}}\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L^{3/2}} + 3(M_{L}-1)\frac{\log^{2}\left(2.77/u_{\alpha}^{(2)}\right)}{L^{2}}\right) + \sqrt{2\mathrm{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right]/\beta} \quad (271)$$

Furthermore, the inequality

$$P_{\lambda}\left(N_1 \ge LI(\lambda) + \sqrt{\frac{2LI(\lambda)}{\beta}}\right) \le \frac{\beta}{2}$$
,

combined with Lemma 53 and the upper bound $I(\lambda) \leq R$, leads to

$$P_{\lambda}\left(t'_{N_{1},\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}\left(1-u_{\alpha}^{(2)}\right) \ge Q(\alpha,\beta,L,k'_{0})\right) \le \frac{\beta}{2} , \qquad (272)$$

with

$$Q(\alpha, \beta, L, k'_{0}) = C \left(5R \frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L} + 5\sqrt{\frac{2R}{\beta}} \frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{L^{3/2}} + 3\max\left(\frac{k'_{0}/M_{L}}{1 - k'_{0}/M_{L}}, \frac{1 - k'_{0}/M_{L}}{k'_{0}/M_{L}}\right) \frac{\log^{2}\left(2.77/u_{\alpha}^{(2)}\right)}{L^{2}} \right) . \quad (273)$$

Since $k'_0/M_L \leq \ell \leq 1/2$, one has

$$\max\left(\frac{k_0'/M_L}{1-k_0'/M_L}, \frac{1-k_0'/M_L}{k_0'/M_L}\right) = \frac{1-k_0'/M_L}{k_0'/M_L} = \frac{M_L}{k_0'} - 1 \le M_L - 1 ,$$

and the inequality (271) leads to

$$E_{\lambda}\left[T'_{\frac{k_0}{M_L},\frac{k_0+k'_0}{M_L}}(N)\right] \ge Q(\alpha,\beta,L,k'_0) + \sqrt{2\operatorname{Var}_{\lambda}\left[T'_{\frac{k_0}{M_L},\frac{k_0+k'_0}{M_L}}(N)\right]/\beta} \quad .$$
(274)

We conclude with the following inequalities

$$\begin{split} P_{\lambda}\left(T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N) \leq t'_{N_{1},\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}\left(1-u_{\alpha}^{(2)}\right)\right) \\ &\leq P_{\lambda}\left(T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N) < Q(\alpha,\beta,L,k'_{0})\right) + P_{\lambda}\left(t'_{N_{1},\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}\left(1-u_{\alpha}^{(2)}\right) \geq Q(\alpha,\beta,L,k'_{0})\right) \\ &\leq P_{\lambda}\left(T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N) < Q(\alpha,\beta,L,k'_{0})\right) + \frac{\beta}{2} \text{ with } (272) \\ &\leq P_{\lambda}\left(T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N) - E_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] < -\sqrt{2\mathrm{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] / \beta}\right) + \frac{\beta}{2} \text{ with } (274) \\ &\leq \beta \text{ with the Bienayme-Chebyshev inequality }. \end{split}$$

Let us then consider the case where $\ell > 1/2$. We define $k_0 = \max(k \in \{0, \dots, M_L - 1\}, \tau \ge k/M_L)$ and $k'_0 = \min(k' \in \{1, \dots, M_L - k_0\}, (k_0 + k')/M_L \ge \tau + \ell)$ so that $k_0/M_L \le \tau < \tau + \ell \le (k_0 + k'_0)/M_L$. Notice that the condition (261) entails

$$1 - \ell > \frac{3}{M_L}$$
 (275)

Since by definition of k_0 and k'_0 , $\tau - k_0/M_L < 1/M_L$ and $(k_0 + k'_0)/M_L - (\tau + \ell) < 1/M_L$, then, with (275),

$$\frac{k_0'}{M_L} = \frac{k_0 + k_0'}{M_L} - (\tau + \ell) + \ell + \tau - \frac{k_0}{M_L} < 1/M_L + (1 - 3/M_L) + 1/M_L < 1 \quad .$$
(276)

In the same way, we can also get with (275)

$$1 - \frac{k_0'}{M_L} = 1 - \ell - \left(\left(\tau - \frac{k_0}{M_L} \right) + \left(\frac{k_0 + k_0'}{M_L} - (\tau + \ell) \right) \right) > 1 - \ell - \frac{2}{M_L} > \frac{1 - \ell}{3} \quad (277)$$

Furthermore, by definition of k_0 and k'_0 again, one has $k'_0/M_L \ge \ell$, so

$$\frac{1 - k_0'/M_L}{k_0'/M_L} \le \frac{1 - \ell}{\ell} \quad . \tag{278}$$

From Lemma 52, we deduce the expectation

$$E_{\lambda}\left[T'_{k_0/M_L,(k_0+k'_0)/M_L}(N)\right] = \delta^2 \ell^2 \left(1 - k'_0/M_L\right) / \left(k'_0/M_L\right) ,$$

which satisfies with (277) and $\ell > 1/2$

$$E_{\lambda}\left[T'_{k_0/M_L,(k_0+k'_0)/M_L}(N)\right] > \frac{\delta^2}{6}\ell(1-\ell)$$
.

Lemma 52 also gives the variance

$$\operatorname{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] = \frac{2}{L^{2}}\left(\lambda_{0} + \delta\ell \frac{1-k'_{0}/M_{L}}{k'_{0}/M_{L}}\right)^{2} + \frac{4}{L}\delta^{2}\ell^{2}\frac{1-k'_{0}/M_{L}}{k'_{0}/M_{L}}\left(\lambda_{0} + \delta\ell \frac{1-k'_{0}/M_{L}}{k'_{0}/M_{L}}\right) ,$$

which satisfies with (278)

$$\begin{aligned} \operatorname{Var}_{\lambda} \left[T'_{\frac{k_{0}}{M_{L}}, \frac{k_{0}+k'_{0}}{M_{L}}}(N) \right] &\leq \left(\frac{2}{L^{2}} (\lambda_{0}+\delta)^{2} + \frac{4}{L} \delta^{2} \ell (1-\ell) (\lambda_{0}+\delta) \right) \mathbb{1}_{\delta>0} \\ &+ \left(\frac{2}{L^{2}} \lambda_{0}^{2} + \frac{4}{L} \lambda_{0} \delta^{2} \ell (1-\ell) \right) \mathbb{1}_{\delta<0} \end{aligned}$$

thereby

$$\operatorname{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] \leq \frac{2R^{2}}{L^{2}} + \frac{4R}{L}\delta^{2}\ell(1-\ell)$$

Finally, since $k'_0/M_L \ge \ell > 1/2$, (276) leads to

$$\max\left(\frac{k_0'/M_L}{1-k_0'/M_L}, \frac{1-k_0'/M_L}{k_0'/M_L}\right) = \frac{k_0'/M_L}{1-k_0'/M_L} = \frac{k_0'}{M_L-k_0'} \le M_L - 1$$

As in the above case where $\ell \leq 1/2$, we can therefore prove that the assumption (260) leads to (274), that is

$$E_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right] \geq Q(\alpha,\beta,L,k'_{0}) + \sqrt{2\mathrm{Var}_{\lambda}\left[T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N)\right]/\beta} \ ,$$

with $Q(\alpha, \beta, L, k'_0)$ is defined by (273). We then use the same final arguments as in this case, to conclude that

$$P_{\lambda}\left(T'_{\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}(N) \leq t'_{N_{1},\frac{k_{0}}{M_{L}},\frac{k_{0}+k'_{0}}{M_{L}}}\left(1-u_{\alpha}^{(2)}\right)\right) \leq \beta \quad .$$

Coming back to the assumption (260), we can thus affirm that

$$SR_{\beta}\left(\phi_{9/10,\alpha}^{u(2)}, \mathcal{S}_{\cdot,\cdots,\cdots}^{u}[R]\right) \leq \max\left(\sqrt{\frac{60CR\log\left(2.77/u_{\alpha}^{(2)}\right)}{L}} + \left(\frac{2R}{\beta}\right)^{1/4} \frac{\sqrt{60C\log\left(2.77/u_{\alpha}^{(2)}\right)}}{L^{3/4}} + 6\sqrt{C}\frac{\log\left(2.77/u_{\alpha}^{(2)}\right)}{\sqrt{L\log L}} + 2\sqrt{\frac{6R}{L\sqrt{\beta}}}, \ 24\sqrt{\frac{2R}{\beta L}}, \ R\sqrt{\frac{3\log L}{L}}\right), \ (279)$$

which, since $M_L = \lceil L/\log L \rceil$ and $u_{\alpha}^{(2)} = 2\alpha/(M_L(M_L+1)-2)$, leads to the result stated in Proposition 41 for $\phi_{9/10,\alpha}^{u(2)}$ and the set $\mathcal{S}^u_{\cdot,\cdot,\cdot,\cdots}[R]$ of the alternative [Alt^u.10].

Since $\mathcal{S}^{u}_{\delta^*,..,..}[R] \subset \mathcal{S}^{u}_{\cdot,..,..}[R]$ for any δ^* in $(-R,R) \setminus \{0\}$, the same result holds for $\phi^{u(2)}_{9/10,\alpha}$ and the set $\mathcal{S}^{u}_{\delta^*,..,..}[R]$ of the alternative [Alt^u.9]. Again, notice that in this case, the constant $C(\alpha, \beta, R)$ could be refined thanks to the knowledge of δ^* , which justifies the formulation with $C(\alpha, \beta, R, \delta^*)$ instead of $C(\alpha, \beta, R)$ in Proposition 41.

6 Key arguments for minimax separation rates lower bounds and technical results for minimax separation rates upper bounds

6.1 Key arguments for nonasymptotic minimax separation rates lower bounds

The arguments given below are repeated from [8], who adapted the asymptotic Bayesian approach of Ingster [65] to obtain lower bounds for minimax rates of testing in a nonasymptotic perspective, and [50] who derived these lower bounds in the Poisson framework. We only recall these arguments without the proofs for the sake of clarity and completeness. Let us recall the notation of the introduction, where $N = (N_t)_{t \in [0,1]}$ is a Poisson process observed on [0, 1], with intensity λ w.r.t. some measure Λ on [0, 1], and whose distribution is denoted by P_{λ} , and where S_0 is either the set $S_0[\lambda_0] = {\lambda_0}$ of a single known constant intensity λ_0 on [0, 1], or the set $S_0^u[R]$ of all constant intensities on [0, 1] bounded by R. We consider the problem of testing (H_0) " $\lambda \in S_0$ " versus (H_1) " $\lambda \in S$ ", where S is a set of possible alternative intensities, from the nonasymptotic minimax point of view based on the definition of mSR_{α,β}(S) given in (2). **Lemma 42.** For r > 0 and any subspace S of $\mathbb{L}_2([0,1])$, set $S_r = \{\lambda \in S, d_2(\lambda, S_0) \ge r\}$. For α in (0,1), we define

$$\rho_{\alpha}(\mathcal{S}_{r}) = \inf_{\phi_{\alpha}, \sup_{\lambda \in \mathcal{S}_{0}} P_{\lambda}(\phi_{\alpha}(N)=1) \leq \alpha} \quad \sup_{\lambda \in \mathcal{S}_{r}} P_{\lambda}(\phi_{\alpha}(N)=0) .$$

(i) If $\rho_{\alpha}(\mathcal{S}_r) \geq \beta$ then $\operatorname{mSR}_{\alpha,\beta}(\mathcal{S}) \geq r$. (ii) For all subsets \mathcal{S}' and \mathcal{S} of $\mathbb{L}_2([0,1])$ such that $\mathcal{S}' \subset \mathcal{S}$, then $\rho_{\alpha}(\mathcal{S}'_r) \leq \rho_{\alpha}(\mathcal{S}_r)$ and $\operatorname{mSR}_{\alpha,\beta}(\mathcal{S}') \leq \operatorname{mSR}_{\alpha,\beta}(\mathcal{S})$.

Lemma 43 (Minimax lower bounds and Bayesian approach). Let μ be a probability measure on S_r and define P_{μ} the mixture probability by

$$P_{\mu} = \int P_{\lambda} d\mu(\lambda).$$

Then

$$\rho_{\alpha}(\mathcal{S}_{r}) \geq 1 - \alpha - \frac{1}{2} \left(\inf_{\lambda_{0} \in \mathcal{S}_{0}} E_{\lambda_{0}} \left[\left(\frac{dP_{\mu}}{dP_{\lambda_{0}}} \right)^{2} (N) \right] - 1 \right)^{1/2}$$

As a consequence, if $\alpha + \beta < 1$ and $\inf_{\lambda_0 \in S_0} E_{\lambda_0} \left[(dP_{\mu}/dP_{\lambda_0})^2 (N) \right] \le 1 + 4(1 - \alpha - \beta)^2$ then

$$\rho_{\alpha}(\mathcal{S}_r) \geq \beta \quad and \quad \mathrm{mSR}_{\alpha,\beta}(\mathcal{S}) \geq r \ .$$

6.2 Technical results for minimax separation rates upper bounds

Lemma 44 (Moments of a Poisson distribution). Let X be a Poisson distributed random variable with constant intensity $\xi > 0$.

1. The Laplace transform for X is given by

$$\forall c \in \mathbb{R}, \quad \mathbb{E}\left[\exp(cX)\right] = \exp[\xi(\exp(c) - 1)].$$

2. The first moments of X are given by the following formulas:

$$\mathbb{E}[X] = \xi, \ \mathbb{E}[X^2] = \xi^2 + \xi, \ \mathbb{E}[X^3] = \xi^3 + 3\xi^2 + \xi, \ \mathbb{E}[X^4] = \xi^4 + 6\xi^3 + 7\xi^2 + \xi,$$

and its central moments are given by $\mathbb{E}[(X - \mathbb{E}[X])^3] = \xi$ and $\mathbb{E}[(X - \mathbb{E}[X])^4] = 3\xi^2 + \xi$.

Lemma 45 (Rough quantile bounds for the Poisson distribution). The u-quantile $p_{\xi}(u)$ of the Poisson distribution with parameter ξ satisfies

$$\xi - \sqrt{\xi/u} \le p_{\xi}(u) \le \xi + \sqrt{\xi/(1-u)}$$
 (280)

Proof. Let N be a Poisson random variable with parameter ξ . Using the Bienayme-Chebyshev inequality, we obtain for all ε in (0, u)

$$\mathbb{P}\left(N \leq \xi - \sqrt{\frac{\xi}{u - \varepsilon}}\right) \leq u - \varepsilon < u \text{ and } \mathbb{P}\left(N > \xi + \sqrt{\frac{\xi}{u}}\right) \leq u,$$

which leads to the expected result by letting ε tend to zero.

Lemma 46 (Sharp quantile bounds for the Poisson distribution). The u-quantile $p_{\xi}(u)$ of the Poisson distribution with parameter ξ satisfies

$$\left(\xi - \xi g^{-1}\left(\frac{\log(1/u)}{\xi}\right)\right) \vee 0 \le p_{\xi}(u) \le \xi + \xi g^{-1}\left(\frac{\log(1/(1-u))}{\xi}\right) \quad .$$
(281)

Proof. Let N be a Poisson random variable with parameter ξ . The Cramér-Chernov inequality applied to this Poisson random variable gives that for all x > 0,

$$\mathbb{P}(N-\xi > x) \le \exp\left(-\xi g(x/\xi)\right)$$

and for $0 < x < \xi$,

$$\mathbb{P}(N-\xi \le -x) \le \exp\left(-\xi g(-x/\xi)\right) .$$

The first exponential inequality gives

$$\mathbb{P}\left(N > \xi + \xi g^{-1}\left(\frac{\log(1/(1-u))}{\xi}\right)\right) \le u ,$$

which leads to the upper bound of Lemma 46. The second inequality gives that for all $u > e^{-\xi}$ and all ε such that $u > u - \varepsilon > e^{-\xi}$, using (81),

$$\mathbb{P}\left(N \leq \xi - \xi g^{-1}\left(\frac{\log(1/(u-\varepsilon))}{\xi}\right)\right) \leq \mathbb{P}\left(N \leq \xi + \xi g_{|]-1,0[}^{-1}\left(\frac{\log(1/(u-\varepsilon))}{\xi}\right)\right)$$
$$\leq u - \varepsilon < u \ .$$

By letting ε tend to zero and by continuity of g^{-1} , this leads to the lower bound of Lemma 46 for any $u > e^{-\xi}$. Since $0 < g(1) \leq 1$ and g^{-1} is increasing on $]0, +\infty[$, $1 = g^{-1}(g(1)) \leq g^{-1}(1)$, hence for $0 \leq u \leq e^{-\xi}$,

$$g^{-1}\left(\frac{\log(1/u)}{\xi}\right) \ge g^{-1}(1) \ge 1$$

Therefore in this case $\xi - \xi g^{-1} \left(\frac{\log(1/u)}{\xi} \right) \leq 0$. The conclusion follows from the remark that for any u in [0, 1], $p_{\xi}(u) \geq 0$ as some quantile of a Poisson distribution.

Lemma 47 (Expectation and variance of $T_{\tau_1,\tau_2}(N)$). Let τ_1,τ_2 be in (0,1) such that $0 \leq \tau_1 < \tau_2 \leq 1, \lambda_0 > 0$ and $T_{\tau_1,\tau_2}(N)$ be defined by (12). Assume that $N(\tau_1,\tau_2]$ follows a Poisson distribution with parameter Lx with x > 0. Then

$$\mathbb{E}[T_{\tau_1,\tau_2}(N)] = \left(\frac{x}{\sqrt{\tau_2 - \tau_1}} - \lambda_0 \sqrt{\tau_2 - \tau_1}\right)^2 , \qquad (282)$$

and

$$\operatorname{Var}(T_{\tau_1,\tau_2}(N)) = \frac{4x}{L} \left(\frac{x}{\tau_2 - \tau_1} - \lambda_0\right)^2 + \frac{2}{L^2} \frac{x^2}{(\tau_2 - \tau_1)^2} \quad (283)$$

Proof. Set m_i the moment of order i of $N(\tau_1, \tau_2]$. Since $N(\tau_1, \tau_2]$ follows a Poisson distribution, $m_2 = m_1 + m_1^2$ and

$$\mathbb{E}[T_{\tau_1,\tau_2}(N)] = \frac{1}{L^2(\tau_2 - \tau_1)} m_1^2 - \frac{2\lambda_0}{L} m_1 + \lambda_0^2(\tau_2 - \tau_1),$$

with $m_1 = Lx$. This straightforwardly leads to the first statement of Lemma 47. Notice now that

$$T_{\tau_1,\tau_2}(N) = \frac{1}{L^2(\tau_2 - \tau_1)} \left((N(\tau_1, \tau_2] - m_1)^2 + m_1^2 + (2m_1 - 1)(N(\tau_1, \tau_2] - m_1) - m_1 \right) - \frac{2\lambda_0}{L} (N(\tau_1, \tau_2] - m_1) - \frac{2\lambda_0}{L} m_1 + \lambda_0^2(\tau_2 - \tau_1) ,$$

which entails

$$T_{\tau_1,\tau_2}(N) - \mathbb{E}[T_{\tau_1,\tau_2}(N)] = \frac{1}{L^2(\tau_2 - \tau_1)} \left((N(\tau_1,\tau_2] - m_1)^2 + (2m_1 - 1)(N(\tau_1,\tau_2] - m_1) - m_1 \right) - \frac{2\lambda_0}{L} (N(\tau_1,\tau_2] - m_1).$$

Considering the moment c_i of order *i* of the centered variable $N(\tau_1, \tau_2] - m_1$, one obtains

$$\operatorname{Var}(T_{\tau_1,\tau_2}(N)) = \frac{1}{L^4(\tau_2 - \tau_1)^2} \left(c_4 + (2m_1 - 1)^2 c_2 + m_1^2 + 2(2m_1 - 1)c_3 - 2m_1 c_2 \right) + \frac{4\lambda_0^2}{L^2} c_2 - \frac{4\lambda_0}{L^3(\tau_2 - \tau_1)} \left(c_3 + (2m_1 - 1)c_2 \right).$$

Applying Lemma 44 finally leads to the second statement of Lemma 47.

Lemma 48 (Quantile bound for $T_{\tau_1,\tau_2}(N)$). Let $\lambda_0 > 0$, u in (0,1) and assume that N is a homogeneous Poisson process of intensity λ_0 with respect to the measure Λ . Let $t_{\lambda_0,\tau_1,\tau_2}(1-u)$ be the (1-u)-quantile of $T_{\tau_1,\tau_2}(N)$ defined by (12). Then for all $0 \leq \tau_1 < \tau_2 \leq 1$,

.

$$t_{\lambda_0,\tau_1,\tau_2}(1-u) \le 2\lambda_0^2(\tau_2-\tau_1)\left(g^{-1}\left(\frac{\log(3/u)}{\lambda_0 L(\tau_2-\tau_1)}\right)\right)^2$$
,

where g is defined by (79).

Proof. Notice first that under the assumption of Lemma 48, $T_{\tau_1,\tau_2}(N)$ can be written as

$$T_{\tau_1,\tau_2}(N) = \left(\int_0^1 \frac{\varphi_{(\tau_1,\tau_2]}(t)}{L} (dN_t - \lambda_0 L dt)\right)^2 - \int_0^1 \left(\frac{\varphi_{(\tau_1,\tau_2]}(t)}{L}\right)^2 dN_t$$

with $\varphi_{(\tau_1,\tau_2]} = \mathbb{1}_{(\tau_1,\tau_2]}/\sqrt{\tau_2 - \tau_1}$. Applying the exponential inequality of Theorem 2 in [71], we obtain for all x > 0

$$\mathbb{P}(T_{\tau_1,\tau_2}(N) > x) \le 3 \exp\left(-\frac{\|H_{\tau_1,\tau_2}\|_{2,L}^2}{\|H_{\tau_1,\tau_2}\|_{\infty}^2}g\left(\frac{\|H_{\tau_1,\tau_2}\|_{\infty}}{\|H_{\tau_1,\tau_2}\|_{2,L}^2}\sqrt{\frac{x}{2}}\right)\right) ,$$

where $H_{\tau_1,\tau_2}(t) = \varphi_{(\tau_1,\tau_2]}(t)/L$, g is defined by (79) and $||H_{\tau_1,\tau_2}||_{2,L}$ is the \mathbb{L}_2 -norm of H_{τ_1,τ_2} in $\mathbb{L}^2([0,1], \lambda_0 L dt)$, that is $||H_{\tau_1,\tau_2}||_{2,L}^2 = \int_0^1 |H_{\tau_1,\tau_2}(t)|^2 \lambda_0 L dt = \lambda_0/L$. This leads to

$$\mathbb{P}\left(T_{\tau_1,\tau_2}(N) > x\right) \le 3\exp\left(-\lambda_0 L(\tau_2 - \tau_1)g\left(\frac{1}{\lambda_0\sqrt{\tau_2 - \tau_1}}\sqrt{\frac{x}{2}}\right)\right)$$

We therefore obtain $\mathbb{P}(T_{\tau_1,\tau_2}(N) > x) \le u$ if $x \ge 2\lambda_0^2(\tau_2 - \tau_1) \left(g^{-1} \left(\log(3/u)/(\lambda_0 L(\tau_2 - \tau_1))\right)\right)^2$ and the result follows. **Lemma 49** (Quantile bound for max / min_{\tau \in [0,1-\ell^*]} N(\tau, \tau + \ell^*]). Let $L \ge 1$. The $(1 - \alpha)$ -quantile $p^+_{\lambda_0,\ell^*}(1 - \alpha)$ of max_{\tau \in [0,1-\ell^*]} N(\tau, \tau + \ell^*] under (H_0) satisfies

$$p_{\lambda_0,\ell^*}^+(1-\alpha) \le \lambda_0 L \ell^* + 2\lambda_0 L g^{-1} \left(\frac{\log(2/\alpha)}{\lambda_0 L}\right)$$
.

The α -quantile $p_{\lambda_0,\ell^*}(\alpha)$ of $\min_{\tau\in[0,1-\ell^*]} N(\tau,\tau+\ell^*]$ under (H_0) satisfies

$$p_{\lambda_0,\ell^*}^-(\alpha) \ge \lambda_0 L \ell^* - 2\lambda_0 L g^{-1} \left(\frac{\log(2/\alpha)}{\lambda_0 L}\right) ,$$

where g is defined by (79).

Proof. We define $M_s^t = \int_s^t (dN_u - \lambda_0 L du)$ for all s, t > 0. Let x > 0 be such that

$$x \ge \lambda_0 L\ell^* + 2\lambda_0 Lg^{-1} \left(\frac{\log(2/\alpha)}{\lambda_0 L}\right) \quad . \tag{284}$$

Since

$$P_{\lambda_0}\left(\max_{\tau\in[0,1-\ell^*]}N(\tau,\tau+\ell^*]>x\right) = P_{\lambda_0}\left(\max_{\tau\in[0,1-\ell^*]}M_{\tau}^{\tau+\ell^*}>x-\lambda_0L\ell^*\right)$$
$$\leq P_{\lambda_0}\left(\max_{s,t\in[0,1]}|M_s^t|>x-\lambda_0L\ell^*\right) ,$$

Theorem 4 in [71] ensures that

$$P_{\lambda_0}\left(\max_{s,t\in[0,1]}|M_s^t| > x - \lambda_0 L\ell^*\right) \le 2\exp\left(-\lambda_0 Lg\left(\frac{x - \lambda_0 L\ell^*}{2\lambda_0 L}\right)\right)$$

Then (284) entails

$$2\exp\left(-\lambda_0 Lg\left(\frac{x-\lambda_0 L\ell^*}{2\lambda_0 L}\right)\right) \le \alpha \quad ,$$

leading to $P_{\lambda_0}(\max_{\tau \in [0,1-\ell^*]} N(\tau, \tau + \ell^*] > x) \leq \alpha$. The $(1-\alpha)$ -quantile $p^+_{\lambda_0,\ell^*}(\alpha)$ of $\max_{\tau \in [0,1-\ell^*]} N(\tau, \tau + \ell^*]$ under (H_0) therefore satisfies $p^+_{\lambda_0,\ell^*}(1-\alpha) \leq x$ for every x such that (284) holds. In particular,

$$p_{\lambda_0,\ell^*}^+(1-\alpha) \le \lambda_0 L\ell^* + 2\lambda_0 Lg^{-1}\left(\frac{\log(2/\alpha)}{\lambda_0 L}\right)$$

Let us consider now x in \mathbb{R} and ε in (0,1) satisfying

$$x \le \lambda_0 L \ell^* - 2\lambda_0 L g^{-1} \left(\frac{\log(2/(\alpha(1-\varepsilon)))}{\lambda_0 L} \right) \quad .$$
(285)

Using (285) and Theorem 4 in [71] again, we obtain

$$P_{\lambda_0}\left(\min_{\tau\in[0,1-\ell^*]} N(\tau,\tau+\ell^*] \le x\right) = P_{\lambda_0}\left(\max_{\tau\in[0,1-\ell^*]} -M_{\tau}^{\tau+\ell^*} \ge \lambda_0 L\ell^* - x\right)$$
$$\le P_{\lambda_0}\left(\max_{s,t\in[0,1]} |M_s^t| \ge \lambda_0 L\ell^* - x\right)$$
$$\le 2\exp\left(-\lambda_0 Lg\left(\frac{\lambda_0 L\ell^* - x}{2\lambda_0 L}\right)\right)$$
$$\le \alpha(1-\varepsilon)$$
$$< \alpha .$$

Thus the α -quantile $p_{\lambda_0,\ell^*}(\alpha)$ of $\min_{\tau \in [0,1-\ell^*]} N(\tau,\tau+\ell^*]$ under (H_0) satisfies $p_{\lambda_0,\ell^*}(\alpha) > x$ for every x such that (285) holds. In particular

$$p_{\lambda_0,\ell^*}^-(\alpha) > \lambda_0 L \ell^* - 2\lambda_0 L g^{-1} \left(\frac{\log(2/(\alpha(1-\varepsilon)))}{\lambda_0 L} \right)$$

for every ε in (0, 1). The result then follows by continuity of g^{-1} .

Lemma 50 (Quantile bound for $\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N)$). Let $\gamma > 0$ and $L \ge 1$. Let $(N_t^{\lambda_0})_{t\ge 0}$ be an homogeneous Poisson process with a known constant intensity $\lambda_0 L > 0$ w.r.t. the Lebesgue measure. Then, the u-quantile of the supremum $\sup_{t\ge 0} (N_t^{\lambda_0} - (\lambda_0 + \gamma)Lt)$ does not depend on L, and will therefore be denoted by $s_{\lambda_0,\gamma}^+(u)$. Now considering $s_{\lambda_0,\delta^*,\tau^*,t}^+(u)$, the u-quantile under (H_0) of the statistic $\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N)$ with $S_{\delta^*,\tau^*,\tau^*+\ell}(N)$ defined by (21), we have for all $L \ge 1$

$$\begin{cases} s^{+}_{\lambda_{0},\delta^{*},\tau^{*},L}(u) \leq s^{+}_{\lambda_{0},\delta^{*}/2}(u) & \text{when } \delta^{*} > 0 , \\ s^{+}_{\lambda_{0},\delta^{*},\tau^{*},L}(u) \leq \frac{\log(1/(1-u))}{\log(\lambda_{0}/(\lambda_{0}-|\delta^{*}|/2))} & \text{when } \delta^{*} \in (-\lambda_{0}^{*},0) . \end{cases}$$
(286)

Proof. Equation (7) in [89] directly enables to state the first part of the result. Under (H_0) , since the processes $(N(\tau^*, \tau^* + \ell])_{\ell \in (0, 1-\tau^*]}$ and $(N(0, \ell])_{\ell \in (0, 1-\tau^*]}$ are left-continuous and have the same finite dimensional laws, we get

$$\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N) \stackrel{d}{=} \sup_{\ell \in (0,1-\tau^*)} \left(\operatorname{sgn}(\delta^*)(N(0,\ell] - \lambda_0 \ell L) - \frac{|\delta^*|}{2} \ell L \right).$$

Assume first that $\delta^* > 0$. We compute

$$P_{\lambda_0} \Big(\sup_{\ell \in (0, 1 - \tau^*)} S_{\delta^*, \tau^*, \tau^* + \ell}(N) > s^+_{\lambda_0, \delta^*/2}(u) \Big) \le \mathbb{P} \left(\sup_{t \in [0, +\infty)} \left(N^{\lambda_0}(0, t] - \left(\lambda_0 + \frac{\delta^*}{2} \right) Lt \right) > s^+_{\lambda_0, \delta^*/2}(u) \right) \le 1 - u ,$$

by definition of $s^+_{\lambda_0,\delta^*/2}(u)$. This allows to conclude that the first part of (286) holds. Assume then that δ^* belongs to $(-\lambda_0, 0)$. For all x > 0,

$$P_{\lambda_0}\left(\sup_{\ell\in(0,1-\tau^*)}S_{\delta^*,\tau^*,\tau^*+\ell}(N) > x\right) = P_{\lambda_0}\left(\sup_{\ell\in(0,1-\tau^*)}\left(\left(\lambda_0 - \frac{|\delta^*|}{2}\right)\ell L - N(0,\ell]\right) > x\right)$$

Theorem 3 in [89] then entails

$$P_{\lambda_0}\left(\sup_{\ell\in(0,1-\tau^*)}S_{\delta^*,\tau^*,\tau^*+\ell}(N)>x\right)\leq \exp\left(-\omega x\right) ,$$

where ω is the largest real root of the equation $\lambda_0(1 - e^{-\omega}) = \omega (\lambda_0 - |\delta^*|/2)$. Notice that $\omega > \log (\lambda_0/(\lambda_0 - |\delta^*|/2))$. Then, correctly choosing x in the above exponential inequality leads to

$$P_{\lambda_0}\left(\sup_{\ell \in (0,1-\tau^*)} S_{\delta^*,\tau^*,\tau^*+\ell}(N) > \frac{\log\left(1/(1-u)\right)}{\log\left(\lambda_0/(\lambda_0-|\delta^*|/2)\right)}\right) \le 1-u ,$$

which implies the second part of (286).

Lemma 51 (Quantile bound for $\sup_{\tau \in (0,1)} S_{\delta^*,\tau,1}(N)$). Let $L \ge 1$. With the same notation as in Lemma 50 and $S_{\delta^*,\tau,1}(N)$ defined by (21), the u-quantile $s^+_{\lambda_0,\delta^*,L}(u)$ of the statistic $\sup_{\tau \in (0,1)} S_{\delta^*,\tau,1}(N)$ under (H_0) satisfies

$$\begin{cases} s^+_{\lambda_0,\delta^*,L}(u) \leq s^+_{\lambda_0,\delta^*/2}(u) \quad when \ \delta^* > 0 \\ s^+_{\lambda_0,\delta^*,L}(u) \leq \frac{\log(1/(1-u))}{\log(\lambda_0/(\lambda_0-|\delta^*|/2))} \quad when \ \delta^* \in (-\lambda_0^*,0) \end{cases}.$$

$$(287)$$

Proof. Under (H_0) , since the processes $(N(\tau, 1])_{\tau \in (0,1)}$ and $(N(0, 1 - \tau])_{\tau \in (0,1)}$ are leftcontinuous and have the same finite dimensional laws, we get

$$\sup_{\tau \in (0,1)} S_{\delta^*,\tau,1}(N) \stackrel{d}{=} \sup_{\tau \in (0,1)} \left(\operatorname{sgn}(\delta^*)(N(0,\tau] - \lambda_0 \tau L) - \frac{|\delta^*|}{2} \tau L \right)$$

The result then follows from the same arguments as in the proof of Lemma 50, by noticing that when $\delta^* > 0$,

$$P_{\lambda_0}\left(\sup_{\tau\in(0,1)}S_{\delta^*,\tau,1}(N) > s^+_{\lambda_0,\delta^*/2}(u)\right) \le \mathbb{P}\left(\sup_{t\in[0,+\infty)}\left(N^{\lambda_0}(0,t] - \left(\lambda_0 + \frac{\delta^*}{2}\right)Lt\right) > s^+_{\lambda_0,\delta^*/2}(u)\right)$$

and when δ^* belongs to $(-\lambda_0, 0)$,

$$P_{\lambda_0}\left(\sup_{\tau\in(0,1)}S_{\delta^*,\tau,1}(N) > x\right) = P_{\lambda_0}\left(\sup_{t\in(0,1)}\left(\left(\lambda_0 - \frac{|\delta^*|}{2}\right)tL - N(0,t]\right) > x\right)$$

for all x > 0.

Lemma 52 (Expectation and variance of $T'_{\tau_1,\tau_2}(N)$). Let τ_1 and τ_2 in (0,1] such that $0 < \tau_1 < \tau_2 \leq 1$ and let N be a Poisson process such that $N(0,\tau_1]$, $N(\tau_1,\tau_2]$, and $N(\tau_2,1]$ follow a Poisson distribution with respective parameters Lx > 0, Ly > 0 and Lz > 0. Considering the statistic T'_{τ_1,τ_2} defined by (46), one has

$$\mathbb{E}[T'_{\tau_1,\tau_2}(N)] = \left(\sqrt{\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1}}(x + z) - \sqrt{\frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1}}y\right)^2 \quad , \tag{288}$$

and

$$\operatorname{Var}(T_{\tau_1,\tau_2}'(N)) = \frac{2}{L^2} \left(\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} (x + z) + \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} y \right)^2 + \frac{4}{L} \left(\frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \right)^2 \left(y - \frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} (x + z) \right)^2 \left(\left(\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \right)^2 (x + z) + y \right) \quad . \quad (289)$$

Proof. Notice that the statistic T' can be written as

$$\begin{aligned} T'_{\tau_1,\tau_2}(N) &= \frac{1}{L^2} \Biggl[\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \left((N(0,\tau_1] + N(\tau_2,1])^2 - (N(0,\tau_1] + N(\tau_2,1]) \right) \\ &+ \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \left(N(\tau_1,\tau_2]^2 - N(\tau_1,\tau_2] \right) - 2N(\tau_1,\tau_2] \left(N(0,\tau_1] + N(\tau_2,1] \right) \Biggr] \end{aligned}$$

Set, as in the proof of Lemma 47, m_i and \bar{m}_i the moments of order *i* of $N(\tau_1, \tau_2]$ and $(N(0, \tau_1] + N(\tau_2, 1])$ respectively, and c_i and \bar{c}_i the corresponding centered moments of order *i*. Then, by independence of $N(0, \tau_1]$, $N(\tau_1, \tau_2]$ and $N(\tau_2, 1]$, and since $m_2 = m_1^2 + m_1$, $\bar{m}_2 = \bar{m}_1^2 + \bar{m}_1$ with $m_1 = Ly$ and $\bar{m}_1 = L(x + z)$,

$$\mathbb{E}[T_{\tau_1,\tau_2}'(N)] = \frac{1}{L^2} \left[\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} (\bar{m}_2 - \bar{m}_1) + \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} (m_2 - m_1) - 2m_1 \bar{m}_1 \right]$$
$$= \frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} (x + z)^2 + \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} y^2 - 2y(x + z) ,$$

which gives (288). Moreover,

$$T_{\tau_1,\tau_2}'(N) = \frac{1}{L^2} \left[\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \left((N(0,\tau_1] + N(\tau_2,1])^2 - (N(0,\tau_1] + N(\tau_2,1]) \right) + \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \left(N(\tau_1,\tau_2]^2 - N(\tau_1,\tau_2] \right) - 2N(\tau_1,\tau_2] \left(N(0,\tau_1] + N(\tau_2,1] \right) \right],$$

and then

$$\begin{aligned} T'_{\tau_1,\tau_2}(N) &- \mathbb{E}[T'_{\tau_1,\tau_2}(N)] = \frac{1}{L^2} \Bigg[\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \Big((N(0,\tau_1] + N(\tau_2,1] - \bar{m}_1)^2 \\ &+ (2\bar{m}_1 - 1) \left(N(0,\tau_1] + N(\tau_2,1] - \bar{m}_1) - \bar{m}_1 \right) \\ &+ \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \Big((N(\tau_1,\tau_2] - m_1)^2 + (2m_1 - 1)(N(\tau_1,\tau_2] - m_1) - m_1 \Big) \\ &- 2(N(\tau_1,\tau_2] - m_1) \Big((N(0,\tau_1] + N(\tau_2,1] - \bar{m}_1) + \bar{m}_1 \Big) \\ &- 2m_1 \left(N(0,\tau_1] + N(\tau_2,1] - \bar{m}_1 \right) \Bigg] . \end{aligned}$$

This entails

$$\begin{aligned} \operatorname{Var}(T'_{\tau_1,\tau_2}(N)) &= \frac{1}{L^4} \left[\left(\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \right)^2 \left(\bar{c}_4 + (2\bar{m}_1 - 1)^2 \bar{c}_2 + \bar{m}_1^2 + 2(2\bar{m}_1 - 1)\bar{c}_3 - 2\bar{m}_1 \bar{c}_2 \right) \\ &+ \left(\frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \right)^2 \left(c_4 + (2m_1 - 1)^2 c_2 + m_1^2 + 2(2m_1 - 1)c_3 - 2m_1 c_2 \right) \\ &+ 4 \left(c_2 \bar{m}_2 + m_1^2 \bar{c}_2 \right) + 2(c_2 - m_1) (\bar{c}_2 - \bar{m}_1) - 4m_1 \frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \left(\bar{c}_3 + (2\bar{m}_1 - 1)\bar{c}_2 \right) \\ &- 4\bar{m}_1 \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \left(c_3 + (2m_1 - 1)c_2 \right) \right] . \end{aligned}$$

This leads using Lemma 44 to

$$\begin{aligned} \operatorname{Var}(T_{\tau_1,\tau_2}'(N)) &= \frac{1}{L^4} \left[\left(\frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \right)^2 \left(4L^3(x+z)^3 + 2L^2(x+z)^2 \right) \\ &+ \left(\frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \right)^2 \left(4L^3y^3 + 2L^2y^2 \right) \\ &+ 4 \left(Ly(L^2(x+z)^2 + L(x+z)) + L^2y^2L(x+z) \right) \\ &- 4Ly \frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \left(L(x+z) + (2L(x+z) - 1)L(x+z) \right) \\ &- 4L(x+z) \frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} \left(Ly + (2Ly - 1)Ly \right) \right]. \end{aligned}$$

The second statement of Lemma 52 given in (289) finally follows from direct computations. $\hfill \Box$

Lemma 53 (Conditional quantile bound for $T'_{\tau_1,\tau_2}(N)$). Assume that N is a homogeneous Poisson process with a constant intensity λ_0 with respect to the measure $\Lambda = Ldt$. For τ_1 and τ_2 in (0,1) such that $0 < \tau_1 < \tau_2 \leq 1$, u in (0,1) and n in \mathbb{N} , let $t'_{n,\tau_1,\tau_2}(1-u)$ the (1-u)-quantile of the conditional distribution of $T'_{\tau_1,\tau_2}(N)$ defined by (46) given the event $\{N_1 = n\}$. Then

$$t'_{n,\tau_1,\tau_2}(1-u) \leq \frac{C}{L^2} \left(5n \log\left(\frac{2.77}{u}\right) + 3 \max\left(\frac{1-\tau_2+\tau_1}{\tau_2-\tau_1}, \frac{\tau_2-\tau_1}{1-\tau_2+\tau_1}\right) \log^2\left(\frac{2.77}{u}\right) \right) ,$$

with $C = \min_{\varepsilon>0} \left(e(1+\varepsilon^{-1})^2 (2.5+32\varepsilon^{-1}) + \left((2\sqrt{2}(2+\varepsilon+\varepsilon^{-1})) \vee ((1+\varepsilon)^2)/\sqrt{2} \right) \right) .$

Proof. For $n \geq 2$ and conditionally on the event $\{N_1 = n\}$, the points of the process N obey the same law as a n-sample (U_1, \ldots, U_n) of i.i.d. random variables uniformly distributed on (0, 1). $t'_{n,\tau_1,\tau_2}(1-u)$ is thus equal to the (1-u)-quantile of the following U-statistic of order 2

$$T'_{n,L,\tau^1,\tau^2} = \frac{1}{L^2} \sum_{i \neq j=1}^n \psi_{\tau_1,\tau_2}(U_i)\psi_{\tau_1,\tau_2}(U_j) = \sum_{i=2}^n \sum_{j=1}^{i-1} H_{L,\tau_1,\tau_2}(U_i,U_j)$$

where $H_{L,\tau_1,\tau_2}(x,y) = 2\psi_{\tau_1,\tau_2}(x)\psi_{\tau_1,\tau_2}(y)/L^2$ for any x and y in [0, 1].

Since for all $0 \leq \tau_1 < \tau_2 \leq 1$, ψ_{τ_1,τ_2} is orthogonal to ψ_0 (in $\mathbb{L}^2([0,1])$), the variables $\psi_{\tau_1,\tau_2}(U_i)$ are centered and we can apply Theorem 3.4 in [62]. We obtain that there exists some absolute constant C > 0 such that for all x > 0 and for all $n \geq 2$

$$\mathbb{P}\left(T'_{n,L,\tau^{1},\tau^{2}} \ge C(A_{1}\sqrt{x} + A_{2}x + A_{3}x^{3/2} + A_{4}x^{2})\right) \le 2.77e^{-x} ,$$

where

$$\begin{split} A_1^2 &= n^2 \mathbb{E}[H_{L,\tau_1,\tau_2}(U_1, U_2)^2] \ , \\ A_2 &= \sup \left(\left| \mathbb{E} \left[\sum_{i=2}^n \sum_{j=1}^{i-1} H_{L,\tau_1,\tau_2}(U_i, U_j) f_i(U_i) g_j(U_j) \right] \right| \ , \\ & \mathbb{E} \left[\sum_{i=2}^n f_i^2(U_i) \right] \leq 1, \ \mathbb{E} \left[\sum_{j=1}^{n-1} g_j^2(U_j) \right] \leq 1, \ f_i \text{ and } g_i \text{ Borel measurable functions} \right) \ , \\ A_3^2 &= n \sup_{y \in [0,1]} \int_0^1 H_{L,\tau_1,\tau_2}^2(x, y) dx \ , \\ A_4 &= \sup_{x,y \in [0,1]} |H_{L,\tau_1,\tau_2}(x, y)| \ . \end{split}$$

From Theorem 3.4 in [62], notice that the constant C can be taken equal to $C = \min_{\varepsilon>0} \left(e(1+\varepsilon^{-1})^2 (2.5+32\varepsilon^{-1}) + \left((2\sqrt{2}(2+\varepsilon+\varepsilon^{-1})) \vee ((1+\varepsilon)^2/\sqrt{2}) \right) \right)$. Let us now evaluate A_1, A_2, A_3 and A_4 .

Since the function ψ_{τ_1,τ_2} has a L₂-norm equal to 1 and the U_i 's are independent, we get

$$A_1^2 = \frac{4n^2}{L^4} \mathbb{E}\left[\psi_{\tau_1,\tau_2}^2(U_1)\right]^2 = \frac{4n^2}{L^4}$$

.

The independence of the U_i 's and the Cauchy-Schwarz inequality applied twice also yields

$$\begin{split} \left| \mathbb{E} \left[\sum_{i=2}^{n} \sum_{j=1}^{i-1} H_{L,\tau_{1},\tau_{2}}(U_{i},U_{j})f_{i}(U_{i})g_{j}(U_{j}) \right] \right| \\ &\leq \frac{2}{L^{2}} \sum_{i=2}^{n} \sum_{j=1}^{i-1} \left| \int_{0}^{1} \psi_{\tau_{1},\tau_{2}}(x)f_{i}(x)dx \right| \left| \int_{0}^{1} \psi_{\tau_{1},\tau_{2}}(y)g_{j}(y)dy \right| \\ &\leq \frac{2}{L^{2}} \sum_{i=2}^{n} \sqrt{\int_{0}^{1} f_{i}^{2}(x)dx} \sum_{j=1}^{n-1} \sqrt{\int_{0}^{1} g_{j}^{2}(y)dy} \\ &\leq \frac{2(n-1)}{L^{2}} \sqrt{\sum_{i=2}^{n} \int_{0}^{1} f_{i}^{2}(x)dx} \sqrt{\sum_{j=1}^{n-1} \int_{0}^{1} g_{j}(y)^{2}dy} , \end{split}$$

hence $A_2 \leq 2n/L^2$. Moreover,

$$A_3^2 = \frac{4n}{L^4} \sup_{y \in [0,1]} \int_0^1 \psi_{\tau_1,\tau_2}^2(x) \psi_{\tau_1,\tau_2}^2(y) dx$$

= $\frac{4n}{L^4} \sup_{y \in [0,1]} \psi_{\tau_1,\tau_2}^2(y)$
= $\frac{4n}{L^4} \max\left(\frac{1-\tau_2+\tau_1}{\tau_2-\tau_1}, \frac{\tau_2-\tau_1}{1-\tau_2+\tau_1}\right)$.

To conclude,

$$A_4 = \frac{2}{L^2} \left(\sup_{x \in [0,1]} |\psi_{\tau_1,\tau_2}(x)| \right)^2 = \frac{2}{L^2} \max \left(\frac{1 - \tau_2 + \tau_1}{\tau_2 - \tau_1} , \frac{\tau_2 - \tau_1}{1 - \tau_2 + \tau_1} \right) .$$

We finally obtain for all x > 0 and for all $n \ge 2$

$$\mathbb{P}\left(T'_{n,L,\tau^{1},\tau^{2}} \geq \frac{C}{L^{2}}\left(2n\sqrt{x} + 2nx + 2\sqrt{n\max\left(\frac{1-\tau_{2}+\tau_{1}}{\tau_{2}-\tau_{1}}, \frac{\tau_{2}-\tau_{1}}{1-\tau_{2}+\tau_{1}}\right)}x^{3/2} + 2\max\left(\frac{1-\tau_{2}+\tau_{1}}{\tau_{2}-\tau_{1}}, \frac{\tau_{2}-\tau_{1}}{1-\tau_{2}+\tau_{1}}\right)x^{2}\right)\right) \leq 2.77e^{-x}.$$

Then notice that for all x > 0

$$2\sqrt{n\max\left(\frac{1-\tau_2+\tau_1}{\tau_2-\tau_1}, \frac{\tau_2-\tau_1}{1-\tau_2+\tau_1}\right)}x^{3/2} \le nx + \max\left(\frac{1-\tau_2+\tau_1}{\tau_2-\tau_1}, \frac{\tau_2-\tau_1}{1-\tau_2+\tau_1}\right)x^2,$$

whereby

$$\mathbb{P}\left(T'_{n,L,\tau^{1},\tau^{2}} \geq \frac{C}{L^{2}}\left(2n\sqrt{x} + 3nx + 3\max\left(\frac{1-\tau_{2}+\tau_{1}}{\tau_{2}-\tau_{1}}, \frac{\tau_{2}-\tau_{1}}{1-\tau_{2}+\tau_{1}}\right)x^{2}\right)\right) \leq 2.77e^{-x}$$
(290)

By convention, (290) also holds for n in $\{0,1\}$ such that $T'_{n,L,\tau^1,\tau^2} = 0$. We then obtain for all n in N and $x = \log(2.77/u)$ (which satisfies $x \ge 1$ for all u in (0,1))

$$\mathbb{P}\left(T'_{n,L,\tau^1,\tau^2} \ge \frac{C}{L^2} \left(5n \log\left(\frac{2.77}{u}\right) + 3 \max\left(\frac{1-\tau_2+\tau_1}{\tau_2-\tau_1}, \frac{\tau_2-\tau_1}{1-\tau_2+\tau_1}\right) \log^2\left(\frac{2.77}{u}\right)\right)\right) \le u \quad .$$
This allows to end the proof.

This allows to end the proof.

Lemma 54 (Conditional quantile bound for $\max / \min_{\tau \in [0, 1-\ell]} N(\tau, \tau + \ell]$). Let $L \ge 1$ and $0 < \ell < 1$. For any n in $\mathbb{N} \setminus \{0\}$, the $(1 - \alpha)$ -quantile $b_{n,\ell}^+(1 - \alpha)$ of the conditional distribution of $\max_{\tau \in [0,1-\ell]} N(\tau, \tau + \ell]$ given $N_1 = n$ under (H_0) satisfies

$$b_{n,\ell}^+(1-\alpha) \le \ell n + \frac{n}{2}g^{-1}\left(\frac{32}{n}\log\left(\frac{320}{\alpha}\right)\right)$$
,

where g is defined by (79). Moreover, $b_{0,\ell}^+(1-\alpha) = 0$. For any n in \mathbb{N} , the α -quantile $b_{n,\ell}^{-}(\alpha)$ of the conditional distribution of $\min_{\tau \in [0,1-\ell]} N(\tau, \tau+\ell]$ given $N_1 = n$ under (H_0) satisfies

$$b_{n,\ell}^{-}(\alpha) \ge \ell n - 4\sqrt{2n\log\left(\frac{320}{\alpha}\right)}$$

Proof. Let n in $\mathbb{N} \setminus \{0\}$. If U_1, \ldots, U_n denote independent uniform random variables on (0,1), let us consider the empirical process \mathbb{U}_n associated with these random variables, defined for $0 \le t \le 1$ by

$$\mathbb{U}_n(t) = \sqrt{n} \left(\frac{1}{n} \sum_{i=1}^n \mathbb{1}_{X_i \le t} - t \right) \quad .$$

Let ℓ in (0, 1/2] and x > 0 satisfying

$$x \ge n\ell + \frac{n}{2}g^{-1}\left(\frac{32}{n}\log\left(\frac{320}{\alpha}\right)\right) \quad . \tag{291}$$

Then, we may compute for all λ_0 in (0, R)

$$\begin{aligned} P_{\lambda_0} \left(\max_{t \in [0, 1-\ell]} N(t, t+\ell] > x \mid N_1 = n \right) \\ &= P_{\lambda_0} \left(\max_{t \in [0, 1-\ell]} \left(\sum_{i=1}^n \mathbbm{1}_{X_i \le t+\ell} - \sum_{i=1}^n \mathbbm{1}_{X_i \le t} \right) > x \mid N_1 = n \right) \\ &= P_{\lambda_0} \left(\max_{t \in [0, 1-\ell]} \sqrt{n} \left(\frac{1}{n} \sum_{i=1}^n \mathbbm{1}_{X_i \le t+\ell} - \frac{1}{n} \sum_{i=1}^n \mathbbm{1}_{X_i \le t} - \ell \right) > \sqrt{n} \left(\frac{x}{n} - \ell \right) \mid N_1 = n \right) \\ &\leq P_{\lambda_0} \left(\max_{t \in [0, 1-\ell]} |\mathbb{U}_n(t+\ell) - \mathbb{U}_n(t)| > \sqrt{n} \left(\frac{x}{n} - \ell \right) \mid N_1 = n \right) \\ &\leq P_{\lambda_0} \left(\max_{0 \le z \le \frac{1}{2}} \max_{t \in [0, 1-z]} |\mathbb{U}_n(t+z) - \mathbb{U}_n(t)| > \sqrt{n} \left(\frac{x}{n} - \ell \right) \mid N_1 = n \right) \end{aligned}$$

We shall use now an inequality for controlling the oscillations of the empirical process which is due to Mason, Shorack and Wellner, and which can be found in Chapter 14 of [95] page 545 under the name of Inequality 1. We get

$$\sup_{\lambda_0 \in (0,R)} P_{\lambda_0} \left(\max_{t \in [0,1-\ell]} N(t,t+\ell] > x \mid N_1 = n \right) \le 320 \exp\left(-\frac{n}{32}g\left(2\left(\frac{x}{n}-\ell\right)\right)\right) \le \alpha \quad \text{using } (291) \quad .$$

This yields $b_{n,\ell}^+(1-\alpha) \leq n\ell + ng^{-1} (32 \log (320/\alpha)/n)/2$ for every n in $\mathbb{N} \setminus \{0\}$, which is the first statement of Lemma 54. The fact that $b_{0,\ell}^+(1-\alpha) = 0$ is obvious. Now let again n in $\mathbb{N} \setminus \{0\}$, $\varepsilon > 0$ and y in \mathbb{R} satisfying

$$y \le n\ell - 4\sqrt{2n\log\left(\frac{320}{\alpha - \varepsilon}\right)}$$
 (292)

We compute for all λ_0 in (0, R)

$$\begin{aligned} P_{\lambda_0} \left(\min_{t \in [0, 1-\ell]} N(t, t+\ell] &\leq y \mid N_1 = n \right) \\ &= P_{\lambda_0} \left(\min_{t \in [0, 1-\ell]} \left(\mathbb{U}_n(t+\ell) - \mathbb{U}_n(t) \right) \leq \sqrt{n} \left(\frac{y}{n} - \ell \right) \mid N_1 = n \right) \\ &= P_{\lambda_0} \left(\max_{t \in [0, 1-\ell]} \left(\mathbb{U}_n(t) - \mathbb{U}_n(t+\ell) \right) \geq \sqrt{n} \left(\ell - \frac{y}{n} \right) \mid N_1 = n \right) \\ &\leq P_{\lambda_0} \left(\max_{0 \leq z \leq \frac{1}{2}} \max_{t \in [0, 1-z]} \left(\mathbb{U}_n(t) - \mathbb{U}_n(t+z) \right) \geq \sqrt{n} \left(\ell - \frac{y}{n} \right) \mid N_1 = n \right) \end{aligned}$$

Applying now the second part of Inequality 1 page 545 in [95], we obtain

$$\sup_{\lambda_0 \in (0,R)} P_{\lambda_0} \left(\min_{t \in [0,1-\ell]} N(t,t+\ell] \le y \mid N_1 = n \right) \le 320 \exp\left(-\frac{n}{32} \left(\ell - \frac{y}{n} \right)^2 \right) \\ \le \alpha - \varepsilon < \alpha \quad \text{using (292)} \quad .$$

This entails $b_{n,\ell}^-(\alpha) > n\ell - 4\sqrt{2n\log(320/(\alpha - \varepsilon))}$ for all $\varepsilon > 0$. With ε tending to 0, we get $b_{n,\ell}^-(\alpha) \ge n\ell - 4\sqrt{2n\log(320/\alpha)}$, which leads to the second statement of Lemma 54 with the obvious fact that $b_{0,\ell}^-(\alpha) = 0$.

Lemma 55 (Conditional quantile bound for $\sup_{\ell \in (0,1-\tau^*)} S'_{\delta^*,\tau^*,\tau^*+\ell}(N)$). Let $L \ge 1$ and $n_0 \ge 1$. For all $0 \le n \le n_0 L$, the $(1-\alpha)$ -quantile $s'_{n,\delta^*,\tau^*,L}(1-\alpha)$ of the conditional distribution of $\sup_{\ell \in (0,1-\tau^*)} S'_{\delta^*,\tau^*,\tau^*+\ell}(N)$ given $N_1 = n$ under (H_0) with $S'_{\delta^*,\tau^*,\tau^*+\ell}(N)$ defined by (55) satisfies

$$s_{n,\delta^*,\tau^*,L}^{'+}(1-\alpha) \le Q(n_0,\delta^*,\alpha)$$
,

where

$$Q(n_0, \delta^*, \alpha) = \log(2) \frac{(6n_0 + |\delta^*|)(6n_0 + 2|\delta^*|/3)}{\delta^{*2}} + \log\left(\frac{\pi^2}{3\alpha}\right) \frac{9n_0 + |\delta^*|}{3|\delta^*|}$$

Proof. First recall that (see (55))

$$S'_{\delta^*,\tau^*,\tau^*+\ell}(N) = \operatorname{sgn}(\delta^*) \Big(N(\tau^*,\tau^*+\ell] - \ell N_1 \Big) - |\delta^*|\ell(1-\ell)L/2$$

Since N is an homogeneous Poisson process, the processes $(N_1, N(\tau^*, \tau^* + \ell])_{\ell \in (0, 1-\tau^*]}$ and $(N_1, N(0, \ell])_{\ell \in (0, 1-\tau^*]}$ have the same finite dimensional law and since N is a right continuous process, we get that $\sup_{\ell \in (0, 1-\tau^*)} S'_{\delta^*, \tau^*, \tau^* + \ell}(N)$ is distributed as

$$\sup_{\ell \in (0,1-\tau^*)} \left(\operatorname{sgn}\left(\delta^*\right) \left(N(0,\ell] - \ell N_1\right) - \frac{|\delta^*|}{2}\ell(1-\ell)L \right)$$

As seen above, for $n \ge 1$ and conditionally on the event $\{N_1 = n\}$, the points of the process N obey the same law as a n-sample (U_1, \ldots, U_n) of i.i.d. random variables uniformly distributed on (0, 1). Therefore, considering the empirical distribution function F_n associated with this sample defined for $0 \le t \le 1$ by

$$F_n(t) = \frac{1}{n} \sum_{i=1}^n \mathbb{1}_{X_i \le t} ,$$

conditionally on the event $\{N_1 = n\}$, $\sup_{\ell \in (0, 1-\tau^*)} S'_{\delta^*, \tau^*, \tau^*+\ell}(N)$ is distributed as

$$\sup_{\ell \in (0,1-\tau^*)} \left(\operatorname{sgn}(\delta^*) \left(nF_n(\ell) - n\ell \right) - \frac{|\delta^*|}{2} \ell(1-\ell)L \right) .$$

Notice first that for all $n \ge 1$ and for all x > 0

$$\mathbb{P}\left(\sup_{\ell\in(0,1-\tau^*)}\left(\operatorname{sgn}\left(\delta^*\right)\left(nF_n(\ell)-n\ell\right)-\frac{|\delta^*|}{2}\ell(1-\ell)L\right)>x\right) \\
\leq \mathbb{P}\left(\sup_{\ell\in(0,1)}\left(\operatorname{sgn}\left(\delta^*\right)\left(nF_n(\ell)-n\ell\right)-\frac{|\delta^*|}{2}\ell(1-\ell)L\right)>x\right).$$

Moreover, since the process $(-(nF_n(\ell) - n\ell))_{\ell \in (0,1)}$ has the same distribution as the process $(nF_n(1-\ell) - n(1-\ell))_{\ell \in (0,1)}$, one has when $\delta^* < 0$

$$\mathbb{P}\left(\sup_{\ell\in(0,1)}\left(\operatorname{sgn}\left(\delta^{*}\right)\left(nF_{n}(\ell)-n\ell\right)-\frac{|\delta^{*}|}{2}\ell(1-\ell)L\right)>x\right)$$
$$=\mathbb{P}\left(\sup_{\ell\in(0,1)}\left(nF_{n}(1-\ell)-n(1-\ell)-\frac{|\delta^{*}|}{2}\ell(1-\ell)L\right)>x\right)$$
$$=\mathbb{P}\left(\sup_{\ell\in(0,1)}\left(nF_{n}(\ell)-n\ell-\frac{|\delta^{*}|}{2}\ell(1-\ell)L\right)>x\right).$$

Therefore, whatever the sign of δ^* ,

$$\mathbb{P}\left(\sup_{\ell\in(0,1-\tau^*)}\left(\operatorname{sgn}\left(\delta^*\right)\left(nF_n(\ell)-n\ell\right)-\frac{|\delta^*|}{2}\ell(1-\ell)L\right) > x\right) \\
\leq \mathbb{P}\left(\sup_{\ell\in(0,1)}\left(nF_n(\ell)-n\ell-\frac{|\delta^*|}{2}\ell(1-\ell)L\right) > x\right),$$

and

$$\mathbb{P}\left(\sup_{\ell\in(0,1-\tau^*)} \left(\operatorname{sgn}\left(\delta^*\right) \left(nF_n(\ell) - n\ell\right) - \frac{|\delta^*|}{2}\ell(1-\ell)L\right) > x\right) \\
\leq \mathbb{P}\left(\sup_{\ell\in(0,1/2]} \left(nF_n(\ell) - n\ell - \frac{|\delta^*|}{2}\ell(1-\ell)L\right) > x\right) \\
+ \mathbb{P}\left(\sup_{\ell\in(1/2,1)} \left(nF_n(\ell) - n\ell - \frac{|\delta^*|}{2}\ell(1-\ell)L\right) > x\right) \quad . (293)$$

Let us now prove that on the one hand

$$\mathbb{P}\left(\sup_{\ell\in(0,1/2]}\left(nF_n(\ell)-n\ell-\frac{|\delta^*|}{2}\ell(1-\ell)L\right)>Q(n_0,\delta^*,\alpha)\right)\leq\frac{\alpha}{2},\qquad(294)$$

and on the other hand

$$\mathbb{P}\left(\sup_{\ell\in(1/2,1)}\left(nF_n(\ell)-n\ell-\frac{|\delta^*|}{2}\ell(1-\ell)L\right)>Q(n_0,\delta^*,\alpha)\right)\leq\frac{\alpha}{2}.$$
(295)

Let $1 \le n \le n_0 L$. Set $C(\delta^*, n_0) = |\delta^*| / (6n_0)$ and $C'(\delta^*, n_0, n) = \left[\exp\left(\frac{3n}{2} \frac{C(\delta^*, n_0)^2}{3 + 2C(\delta^*, n_0)}\right) \right]$. For all k in $\{0, \dots, C'(\delta^*, n_0, n) - 1\}$, we define

$$\ell_k = \frac{3 + 2C(\delta^*, n_0)}{3C(\delta^*, n_0)^2} \frac{\log(k+1)}{n} ,$$

and

$$\ell_{C'(\delta^*, n_0, n)} = \left(\frac{3 + 2C(\delta^*, n_0)}{3C(\delta^*, n_0)^2} \frac{\log(C'(\delta^*, n_0, n) + 1)}{n}\right) \wedge 1 \ .$$

Notice that with such definitions, ℓ_k belongs to [0, 1/2] for all k in $\{0, \ldots, C'(\delta^*, n_0, n) - 1\}$ and $\ell_{C'(\delta^*, n_0, n)}$ belongs to (1/2, 1].

Applying Bernstein's inequality, as stated in equation (2.6) in [12][page 12], one obtains for every x > 0 and every k in $\{1, \ldots, C'(\delta^*, n_0, n)\}$

$$\mathbb{P}\left(nF_n(\ell_k) > n\ell_k + \sqrt{2n\ell_k(1-\ell_k)(x+2\log k)} + \frac{x+2\log k}{3}\right) \le \frac{e^{-x}}{k^2} .$$

A union bound therefore gives for every x > 0

$$\mathbb{P}\left(\forall k \in \{1, \dots, C'(\delta^*, n_0, n)\}, \ nF_n(\ell_k) \le n\ell_k + \sqrt{2n\ell_k(1 - \ell_k)(x + 2\log k)} + \frac{x + 2\log k}{3}\right)$$
$$\ge 1 - \frac{\pi^2}{6}e^{-x} .$$

Using the inequality $\sqrt{2ab} \leq aC(\delta^*, n_0) + b/(2C(\delta^*, n_0))$ and the fact that $\ell \mapsto nF_n(\ell)$ is nondecreasing, we get

$$\mathbb{P}\left(\forall k \in \{1, \dots, C'(\delta^*, n_0, n)\}, \ \forall \ell \in (\ell_{k-1}, \ell_k], \ nF_n(\ell) \le n\ell_k + C(\delta^*, n_0)n\ell_k(1 - \ell_k) + \frac{x + 2\log k}{2C(\delta^*, n_0)} + \frac{x + 2\log k}{3}\right) \ge 1 - \frac{\pi^2}{6}e^{-x}$$

Therefore, with probability larger than $1 - \pi^2 e^{-x}/6$, for all k in $\{1, \ldots, C'(\delta^*, n_0, n)\}$ and all ℓ in $(\ell_{k-1}, \ell_k]$,

$$\begin{split} nF_n(\ell) &\leq n\ell_k + C(\delta^*, n_0)n\ell_k(1-\ell_k) + x \frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + \log k \frac{3+2C(\delta^*, n_0)}{3C(\delta^*, n_0)} \\ &= n\ell_k + C(\delta^*, n_0)n\ell_k(1-\ell_k) + x \frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + C(\delta^*, n_0)n\ell_{k-1} \\ &< n\ell_k + C(\delta^*, n_0)n\ell_k(1-\ell_k) + x \frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + C(\delta^*, n_0)n\ell \\ &= n\ell + C(\delta^*, n_0)n\ell(1-\ell) + C(\delta^*, n_0)n\ell + n(\ell_k - \ell) + C(\delta^*, n_0)n\left(\ell_k(1-\ell_k) - \ell(1-\ell)\right) \\ &+ x \frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} \\ &\leq n\ell + C(\delta^*, n_0)n\ell(1-\ell) + C(\delta^*, n_0)n\ell + n(\ell_k - \ell) + C(\delta^*, n_0)n(1-\ell)\left(\ell_k - \ell\right) \\ &+ x \frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} \\ &< n\ell + C(\delta^*, n_0)n\ell(1-\ell) + C(\delta^*, n_0)n\ell + (1+C(\delta^*, n_0)(1-\ell))\frac{3+2C(\delta^*, n_0)}{3C(\delta^*, n_0)^2} \log 2 \\ &+ x \frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} . \end{split}$$

Finally, we have proved that with probability larger than $1 - \pi^2 e^{-x}/6$, for all ℓ in (0, 1/2],

$$nF_n(\ell) \le n\ell + 3C(\delta^*, n_0)n_0L\ell(1-\ell) + (1+C(\delta^*, n_0))\frac{3+2C(\delta^*, n_0)}{3C(\delta^*, n_0)^2}\log 2 + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + \frac{1}{3C(\delta^*, n_0)}\log 2 + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + \frac{1}{3C(\delta^*, n_0)^2}\log 2 + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + \frac{1}{3C(\delta^*, n_0)^2} + \frac{1}{3C(\delta^*, n_0)} + \frac{1}{3C(\delta^*, n_0)^2} + \frac{1}{3C(\delta^*, n_0)} + \frac{1}{3C(\delta^*, n_0)} + \frac{1}{3C(\delta^*, n_0)^2} + \frac{1}{3C(\delta^*, n_0)} + \frac{1}{3C$$

hence

$$\mathbb{P}\left(\sup_{\ell\in(0,1/2]}\left(nF_n(\ell)-n\ell-\frac{|\delta^*|}{2}\ell(1-\ell)L\right)\leq Q(n_0,\delta^*,\alpha)\right)\geq 1-\frac{\alpha}{2},$$

that is (294).

Define now $\ell'_k = 1 - \ell_k$ for all k in $\{0, \ldots, C'(\delta^*, n_0, n)\}$ and notice that ℓ'_k belongs to [1/2, 1] for all k in $\{0, \ldots, C'(\delta^*, n_0, n) - 1\}$ and $\ell'_{C'(\delta^*, n_0, n)}$ belongs to [0, 1/2). Applying Bernstein's inequality again, one obtains for every x > 0

$$\mathbb{P}\left(\forall k \in \{1, \dots, C'(\delta^*, n_0, n)\}, \ nF_n(\ell'_{k-1}) \le n\ell'_{k-1} + \sqrt{2n\ell'_{k-1}(1 - \ell'_{k-1})(x + 2\log k)} + \frac{x + 2\log k}{3}\right) \ge 1 - \frac{\pi^2}{6}e^{-x}$$

With the same computations as in the above case, we get

$$\mathbb{P}\left(\forall k \in \{1, \dots, C'(\delta^*, n_0, n)\}, \forall \ell \in [\ell'_k, \ell'_{k-1}), \ nF_n(\ell) \le n\ell'_{k-1} + C(\delta^*, n_0)n\ell'_{k-1}(1 - \ell'_{k-1}) + \frac{x + 2\log k}{2C(\delta^*, n_0)} + \frac{x + 2\log k}{3}\right) \ge 1 - \frac{\pi^2}{6}e^{-x} .$$

Therefore, with probability larger than $1 - \pi^2 e^{-x}/6$, for all k in $\{1, \ldots, C'(\delta^*, n_0, n)\}$ and all ℓ in $[\ell'_k, \ell'_{k-1})$,

$$\begin{split} nF_n(\ell) &\leq n\ell'_{k-1} + C(\delta^*, n_0)n\ell'_{k-1}(1-\ell'_{k-1}) + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + \log k\frac{3+2C(\delta^*, n_0)}{3C(\delta^*, n_0)} \\ &= n\ell'_{k-1} + C(\delta^*, n_0)n\ell'_{k-1}(1-\ell'_{k-1}) + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + C(\delta^*, n_0)n(1-\ell'_{k-1}) \\ &< n\ell'_{k-1} + C(\delta^*, n_0)n\ell'_{k-1}(1-\ell'_{k-1}) + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} + C(\delta^*, n_0)n(1-\ell) \\ &< n\ell + C(\delta^*, n_0)n\ell(1-\ell) + C(\delta^*, n_0)n(1-\ell) + n(\ell'_{k-1}-\ell) + C(\delta^*, n_0)n(1-\ell) \left(\ell'_{k-1}-\ell\right) \\ &+ x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} \\ &\leq n\ell + C(\delta^*, n_0)n\ell(1-\ell) + C(\delta^*, n_0)n(1-\ell) + (1+C(\delta^*, n_0)(1-\ell))\frac{3+2C(\delta^*, n_0)}{3C(\delta^*, n_0)^2} \log 2 \\ &+ x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} \end{split}$$

Finally, we have proved that with probability larger than $1 - \pi^2 e^{-x}/6$, for all ℓ in [1/2, 1),

$$nF_n(\ell) \le n\ell + 3C(\delta^*, n_0)n_0L\ell(1-\ell) + (1+C(\delta^*, n_0))\frac{3+2C(\delta^*, n_0)}{3C(\delta^*, n_0)^2}\log 2 + x\frac{3+2C(\delta^*, n_0)}{6C(\delta^*, n_0)} ,$$

hence

$$\mathbb{P}\left(\sup_{\ell\in[1/2,1)}\left(nF_n(\ell)-n\ell-\frac{|\delta^*|}{2}\ell(1-\ell)L\right)\leq Q(n_0,\delta^*,\alpha)\right)\geq 1-\frac{\alpha}{2},$$

that is (295). Combined with (293), the equations (294) and (295) conclude the proof using the obvious fact that $s_{0,\delta^*,\tau^*,L}^{'+}(1-\alpha) = 0$.

Lemma 56 (Conditional quantile bounds for $N(\tau_1, \tau_2] - (\tau_2 - \tau_1) N_1$). Let $L \ge 1$, u in $(0, 1), \tau_1$ and τ_2 such that $0 < \tau_1 < \tau_2 \le 1$ and n in \mathbb{N} . The u-quantile $\bar{b}_{n,\tau_2-\tau_1}(u)$ of the conditional distribution of $N(\tau_1, \tau_2] - (\tau_2 - \tau_1)N_1$ given $N_1 = n$ under (H_0) satisfies

$$-\frac{2}{3}\log(1/u) - \sqrt{n(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)}\sqrt{2\log(1/u)} \le \bar{b}_{n,\tau_2 - \tau_1}(u)$$
$$\le \frac{2}{3}\log(1/(1 - u)) + \sqrt{n(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)}\sqrt{2\log(1/(1 - u))} ,$$

Proof. Let $n \geq 1$. Under (H_0) and conditionally on $N_1 = n$, $N(\tau_1, \tau_2] - (\tau_2 - \tau_1)N_1$ follows a recentered binomial distribution with parameters $(n, \tau_2 - \tau_1)$. Applying Bennett's inequality as stated in Theorem 2.28 in [12] to $\sum_{i=1}^{n} (X_i - (\tau_2 - \tau_1))$ where (X_1, \ldots, X_n) is a sample of i.i.d. random variables with a Bernoulli distribution with parameter $\tau_2 - \tau_1$, we obtain for all x > 0

$$\sup_{\lambda_0 \in \mathcal{S}_0^u[R]} P_{\lambda_0} \left(N(\tau_1, \tau_2] - (\tau_2 - \tau_1) N_1 > x \mid N_1 = n \right) \\ \leq \exp\left(-n(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)g\left(\frac{x}{n(\tau_2 - \tau_1)(1 - \tau_2 + \tau_1)}\right) \right) .$$

It directly follows that

$$\bar{b}_{n,\tau_2-\tau_1}(1-u) \le n(\tau_2-\tau_1)(1-\tau_2+\tau_1)g^{-1}\left(\frac{\log(1/u)}{n(\tau_2-\tau_1)(1-\tau_2+\tau_1)}\right) \quad .$$

The upper bound (80) for g^{-1} and the fact that $\bar{b}_{0,\tau_2-\tau_1}(1-u) = 0$ allow to conclude for the first part of Lemma 56. Then, following similar arguments, but applying Bennett's inequality to $\sum_{i=1}^{n} ((\tau_2 - \tau_1) - X_i)$ and using the continuity of g^{-1} as in Lemma 46, one obtains the second part of Lemma 56.

Lemma 57 (Conditional quantile bound for $\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)$). Let $L \ge 1$ and $n_0 \ge 1$. For all $0 \le n \le n_0 L$, the $(1 - \alpha)$ -quantile $s'^+_{n,\delta^*,L}(1 - \alpha)$ of the conditional distribution of $\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)$ given $N_1 = n$ under (H_0) with $S'_{\delta^*,\tau,1}(N)$ defined by (55) satisfies

$$s_{n,\delta^*,L}^{\prime+}(1-\alpha) \le Q(n_0,\delta^*,\alpha)$$

where $Q(n_0, \delta^*, \alpha)$ is defined in Lemma 55.

Proof. Notice first that $\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)$ is distributed as

$$\sup_{\tau \in (0,1)} \left(\operatorname{sgn}\left(\delta^*\right) \left(N(0, 1-\tau] - (1-\tau)N_1 \right) - \frac{|\delta^*|}{2} \tau(1-\tau)L \right)$$

Now, recall that for $n \ge 1$ and conditionally on the event $\{N_1 = n\}$, the points of the process N obey the same law as a n-sample (U_1, \ldots, U_n) of i.i.d. random variables uniformly distributed on (0, 1) and consider the empirical distribution function F_n associated with this sample defined for $0 \le t \le 1$ as in the proof of Lemma 55. Then, conditionally on the event $\{N_1 = n\}$, $\sup_{\tau \in (0,1)} S'_{\delta^*,\tau,1}(N)$ is distributed as

$$\sup_{\tau \in (0,1)} \left(\operatorname{sgn}(\delta^*) \left(nF_n(1-\tau) - n(1-\tau) \right) - \frac{|\delta^*|}{2} \tau(1-\tau)L \right) .$$

Since the process $(-(nF_n(1-\tau)-n(1-\tau)))_{\tau\in(0,1)}$ has the same distribution as the process $(nF_n(\tau)-n\tau)_{\tau\in(0,1)}$, whatever the sign of δ^* , $\sup_{\tau\in(0,1)} S'_{\delta^*,\tau,1}(N)$ is distributed as

$$\sup_{\tau \in (0,1)} \left(nF_n(\tau) - n\tau - \frac{|\delta^*|}{2} \tau(1-\tau)L \right) .$$

The proof now follows the same line as the proof of Lemma 55, just replacing ℓ by τ and using the fact that $s_{0,\delta^*,L}^{'+}(1-\alpha) = 0$.

Acknowledgements

The authors want to thank Nicolas Verzelen for fruitful discussions. The PhD grant of Fabrice Grela is funded by the french Région Bretagne and Direction Générale des Armées.

References

- Agustin, M. Z. N. and Peña, E. A. (1999). Order statistic properties, random generation, and goodness-of-fit testing for a minimal repair model. *Journal of the American Statistical Association*, 94(445):266–272.
- [2] Akman, V. E. and Raftery, A. E. (1986a). Asymptotic inference for a change-point Poisson process. *The Annals of Statistics*, 14(4):1583–1590.
- [3] Akman, V. E. and Raftery, A. E. (1986b). Bayes factors for non-homogeneous Poisson processes with vague prior information. *Journal of the Royal Statistical Society: Series B (Methodological)*, 48(3):322–329.
- [4] Antoch, J. and Jarušková, D. (2007). Testing a homogeneity of stochastic processes. *Kybernetika*, 43(4):415–430.
- [5] Arias-Castro, E., Donoho, D. L., and Huo, X. (2005). Near-optimal detection of geometric objects by fast multiscale methods. *IEEE Transactions on Information Theory*, 51(7):2402–2425.
- [6] Bahadur, R. R. (1967). An optimal property of the likelihood ratio statistic. In Proceedings of the Fifth Berkeley Symposium on Mathematical Statistics and Probability, volume 1, pages 13–26.

- [7] Bain, L. J., Engelhardt, M., and Wright, F. T. (1985). Tests for an increasing trend in the intensity of a Poisson process: a power study. *Journal of the American Statistical Association*, 80(390):419–422.
- [8] Baraud, Y. (2002). Non-asymptotic minimax rates of testing in signal detection. Bernoulli, 8(5):577-606.
- [9] Baraud, Y., Huet, S., and Laurent, B. (2005). Testing convex hypotheses on the mean of a Gaussian vector. Application to testing qualitative hypotheses on a regression function. *The Annals of Statistics*, 33(1):214 – 257.
- [10] Basseville, M. and Nikiforov, I. V. (1993). Detection of abrupt changes: theory and application, volume 104. Prentice Hall Englewood Cliffs.
- [11] Bayraktar, E., Dayanik, S., and Karatzas, I. (2006). Adaptive Poisson disorder problem. *The Annals of Applied Probability*, 16(3):1190–1261.
- [12] Bercu, B., Delyon, B., and Rio, E. (2015). Concentration Inequalities for Sums and Martingales. Springer Briefs in Mathematics. Springer.
- [13] Bhattacharjee, M., Deshpande, J. V., and Naik-Nimbalkar, U. V. (2004). Unconditional tests of goodness of fit for the intensity of time-truncated nonhomogeneous Poisson processes. *Technometrics*, 46(3):330–338.
- [14] Brémaud, P. (1981). Point Processes and Queues. Martingale Dynamics. Springer-Verlag, Berlin-Heidelberg-New York.
- [15] Brodsky, E. and Darkhovsky, B. S. (2013a). Non-parametric statistical diagnosis: problems and methods, volume 509. Springer Science & Business Media.
- [16] Brodsky, E. and Darkhovsky, B. S. (2013b). Nonparametric methods in change point problems, volume 243. Springer Science & Business Media.
- [17] Brown, L. D. (1971). Non-local asymptotic optimality of appropriate likelihood ratio tests. The Annals of Mathematical Statistics, 42(4):1206–1240.
- [18] Brown, M. and Zacks, S. (2006). A note on optimal stopping for possible change in the intensity of an ordinary Poisson process. *Statistics and Probability Letters*, 76(13):1417– 1425.
- [19] Brunel, V.-E. (2014). Convex set detection. arXiv:1404.6224.
- [20] Cao, J., Cleveland, W. S., Lin, D., and Sun, D. X. (2003). Internet traffic tends toward poisson and independent as the load increases. In *Nonlinear estimation and classification*, pages 83–109. Springer.
- [21] Carlstein, E. G., Müller, H.-G., and Siegmund, D. (1994). Change-point problems.
- [22] Chan, H. P. and Walther, G. (2013). Detection with the scan and the average likelihood ratio. *Statistica Sinica*, 23:409–428.
- [23] Cohen, A. and Sackrowitz, H. B. (1993). Evaluating tests for increasing intensity of a poisson process. *Technometrics*, 35(4):446–448.

- [24] Cox, D. R. (1955). Some statistical methods connected with series of events. Journal of the Royal Statistical Society: Series B (Methodological), 17(2):129–157.
- [25] Crow, L. H. (1974). Reliability analysis for complex repairable systems, soc. industrial and applied mathematics, reliability and biometry. *Proceedings of Statistical Analysis* of Life Length, 25:248–253.
- [26] Csorgo, M. and Horváth, L. (1997). Limit theorems in change-point analysis. John Wiley & Sons Chichester.
- [27] Dachian, S. and Kutoyants, Y. A. (2006). Hypotheses testing: Poisson versus selfexciting. Scandinavian Journal of Statistics, 33(2):391–408.
- [28] Dachian, S. and Kutoyants, Y. A. (2009). Hypotheses testing: Poisson versus stressrelease. Journal of Statistical Planning and Inference, 139(5):1668–1684.
- [29] Dachian, S., Kutoyants, Y. A., and Yang, L. (2016a). On hypothesis testing for Poisson processes. Regular case. Communications in Statistics - Theory and Methods, 45(23):6816-6832.
- [30] Dachian, S., Kutoyants, Y. A., and Yang, L. (2016b). On hypothesis testing for Poisson processes. Singular cases. Communications in Statistics - Theory and Methods, 45(23):6833-6859.
- [31] Dachian, S. and Yang, L. (2015). On a Poissonian change-point model with variable jump size. *Statistical Inference for Stochastic Processes*, 18(2):127–150.
- [32] Daley, D. J. and Vere-Jones, D. (2007). An Introduction to the Theory of Point Processes. Springer.
- [33] Daras, N. J. (2014). Stochastic analysis of cyber-attacks. In Applications of Mathematics and Informatics in Science and Engineering, pages 105–129. Springer.
- [34] Davies, R. B. (1977). Testing the hypothesis that a point process is Poisson. Advances in Applied Probability, 9(4):724–746.
- [35] Deshayes, J. (1984). Rupture de modèles pour des processus de poisson. Annales scientifiques de l'Université de Clermont-Ferrand 2. Série Probabilités et applications, 78(2):1–7.
- [36] Deshayes, J. and Picard, D. (1985). Off-line statistical analysis of change-point models using non parametric and likelihood methods. In *Detection of Abrupt Changes* in Signals and Dynamical Systems, pages 103–168. Springer.
- [37] Dümbgen, L. and Walther, G. (2008). Multiscale inference about a density. The Annals of Statistics, 36(4):1758–1785.
- [38] Dvoretzky, A., Kiefer, J., and Wolfowitz, J. (1953a). Sequential decision problem for processes with continuous time parameter. problems of estimation. *The Annals of Mathematical Statistics*, 24(3):403–415.

- [39] Dvoretzky, A., Kiefer, J., and Wolfowitz, J. (1953b). Sequential decision problem for processes with continuous time parameter. testing hypotheses. *The Annals of Mathematical Statistics*, 24(2):254–264.
- [40] El Karoui, N., Loisel, S., and Salhi, Y. (2015). Minimax optimality in robust detection of a disorder time in Poisson rate.
- [41] El Karoui, N., Loisel, S., Salhi, Y., et al. (2017). Minimax optimality in robust detection of a disorder time in doubly-stochastic Poisson processes. *The Annals of Applied Probability*, 27(4):2515–2538.
- [42] Engelhardt, M., Guffey, J. M., and Wright, F. T. (1990). Tests for positive jumps in the intensity of a Poisson process: a power study. *IEEE transactions on reliability*, 39(3):356–360.
- [43] Enikeeva, F. and Harchaoui, Z. (2019). High-dimensional change-point detection under sparse alternatives. *The Annals of Statistics*, 47(4):2051 2079.
- [44] Enikeeva, F., Munk, A., Pohlmann, M., and Werner, F. (2020). Bump detection in the presence of dependency: Does it ease or does it load? *Bernoulli*, 26(4):3280 3310.
- [45] Enikeeva, F., Munk, A., and Werner, F. (2018). Bump detection in heterogeneous Gaussian regression. *Bernoulli*, (24):1266–1306.
- [46] Farinetto, C. (2017). On hypothesis tests in misspecified change-point problems for a Poisson process. Communications in Statistics - Theory and Methods, 46(20):10103– 10115.
- [47] Fazli, K. (2007). Second-order efficient test for inhomogeneous Poisson processes. Statistical Inference for Stochastic Processes, 10(2):181–208.
- [48] Fazli, K. and Kutoyants, Y. A. (2005). Two simple hypotheses testing for Poisson process. Far East Journal of Theoretical Statistics, 15(2):251–290.
- [49] Fisher, W. D. (1958). On grouping for maximum homogeneity. Journal of the American statistical Association, 53(284):789–798.
- [50] Fromont, M., Laurent, B., and Reynaud-Bouret, P. (2011). Adaptive tests of homogeneity for a poisson process. In Annales de l'Institut Henri Poincaré Probabilités et Statistiques, volume 47, pages 176–213.
- [51] Fromont, M., Laurent, B., and Reynaud-Bouret, P. (2013). The two-sample problem for poisson processes: Adaptive tests with a nonasymptotic wild bootstrap approach. *The Annals of Statistics*, 41(3):1431–1461.
- [52] Fromont, M., Lerasle, M., and Reynaud-Bouret, P. (2016). Family-wise separation rates for multiple testing. *The Annals of Statistics*, 44(6):2533–2563.
- [53] Galeano, P. (2007). The use of cumulative sums for detection of changepoints in the rate parameter of a Poisson process. *Computational Statistics & Data Analysis*, 51(12):6151–6165.

- [54] Gao, C., Han, F., and Zhang, C.-H. (2020). On estimation of isotonic piecewise constant signals. *The Annals of Statistics*, 48(2):629 – 654.
- [55] Girshick, M. A. and Rubin, H. (1952). A bayes approach to a quality control model. The Annals of Mathematical Statistics, 23(1):114–125.
- [56] Green, P. J. (1995). Reversible jump Markov chain Monte Carlo computation and Bayesian model determination. *Biometrika*, 82(4):711–732.
- [57] Herberts, T. and Jensen, U. (2004). Optimal detection of a change point in a Poisson process for different observation schemes. *Scandinavian Journal of Statistics*, 31(3):347– 366.
- [58] Hinkley, D. V. and Hinkley, E. A. (1970). Inference about the change-point in a sequence of binomial variables. *Biometrika*, 57(3):477–488.
- [59] Ho, C.-H. (1993). Forward and backward tests for an abrupt change in the intensity of a Poisson process. Journal of statistical computation and simulation, 48(3-4):245–252.
- [60] Ho, C.-H. (1995). A simulation study of a change-point Poisson process based on two well-known test statistics. In *Monte Carlo and quasi-Monte Carlo methods in scientific computing*, pages 228–238. Springer.
- [61] Holm, H. (2013). A large-scale study of the time required to compromise a computer system. *IEEE Transactions on Dependable and Secure Computing*, 11(1):2–15.
- [62] Houdré, C. and Reynaud-Bouret, P. (2003). Exponential inequalities, with constants, for U-statistics of order two. In *Stochastic inequalities and applications*, pages 55–69. Springer.
- [63] Ingster, Y. I. (1982). On the minimax nonparametric detection of signals in white gaussian noise. *Problemy Peredachi Informatsii*, 18(2):61–73.
- [64] Ingster, Y. I. (1984). Asymptotic minimax nonparametric testing for independent sample density hypothesis. Zapiski Nauchnykh Seminarov POMI, 136:74–96.
- [65] Ingster, Y. I. (1993). Asymptotically minimax hypothesis testing for nonparametric alternatives. i, ii, iii. Mathematical Methods of Statistics, 2,3,4:85–114, 171–189, 249– 268.
- [66] Ingster, Y. I. and Kutoyants, Y. A. (2007). Nonparametric hypothesis testing for intensity of the Poisson process. *Mathematical Methods of Statistics*, 16(3):217–245.
- [67] Karagiannis, T., Molle, M., Faloutsos, M., and Broido, A. (2004). A nonstationary poisson view of internet traffic. In *IEEE INFOCOM 2004*, volume 3, pages 1558–1569. IEEE.
- [68] Kendall, D. G. and S., K. W. (1980). Alignments in two-dimensional random sets of points. Advances in Applied Probability, 12(2):380–424.
- [69] Lai, T. L. (2001). Sequential analysis: some classical problems and new challenges. Statistica Sinica, pages 303–351.

- [70] Le Cam, L. (1970). On the assumptions used to prove asymptotic normality of maximum likelihood estimates. *The Annals of Mathematical Statistics*, 41(3):802–828.
- [71] Le Guével, R. (2021). Exponential inequalities for the supremum of some counting processes and their square martingales. To appear in Comptes Rendus de l'Académie des Sciences, preprint hal-02275583.
- [72] Lee, T.-S. (2010). Change-point problems: bibliography and review. *Journal of Statistical Theory and Practice*, 4(4):643–662.
- [73] Lehmann, E. L. and Romano, J. P. (2006). Testing statistical hypotheses. Springer Science & Business Media.
- [74] Lepski, O. V. and Tsybakov, A. B. (2000). Asymptotically exact nonparametric hypothesis testing in sup-norm and at a fixed point. *Probability Theory and Related Fields*, 117(1):17–48.
- [75] Lewis, P. A. W. (1965). Some results on tests for Poisson processes. Biometrika, 52(1-2):67–77.
- [76] Liu, H., Gao, C., and Samworth, R. J. (2021). Minimax rates in sparse, highdimensional change point detection. *The Annals of Statistics*, 49(2):1081 – 1112.
- [77] Loader, C. R. (1990). Change point problems for Poisson processes. PhD thesis, Stanford University.
- [78] Maguire, B. A., Pearson, E. S., and Wynn, A. H. A. (1952). The time intervals between industrial accidents. *Biometrika*, 39(1/2):168–180.
- [79] Matthews, D. E., Farewell, V. T., and Pyke, R. (1985). Asymptotic score-statistic processes and tests for constant hazard against a change-point alternative. *The Annals* of *Statistics*, 13(2):583–591.
- [80] Mei, Y., Han, S. W., and Tsui, K.-L. (2011). Early detection of a change in Poisson rate after accounting for population size effects. *Statistica Sinica*, pages 597–624.
- [81] Moustakides, G. V. (2008). Sequential change detection revisited. The Annals of Statistics, 36(2):787–807.
- [82] Neyman, J. and Pearson, E. S. (1928). On the use and interpretation of certain test criteria for purposes of statistical inference: Part i, part ii. *Biometrika*, pages 175–240,263–294.
- [83] Niu, Y. S., Hao, N., and Zhang, H. (2016). Multiple change-point detection: A selective overview. *Statistical Science*, 31(4):611–623.
- [84] Page, E. S. (1954). Continuous inspection schemes. Biometrika, 41(1/2):100–115.
- [85] Peña, E. A. (1998). Smooth goodness-of-fit tests for composite hypothesis in hazard based models. The Annals of Statistics, 26(5):1935–1971.
- [86] Peskir, G. and Shiryaev, A. N. (2002). Solving the Poisson disorder problem. In Advances in finance and stochastics, pages 295–312. Springer.

- [87] Poisson, S. D. (1837). Recherches sur la probabilité des jugements en matière criminelle et en matière civile. Bachelier.
- [88] Polunchenko, A. S. and Tartakovsky, A. G. (2012). State-of-the-art in sequential change-point detection. *Methodology and computing in applied probability*, 14(3):649– 684.
- [89] Pyke, R. (1959). The supremum and infimum of the poisson process. Ann. Math. Statist., 30.
- [90] Raftery, A. E. (1994). Change point and change curve modeling in stochastic processes and spatial statistics. *Journal of Applied Statistical Science*, 1(4):403–423.
- [91] Raftery, A. E. and Akman, V. E. (1986). Bayesian analysis of a Poisson process with a change-point. *Biometrika*, pages 85–89.
- [92] Rivera, C. and Walther, G. (2013). Optimal detection of a jump in the intensity of a Poisson process or in a density with likelihood ratio statistics. *Scandinavian Journal* of *Statistics*, 40(4):752–769.
- [93] Rubin, H. (1961). The estimation of discontinuities in multivariate densities, and related problems in stochastic processes. In Proceedings of the Fourth Berkeley Symposium on Mathematical Statistics and Probability, Volume 1: Contributions to the Theory of Statistics.
- [94] Shen, J. J. and Zhang, N. R. (2012). Change-point model on nonhomogeneous Poisson processes with application in copy number profiling by next-generation DNA sequencing. *The Annals of Applied Statistics*, 6(2):476–496.
- [95] Shorack, G. and Wellner, J. (1986). Empirical Processes with Applications to Statistics. Wiley Series in Probability and Mathematical Statistics: Probability and Mathematical Statistics. John Wiley and Sons, Inc., New York.
- [96] Siegmund, D. (1988). Confidence sets in change-point problems. International Statistical Review, pages 31–48.
- [97] Soltani, R., Goeckel, D., Towsley, D., and Houmansadr, A. (2015). Covert communications on poisson packet channels. In 2015 53rd Annual Allerton Conference on Communication, Control, and Computing (Allerton), pages 1046–1052. IEEE.
- [98] Soltani, R., Goeckel, D., Towsley, D., and Houmansadr, A. (2020). Fundamental limits of covert packet insertion. *IEEE Transactions on Communications*.
- [99] Soltani, S., Khayam, S. A., and Radha, H. (2008). Detecting malware outbreaks using a statistical model of blackhole traffic. In 2008 IEEE International Conference on Communications, pages 1593–1597. IEEE.
- [100] Sonesson, C. and Bock, D. (2003). A review and discussion of prospective statistical surveillance in public health. *Journal of the Royal Statistical Society: Series A* (*Statistics in Society*), 166(1):5–21.

- [101] Spokoiny, V. G. (1996). Adaptive hypothesis testing using wavelets. The Annals of Statistics, 24(6):2477–2498.
- [102] Sukhatme, P. V. (1936). On the analysis of k samples from exponential populations with especial reference to the problem of random intervals, volume 1.
- [103] Tartakovsky, A., Nikiforov, I., and Basseville, M. (2014). Sequential analysis: Hypothesis testing and changepoint detection. Chapman and Hall/CRC.
- [104] Tsybakov, A. B. (2008). Introduction to nonparametric estimation. Springer Science & Business Media.
- [105] Verzelen, N., Fromont, M., Lerasle, M., and Reynaud-Bouret, P. (2021). Optimal change-point detection and localization. arXiv 2010.11470.
- [106] Vishwanath, A., Sivaraman, V., and Ostry, D. (2009). How poisson is tcp traffic at short time-scales in a small buffer core network? In 2009 IEEE 3rd International Symposium on Advanced Networks and Telecommunication Systems (ANTS), pages 1– 3. IEEE.
- [107] Wald, A. (1945). Sequential tests of statistical hypotheses. The Annals of Mathematical Statistics, 16(2):117–186.
- [108] Wang, L. (2018). The continuous-time poisson channel has infinite covert communication capacity. In 2018 IEEE International Symposium on Information Theory (ISIT), pages 756–760. IEEE.
- [109] Worsley, K. J. (1986). Confidence regions and tests for a change-point in a sequence of exponential family random variables. *Biometrika*, 73(1):91–104.
- [110] Yang, L. (2014). Hypotheses testing problems for inhomogeneous Poisson processes. PhD thesis, Université du Maine.
- [111] Yang, L. (2020). Multiple hypothesis testing for Poisson processes with variable change-point intensity. *Communications in Statistics - Theory and Methods*, pages 1–23.
- [112] Young Yang, T. and Kuo, L. (2001). Bayesian binary segmentation procedure for a Poisson process with multiple changepoints. *Journal of Computational and Graphical Statistics*, 10(4):772–785.